The Impacts of Climate Change on Rice Production and Small Farmers' Adaptation: A Case of Guyana

Omchand Mahdu

Dissertation submitted to the faculty of the Virginia Polytechnic Institute and State University in partial fulfillment of the requirements for the degree of

> Doctor of Philosophy in Planning, Governance, and Globalization

> > Todd Schenk, Co-Chair Yang Zhang, Co-Chair Andrew Ellis Bradford Mills

April 21, 2019 Blacksburg, Virginia

Keywords: climate change, climate variability, rice production, small farmers, climate adaptation, Guyana

Copyright © 2019, Omchand Mahdu

The Impacts of Climate Change on Rice Production and Small Farmers' Adaptation: A Case of Guyana

Omchand Mahdu

Abstract

Prior research has concluded that climate change is having an overall negative impact on rice production worldwide. The vast majority of climate change impacts on rice production result from fluctuations in precipitation and temperature, which lead to flooding, water scarcity, and increases in insects and pests, diseases, and weeds. As a small developing country, Guyana is highly vulnerable to climate change despite its insignificant contribution to global warming. Guyana heavily relies on rice cultivation for food, employment, and export earnings. While generally increasing, rice yields have fluctuated over the last two decades. For example, in 2016, rice yields declined by 12.7 percent due to a drought. This dissertation explores the relationship between fluctuating yields and climate change, and how farmers are adapting.

Of particular importance are the impacts of climate change on small farmers (those cultivating less than 4.45 hectares or 11 acres) and their ability to successfully adapt. Small farmers are especially vulnerable to a changing climate because they often lack the necessary knowledge, support, and resources to effectively respond and adapt. Given the large percentage of rice farmers engaging in small-scale production in Guyana, this study investigates the impacts of climate variability on rice production and the extent to which the production and productivity of small farmers are affected. It also identifies the coping strategies small farmers employ to combat the effects of climate change and the extent to which these strategies are successful.

Given that climate change is expected to vary across different regions of the world, the first aim of this study is to show how the climate in Guyana has changed. At the country level, evidence from descriptive statistics, a linear trend model, and a two-sample t-test shows that minimum and maximum temperatures have increased over the last 111 years. The aggregate data is less clear on changes in precipitation over the last 111 years.

However, analysis of farm-level data provides strong evidence of shifts in rainfall patterns. Among 189 small farmers interviewed, 182 (96.3%) perceived changes in rainfall patterns, 170 (89.9%) perceived changes in temperature, 169 (89.4%) perceived changes in extreme weather events, 185 (97.9%) perceived changes in insects and pests, 73 (38.6%) perceived changes in diseases, and 168 (88.9%) perceived changes in weeds.

Changes in precipitation have included an increase in intensity and out of season rainfall, which has impacted harvesting due to poor dams, wet fields, and the lodging of plants. The primary responses farmers have adopted include adjusting planting dates based on water availability and the cultivation of different rice varieties. Changes in temperature have resulted in hotter days, accelerating the evaporation of water from fields. In response, farmers replenish water in their fields, when available. Excess rainfall and resulting flooding, drought, and heavy winds have been the primary extreme weather events observed. Excess rainfall and associated flooding submerges, uproots, and/or kills young plants. The lodging of plants due to heavy winds and flooding has been the main impact. In response to flooding, farmers have pumped water out of their fields. There is very little that farmers can do in response to heavy winds.

The primary change in insects and pests reported by farmers has been an increase in paddy bug infestations, which cause damage to the grains resulting in lower quality and quantity at harvest. As a result, farmers are engaging in more preventative spraying. An increase in brown spot disease was also reported. Brown spots are primarily found on the leaves, damaging and/or stunting the growth of the plants by reducing the amount of food they manufacture through photosynthesis. Farmers have responded by engaging in preventative spraying and the rotation of fungicides. Increases in red rice and duckweed have been the major changes in weeds observed. Both weeds compete with rice for space, sunlight, nutrients, and water. Additionally, red rice reduces the quality and by extension the price farmers receive. Farmers are responding by spraying more herbicide and using a contact chemical to burn red rice.

Multivariate analysis of farm-level data found that land tenure, tractor ownership, membership in an agricultural organization(s), secondary non-agricultural income, and farms located in regions two and four have positive correlations with annual yields. Perceived changes in rainfall, farm size, livestock ownership, participation in rice extension training, and household members help with rice farming were found to have negative correlations with annual yields.

Policy recommendations to improve rice production and farmers' resilience include improving research and development capacity; tax exemption for agricultural inputs and equipment; improving extension services; improving the management of irrigation systems and water resources; enhanced access to credit, insurance, and subsidies; improving weather forecasting and climate monitoring; and improving the management of drainage infrastructure.

The analytical framework used in this research produced a rich dataset and interesting results that are important to our understanding of farm-level impacts and responses to climate change. As such, it may prove useful for studying climate change impacts in other developing countries that have similar characteristics and face similar risks from climate change as Guyana.

The Impacts of Climate Change on Rice Production and Small Farmers' Adaptation: A Case of Guyana

Omchand Mahdu

General Audience Abstract

The vast majority of climate change impacts on rice production result from variations in rainfall and temperature that lead to flooding, water shortage, and increases in insects and pests, diseases, and weeds. Guyana is highly exposed to climate change. More importantly, the country relies heavily on rice farming for food, employment, and foreign income. Of particular importance are the impacts of climate change on small farmers (growing less than 4.45 hectares) and their ability to successfully adapt. Small farmers are especially helpless because they often lack the necessary knowledge, support, and resources to effectively respond and adapt. Given the large percentage of rice farmers engaged in small-scale production in Guyana, this study explores the impacts of climate variability on rice production and the extent to which the production and output of small farmers are affected.

Analysis of farm-level data shows that changes in rainfall have included an increase in intensity and out of season rainfall which has affected harvesting due to poor farm-tomarket roads, wet fields, and lodging of plants. The main responses involved adjusting planting dates based on water availability and the cultivation of different rice varieties. Changes in temperature resulted in hotter days which increased the loss of water from the field. In response, farmers replenish water in their fields, when available. Excess rainfall and resulting flooding, drought, and heavy winds have been the main extreme weather events observed. Excess rainfall and associated flooding submerges, uproots, and/or kills young plants. The lodging of plants due to heavy winds and flooding has been the main impact. In response to flooding, farmers have pumped water out of their fields. There is very little that farmers can do in response to heavy winds. An increase in paddy bug infestations damaged the grains resulting in lower grain quality while an increase in red rice and duckweed increased the competition for space, sunlight, nutrients, and water. Farmers engaged in more defensive spraying and used a contact chemical to burn red rice.

Dedication

I dedicate this dissertation to my wife for her unconditional love and support throughout this entire journey. You have been a great source of strength, motivation, and inspiration and I am grateful for your patience, understanding, and constant words of encouragement. You have never wasted an opportunity to remind me to stay positive when progress was slow, and I am thankful for having you in my corner. To my daughter Sofia for reminding me each day that there are more important things to life than writing a dissertation. I hope you will understand one day the sacrifice I have had to make between spending more time with you and working on this research. To my mother who taught me never to leave for tomorrow what can be done today. Your support over the last several years is immeasurable and I am very grateful for all you have done to help get me through this arduous process. To my father for helping to ensure my early academic endeavors were realized and to my brother and sister for their loyalty, friendship, and responsiveness over my academic career and in my times of need. You have all contribute to my growth and development as a person and my achievements could not have been possible without your commitment, guidance, and involvement. In one way or the other, you have enabled me to achieve so much in such a short period of time.

Acknowledgement

The author acknowledges several individuals whose contributions in various capacities have made this research a reality. I owe my deepest gratitude to Mr. Ricky Roopchand for introducing me to key stakeholders in Guyana's rice industry. Special thanks to the staff of the Guyana Rice Development Board (GRDB) especially the district rice extension officers that helped identify small farmers and accompanied me during my fieldwork and to the small farmers that participated in this research without fear and/or reservation. To my advisors Dr. Todd Schenk and Dr. Yang Zhang for providing invaluable advice and constructive feedback throughout this research process. Special thanks to Dr. Brad Mills and Dr. Andrew Ellis for agreeing to serve on my committee and for responding to my questions and concerns in a timely manner. I am also grateful to the late Dr. John Browder who provided innumerable suggestions and steadfast support in the very beginning of this research project. Special thanks to my wife Samantha Mahdu for helping to refine my thoughts and making valuable suggestions throughout this research. My gratitude is also extended to my Planning, Governance, and Globalization (PGG) cohort for sharing their ideas, knowledge, and experience over the last three and a half years. I greatly appreciate your candor and insights.

List of Figures	XV
List of Tables	xvii
List of Abbreviations	xix
Chapter 1 Introduction	1
1.1 Background and Motivation	2
1.1.1 Research Location	3
1.1.2 Guyana's Vulnerability to Climate Change	4
1.1.3 Climate Variability, Insect Infestation, and Rice Cultivation in Guyana	6
1.1.4 Importance of Rice Production to Guyana	8
1.1.5 Small Farmers	9
1.2 Brief History of the Rice Industry in Guyana	10
1.3 Contemporary Rice Production in Guyana	14
1.4 Other Challenges Facing the Rice Industry in Guyana	15
1.5 Research Questions	17
1.6 Contributions and Anticipated Impact	18
1.7 Study Overview	18
	20
Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20
Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20
Chapter 2 Review of Literature: Climate Change and Crop Agriculture 2.1 Introduction 2.2 Evidence of Global Climate Change	20 21 21
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	21 21 21 22 23
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	21 21 22 22 23 24
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	21 21 22 22 23 24 25
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	21 21 22 22 23 24 25 26
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	21 21 21 22 23 23 24 25 26 27
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 24 25 26 27 30
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 24 25 26 27 30 33
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 23 23 25 26 27 30 33 33
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 23 23 25 26 27 30 33 33 34
Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 23 23 23 23 23 23 24 25 26 30 33 33 34 35
 Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 23 23 23 23 23 25 26 27 30 33 33 34 35 35
Chapter 2 Review of Literature: Climate Change and Crop Agriculture	20 21 21 22 23 24 25 26 27 30 33 33 34 35 35

Table of Contents

2.4.7 Climate Change and Future Rice Production	37
2.4.8 Rice Cultivation Impact on Climate Change	37
2.5 Approaches to Assessing Climate Change Impacts on Crop Agriculture	38
2.5.1 General Equilibrium Models	38
2.5.2 Partial Equilibrium Models	39
2.6 Climate Data	44
2.7 Empirical Evidence of Climate Change Impacts on Crop Agriculture	46
2.7.1 Evidence from Global Agriculture	46
2.7.2 Evidence from Africa, Western Europe, and South America	49
2.7.3 Evidence from the Brazil, Canada, China, India and United States	51
2.7.4 Evidence from Other Developed and Developing Countries	53
2.8 Empirical Evidence of Farmers' Perceptions and Adaptation to Climate Change.	56
2.9 Gaps in the Existing Literature	59
2.10 Conceptual Framework	60
2.11 Conclusion	61
Chapter 3 Data and Methods	64
3.1 Introduction	65
3.2 Analyzing Climate Variability and Climate Change in Guyana	66
3.2.1 Data Source and Description – Climate Data	66
3.2.2 Methods of analyzing climate data	67
3.3 Analyzing Small Farmers' and Key Informants' Perceptions, Impacts, and Adaptation	.72
3.3.1 Theoretical Model and Empirical Approach	73
3.4 Research Design	84
3.4.1 Study Area	84
3.4.2 Study Population	88
3.4.3 Sample Size and Selection	88
3.4.4 Data Source and Description of Survey Instrument	89
3.4.5 Validity and Reliability of the Survey Instrument	91
3.4.6 Pre-Pilot Testing	92
3.4.7 Pilot Testing	92
3.4.8 Interviews	93
3.4.9 Coding and Tabulation Process	97
3.5 Method of Analyzing Rice Yield and Non-climatic Data at the Farm-level	98
3.5.1 Multiple Regression	98

3.5.2 Multiple Regression Equation	
3.6 Method for Analyzing Small Farmers' and Key Informants' Perceptions and Adaptation	s, Impacts, 101
3.7 Conclusion	
Chapter 4 An Overview of Climate Variability and Climate Change in G	iyana 104
4.1 Introduction	
4.2 Weather and Climate	105
4.3 Climate Variability and Climate Change	106
4.4 Climate of Guyana	
4.4.1 Rainfall and Climate Types	108
4.4.2 Temperature, Wind, Sunshine, and Humidity	111
4.5 Characterizing Climate Variability in Guyana	
4.6 Empirical Results of Climate Change and Climate Variability in Guyana	a 115
4.6.1 Evidence from Descriptive Statistics	115
4.6.2 Evidence from Linear Trend Model	123
4.6.3 Evidence from Two Sample T-test	128
4.7 Future Climate Change Projections	129
4.8 Conclusion	134
Chapter 5 Multivariate Analysis of Farm-level Data	
5.1 Introduction	137
5.2 Small Farmers' Perceptions: Rainfall, Temperature and Extreme Weath and Pests, Diseases, and Weeds	er, Insects 138
5.3 Socio-economic Characteristics of Small Farmers	
5.3.1 Age, Gender, and Education	138
5.3.2 Primary and Secondary Sources of Income	139
5.4 Farm-level Characteristics	
5.4.1 Farm Size and Tenure Status	140
5.4.2 Farming Experience, Household Participation, and Seasonal Labor	r 141
5.4.3 Equipment Ownership and Rental	142
5.4.4 Livestock Ownership	142
5.5 Institutional Accessibility	
5.5.1 Farmers' Participation in Extension Training	143
5.5.2 Weather Information in Advance	145
	145
5.5.3 Input Credit and Agricultural Insurance	145

5.5.5 Membership in an Agricultural Organization	146
5.6 Inputs and Equipment	146
5.6.1 Seed Variety	147
5.6.2 Quality of Inputs and Equipment	148
5.7 Yields and its Determinants: Multivariate Analyses	149
5.7.1 Estimation Procedure: Heteroskedasticity and Multicollinearity	150
5.7.2 Results and Discussion of the Multivariate Analysis	150
5.8 Conclusion	164
Chapter 6 Farmers' Perceptions, Impacts, and Adaptation	166
6.1 Introduction	167
6.2 Small Farmers' Understanding of Climate Change	168
6.3 Changes in Rainfall	170
6.3.1 Perceptions of Changes in Rainfall	170
6.3.2 Perceived Impacts of Changes in Rainfall	175
6.3.3 Responses to Perceived Changes in Rainfall	183
6.4 Changes in Temperature	189
6.4.1 Perceptions of Changes in Temperature	189
6.4.2 Perceived Impacts of Changes in Temperature	191
6.4.3 Responses to Perceived Changes in Temperature	195
6.5 Changes in Extreme Weather Events	198
6.5.1 Perceived Changes in Extreme Weather Events	198
6.5.2 Perceived Impacts of Changes in Extreme Weather Events	200
6.5.3 Responses to Perceived Changes in Extreme Weather Events	206
6.6 Changes in Insects and Pests	209
6.6.1 Perceived Changes in Insect and Pest Populations and Impacts	209
6.6.2 Perceived Impacts of Changes in Insects and Pests	215
6.6.3 Responses to Perceived Changes in Insects and Pests	218
6.7 Changes in Diseases	222
6.7.1 Perceived Changes in Diseases	223
6.7.2 Impacts of Perceived Changes in Diseases	226
6.8 Changes in Weeds	
6.8.1 Perceived Changes in Weeds	231
6.8.2 Impacts of Perceived Changes in Weeds	236
6.8.3 Responses to Perceived Changes in Weeds	238

APPENDIX D – MEAN, STANDARD DEVIATION, AND COEFFICIENT OF VARIATION (1905 – 2015) ^a	351
APPENDIX C – CODEBOOK	. 345
APEENDIX B – SURVEY INSTRUMENT: KEY INFORMANTS	. 340
APPENDIX A – SURVEY INSTRUMENT: SMALL FARMERS	. 329
References	. 305
7.4 Future Work	. 302
7.3.5 Other Policy Recommendations	. 298
7.3.4 Disaster Risk Management	. 296
7.3.3 Crop and Income Loss Risk Management	. 293
7.3.2 Irrigation and Water Resource Management	. 292
7.3.1 Crop Development and Farming Practices	. 289
7.3 Policy Implications and Recommendations	. 288
7.2.6 Other Findings	. 287
7.2.5 Research Question 5:	. 285
7.2.4 Research Question 4:	. 284
7.2.3 Research Question 3:	. 282
7.2.2 Research Question 2:	. 280
7.2.1 Research Question 1:	. 279
7.2 Summary of Key Findings	. 279
Chapter 7 Conclusion and Recommendations	. 278
6.16 Conclusion	. 273
6.15.3 Limits to Adaptation	. 268
6.15.2 Maladaptation	. 267
6.15.1 Co-benefits	. 266
6.15 Co-benefits, Maladaptation, and Limits to Adaptation	. 266
6.14 Barriers to Adaptation	. 257
6.13 Rice Extension Services and Small Farmers' Adaptation	. 254
6.12 Paying for Changes in Farming Practice	. 250
6.11 Crop Failure and Coping Strategies	. 247
6.10 Seasonal Changes and Impacts on Yields	. 247
6.9 Changes to Farming Practices	. 244

List of Figures

Figure 1.1 Administrative map of Guyana	4
Figure 1.2 Map showing the low coastal zone of Guyana	5
Figure 1.3 Aggregate rice production for Guyana (1970-2017)	. 14
Figure 2.1 Global atmospheric concentrations of carbon dioxide over time	. 22
Figure 2.2 Global greenhouse gas emissions by gas (1990-2010)	. 23
Figure 2.3 Temperature worldwide (1901-2015)	. 24
Figure 2.4 Average global sea surface temperature (1880–2015)	. 25
Figure 2.5 Precipitation worldwide (1901-2015)	. 26
Figure 2.6 Global average absolute sea level change (1880–2015)	. 27
Figure 2.7 March and September monthly average Arctic Sea ice extent (1979–2016).	. 28
Figure 2.8 Dwindling Arctic Sea ice.	. 29
Figure 2.9 Impacts of climate change on the agricultural sector	. 32
Figure 2.10 Conceptual framework of climate change impacts on rice yields and	
adaptation strategies	. 61
Figure 3.1 Map of natural regions in Guyana	. 85
Figure 3.2 Map of primary rice producing regions in Guyana	. 86
Figure 3.3 Map of Guyana showing rice producing areas	. 87
Figure 3.4 Map of locations of farm-level interviews	. 96
Figure 4.1 Timescale of weather, climate variability, and climate change	107
Figure 4.2 Köppen-Geiger climate type map of Guyana	109
Figure 4.3 Climatic regions of Guyana	110
Figure 4.4 Mean temperature and rainfall for Guyana (1901-2015)	112
Figure 4.5 Rainfall Stations in Guyana	114
Figure 4.6 Moving average of mean annual rainfall for Guyana	117
Figure 4.7 Standard deviation of mean annual rainfall for Guyana	118
Figure 4.8 Coefficient of variation of annual rainfall for Guyana	119
Figure 4.9 Moving average of mean annual minimum temperature for Guyana	119
Figure 4.10 Standard deviation of mean annual minimum temperature for Guyana	120
Figure 4.11 Coefficient of variation of annual minimum temperature for Guyana	121
Figure 4.12 Moving average of mean annual maximum temperature for Guyana	121
Figure 4.13 Standard deviation of mean annual maximum temperature for Guyana	122
Figure 4.14 Coefficient of variation of annual maximum temperature for Guyana	123
Figure 4.15 Projected change in precipitation (mm/day) relative to 1986–2005	132
Figure 4.16 Projected change in near surface temperature (0C) relative to 1986–2005.	133
Figure 6.1 Poor condition of access dam	176
Figure 6.2 Harvesting under wet (left) and dry (right) field conditions	177
Figure 6.3 Combine harvester and transport truck (left) and grain cart (right)	178
Figure 6.4 Unlevelled field resulting in uneven water distribution	181
Figure 6.5 Early sowing in region 2 - Pomeroon Supenaam	184
Figure 6.6 A farmer cleaning a drainage canal (left) and drainage outlet pipe (right)	185
Figure 6.7 A rice field after flood waters receded	202
Figure 6.8 Heavy winds caused lodging of rice plants	203
Figure 6.9 Drainage canal (right) overflowing rice field (left)	207

Figure 6.10 Paddy bug (top left), snail (top right), adult water weevil (center left), w	ater
weevil larvae (center right), caterpillar (bottom left) and leaf miner (bottom right)	212
Figure 6.11 Brown planthopper (left) and leaffolder (right)	213
Figure 6.12 Farmer sweeping (left) and spraying for paddy bugs (right)	219
Figure 6.13 Weeds on the embankment of a rice field sanitized	220
Figure 6.14 Brown spot (left) and blast disease (right)	224
Figure 6.15 Sheath blight (left) and sheath rot (right)	225
Figure 6.16 Red rice (left) and duckweed (right)	233
Figure 6.17 Jhussia (left) and monkeytail (right)	234
Figure 6.18 Muraina (left) and water sedge (right)	235
Figure 6.19 Red rice grains after milling	237
Figure 6.20 Red rice plants controlled using the rope and stick method	240
Figure 6.21 Water rogued rice field	241
Figure 6.22 A farmer engaged in dry rogueing	241
Figure 6.23 A farmer leveling the land prior to sowing	242

List of Tables

Table 2.1 Rice Plant Response to Varying Daily Mean Temperature at Different Growth	
Stages	3
Table 3.1 Definitions, Expected Signs and Source of Socioeconomic, Farm-level, and	
Institutional Variables	2
Table 3.2 Summary of Farmers by Acreage and Region	7
Table 3.3 Small Farmers by Region	8
Table 3.4 Reasons for Replacing Farmers	5
Table 3.5 Sample Size by Region	5
Table 4.1 Climate Variability in Guyana (1901-2015) ^a 11	5
Table 4.2 Trend Analysis of the Variability in Annual Rainfall	5
Table 4.3 Trend Analysis of the Variability in Annual Minimum Temperature 12	6
Table 4.4 Trend Analysis of the Variability in Annual Maximum Temperature 12	8
Table 4.5 Future Climate Scenarios for Guyana 13	0
Table 5.1 A ge of Small Farmers	8
Table 5.2 Education Level of Small Farmers 13	0
Table 5.3 Income Source of Small Farmers	á
Table 5.4 Other Agricultural Sources of Income of Small Farmers	0
Table 5.5 Non Agricultural Sources of Income of Small Formers	0
Table 5.5 Non-Agricultural Sources of filcome of Sman Farmers	1
Table 5.0 Farming Experience of Small Economy 14	1 1
Table 5.7 Farming Experience of Small Farmers	1 2
Table 5.8 Tasks Performed by Seasonal Labor	2
Table 5.9 Type of Livestock Owned	3
Table 5.10 Small Farmers' Knowledge of the Six-Point Practice	4
Table 5.11 Seed Variety Planted	8
Table 5.12 Quality of Inputs and Equipment	.9
Table 5.13 Definitions, Expected Signs, and Descriptive Statistics of Independent	_
Variables	2
Table 5.14 Multivariate Analysis of Small Farmers' Yields ^a 15	4
Table 5.15 Multivariate Analysis of Small Farmers' Yield by Season ^a 16	1
Table 6.1 Meaning of Climate Change to Small Farmers 16	9
Table 6.2 Changes in Rainfall Reported by Small Farmers and Key Informants 17	4
Table 6.3 Impacts of Changes in Rainfall Reported by Small Farmers and Key Informant	ts
	2
Table 6.4 Responses to Perceived Changes in Rainfall	7
Table 6.5 Changes in Temperature Reported by Small Farmers and Key Informants 19	0
Table 6.6 Impacts of Changes in Temperature 19	4
Table 6.7 Responses to Perceived Changes in Temperature 19	7
Table 6.8 Changes in Extreme Weather Events Reported by Small Farmers and Key	
Informants	9
Table 6.9 Impacts of Changes in Extreme Weather Events	5
Table 6.10 Responses to Perceived Changes in Extreme Weather Events	8
Table 6.11 Changes in Insects and Pests Reported by Small Farmers and Key Informants	3
	4
Table 6.12 Impacts of Changes in Insects and Pests	7

Table 6.13 Responses to Perceived Changes in Insects and Pests	221
Table 6.14 Changes in Diseases Reported by Small Farmers and Key Informants	225
Table 6.15 Impacts of Changes in Diseases	227
Table 6.16 Responses to Perceived Changes in Diseases	229
Table 6.17 Changes in Weeds Reported by Small Farmers and Key Informants	235
Table 6.18 Impacts of Changes in Weeds	238
Table 6.19 Responses to Perceived Changes in Weeds	243
Table 6.20 Adaptation by Small Farmers	246
Table 6.21 Coping Strategies Employed by Small Farmers	250
Table 6.22 Paying for Adaptation	253
Table 6.23 Barriers to Adaptation	265

List of Abbreviations

AEZ	Agroecological Zone Analysis
AGE	Applied General Equilibrium Model
AOGCM	Atmosphere-Ocean General Circulation Model
APSIM	Agricultural Production Systems Simulator
AQUACROP	Crop-water Productivity Model
BoG	Bank of Guyana
CAEPNET	Caribbean Agricultural Extension Providers Network
CERES	Crop Estimation through Resource and Environment Synthesis
CGE	Computable General Equilibrium Model
CH ₄	Methane
CIAT	International Center for Tropical Agriculture
CMIP	Coupled Model Intercomparison Project
CO ₂	Carbon Dioxide
CRU	Climate Research Unit
CV	Coefficient of Variation
DEI	Disaster Exposure Index
DSSAT	Decision Support System for Agrotechnology Transfer
ECLAC	Economic Commission for Latin America and the Caribbean
ENSO	El Niño Southern Oscillation
EPA	Environmental Protection Agency
FAO	Food and Agricultural Organization
FLAR	Latin American Fund for Irrigated Rice
GCM	General Circulation Model
GDP	Gross Domestic Product
GMSL	Global Mean Sea Level
GNBS	Guyana National Bureau of Standards
GRA	Guyana Revenue Authority
GRDB	Guyana Rice Development Board
GREB	Guyana Rice Export Board
GRIF	Guyana REDD+ Investment Fund
GRMMA	Guyana Rice Milling and Marketing Authority
GTAP-W	Global Trade Analysis Project-Water
H ₂ O	Water Vapor
HadCM	Hadley Center Coupled Model
HDI	Human Development Index
HYDROMET	Hydrometeorological Service
IDB	Inter-America Development Bank
IFC	International Finance Corporation
	International Food Policy Research Institute
IPUU IDM	Intergovernmental Panel on Climate Change
	Integrated Pest Management
IKKI ITCZ	International Kice Kesearch Institute
HCL	Inter-tropical Convergence Zone

KN	Kaieteur News
KNMI	Royal Netherlands Meteorological Institute
LAC	Latin America and the Caribbean
LCDS	Low Carbon Development Strategy
MA	Moving Average
MMAADA	Mahaica, Mahaicony, Abary Agricultural Development Authority
MoA	Ministry of Agriculture
MoF	Ministry of Finance
MONICA	Model for Nitrogen and Carbon in Agroecosystems
MoP	Ministry of the Presidency
N ₂ O	Nitrous Oxide
NCEP	National Center for Environmental Prediction
NCDC	National Climatic Data Center
ND-GAIN	Notre Dame Global Adaptation Initiative
NDIA	National Drainage and Irrigation Authority
NIS	National Insurance Scheme
NOAA	National Oceanic and Atmospheric Administration
NPRGC	National Padi and Rice Grading Centre
O ₃	Ozone
OLS	Ordinary Least Square
РТССВ	Pesticides and Toxic Chemicals Control Board
RCP	Representative Concentration Pathway
RH	Relative Humidity
RPA	Rice Producer Association
SB	Stabroek News
SRES	Special Report on Emissions Scenarios
UAH	University of Alabama Huntsville
UDEL	University of Delaware
UNDP	United Nations Development Program
UNFCCC	United Nations Framework Convention on Climate Change
USGCRP	United States Global Change Research Program
VAT	Value-added Tax
VIF	Variance Inflation Factor
WB	World Bank
WMO	World Meteorological Organization
WWR	World Weather Records

Chapter 1 Introduction

1.1 Background and Motivation

In the first two decades of the 21st century, the impacts of anthropogenic climate change have attracted considerable attention across the world. Of particular concern are the impacts of climate change on agricultural production and food security in terms of access, consumption, and stable prices (Dinar and Mendelsohn 2011; IPCC 2014b; FAO 2016; FAO 2018). The Intergovernmental Panel on Climate Change (IPCC) fifth assessment report notes with high confidence the negative effects of climate change on agricultural production (IPCC 2014a). Crop agriculture is especially sensitive to a changing climate. Variations and changes in rainfall and temperature, and the frequency and intensity of extreme weather events such as floods, droughts, and windstorms are expected to negatively affect future yields in some regions (WB 2010b; IPCC 2014b; FAO 2018).

Although climate change respects no borders, vulnerability remains disproportionate across different countries, groups, and crops. For developing countries like Guyana that depend heavily on agriculture for food, employment, and export earnings (ECLAC 2011; MoA 2013; MoP 2015), climate change represents a clear and present danger with far-reaching socio-economic implications. The effects of climate change usually translate into lower incomes at the household and national levels (FAO 2016).

More importantly, small farmers are especially vulnerable to a changing climate (Frank and Buckley 2012; Harvey et al. 2014; Holland et al. 2017; Harvey et al. 2018). Although they may boast significant experience in responding to climate variability, the sheer magnitude of the variability associated with long-term climate change is beyond traditional coping strategies (Pettengell 2010). In addition, their marginalized status often prevents them from accessing the resources needed to successfully cope with climate shocks (Harvey et al. 2014). According to Vorley and Chan (2012), small farmers usually lack access to technology, credit, and the institutional support that are crucial for helping them respond to climate change.

1.1.1 Research Location

The geographic focus of this research is the five rice-producing regions of the Cooperative Republic of Guyana. Guyana is located on the northeastern corner of South America between 2°N and 8°N Latitude and 57°W and 61.5°W Longitude and shares borders with Suriname to the east, Brazil to the south and southwest, Venezuela to the west, and the Atlantic Ocean to the north. Figure 1.1 shows the administrative regions of Guyana. Guyana has a land area of 83,000 square miles and a population of approximately 757,000 people, the majority of whom live along the agricultural belt on the Atlantic coast. Rice is primarily cultivated in administrative regions 2, 3, 4, 5, and 6¹ spanning the three counties of Essequibo, Demerara, and Berbice.

¹ Rice is also grown at Moco Moco in Upper Takutu-Upper Essequibo (Region 9).



Figure 1.1 Administrative map of Guyana Source: GLSC (2006)

1.1.2 Guyana's Vulnerability to Climate Change

Guyana is extremely vulnerable to climate change. Approximately 90 percent of the population is concentrated on the low coastal plain natural region (MoP 2015) which forms the heart of the country's economic activities (ECLAC 2011; NAPG 2016), key among them is the

agricultural sector (Hickey and Weis 2012; Velasco 2014). The coastal zone ranges from 8 to 65 kilometers in width (Velasco 2014) and runs 425 kilometers along the Atlantic coast (Hickey and Weis 2012). The majority of this relatively flat region is located below sea level and highly susceptible to flooding due to rising seas and excess rainfall. Figure 1.2 shows the coastal zone of Guyana.



Figure 1.2 Map showing the low coastal zone of Guyana Source: Velasco (2014)

According to the Inter-American Development Bank (IDB) Disaster Exposure Index (DEI), Guyana is the fourth most exposed country to natural disasters in Latin America and the Caribbean (LAC) (Garlati 2013). Barr, Fankhauser, and Hamilton (2010) grouped Guyana among countries that face the highest risks related to climate change. These rankings are especially troubling considering that Guyana is spared from major tropical storms and lies outside of the Caribbean hurricane corridor (MoP 2015). In the LAC region, Guyana is ranked number one in terms of flood risk and faces a very high risk of droughts (Garlati 2013). In the last two decades, Guyana has experienced both floods (2005, 2006, 2008, 2010, 2011, 2013, 2014, and 2015) and droughts (1997-1998, 2009-2010, and 2015-2016) (NAPG 2016). Extreme rainfall events coupled with inadequate infrastructure were chiefly responsible for these flood events.

Equally important is the country's resilience and adaptive capacity. As a developing country, Guyana is one of the poorest countries in the LAC region. It is ranked 125th on the United Nations Development Program (UNDP) Human Development Index (HDI) (UNDP 2018). The Notre Dame Global Adaptation Initiative (ND-GAIN)², ranks Guyana 111th in the world (ND-GAIN 2016). Thus, considering Guyana's limited capacity to adapt, a changing climate poses significant threats to the country's economic base.

1.1.3 Climate Variability, Insect Infestation, and Rice Cultivation in Guyana

Although there is a lack of empirical evidence of climatic impacts on rice production in Guyana, articles in the local media and letters to the editor highlight the effects of climate variability on rice farming in recent years. As recent as January 2019, rice farmers in Region 2 were advised to conserve water since low rainfall during the short rainy season [December 2018–

² Summarizes a country's vulnerability to climate change and other global challenges in combination with its readiness to improve resilience (ND-GAIN 2016).

January 2019] resulted in lower water levels in the main water conservancy (SN 2019). Similarly, the start of the 2019 spring season [December 2018- January 2019] has been severely affected by drought like conditions in Region 6. Reports suggest that over 3,237 hectares [8,000 acres] of rice are at risk while another 3,237 hectares [8,000 acres] remain uncultivated due to the lack of water (KN 2019).

In June 2018, heavy rainfall coupled with the lack of maintenance prompted emergency work by the Mahaica, Mahaicony, Abary-Agricultural Development Authority (MMA-ADA) on the Onverwagt access dam [farm-to-market road] in Region 5. This dam is a major access point to rice farms in the region. Yusuf (2018) reported in July 2018 that impassable dams made worse by rainy conditions resulted in over 80 hectares [200 acres] of rice left unharvested in Region 6. Inclement weather and inaccessible dams negatively affect harvesting and increased the costs of transporting the grains from the field to the mills (SN 2018). In September 2018, heavy rainfall and poor access dams delayed harvesting by a week in Region 2 (KN 2018a). Delays in harvesting usually result in lower than expected yields. According to Khan (2018a), waterlogged land affected the harvest of approximately 8,093 hectares [20,000 acres] of rice on the northern side of Region 2.

In January 2017, heavy rainfall inundated over 1,214 hectares [3,000 acres] of rice in Region 2 forcing farmers to incur additional costs to pump out excess water from their fields (iNews 2017a). In May 2017, poor dams attributed to heavy rainfall restricted farmers' access to their fields in Region 2 (iNews 2017b) while heavy rainfall threatened young rice plants in Regions 5 and 6 in June 2017 (iNews 2017c). The U.S. Department of Agriculture (USDA) Foreign Agricultural Service reported that heavy rainfall slowed sowing and field activities in Regions 2, 3, and 5 in January 2017 and dry condition in October 2017 accelerated plan maturation leading to lower yields (USDA-FAS 2018).

El Niño conditions in 2015-2016 resulted in rice production declining by approximately 13 percent (Fanfai 2016). Young rice plants on the Essequibo coast [Region 2] were affected by flooding as a result of persistent rains, poor drainage, and non-functioning sluices and pumps in at the start of the spring crop in December 2016 (SN 2016). In May 2015, 404 hectares [1,000 acres] of rice between Queenstown and Devonshire Castle in Region 2 were threatened by flooding attributed to heavy rainfall which was exacerbated by a non-functioning pump (SN 2015).

Reported incidences of insects and pests have also appeared in the local media. In March 2018, rice farmers on the Corentyne [Region 6] reported that their fields were being attacked by worms and flies (iNews 2018). In November 2018, farmers in Region 2 that planted late in the autumn season were severely affected by paddy bug infestations (Khan 2018). The late planting is usually due to heavy rains that delayed harvesting in the previous season and/or the lack of water at the beginning of the current season. As a result, farmers were forced to burn 404 hectares [1,000 acres] of rice in the region because of the paddy bug infestation (KN 2018b). Confirmed reports of paddy bug infestation in Region 2 were also reported in 2015 (Khan 2015). Despite such evidence, empirical research in this area is significantly lacking. As a result, information on how farmers perceive, are impacted by, and respond to climate change remains largely unknown.

1.1.4 Importance of Rice Production to Guyana

Rice (Oryza sativa L.) is important to Guyana because it is a major source of nutrition and rural livelihood. Rice is the main staple in the diet of most Guyanese, with consumption of approximately 50 kg per capita annually (ECLAC 2011). The industry supports about 6,300

farmers and their families directly (GRDB 2018) and 150,000 people indirectly (Ragnauth et al. 2014). The rice sector is the largest user of agricultural land with approximately 87,400 hectares [215,970 acres] under cultivation (GRDB 2018) making it a major source of income and employment for Guyanese. Rice production is also the primary agricultural sub-sector. In 2017, rice contributed 3.4 percent to the gross domestic product (GDP), 20.7 percent to agriculture GDP, and 33.7 percent to crop agricultural GDP (BoG 2017). In 2017, approximately 86 percent of the rice produced in Guyana was exported thus contributing about 14 percent to total export earnings (BoG 2017).

1.1.5 Small Farmers

Morton (2007) suggests that there is no commonly accepted definition of small farmers. One definition is that they are subsistence farmers, consuming the majority of their output within the household (Barnett, Blas, and Whiteside 1996). The scale of production is another often used criteria to identify small farmers. For example, Lowder, Skoet, and Raney (2016) used a threshold of less than 2 hectares to estimate that there are 475 million small farmers in the world. It should be noted that the use of production scale in terms of farm size or farm income is subjective to the national circumstance; small farmers in developed countries are very different from small farmers in developing countries (Morton 2007). Additionally, the size of farms and circumstances of small farmers are likely different among developing countries.

In Guyana, small rice farmers are not considered subsistence farmers since they sell their entire harvest on the local market (to rice millers). As such, a production scale approach is used to define small farmers in the Guyana context. For the purpose of this research, a small farmer is defined as someone who cultivates 4.45 hectares (11 acres) or less in rice each season. There are approximately 2,714 (42.4% of all rice farmers) small farmers cultivating 6,205 hectares [15,335 acres] in Guyana (GRDB 2017). The average farm size is 2.29 hectares [5.65 acres]. Although small farmers are fully integrated with the local markets, several characteristics distinguish them from large farmers. A great majority of small farmers have a secondary source of income from either off-farm employment or receive a government pension. Off-farm employment sources of income may include agricultural labor with large farmers or non-agricultural employment (e.g., taxi driver).

Farmers engaged in small-scale production in Guyana are also historically disadvantaged in terms of access to credit. Many small farmers do not possess the collateral or production scale needed to secure credit from lending institutions. In instances where they are able to obtain credit from millers, it is often at exorbitant interest rates and/or unfavorable terms. For example, farmers that secure fertilizer credit from a rice miller are contractually required to sell their grains to that miller in addition to paying a fee (usually a percentage) for the fertilizer credit received. While some small farmers may own a tractor and implements, none of them own a combine harvester. As such, small farmers are perpetually at the mercy of large farmers when it comes to timely land preparation and harvesting of their grains. As a result, they are exposed to climate variability due to out of season sowing and harvesting.

1.2 Brief History of the Rice Industry in Guyana

In 2008, Guyana celebrated 100 years as a rice exporter. However, the history of rice cultivation dates back to the days of slavery under Dutch, French, and British rule. While it is unknown when rice was first grown in Guyana, historical records suggest it was introduced in the mid-18th century by the Dutch as a source of food for enslaved Africans (Madramootoo 1973).

According to McGowan (2008), enslaved Africans working on the sugar plantations first planted rice to supplement inadequate food rations while runaway slaves planted rice at their hinterland settlements. The abolition of slavery in 1834 saw a decline in rice cultivation by Africans but the arrival of indentured servants from India beginning in 1838 precipitated increased demand which was met by imports from India (McGowan 2008).

Increased imports from India gave the impetus to produce rice locally. However, the lack of capital, knowledge, and irrigation resulted in several unsuccessful attempts to cultivate rice to satisfy local consumption (Homenauth 1997; McGowan 2008;). Starting in the early 1860s, East Indians began to successfully engage in subsistence rice farming and cultivation expanded throughout the colony in subsequent years, albeit in a haphazard manner (McGowan 2008).

The discontinuation of re-indentureship (renewal of contracts) in 1873 paved the way for East Indians to leave the sugar plantations and engage in small-scale rice cultivation (McGowan 2008). Beginning in the early 1880s, economic depression in the once dominant sugar industry led to agricultural diversification with land, labor, and capital diverted to rice cultivation (Homenauth 1997; McGowan 2008c). It is at this point that the rice industry in what was then British Guiana began to grow reaching unprecedented heights in the decades to come.

Between the 1880s and 1920s, the land area under rice cultivation increased rapidly while rice imports decreased. In 1893 about 1,011 hectares [2,500 acres] were under cultivation but this number grew to 24,776 hectares [61,222 acres] by 1919 (GRDB 2008). During this period, rice cultivation was stimulated by the availability of cheap land in lieu of costly repatriation of East Indians, local shortages of ground provisions (e.g., cassava, eddo³, and yams), production shortfalls in India, and high importation costs of wheat flour (McGowan 2008b). Above all, it was

³ Taro root.

the persistence and diligence of East Indians that defied the constant threat of malaria to engage in rice production (McGowan 2008b). Thus, Homenauth (1997) credits the success of rice cultivation in Guyana to the arrival of East Indians who brought with them indigenous knowledge from India.

For much of the first half of the 20th century, land preparation, harvesting, threshing (separating grains from the stalk), and transportation were done independent of modern technology and cultivation was almost entirely under the control of small farmers (GRDB 2008). In 1946, two important laws were introduced to further develop the industry. First, the British Guiana Rice Marketing Board was created to oversee the purchase and sale of rice paddy. Second, the Rice Producer Association (RPA) was established to protect, promote, and advance the interest of rice farmers (GRDB 2008).

The post-World War II years saw the introduction of new techniques, varieties, and mechanization which led to rice exports almost doubling between 1939 and 1957 (GRDB 2008). Starting in 1957, several land development schemes were established. These include the Black Bush Polder scheme [Region 6] which opened-up 12,545 hectares [31,000 acres] of land and the Tapakuma [Region 2] and Boerasirie Extension Project [Region 3] which helped boost drainage and irrigation and access to additional land (GRDB 2008).

In 1958, there were approximately 27,000 small farmers whose fields ranged from 0.8 hectare [2 acres] to 6.1 hectares [15 acres] and less than 300 farmers held more than 40.5 hectares [100 acres] (GRDB 2008). Additional mechanization in the 1960s and 1970s transformed the industry at the expense of small farmers which began to decline sharply (GRDB 2008). In addition, the establishment of the Mahaica, Mahaicony, Abary Agricultural Development Authority (MMA-ADA) in 1977 made available another 14,163 hectares [35,000 acres] of land (GRDB 2008). However, the prolonged economic troubles that began in the late 1970s would have far-reaching

implications for the industry and the country as a whole. Production declined from an average of approximately 150,000 tons in the 1980s to 93,400 tons in 1990 but rebounded to 151,000 tons in 1991 (GRDB 2008).

Until 1995, the rice industry was overseen by three state entities: the Guyana Rice Export Board (GREB), Guyana Rice Milling and Marketing Authority (GRMMA), and the National Padi and Rice Grading Centre (NPRGC). To streamlined operations and improve efficiency, these three entities were replaced with the creation of Guyana Rice Development Board (GRDB) in 1995 (GRDB 2008). The primary objectives of the GRDB are to develop the industry and expand exports, conduct research and disseminate information through extension services, and other promotional and development activities mandated by the Board.

With the establishment of the GRDB, the last decade of the 20th century saw a steady increase in rice production reaching 365,500 tons in 1999 before declining to 291,800 tons in 2000 due to political instability, debt repayment problems, and adverse weather conditions (GRDB 2008). The first decade of the 21st century saw fluctuations in rice production due in part to several floods and droughts, including a devastating flood in 2005. Despite these setbacks, rice production increased steadily starting in 2007. This was due in part to the introduction of several high yielding varieties since 2005. In 2017, aggregate production was 630,104 metric tons of rice, earning approximately US \$201M in revenues (BoG 2017). Figure 1.3 illustrates aggregate rice production for the period 1970-2017.



Figure 1.3 Aggregate rice production for Guyana (1970-2017)

1.3 Contemporary Rice Production in Guyana

Today, rice is still cultivated under irrigation; rainwater stored in the many canals and water conservancies that crisscross the agricultural belt is released into irrigation canals⁴. However, water levels are highly susceptible to variation in rainfall patterns and timing. There are two seasons per year: spring and autumn. For the spring season, sowing is done primarily between December and January and harvesting between March and May. For the autumn season, sowing is done primarily between May and June and harvesting between September and November.

Rice is grown under a highly mechanized system where tractors and combines are used for land preparation (plowing, harrowing, and puddling) and harvesting, respectively. The crop is directly seeded using pre-germinated seeds sown onto lightly flooded fields. On small farms, sowing, fertilizer application, and the control of insects and pests, diseases, and weeds are strictly

⁴ The islands of Hogg, Leguan, and Wakenaam in Region 3 rely exclusively on rainfall since there is no water conservancy on the islands while in some areas in other regions farmers depend on nearby rivers in the absence of irrigation systems.

done by physical labor⁵ while on some large farms, aircraft are used for sowing, applying fertilizer, and controlling insects and pests, diseases, and weeds.

Harvested rice paddies are usually transported in bulk to the mills. The processing of paddy into rice is done at rice mills located throughout Guyana's rice belt. Smaller mills produce rice mainly for domestic consumption while larger mills concentrate mainly on production for export. Export markets include the Caribbean, Central and South America, and the Europe Union (GRDB 2015a). Since 1997, 15 new rice varieties have been released by the GRDB Rice Research Station (GRDB 2015b). In April 2018, GRDB 15 became the latest variety to be released. Among other characteristics, it promises higher yields and the ability to withstand lodging. However, such assurances can be easily offset by poor infrastructure exacerbated by unfavorable weather conditions attributed to climate change.

1.4 Other Challenges Facing the Rice Industry in Guyana

Apart from the vagaries of weather, the rice industry in Guyana is plagued with several other challenges that compound the impacts of increased climate variability on farmers in general and small farmers in particular. The following briefly describes several of these challenges reported in the media and by key informants interviewed as part of this research.

 High cost of production – The high cost of inputs such as fertilizer, chemicals, and fuel are further exacerbated by the value-added tax (VAT) farmers are required to pay on inputs. In addition, import duties on equipment and machinery are usually passed on to small farmers that rely on large farmers to prepare their land and harvest their grains. When the costs of climate change adaptation are factored, profit margins are usually thin.

⁵ Laborers apply fertilizer by walking through the field and throwing by hand and use a spray can or motor blower to spray for insects and pests, diseases, and weeds.

- 2. Unstable prices The price received by farmers is dictated by the free market system. In the absence of a minimum established price, farmers are price takers who are at the mercy of the market. In addition, the loss of the lucrative Venezuelan market in 2015 has forced millers and by extension, farmers to accept lower prices from markets elsewhere.
- 3. Poor infrastructure In recent years, the drainage system has been in deplorable condition due to poor maintenance by the government. This has worsened the impacts of excess rainfall which often leads to flooding. Untimely release of irrigation water by some Water User Associations also affects planting in the season. In addition, farm-to-market roads are often impassable due to the lack of maintenance and repairs. This causes delays in accessing the fields and/or transporting harvested grains.
- 4. Generic weather forecast Although weather forecast information is readily available, it lacks specificity as it relates to rainfall patterns across the agricultural belt. Hence, the lack of details limits farmers' ability to consider such information when making farm management decisions.
- 5. Late payments Farmers in some areas of the country often face significant delays in receiving their payments which is in contravention of the Rice Factories Act of 1998. The law requires rice millers to pay 50 percent of the amount due to farmers within two weeks of receipt of the grains and the remainder within 42 days. Significant delays of payments deny farmers vital working capital needed to replant in the subsequent season. Furthermore, late payments are not accompanied by interest on the outstanding amounts owed to farmers.
- 6. *Rent and service charge* In 2017, the government raised the rental and drainage and irrigation charges for lease land to in excess of 600 percent for some charges. This is a
significant burden on rice farmers already struggling to cope with low prices and high cost of production.

 Roaming cattle – In some areas, unattended cattle cause significant damage to rice fields. Cattle usually enter the rice fields at night and consume the plants thus leaving a trail of destruction in the process.

1.5 Research Questions

To the researcher's knowledge, there has been no previous study on the impacts of climate change on rice production for Guyana. As such, this research is both novel and timely given the reports in the media regarding changing climatic conditions and insect infestation problems farmers are facing. In addition, the socioeconomic implications for small farmers, the rice industry, and the country as a whole compels a closer examination of this problem. Therefore, the objective of this study is to examine the impacts of climate change on rice production in Guyana and to assess how small farmers are adapting. To achieve this objective, the following research questions are proposed and addressed by this research:

- 1. How has the climate in Guyana changed?
- 2. What non-climatic factors influence rice yields of small farmers?
- 3. How do small rice farmers and key informants perceive changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?
- 4. What impacts are farmers and key informants seeing due to the observed changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?
- 5. What adaptive measures are they adopting in response to these impacts?

1.6 Contributions and Anticipated Impact

Given the lack of empirical evidence on the impacts of climate change on rice production in Guyana, this study will enrich our knowledge and understanding of how farmers in general and small farmers specifically perceive, are impacted by, and respond to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. Since most studies are predominantly quantitative in nature, another contribution of this study is the use of qualitative methods to explore cognitive dimensions of and adaptation to climate change.

This study also has broader implications for regions of the world where small farmers that are important to agricultural production may be facing similar challenges. Since little is known about how small rice farmers in Latin America and the Caribbean (LAC) region respond and adapt to climate change, this study will contribute to that literature. This research will also contribute to the broader body of knowledge of how small farmers in the global south perceive, are impacted by, and respond to climate change. The findings will highlight novel strategies and best management practices that can be used to develop appropriate adaptation measures and institutional responses based on local conditions and needs.

1.7 Study Overview

Including this initial chapter, this study comprises of seven chapters. Chapter 2 provides an overview of the relevant literature on climate change and crop agriculture. It includes a discussion on the interaction between climate change and crop agriculture, the common methods used to study the relationship, and a summary of studies on climate change impacts from across the world. Chapter 3 describes the data source and collection, and methods used in the analyses. Chapter 4 examines the changes in rainfall, minimum and maximum temperature over the last 115 years for Guyana. Chapter 5 explores the effects of perceived changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds together with socio-economic, farm-level, and institutional characteristics on rice yields at the farm-level. Chapter 6 documents small farmers' and key informants' perceptions of changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds and the impacts and responses to the observed changes. This chapter also presents the adaptation practices implemented and small farmers' ability to pay for these changes in farming practices. A summary of the service and support provided to farmers by district rice extension officers is also discussed. Chapter 7 summarizes the key findings and explores the overall policy implications of this study. Opportunities for future research are subsequently identified.

Chapter 2 Review of Literature: Climate Change and Crop Agriculture

2.1 Introduction

Agricultural production is vulnerable to a changing climate. As such, the purpose of this chapter is to provide an overview of the relevant literature at the climate change – crop agriculture nexus. Section 2.2 provides multiple pieces of evidence of global climate change. The relationship between crop agriculture and climate change is presented in section 2.3 while specific links between climate change and rice are explored in section 2.4. Section 2.5 describes the different approaches to estimating climate change impacts on crop agriculture, while section 2.6 discusses the sources and types of climate data used in climate impact studies. Section 2.7 summarizes empirical evidence of the impacts of climate change on crop agriculture from three perspectives: globally, developed countries, and developing countries. Empirical evidence of farmers' perceptions and adaptation to climate change is presented in section 2.8. Section 2.9 presents the conceptual framework while the gaps in the existing literature are outlined in section 2.10. Section 2.11 provides a summary of key points covered in this chapter.

2.2 Evidence of Global Climate Change

Observed changes attributed to climate change are prominent across the world. Signs of climate change include soaring temperatures, shifting rainfall patterns, more extreme weather events, warming, rising, and more acidic oceans, declining Arctic Sea ice, shrinking ice sheets, and retreating glaciers. These and many other changes provide ample evidence of a changing climate. Perhaps more importantly, many of these changes are linked to the increasing levels of greenhouse gases⁶ in our atmosphere, attributed to human activities.

⁶ Gases in Earth's atmosphere that absorb and emit radiation thus causing a warming effect on Earth's surface (Mann and Kump 2015). Water vapor (H₂O), carbon dioxide (CO₂), nitrous oxide (N₂O), methane (CH₄) and ozone (O₃) are the primary greenhouse gases in the Earth's atmosphere (IPCC 2012).

2.2.1 Greenhouse Gases and Climate Change

Earth's climate is driven by the amount of solar energy that is absorbed at the surface. Increased concentrations of greenhouse gases, primarily carbon dioxide, methane, and nitrous oxide in the atmosphere absorb the outgoing energy radiated by the surface, resulting in Earth's atmosphere becoming warmer (Bast 2013). In 2017, the average global carbon dioxide concentration was approximately 405 parts per million (ppm) (Dlugokencky et al. 2018). Figure 2.1 illustrates the concentration of carbon dioxide in the atmosphere over time while Figure 2.2 shows the global emissions of carbon dioxide, methane, nitrous oxide, and several fluorinated gases from 1990 to 2010.



Figure 2.1 Global atmospheric concentrations of carbon dioxide over time **Source:** EPA (2016b)



Figure 2.2 Global greenhouse gas emissions by gas (1990-2010) Source: EPA (2016c)

2.2.2 Global Land Surface Temperature

The global land surface temperature encompasses measures of average near-surface air temperature over land and sea-ice (Mach, Planton, and von Stechow 2014). According to the Intergovernmental Panel on Climate Change (IPCC) Fifth Assessment Report, each of the last three decades has been successively warmer at the Earth's surface than any preceding decade since 1850 (IPCC 2014b). Ten of the warmest years on record have happened since 1998 with the four warmest years occurring since 2014 (Sánchez-Lugo et al. 2018). There is also high confidence that human-influenced global warming reached approximately 1°C above pre-industrial levels in 2017 (IPCC 2018). Using 1901-2000 average as a baseline, Figure 2.3 shows the change in global annual average temperatures.



Figure 2.3 Temperature worldwide (1901-2015) Source: (EPA 2016g)

2.2.3 Global Sea Surface Temperature

Sea surface temperature refers to the temperature of the water just below the surface of the ocean as measured by ships, buoys, and drifters (Mach, Planton, and von Stechow 2014). Although there was a slight decline in the global sea surface temperature in 2017, the long-term warming trend continues to point upwards (Huang et al. 2018). With the average of 1971-2000 as a baseline, Figure 2.4 depicts the changes in the average surface temperature of the world's oceans.



Figure 2.4 Average global sea surface temperature (1880–2015) Source: EPA (2016c)

2.2.4 Global Precipitation

Changes in precipitation patterns are also evident with more frequent heavy precipitation events (USGCRP 2018). For example, in 2017 precipitation over global land areas exceeded longterm averages, with some areas wetter than others (Vose et al. 2018). Precipitation deficits on the other hand saw a sharp decline in early 2017, then rising above average later in the year (Osborn et al. 2018). Using the 1901-2000 average as a baseline, Figure 2.5 illustrates the global total annual amount of precipitation over land.



Source: EPA (2016f)

2.2.5 Global Mean Sea Level (GMSL)

Rising seas are another result of a warming world. In 2017, global mean sea level (GMSL) was approximately three inches higher than it was in 1993⁷, rising at an average rate of 1.2 inches per decade (Thompson et al. 2018). Based on long-term tidal gauge measurements and satellite altimetry, Figure 2.6 shows the cumulative changes in sea level for the world's oceans.

⁷ Satellite record began in 1993.



Figure 2.6 Global average absolute sea level change (1880–2015) Source: EPA (2016d)

2.2.6 Artic Sea Ice

The extent of Arctic Sea ice varies considerably across seasons with minimum and maximum extent typically occurring in September and March, respectively (Meier et al. 2014). In March 2017, the maximum extent of Arctic Sea ice was the lowest ever recorded while in September 2017, the minimum extent of Arctic Seas ice was 25 percent below the long-term average (Perovich et al. 2018). Figure 2.7 shows the extent of Arctic Sea ice for the months of September and March from 1979 to 2016. Figure 2.8 captures the dwindling Arctic Sea ice between September 1979 and September 2015.



Figure 2.7 March and September monthly average Arctic Sea ice extent (1979–2016) Source: EPA (2016a)



Figure 2.8 Dwindling Arctic Sea ice Source: EPA (2016a)

2.3 Overview of Climate Change and Crop Agriculture

The agricultural sector is perhaps the most vulnerable to a constantly changing climate. This is because climate is a key determinant of agricultural output (Adams et al. 1998), given that outdoor production practices depend heavily on particular levels of temperature and precipitation (Ackerman and Stanton 2013). A changing climate suggests changing conditions for agriculture, including shifts in growing seasons, seasonal temperatures, rainfall timing, quantity, intensity, and distribution, and extreme weather events (Rose 2015). Hence, climate change impact crop yields in a myriad of ways including temperature extremes, water availability and usage, and soil health. Waggoner (1983) noted that plant systems and hence, crop yields, are influenced by many environmental factors such as moisture and temperature that may act either synergistically or antagonistically with other factors in determining yields.

A quarter of a century ago, Mendelsohn, Nordhaus, and Shaw (1994) concluded that warmer weather was expected to bring net benefits to global agriculture. Today, rising temperatures, variable rainfall, and extreme weather events (e.g., flood and drought), are expected to increasingly disrupt agricultural production. Although climate change can have both positive and negative effects on crop yields, in general, negative impacts have been more common (IPCC 2014a).

Temperature increases have been found to reduce yields and the quality of crops (Adams et al. 1998). Warming generally causes some plants that are below their optimum temperature to grow faster. However, for other plants, faster growth means there is less time to mature. Simpson (2016) notes that rising average temperatures cause stress leading to early maturity. This means that plants spend less time in each development stage including reproduction which causes yields to decline. Additionally, increased nighttime temperature leads to respiration that consumes larger amounts of plants' energy created during the day resulting in less energy available for producing grains; heat waves cause pollen to become sterile during flowering, which means no grain is produced (Simpson 2016).

Temperature increase, particularly in regions where agricultural production is currently limited by a colder climate, extends the growing season available for plants and reduces the growing period required by crops to mature. However, changes in the frequency of extreme climatic events can also be damaging and costly for agriculture (Parry 1990). Extreme weather events that include heat waves, droughts, strong winds, and heavy rainfall can lead to crop failures, soil erosion, flooding, and the occurrence of pest and diseases (Motha 2011).

While plants exposed to increased levels of carbon dioxide (CO₂) are expected to benefit from carbon fertilization, the extent and level of fertilization depend on the species of plant. Increased atmospheric CO₂ concentrations are vital to photosynthesis and can have a direct effect on the growth rate of crop plants and weeds (Parry 1990). Higher concentrations of CO₂

would also result in reduced transpiration (i.e. water loss) as plants reduce their stomatal apertures (Adams et al. 1998). Despite these benefits, elevated CO₂ levels also suggest the potential for increased weed pressure. Weeds are likely to become more prolific and are expected to invade new habitats as global warming increases. As a result, overall yields are likely to decline for crops that compete for vital nutrients with an increased weed population. Additionally, carbon fertilization may interact with other environmental influences. For instance, the interaction between atmospheric CO₂ and tropospheric (ground level) ozone may reduce yields of many plants.

Climate change also causes variation in precipitation in terms of the timing, quantity, intensity, and distribution (Simpson 2016). No rainfall causes water deficits, which reduces soil

moisture. This may cause farmers to abandon planting the crop altogether, crop failure, or low yields due to water deficiency. Heavy downpours may lead to flooding which can result in crop failure. Additionally, flooding may result in soil loss and/or degradation, which reduce soil fertility. Less fertile soil is likely to result in lower yields. Rainfall that occurs too early may disrupt harvesting and/or lead to unavailability of water at the start of the planting season. Rainfall that arrives late may lead farmers to abandon the crop altogether. Figure 2.9 illustrates the impacts of climate change on the agricultural sector.



Figure 2.9 Impacts of climate change on the agricultural sector Source: Author (as adopted from (Kim, n.d.).

2.4 Climate Change and Rice Cultivation

Although the global impact of climate change on crop production is expected to be negative, impacts vary across crops and geographic areas (Zhao, Liu, et al. 2017). Rainfall, temperature, atmospheric carbon dioxide (CO₂), and solar radiation play key roles in rice production; the physiological stages of plant growth, development, and grain formation are directly influenced by these variables (Haque, Ali, and Masum 2016).

2.4.1 Temperature

Rice is vulnerable to extremely low and high temperatures. For each growth stage, critical low and high temperatures (typically below 20°C and above 30°C) differs and also depends on the variety, duration, and daily cycle (Yoshida 1981). Table 2.1 presents the rice plant response to varying daily mean temperature at different growth stages.

Critical Temperature (°C)		
Low	High	Optimum
10	45	20-35
12-13	35	25-30
16	35	25-28
7-12	45	31
9-16	33	25-31
15	-	22-23
15-20	38	-
22	35	33-30
12-18	30	20-25
	C Low 10 12-13 16 7-12 9-16 15 15-20 22 12-18	Low High 10 45 12-13 35 16 35 7-12 45 9-16 33 15 - 15-20 38 22 35 12-18 30

 Table 2.1 Rice Plant Response to Varying Daily Mean Temperature at Different Growth Stages

Source: Yoshida (1981)

As Table 2.1 suggests, rice can tolerate maximum temperatures of 45 °C and minimum temperatures of 7 °C. However, depending on the growth stage and geographical location, temperatures outside the optimum range can have negative effects on yields. Flowering and ripening stages are highly sensitive to changes in temperature (Sánchez, Rasmussen, and Porter

2013). During the flowering stage, temperatures greater than 35 °C generally cause sterility (Haque, Ali, and Masum 2016) which results in no grains being produced. However, Yoshida (1981) notes that temperatures have to exceed 40 °C for this to become noticeable. Elevated nighttime temperatures reduce grain filling and yield considerably (Peng et al. 2004), increased chalkiness reduce milling quality (Lanning et al. 2011) and can cause increase sterility in rice spikelet (flower) (Reiner Wassmann and Dobermann 2007). Other studies reported a decline in grain quality due to higher temperatures (Counce et al., 2005, Zhong et al., 2005, Tanaka et al., 2009). Higher transpiration losses linked to higher temperatures make rice less productive and increase the water requirements of rice.

2.4.2 Rainfall

The water requirements of low land rice are significant and vary across each growth stage. As such, variability in rainfall in terms of timing, quantity, intensity, and distribution affects plant growth and development which impacts the quality and quantity of yields. Inadequate rainfall causes moisture stress which hampers root and shoot development, delays flowering, and affect pollination, fertilization, and grain filling; excessive rainfall disrupts farming operations such as land preparation, sowing, harvesting, threshing and processing, and seed drying; incessant rainfall during flowering impedes fertilization and grain formation (Basnayake et al. 2006). Rainfall variability that occurs during sowing affects seed establishment and the duration of the crop (Haque, Ali, and Masum 2016). Untimely rainfall may also bring a premature end to the reproductive or ripening stage which significantly reduces yield (Moomaw and Vergara 1965).

2.4.3 Solar Radiation

Solar radiation refers to energy provided by the sun to facilitate plant growth and development. The sun provides the energy needed for seed germination and the physiological development of plant leaves, stems, and shoots (Khadka 2016). Like rainfall and temperature, the intensity level of solar radiation required by each growth stage is different. Low-intensity solar radiation has a profound impact during the reproductive and ripening stages (Basnayake et al. 2006). Under a changing climate, shifting rainfall patterns will disrupt the trade-off between sunshine and rainfall through increase cloud cover. This will likely affect the intensity level of solar radiation needed for photosynthesis.

2.4.4 Carbon Dioxide

Increases in atmospheric carbon dioxide also affect the quality and quantity of rice production. Higher carbon dioxide levels typically increase biomass production in terms of leaf number, shoot dry weight, and shoot length but not grain yield. (IRRI n.d.). Zhu et al. (2018) stated that elevated atmospheric carbon dioxide will decrease proteins, micronutrients, and vitamins found in rice.

2.4.5 Extreme Weather

Erratic rainfall patterns produce extreme events that lead to flood and water scarcity. Excess rainfall usually causes rice fields to become inundated. Although rice thrives in wet conditions where other crops fail, it cannot survive if submerged under water for an extended period of time. Additionally, floods cause indirect damage to rice production by harming the properties and production means of farmers and infrastructures such as farm-to-market roads, and drainage and irrigation systems that support rice production. Wet conditions arising from heavy downpours increase the moisture content of rice paddy which impacts grain quality at harvest. Auffhammer, Ramanathan, and Vincent (2012) found that drought and extreme rainfall negatively affected rice yield (harvest per hectare) in predominantly rain-fed areas of India.

Prolonged dry spells result in water scarcity. Since rice requires abundant water to grow, drought-like conditions significantly reduce rice yields. Average yield reduction in rain-fed, drought-prone areas has ranged from 17 to 40% in severe drought years, leading to production losses and food scarcity (IRRI n.d.). The intensity and frequency of droughts result in water deficits that are estimated to affect more than 23 million hectares of rain-fed rice production areas in South and Southeast Asia; in Africa, recurring drought affects nearly 80% of the potential 20 million hectares of rain-fed lowland rice (IRRI n.d.). Koc and Ceylan (2013) found that drought had a negative effect on rice production in Turkey.

2.4.6 Pests, Diseases, and Weeds

Climate change also indirectly affects rice yields by altering the types, frequency, and magnitude of pests, diseases, and weeds. Basnayake et al. (2006) note that excess rainfall supports an increased incidence of pests and diseases which reduces yields. Surveys conducted by the International Rice Research Institute (IRRI) found that extreme weather events have led to significant increases in the rodent population as a result of unseasonal and asynchronous cropping; weed infestation and rice-weed completion represents a major challenge for sustainable rice production (IRRI n.d.). Water shortages, irregular rainfall patterns, and related water stresses increased the intensity of some diseases, including brown spot and blast (IRRI n.d.).

2.4.7 Climate Change and Future Rice Production

Projections show that the proportion of global rice producing areas exposed to critically high temperatures is expected to grow from 8.0% in the 2000s to 27.0% by mid-century (Gourdji, Sibley, and Lobell 2013). According to the International Food Policy Research Institute (IFPRI) Food Policy Report, global losses in rice yield attributed to climate change could be between 10 and 15 percent by 2050 (Nelson et al. 2009). Most studies have focused on temperature increases. Excluding CO₂ fertilization, successful adaptation, and genetic improvement, four independent estimates found that rice yields will decrease by 3.2% for every one degree Celsius increase in global temperature (Zhao et al. 2017). Field experiments by Zhao, Shilong, et al. (2017) noted appreciable losses under warming conditions with yields decreasing by 5.2% on average.

2.4.8 Rice Cultivation Impact on Climate Change

While the literature review in the preceding sub-sections focused on the impacts of climate change on rice production, it would be remiss if the impacts of rice production on climate change are overlooked. Since rice is grown under lightly flooded conditions, rice cultivation increases methane (CH₄) and nitrous oxide (N₂O), two very powerful greenhouse gases that contribute to climate change. According to Kritee et al. (2018), global rice cultivation accounts for 50.0% of all crop related methane (CH₄) emissions. Although the atmospheric life of methane is about 12 years, it is 86 times more potent than carbon dioxide (CO₂) (Mann and Kump 2015). Increases in nitrous oxide occur in periods between aeration and flooding of rice fields. Kritee et al. (2018) found high fluctuations of nitrous oxide at average and intense-intermittently flooded rice farms. In comparison to carbon dioxide, the atmospheric life expectancy of nitrous oxide is approximately 121 years and 268 times more powerful (Mann and Kump 2015).

2.5 Approaches to Assessing Climate Change Impacts on Crop Agriculture

The literature cites two broad approaches for estimating the impacts of climate change on agriculture: general equilibrium and partial equilibrium models (Zhai, Lin, and Byambadory 2009; Sarker 2012). The discussion that follows describes each approach including their strengths and weaknesses. While it is important to highlight these approaches, this study ultimately assumes a more descriptive approach to help understand the impacts of climate change on rice production in Guyana, and how small farmers are and may adapt in the future.

2.5.1 General Equilibrium Models

General equilibrium models view the economy as a complete set of interconnected markets where economic shock in one market triggers changes in other markets (Nicholson and Snyder 2012). The computable general equilibrium model (CGE)⁸ is the most common economy-wide model (Sarker 2012) since it uses real economic data to evaluate how changes in a stimulus (e.g. policy) reverberate across the economy.

2.5.1.1 Computable General Equilibrium Models

The CGE model uses a system of equations to describe the entire economy and capture the interactions among all sectors (Burfisher 2016). Since the impacts of climate change are likely to be felt across various segments of the economy, CGE is a useful tool for assessing climate shocks (Zhai, Lin, and Byambadory 2009). The major advantages of CGE models include: conceptually sound and computationally consistent, accounts for inter-industry linkages and economy-wide effects; and sound welfare analysis (Hertel 1990, 1999). However, CGE models are not without

⁸ Also referred to as applied general equilibrium (AGE).

limitations. Common critiques include issues relating to parameter selection, model structure, and functional form (Mckitrick 1998). In addition, there are concerns about data consistency due to calibration problems; lack of statistical tests for model specification; and the complexity and advanced skills required to set up and use CGE models (Gillig and McCarl 2002).

2.5.2 Partial Equilibrium Models

Partial equilibrium models examine a single market or part of an economy in isolation while ignoring interrelationships with other markets or sectors within the economy (Nicholson and Snyder 2012). Partial equilibrium models encompass four primary approaches: agronomic crop models, panel weather studies, and Ricardian cross-sectional climate studies (Blanc and Reilly 2017) and Agroecological Zone Analysis (AEZ) (FAO 1978).

2.5.2.1 Agroeconomic Analysis

This hybrid approach combines crop simulation models⁹ with economic models to estimate crop yields. Simulation models express crop yields as a function of different weather conditions, soil conditions, and crop management practices (e.g., planting dates, fertilization rates, and irrigation use). As such, crops are grown in a laboratory setting that simulates how crop yields respond to different levels of temperature, light, water, nutrients, and carbon dioxide (CO₂) (Zhai, Lin, and Byambadory 2009; Blanc and Reilly 2017; Van Passel, Massetti, and Mendelsohn 2017). By exposing crops to different environmental conditions, crop growth and yield can be isolated based on the treatment received. The results of the crop simulation model are subsequently

⁹ Also called biophysical models.

integrated into an economic model that incorporates information on agricultural production, consumption, and trade (Blanc and Reilly 2017).

The parameters of crop models are adjusted through iterative experiments and comparison with the results from field trials (Li et al. 2012). There are many global and regional crop models currently in existence. Prominent examples include the Agricultural Production Systems Simulator (APSIM), Decision Support System for Agrotechnology Transfer / Crop Estimation through Resource and Environment Synthesis (DSSAT-CERES), Model for Nitrogen and Carbon in Agroecosystems (MONICA), crop-water productivity model (AQUACROP), and CROPSYST. Several recent studies have used an ensamble of crop models to explore crop yeild responses to changes in climate. These include for wheat (Pirttioja et al. 2015), maize (Bassu et al. 2014; Araya et al. 2015; Durand et al. 2018), rice (Li et al. 2015), potato (Fleisher et al. 2017), sugarcane (Dias and Sentelhas 2017), and soybean (Battisti, Sentelhas, and Boote 2018).

Simulation models can explore the impacts of future climatic conditions and, assuming the underlying economic models are accurate, can predict efficient adaptation (Mendelsohn and Massetti 2017). However, such models may not accurately reflect changes in demand and supply due to climate change (Mendelsohn and Massetti 2017). Considering the complexity of natural and social systems, simulation models may not accurately account for all interactions (Rauff and Bello 2015; Kant et al. 2017).

2.5.2.2 Ricardian Cross-sectional Analysis

The Ricardian cross-sectional approach draws on David Ricardo's rent theory. Ricardo $(1817)^{10}$ defined rent as the "portion of the produce of the earth, which is paid to a landlord on

¹⁰ Reprinted in 2004 by Dover Books.

account of the original and indestructible powers of the soil" and that the ownership of more productive land will have a higher rental value. As such, the Ricardian approach assumes a direct cause and effect relationship between climate change and farmland value. The value of the farmland is based on its productivity which depends on inherent characteristics such as climate. Therefore, changes in farmland values would be reflective of the economic impact of various climate induce conditions.

The Ricardian approach uses regression techniques to examine the relationship between various explanatory variables (climate, economic, demographic, and physical factors) and the independent variable (net revenues or land values) (Mendelsohn, Nordhaus, and Shaw 1994). A major advantage of the Ricardian cross-sectional studies is that adaptation is considered implicit and endogenous (Seo et al. 2012). In other words, farmers' long-term response to climate change is captured in the analysis. By treating net revenue (or land value) as the dependent variable, cross-sectional analyses also provide a direct measure of welfare (Mendelsohn and Massetti 2017).

The major critique of Ricardian cross-sectional studies is omitted variable bias (Deschênes and Greenstone 2007; Blanc and Schlenker 2017). Cross-sectional studies do not control for other independent variables that vary across space and are correlated with climate (e.g., topography, soil fertility) (Massetti and Mendelsohn 2018). These independent variables may not be available or difficult to collect. The Ricardian cross-sectional studies also assume prices and carbon dioxide concentrations do not vary across space (Mendelsohn and Massetti 2017). As a result, net revenues (or land value) may be under or over-estimated depending on the output price and level of carbon fertilization. Hence, prices and carbon dioxide concentrations are often introduced exogenously (Massetti and Mendelsohn 2018). Finally, cross-sectional studies assume that current adaptation will continue into the future but ignores the costs of transition (Kelly, Kolstad, and Mitchell 2005). Ricardian studies have been employed far and wide and on different geographic scales. For example, continental studies include Africa (Seo and Mendelsohn 2008c), South America (Seo and Mendelsohn 2008b), and Western Europe (Van Passel, Massetti, and Mendelsohn 2017). Examples of large country studies include Brazil (Mendelsohn, Basist, et al. 2007; Sanghi and Mendelsohn 2008), Canada (Reinsborough 2003; Mendelsohn and Reinsborough 2007), China (Wang et al. 2009; Chen et al. 2013), India (Mendelsohn, Kurukulasuriya, et al. 2007; Sanghi and Mendelsohn 2008), and the United States (Mendelsohn, Nordhaus, and Shaw 1994; Schlenker, Hanemann, and Fisher 2005; Massetti and Mendelsohn. 2011). Yet, other studies have been done for small countries or regions within (Gbetibouo and Hassan 2005; Kurukulasuriya and Ajwad 2007; Kabubo-Mariara and Karanja 2007; Deressa et al. 2009; De Salvo, Raffaelli, and Moser 2013; Wood and Mendelsohn 2014).

2.5.2.3 Panel Data Analysis

According to Houser et al. (2015), some impact assessments prefer to rely exclusively on panel data studies to measure the impacts of climate change on agricultural outcomes. This is driven by concerns over misspecification in terms of omitted variable bias inherent in the Ricardian cross-sectional approach (Deschênes and Greenstone 2007; Blanc and Schlenker 2017). Panel data analysis employs regression techniques to estimate the effect of climate change on crop agricultural using a production or profit function (Blanc and Reilly 2017). By using panel data, one can assess the vulnerability of farmers across different climates, space, and time.

Panel data models can be estimated using a fixed effects or random effects model (Baltagi 2008; Blanc and Schlenker 2017). A fixed effects model contains an unobserved variable in the error term that is constant over time (Wooldridge 2013). A fixed effect model is attractive because

it allows the use of panel data to measure causal relationships (Cameron and Trivedi 2005) by controlling for unobserved heterogeneity that is time invariant. In other words, determinants of the dependent variable that is constant over time but difficult or impossible to measure are absorbed into the model thus, easing concerns over omitted variable bias. For example, fixed effect models may control for unobserved effects that are location specific such as soil quality, fertilizer application, irrigation, and farm household characteristics and that do not change over time. It is assumed that the unobserved effect is correlated with one or more independent variables.

A random effects panel data model assumes no correlation between the unobserved effect and observed independent variables in each time period (Wooldridge 2013). If there is a correlation, random effects model produces inconsistent estimates of the parameters due to serial correlation (Baltagi 2008). Assuming that the unobserved effect is time-constant, fixed effects model are considered more consistent and thus preferred.

The main strengths of panel data analysis is that it overcomes omitted variable bias, uses uncorrelated exogenous weather shocks, accounts for short-term adaptation, has greater degrees of freedom, and can be used for forecasting and validation (Blanc and Schlenker 2017). However, the source of variation, time-varying omitted variables, measurement errors, homogenous seasonal weather, and long-term adaptation are major limitations (Blanc and Schlenker 2017).

2.5.2.4 Agroecological Zone Analysis (AEZ)

Climate, soil, topography, and land cover vary across different landscapes. As a result, some landscapes are more suitable for some crops than others and vice versa. Developed by the Food and Agricultural Organization (FAO) in the late 1970s (FAO 1978), the AEZ method combines crop simulation models with land management decisions to measure crop productivity

across different climatic zones (Darwin et al. 1995; Fischer et al. 2005). Under this approach, land units are mapped according to soil, landform, and climatic characteristics (FAO 1996). By classifying an otherwise complex landscape into smaller discrete homogeneous zones (Kurukulasuriya and Mendelsohn 2008b), AEZ encourages maximizing agricultural yields based on land use suitability and potential.

Similar to the agronomic approach, AEZ underscores the importance of the natural science interaction; unlike the agronomic approach, AEZ models the entire eco-physiological¹¹ process and when combined with soil, climate, and technology state, determines the most productive crop to produce in a given area or zone (Mendelsohn and Tiwari 2000). Since climate is a key input of the model, the impacts of changes in rainfall and temperature can be captured. A major drawback of this method is that all relevant components must be available before final outcomes can be predicted (Mendelsohn and Tiwari 2000).

2.6 Climate Data

The use of climate variables in agricultural economics analyses is not a new phenomenon (e.g., Fisher 1925; Wright 1928). While studies rely on rainfall and temperature data to identify weather shocks, the source and type of weather data used in assessing the economic impacts of climate change are important considerations. Weather data fall into four primary categories: ground station, gridded, satellite, and reanalysis data (Dell, Jones, and Olken 2014).

Ground or land-based station data are the most common type of weather data. It is based on observations of rainfall, temperature, dew point, relative humidity, wind speed and direction, atmospheric pressure, and other weather variables collected through instruments located across the

¹¹ Includes factors that influence plant growth (e.g., length of growing cycle, yield formation period, leaf area index, and harvest index).

world (Mendelsohn, Kurukulasuriya, et al. 2007). Assuming it exists, ground station data can be highly accurate for a specific location (Dell, Jones, and Olken 2014). However, in developing countries like Guyana, sparse station coverage and poor record keeping due to lack of capacity (Colston et al. 2018) can be of concern. Regardless, ground station data are still considered advantageous for locations near the station (Dell, Jones, and Olken 2014).

Concerns over incomplete coverage inherent in ground station data can be mitigated by gridded data. Through interpolation of station data, gridded data provides more comprehensive coverage of weather data (Dell, Jones, and Olken 2014). While gridded data adjust for issues related to missing station data, elevation, and urban heat island effect (Dell, Jones, and Olken 2014), it is not without limitations. Issues raised by Auffhammer et al. (2013) include course resolution, the spatial correlation of climate variables, correlation of weather variables, differences in cross-sectional versus time series variation, and endogeneity of weather coverage. Those from the Climate Research Unit (CRU) at East Anglia University and Matsuura and Willmott (2018) at the University of Delaware (UDEL) are two frequently used gridded datasets.

Satellite measurements, which began in the late 1970s are another source of weather data. While satellite offers a solution in areas with limited ground coverage, such data have lower resolution than gridded data and are usually less accurate since temperature and rainfall are not directly measured (Dell, Jones, and Olken 2014). Toté et al. (2015) found that satellite data overestimated low and underestimated high decadal rainfall. Examples of satellite products include those produced by the University of Alabama Huntsville (UAH) and Remote Sensing Systems (Dell, Jones, and Olken 2014).

Finally, reanalysis data entail combining information from multiple sources (e.g., ground station, satellites, weather balloons), to estimate weather variables across a grid (Dell, Jones, and

Olken 2014). It is different from gridded data in that it uses a climate model instead of interpolation. The chief limitation of this dataset is the oversimplification of reality embedded in climate models (Dell, Jones, and Olken 2014). Auffhammer et al. (2013) found that the correlation of reanalysis data for temperature and average rainfall was slightly below the CRU and UDEL gridded datasets. Examples of reanalysis products include those produced by the National Center for Environmental Prediction (NCEP) in the United States and the European Center for Medium-Range Weather Forecasting (Auffhammer et al. 2013).

2.7 Empirical Evidence of Climate Change Impacts on Crop Agriculture

Empirical studies on the impacts of climate change on agriculture are numerous. Some studies have examined climate change impacts on a global scale while others have looked at impacts across a specific continent or sub-continent. Other studies have focused on large industrialized countries while others have focused on small developing countries. The review of literature that follows highlights the findings of a sample of these studies.

2.7.1 Evidence from Global Agriculture

Early studies on the impacts of climate change on agriculture worldwide are overwhelmed with uncertainties. Parry (1990) evaluated the implications of climate change on global agriculture under existing production management and technology and concluded that a doubling of carbon dioxide (CO₂) could reduce yields by 10-30 percent in northern mid-latitude regions while warming and CO₂ fertilization could enhance yield potential towards the pole-ward edge of current agricultural regions. It was also noted that disagreement on how the amount and distribution of precipitation will be affected by global warming makes it difficult to estimate impacts at lower latitudes.

In assessing the likely impacts of climate change on world food supply, Rosenzweig and Parry (1994) found that global agricultural production will decrease slightly under a twofold increase in CO₂ concentration. The results also indicate that vulnerability to climate change is uneven between developed and developing countries with countries in the global south experiencing greater losses. Interestingly, simulation results suggest that adaptation efforts are ineffective is closing the gap between developed and developing countries.

Darwin et al. (1995) reported that while increases in temperature and changes in precipitation patterns over the next century will impact agriculture across the world, adaptation efforts will mitigate the risk to world food production. They also reported unequal impacts with arctic and alpine regions experiencing an increase in agricultural production while tropical regions and the U.S. Corn Belt and southeast seeing a decline.

Using the Hadley Center Coupled Model (HadCM2) global climate scenarios, Parry et al. (1999) reported that high and mid-latitudes regions will experience an increase in yields while the opposite is true for lower latitudes. They also noted that regional differences will grow with the passage of time leading to a clear division of impacts between developed and developing countries. However, gains in some regions are expected to offset losses in other regions thus resulting in production, prices, and risk of hunger being relatively stable under a changing climate.

Parry et al. (2004) used the HadCM3 global climate model under the Intergovernmental Panel on Climate Change (IPCC) Special Report on Emissions Scenarios (SRES) to estimate the impacts of climate change on global food production. Under a fossil fuel intensive economic growth model, the results indicate that an increase in global temperatures will severely impair crop yields at the regional and global levels. Nelson et al. (2009) reported that global food security will be threatened as the overall impacts of climate change on agriculture are expected to be negative. They also noted that developing countries will be hardest hit with yield declines in the most important crops and corresponding price increases.

Calzadilla et al. (2013) employed the Global Trade Analysis Project – Water (GTAP-W) model which accounts for water (rainfed and irrigated agriculture) to assess the potential impacts of climate change and CO₂ fertilization on global agriculture based on the IPCC SRES A1B and A2 scenarios. They found that future climate scenarios will likely alter regional water supply and soil moisture which would disrupt the distribution of cultivated land. As a result, global food production, welfare, and Gross Domestic Product (GDP) will decline and higher food prices can be expected.

The Food and Agricultural Organization (FAO) reported that global food production patterns will be modified as a result of climate change (Elbehri, Elliott, and Wheeler 2015). Agricultural production is expected to be negative across low latitude and tropical regions while high latitude regions may experience a slight increase. The authors also noted that while water lessens the impacts of climate change, water deficits in many regions of the world impede adaptation.

The underlying theme of these and other studies is that climate change impacts will vary across regions of the world with low latitude developing countries experiencing the brunt of this negative environmental externality. Consequently, global food production and security remain at risk under a changing climate.

2.7.2 Evidence from Africa, Western Europe, and South America

Several studies have examined the impacts of climate change on agriculture in Africa, South America, and Western Europe. These studies employed a Ricardian cross-sectional approach to estimate the impacts of climate change.

Kurukulasuriya et al. (2006) utilized survey data from over 9,000 farmers across 11 African countries to study the impacts of seasonal climate on net farm revenues. They found that revenues declined for dryland crops and livestock and increased for irrigated crops due to changes in temperature. Also, changes in precipitation have a positive effect on revenues for all farm types. Applying future climate data to the same dataset, Kurukulasuriya and Mendelsohn (2008a) predicted that net revenues from dryland farms are very sensitive to climate change. If the future climate is mild and wet, net revenues could increase by over 50 percent or decrease by approximately 40 percent under hot and dry conditions.

To determine the adaptive behavior of farmers, Seo and Mendelsohn (2008c) compared the choices farmers face under different environmental conditions. In warmer areas, they found that African farmers changed from cattle to more heat tolerant goats and sheep while in wetter areas farmers changed from cattle and sheep to goats and chicken. In another study, Seo and Mendelsohn (2008a) employed a cross-sectional approach to study approximately 5,000 livestock farmers across ten African countries. The results indicated that large farms were more vulnerable to changes in temperature and in warmer areas large farms have fewer animals per farm.

Seo et al. (2012) studied the impacts of climate change across sixteen Agroecological Zones (AEZs) in Africa. While net revenue from crop agriculture is more sensitive to climate change, they found that combined net revenues from crops and livestock were more climate resilient. The results suggest climate change impacts will be disparate across the African landscape with existing productive areas being more vulnerable while non-productive areas will become more productive in the future.

Based on a sample of 2,300 farms, Seo and Mendelsohn (2008b) examined the impacts of climate change across seven South American countries. The results suggest that an increase in temperature and precipitation will be harmful with climate change impacts varying across farm types (crop-only, mixed, and livestock-only) and geographical regions. Warning in the hot and wet Amazon and Equatorial regions are likely to suffer loses while cooler altitudes and southern regions will benefit.

Moore and Lobell (2014) employed time series and cross-sectional approaches to assess the efficacy of adaptation to climate change in Europe. They found that under future warming conditions, meaningful adaptation will benefit maize but have less of an impact on wheat and barley. The results suggest that overall, the benefits of adaptation are slight while the costs of nonadaptation are significant. Using farm-level data from 41,030 farms across Western Europe, Van Passel, Massetti, and Mendelsohn (2017) reported that under warming conditions, European farms were marginally more vulnerable than American farms while farms in Southern Europe are relatively more sensitive.

2.7.3 Evidence from the Brazil, Canada, China, India and United States

This section introduces evidence from Brazil, Canada, China, India, and the United States. Comparing the U.S. and Brazil, Mendelsohn, Basist, et al. (2007) found that higher climate variance enhances the chance of land being used for agriculture in both countries while high temperatures reduce cropland values. Using panel data from Brazil and India, Sanghi and Mendelsohn (2008) explored how farm values varied with climate and how farmers responded. Results indicated that Brazil could suffer annual losses as low as 1.0 percent and as high as 39.0 percent.

Reinsborough (2003) reported marginal benefits of climate change on Canadian agriculture. However, the large margin of error encouraged further study. In a comparison of climate change impacts on U.S. and Canadian agriculture, Mendelsohn and Reinsborough (2007) found that warmer weather had no effect on Canadian agriculture but increase precipitation was beneficial.

Using survey data from 8,405 farm households across 28 provinces in China, Wang et al. (2009) studied the effects of climate change on both rainfed and irrigated farms. The results indicate harmful and beneficial effects for rainfed and irrigated farms, respectively. There are also regional impacts with slight effects for farms located in the Southeast and larger damages for farms in the Northeast and Northwest. Another study by Chen et al. (2013) examined the impacts of climate change using farm-level data from 13,379 farm households across 316 villages, distributed in 31 provinces. They found that the marginal effects of increased temperature and rainfall on net revenue per hectare were positive and negative, respectively. They also found regional differences due to changes in precipitation.

In comparing the explanatory power of satellite and ground station data in India, Mendelsohn, Kurukulasuriya, et al. (2007) found surface wetness index produced by satellite were superior even though irrigation may be at work. Temperature measured by satellite appears promising for climate and agricultural studies. Sanghi and Mendelsohn (2008) found that farm values in India could suffer losses ranging from 4.0 percent to 26.0 percent due to climate change.

Perhaps access to reliable agricultural and climate data may have encouraged multiple studies on the impacts of climate change on U.S. agriculture. Early studies relied on crop simulation models to estimate the impacts of climate change on U.S. agriculture (Adams 1989; Rosenzweig 1989; Adams et al. 1995; Kaiser et al. 1993). While these studies highlight the negative impacts of climate change on U.S. agriculture, they also contended that U.S. agriculture is resilient with economic adjustment and adaptation. For instance, Kaiser et al. (1993) found that farmers can effectively adapt to a changing climate by planting long maturity cultivars, changing crop mix, and altering field operations to maximize longer growing seasons.

With the introduction of the Ricardian approach, a new wave of studies followed. Mendelsohn, Nordhaus, and Shaw (1994) used economic and geographical data from approximately 3,000 counties to study the impacts of climate change. The results indicated that higher temperatures in winter, spring, and summer reduced farmer values while increased rainfall independent of autumn increased farm values. They also found that future climate change could be beneficial to U.S. agriculture.

Schlenker, Hanemann, and Fisher (2005) studied the impacts of climate change on 2,197 dryland non-urban counties, in the U.S. They estimated that annual losses of approximately \$5 to \$5.3 billion can be attributed to climate change. Using panel data to estimate the Ricardian
function, Massetti and Mendelsohn (2011) found that mild warming is advantageous to U.S agriculture while extreme conditions would be harmful.

2.7.4 Evidence from Other Developed and Developing Countries

The literature also offers evidence from smaller nations and regions within countries that have significant local climate variation. These studies primarily used the Ricardian approach.

In studying land prices across England and Wales, Maddison (2000) found that climate, soil quality, elevation, and structural attributes of farms were important determinants of farmland values. Seo, Mendelsohn, and Munasinghe (2005) examined the impacts of climate change on net revenues of rice, coconut, rubber, and tea in Sri Lanka. They found that increases in rainfall are expected to be beneficial while higher temperatures are predicted to be harmful.

Using data from 300 districts across South Africa, Gbetibouo and Hassan (2005) studied the impacts of climate, soil, and socioeconomic variables on net revenues of seven field crops (maize, wheat, sorghum, sugarcane, groundnut, sunflower, and soybean). The results indicate that an increase in temperature and a decrease in rainfall have a positive and negative impact on net revenues, respectively. They also reported that future climate change will demand different responses across agro-ecological zones including shifts in cropping calendars, growing seasons, and crops altogether.

Kurukulasuriya and Ajwad (2007) used farm-level data to estimate the impacts of climate change on small farmers' net revenues in Sri Lanka. They found that lower rainfall during key agricultural periods have a significant impact on net revenues and that non-climatic factors explained approximately half of the variation in net revenues. Not surprisingly, they also noted that impacts vary across regions.

Fleischer, Lichtman, and Mendelsohn (2007) found that irrigation is very important in assessing climate change impacts in Israel. The inclusion of irrigation reduces the marginal impacts of temperature on net revenues. With technology, irrigation, cover, and marketing arrangements playing an important role, the study also found that higher temperatures would actually increase net revenues.

Kabubo-Mariara and Karanja (2007) used survey data from 816 small farm households across Kenya to examine the economic impact of climate change on net crop revenue. The findings revealed that increases in precipitation result in higher net revenues while warmer summer and winter temperatures have a negative and positive effect on net revenues, respectively. The study also stressed the importance of effective adaptation in maintaining optimal conditions for agriculture production.

Studying farms in Germany, Lippert, Krimly, and Aurbacher (2009) found that land rent increases with an increase in temperatures and a decrease in spring rainfall, except in eastern Germany. They also found that under severe changes in temperature and precipitation, income losses will become more apparent in the long-run. In Cameroon, Molua (2009) found that net revenues were more susceptible to a decrease in rainfall than an increase in temperature.

Deressa and Hassan (2009) used data collected from different agroecological zones in Ethiopia to assess the impacts of climate, household, and soil variables on net crop revenues. The results show that a slight increase in temperature during summer and winter seasons reduce net revenues significantly while a marginal increase in spring rainfall has a strong positive relationship with net revenues. The authors also noted climate change impacts vary across agroecological zones and that adaptation can help mitigate the consequences of future climate scenarios on net revenue. Mendelsohn, Arellano-Gonzalez, and Christensen (2009) used data collected from 621 farm households in Mexico to measure the impacts of climate change. They found that each additional degree of warming reduces farmland values by 4,000 to 6,000 pesos. They also found that rainfed farms are slightly more sensitive than irrigated farmers. Under future climate scenarios, farmland values will decline between 42 and 54 percent on average.

De Salvo, Raffaelli, and Moser (2013) evaluated the impacts of climate change on apple and grapes production in the Italian Alpine region. They controlled for farm characteristics and strategic decisions (e.g., specialization and quality certification) and tested three functional forms. They concluded that climate change reduces annual net revenues in the study area.

Arguing that climate, soils, and socioeconomic factors are challenging to model over wide geographic areas, Wood and Mendelsohn (2014) studied the impacts of climate change on net revenues in Fouta Djallon, West Africa. The authors found that higher temperatures and rainfall were negatively and positively associated with agricultural revenues during the rainy season and cool dry seasons, respectively.

Bozzola et al. (2017) used a dataset of 16,000 farmers to investigate the impacts of climate change on Italian agriculture. While the results indicate that farm net revenues are sensitive to seasonal changes in temperature and rainfall, differences exist between crop and livestock farms and rainfed and irrigated farms. They also noted that more severe changes in climate will have progressively damaging impacts.

2.8 Empirical Evidence of Farmers' Perceptions and Adaptation to Climate Change

Farmers' decision to and choice of adaptation strategy begins with their ability to perceive and understand climate change risk. This section presents a review of the literature on farmers' perceptions and decisions to adapt to climate change from multiple countries.

Based on in-depth interviews with farmers across two counties in New York State, Takahashi et al. (2016) found polarized views about climate change among farmers. Specifically, some farmers are of the opinion that climate change is a serious problem and are taking measures to adapt while others remain skeptical that climate change is real. The researchers also found that past experience plays a crucial role in influencing farmers' beliefs and perceptions. In addition, the study found that farmers with more years of experience are engaging in short-term adaptation practices as part of their normal operations.

Ayanlade, Radeny, and Morton (2017) compared farmers' perceptions of climate change with meteorological data in southwestern Nigeria. Approximately two-thirds of farmers observed changes in climate which is consistent with observed climate trends for the area. In terms of adaptation, years of farming experience and income strongly influence the choice of adaptation practice undertaken.

Tripathi and Mishra (2017) investigated farmers' perceptions of and adaptation to climate change in Uttar Pradesh, India. The results indicate that while farmers are cognizant of long-term changes in temperature and rainfall, they do not associate these changes with climate change. They also found that farmers are conscious of the risk related to climate variability and extreme weather events and are responding by adjusting sowing and harvesting dates, cultivating short duration crop varieties, inter-cropping, changing cropping pattern, and investing in irrigation and agroforestry.

Using data collected from 150 randomly sampled farmers in three selected provinces in South Africa, Elum, Modise, and Marr (2017) found that farmers' perceptions are consistent with changes in climate parameters. The results also indicate that farmers are engaging in different adaptation practices, chief among these being the use of drought-tolerant varieties. However, the lack of awareness and/or the inability to afford insurance is limiting farmers' uptake of crop insurance.

Mase, Gramig, and Prokopy (2017) studied survey responses of approximately 5,000 corn farmers across 22 Midwestern U.S. Watersheds. The study highlights the critical role of perceptions given the strong positive relationship observed between farmers' belief of climate change and variable weather observed on their farm and across the U.S. Corn Belt. As it relates to adaptation practices, farmers are relying on new technologies, crop insurance, and conservation agricultural practices to manage the risk of weather and climate. Importantly, the researchers suggest that crop insurance may be subsidizing lack of adaptation efforts by farmers.

Gandure, Walker, and Botha (2013) examined small farmers' perceptions and responses to long-term climate change in the rural village of Gladstone, Free State Province, South Africa. They found that farmers' perceptions are correlated with observed data on long-term climate change. The results also indicate that rainfall variability and temperature extremes are not perceived to have a strong negative impact on livelihoods and that farmers are more worried about the impacts of weeds, insects, and worms. Rainfall harvesting techniques are the most common risk management and adaptation strategy. The researchers also suggest that greater awareness and education about climate change needs to be prioritized at different levels of society.

Alam, Alam, and Mushtaq (2017) utilized survey data from 380 resource impoverished farm households in Bangladesh to explored farmers' perceptions and adaptation to climate change and climate hazards. They found that farmers' perceptions of changes in climate are analogous to observed climate data. Adaptation strategies undertaken by farmers include planting new crop varieties, adjusting planting dates, gardening, planting trees, and migration. Better access to finance and more information on suitable adaptation strategies remains paramount to supporting adaptation and building resilience among vulnerable farm households.

Mulwa et al. (2017) found that farmers' decision to adapt to adverse changes in weather patterns are influenced by specific plot characteristics, credit constraints, and the availability of climate-related information. They also found that withstanding financial constraints, farmers remain motivated to engage in adaptation if climate-related information is readily available. The researchers conclude that enhance access to information about relevant adaptation practices, easy access to inexpensive credit, and strengthening access to assets are key policy implications.

In assessing farmers' attitudes and determinants of adaptation across Borana, Ethiopia; Nyando, Kenya; Hoima, Uganda; and Lushoto, Tanzania, Shikuku et al. (2017) found that farmers were more amenable towards planting new crops, using different varieties, and adjusting planting dates instead of soil, land, and water management practices. The findings also suggest that providing climate information is crucial to successful short-term adaptation while enhancing financing is an important long-term strategy.

As the preceding studies highlight, analyzing farmers' perceptions is a practical approach to understanding climate change in a variety of countries and settings. More importantly, farmers' perceptions can be used to test the consistency of observed weather variables in addition to providing more descriptive accounts of how climate variables have changed. These studies also draw attention to different adaptation strategies and barriers to adaptation that is based on local characteristics and contexts.

2.9 Gaps in the Existing Literature

The review of the literature highlights several important gaps. Despite the economic implications for both farmers and the rice industry as a whole, to the researcher's knowledge, no previous study has been undertaken to examine the impacts of climate change on rice production in Guyana. Additionally, there is little understanding of how farmers in general and small farmers specifically perceive, are impacted by, and respond to changes in rainfall, temperature, extreme weather events, insects and pests, diseases and weeds. As such, research in this area is needed to identify and assess the economic ramifications of climate change on rice production and to inform policy at the national and rural levels.

In general, studies tend to group the impacts of climate change as being either positive or negative or high, medium or low. However, little emphasis has been placed on understanding specific impacts at the farm household level. By decoupling impacts, this research provides a deeper understanding of how small farmers are affected by and respond to changes in rainfall, temperature, extreme weather events, insects and pests, diseases and weeds.

Other studies have primarily focused on the direct impacts of climate change in terms of changes in rainfall, temperature, and extreme weather events. However, indirect impacts such as changes in insects and pests, diseases, and weeds have not received similar attention. By considering indirect impacts, this study provides a more comprehensive analysis of the impacts of climate change on rice production in Guyana.

This study will also have broader implications for the Latin America and Caribbean (LAC) region. Although the overall impact of climate change on the agricultural sector in LAC is expected to be negative, impacts vary by crops and countries (Fernandes et al. 2012). Additionally, little is known regarding how small rice farmers in the LAC region respond and adapt to climate

change. Moreover, the results of studies done in other regions of the world are less likely to be homogenous given specific climatic conditions, geographic characteristics, agricultural systems, and technological state (Sarker 2012). By using Guyana as a case study, the proposed research will close the gap in knowledge by providing empirical evidence on: the impacts of climate change on small rice farmers, small farmers' perceptions of climate change, adaptations at the farm-level, and factors that drive and constrain adaptation decisions. The results will serve as catalysts by highlighting novel strategies and best management practices that can be used to develop appropriate adaptation measures and institutional responses based on local conditions and needs.

2.10 Conceptual Framework

Considering the impacts of climate change on crop agriculture, Figure 2.10 illustrates the conceptual framework adopted for this research. Specifically, climate change has both direct and indirect impacts on rice yields. Direct impacts include changes in precipitation patterns which affect the availability and timing of water while temperature extremes cause heat stress and higher evapotranspiration. Indirect impacts include changes in the incidence and range of insects and pests, diseases, and weeds. In addition, excess rainfall can be accompanied by heavy winds and/or cause flooding and soil erosion while drought leads to salt water infiltration in the irrigation system and/or on the farmland. Socioeconomic, farm-level, and institutional characteristics influence rice yields and adaptation strategies.



Figure 2.10 Conceptual framework of climate change impacts on rice yields and adaptation strategies Source: Author

2.11 Conclusion

Our climate is shaped by many complex processes. Observed changes in greenhouse gases, land and sea surface temperatures, precipitation, sea level rise, and Arctic sea ice provide ample evidence of climate change in the Anthropocene. This evidence sets the stage for the rest of this research.

The impacts of climate change on the agricultural sector are well documented. Rising temperatures, shifting rainfall patterns, and extreme weather events have both direct and indirect negative effects on agricultural productivity. However, carbon fertilization and longer growing seasons remain the key positive impacts. In terms of rice production, rainfall, temperature,

atmospheric carbon dioxide (CO₂), and solar radiation influence different stages of plant growth and by extension the quality and quantity of grains harvested. In addition, ongoing climate change is also expected to disrupt future rice production. While climate change affects rice cultivation, the opposite is also true. The flooded conditions that rice is grown under produce methane (CH₄) and nitrous oxide (N₂O), two very powerful greenhouse gases.

There are two broad approaches for estimating the impacts of climate change on agriculture: general equilibrium and partial equilibrium models. Most of the studies undertaken have employed a partial equilibrium approach that encompass four primary methods: agronomic crop models, panel weather studies, Ricardian cross-sectional climate studies, and agroecological zone Analysis. While earlier studies have employed agronomic crop models, the majority of recent studies have employed the Ricardian approach. However, panel data studies are becoming increasingly popular.

A key component of studies on climate change and agriculture is the type of climate data used. There are four primary sources of weather data: ground station, gridded, satellite, and reanalysis data. While ground station data are considered the gold standard, lack of spatial coverage and poor record keeping are concerns. Gridded data overcome issues relating to spatial coverage but is hampered by coarse resolution and spatial correlation of climate variables. Satellite data helps with missing ground coverage but are usually less accurate since temperature and rainfall are not directly measured. While reanalysis data combines information from multiple sources it is critiqued because it oversimplifies the reality embedded in climate models.

A great many empirical studies have been undertaken at the global, continental, country and, agricultural ecological zone levels. At the global level, concerns over food security are omnipresent. Africa appears to be the most negatively affected continent. The individual country

62

impact depends on the type of crop, irrigation, technology, and the overall ability of farmers to effectively adapt. Regardless, the underlining finding of many of these studies is that climate change impacts will vary across regions with countries in the global south being disproportionate affected.

Farmers' perceptions of climate change play an important role in their decisions to adapt. The evidence presented suggests that farmers' perceptions of climate change are in line with meteorological data collected. More importantly, their choice of adaptation is driven by specific perceptions. Based on the review of the literature, the conceptual framework was developed to help answer the research questions posed and to close the gaps in knowledge highlighted.

Chapter 3 Data and Methods

3.1 Introduction

The main purpose of this research is to offer insights into how small farmers¹² perceive, are impacted by, and respond to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. Responses from key informants¹³ are used to triangulate and better understand small farmers' perceptions, impacts, and adaptation. As such, this chapter provides a detailed description of the data and methods used to answer the following research questions:

- 1. How has the climate in Guyana changed?
- 2. What non-climatic factors influence rice yields of small farmers?
- 3. How do small rice farmers and key informants perceive changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?
- 4. What impacts are farmers and key informants seeing due to the observed changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?
- 5. What adaptive measures are they adopting in response to these impacts?

To answer each research question, several data sources and methods are employed. Section 3.2 introduces the data source and methods used to analyze climate variability and climate change in Guyana. Section 3.3 presents the underlying theoretical and empirical framework used to explore the relationship between non-climatic variables and rice yields. Section 3.4 outlines the research design, data source, and methodology used to collect farm-level data. Section 3.5 introduces the method of analyzing rice yield and non-climatic data at the farm-level. Section 3.6

¹² Farmers that cultivate 4.45 hectares (11 acres) or less in rice each season.

¹³ Key informants comprise of district rice extension officers, an experience farmer (+25 years of experience), a rice miller, and senior staff (Chief Scientist/Plant Breeder, Plant Pathologist, Agronomist, Entomologist, and the Rice Extension Manager) at the Burma Rice Research Station in Guyana. Please refer to Chapter 3 – Data and Methods for additional information.

describes the method used for analyzing small farmers' and key informants' perceptions, impacts, and adaptation. Section 3.7 provides a summary of the various data and methods introduced.

3.2 Analyzing Climate Variability and Climate Change in Guyana

To answer research question 1, three methods are used to analyze secondary rainfall and temperature data for Guyana. The sections that follow describe the data source and methods used.

3.2.1 Data Source and Description – Climate Data

Average annual rainfall and temperature data for Guyana were obtained from the Climate Research Unit (CRU) country dataset (CRU CY v.3.24.01) at East Anglia University for the period 1901-2015. The CRU CY dataset is derived directly from the CRU time-series (CRU TS v. 3.21), which is an updated version of the high-resolution monthly datasets initially developed by New, Hulme, and Jones (1999, 2000) and subsequently updated by Mitchell and Jones (2005) and covers all land areas with the exception of Antarctica.

The data used to construct the CRU dataset were obtained from the archives of climate station records that have been subject to extensive quality control measures and augmented with newly acquired data (Harris et al. 2014). The primary data sources include: internationally exchanged monthly data for about 2,400 stations from countries within the World Meteorological Organization (WMO); monthly climate data for the world produced by the National Oceanographic and Atmospheric Administration (NOAA) National Climatic Data Center (NCDC); and World Weather Records (WWR) decadal data publications (Harris et al. 2014).

To construct the CRU time-series, station data are gridded on a global grid with a spatial resolution of 0.5° longitude by 0.5° latitude using the Climate Anomaly Method (CAM). Using

this method, each station series must contain at least 75% of the data for each month in order to calculate the based period (1961-1990) average or normal (Harris et al. 2014). Unless considered outliers¹⁴, normals produced for each month are included in the gridding process and converted to anomalies. This is done by subtracting the 1961-1990 normal from each station's data on a monthly basis (Harris et al. 2014). The gridding operation itself entails triangulated linear interpolation that produces values on a grid with half-degree resolution. The spatial averages are calculated using area-weighted means (Harris et al. 2014). To make the dataset complete, station data located further away are used for infilling and if not possible the 1961-1990 average for the specific grid box is used (Harris et al. 2014). New station data for earlier periods that become available are subsequently incorporated when the dataset is updated.

The CRU CY dataset contains monthly, seasonal, and annual data for mean temperature (TMP), maximum and minimum temperatures (TMX and TMN), precipitation total (PRE), vapor pressure (VAP), cloud cover (CLD), rainday counts (WET), and potential evapotranspiration (PET). Minimum and maximum temperature are measured in degrees Celsius and monthly, seasonal and annual precipitation data measured in millimeters (mm).

3.2.2 Methods of analyzing climate data

To characterize climate variability in Guyana, three statistical methods are employed: descriptive statistics, a simple trend model, and a two-sample t-test.

¹⁴ "Values that fall more than three (3) standard deviations from the normal (4.0 for precipitation)" (Harris et al. 2014).

3.2.2.1 Descriptive statistics

Descriptive statistics entail simplifying data to more understandable forms without compromising the data integrity (Agresti and Finlay 2014). This is done through the collection, organization, summarization, and graphical display of data (Jaisingh 2006). To analyze climate data, descriptive statistics comprising of the moving average, standard deviation, and coefficient of variation (CV) are used to show changes in rainfall and temperature over time.

3.2.2.1.1 Moving average

A moving average is a time series constructed by moving the arithmetic mean values through the time series (Mason, Lind, and Marchal 1999). In other words, it is a running average of a subset of sequential values in a time series. Given a set of values x_t , where t = 1, 2, 3..., i, the moving average of a subsequence of *n* terms can be computed as follows:

$$\overline{x_t} = \frac{1}{n} \sum_{t=t-n+1}^t x_t, \qquad [3.1]$$

where,

 $n \leq t$

 \bar{x}_t = moving average at time (*t*)

n = the number of observations in the subsequence

t = time

For example, if we have ten measurements of maximum temperature (T_{MAX}) in a time series $(T_{MAX1}, T_{MAX2}, T_{MAX3}..., T_{MAX10})$, the moving average (MA) of a sequence of five measurements can be computed as follows:

$$MA_{5} = (T_{MAX1} + T_{MAX2} + T_{MAX3} + T_{MAX4} + T_{MAX5})/5$$
$$MA_{6} = (T_{MAX2} + T_{MAX3} + T_{MAX4} + T_{MAX5} + T_{MAX6})/5$$
$$\vdots$$
$$MA_{10} = (T_{MAX6} + T_{MAX7} + T_{MAX8} + T_{MAX9} + T_{MAX10})/5$$

A moving average is used to smooth out any short-term fluctuations and highlight longterm trends (Kenney and Keeping 1962). While short-term perturbations of rainfall and temperature are important from an individual extreme weather perspective, smoothing out such fluctuations removes noise from long-term trends. As such, a 5-year moving average is used to compute the mean, standard deviation, and CV from Climate Research Unit (CRU) gridded timeseries country dataset (CRU CY v.3.24.01).

3.2.2.1.2 Standard Deviation

The standard deviation is one of the most common and useful measures of variability (Jaisingh 2006). It measures the dispersion of the data from a center value such as the mean (Coolidge 2013). The sample standard deviation, s, is defined as the positive square root of the sample variance, s^2 . Per Agresti and Finlay (2014), the sample standard deviation, s of n observation can be computed as follows:

$$s_y = \sqrt{\frac{\Sigma(y_i - \bar{y})^2)}{n - 1}}$$
 [3.2]

where,

 s_y = standard deviation of the respective climate variable y

 $y_i = i$ th observation of the climate variable y.

 \overline{y} = sample mean

n = sample size

By taking the positive square root of the sample variance, the standard deviation is presented in the same units of the variable. As such, it is easier to interpret any deviation from the mean. Large deviations from the mean indicate that the dataset is more volatile and thus less reliable. Thus, by computing the standard deviation of rainfall and temperature, we can measure the absolute variability with respect to the mean.

3.2.2.1.3 Coefficient of Variation

The more (less) disperse a dataset, the higher (lower) the standard deviation. As such, the standard deviation does not allow a direct comparison of two or more different variables since it is proportional to the mean. Thus, the magnitude of the standard deviation depends on the variable being analyzed (Sarker, 2012). The coefficient of variation (CV) overcomes this problem by allowing us to standardize the variation across variables. This is done by converting the standard deviation (a measure of absolute variability) to a relative value. The sample CV is defined as the sample standard deviation divided by the sample mean expressed as a percentage (Jaisingh 2006). Hence, the CV is computed as follows:

$$CV_{y} = \frac{s_{y}}{\bar{y}}$$
[3.3]

Given that the moving average and the standard deviation have the same units, the CV is unitless which allows for direct comparison of different variables with different units.

3.2.2.2 Linear trend model

The second method uses a simple trend model to assess the changes in rainfall, and minimum and maximum temperature over time. Gujarati, Porter, and Gunasekar (2012) note that the coefficient of the trend variable in a growth model gives the instantaneous (at a point in time) rate of growth as shown in equation 3.4 below.

$$Y_t = \beta_0 + \beta_1 t + e_t \tag{3.4}$$

where, Y = independent climate variable t = time $e_t =$ error term

In equation 3.4, the time variable (*t*) is referred to as the trend variable and its coefficient (β_1) indicates the direction of the trend. A positive (negative) coefficient indicates an (a) upward (downward) trend in the climate variable Y. It must be noted that the linear trend model does not imply causation. Rather it suggests the directional change of the climate variable over time.

3.2.2.3 Two-sample t-test

The third method uses a two-sample t-test to examine the change in average annual total rainfall (mm), average annual maximum temperature (°C), and average annual minimum temperature (°C) between two time periods. Livezey et al. (2007) note that numerous empirical studies and simulations indicate that the latest period of modern global warming began in the mid-

1970s. As such, 1975 is used as the hinge point for a two-sample t-test comparison of climate variables between 1934-1974 and 1975-2015.

In hypothesis testing, a two-sample t-test compares the means of two samples to allow us to make inference about their difference. The null and alternate hypotheses are presented below:

H₀:
$$\mu_1 - \mu_2 = 0$$

H_a: $\mu_1 - \mu_2 \neq 0$

where μ_1 and μ_2 are the means of the respective samples. The null hypothesis (H₀) assumes that statistically significant evidence does not exist to support the difference in the means while the alternate hypothesis (H_a) assumes that there is statistically significant evidence to support the difference in the means. The primary assumptions of the two-sample t-test are that the samples are independent and random, both populations are normally distributed, and the population variances are unknown but assumed equal (Agresti and Finlay 2014).

3.3 Analyzing Small Farmers' and Key Informants' Perceptions, Impacts, and Adaptation

To answer research questions 3 and 4, primary data were collected from small farmers and key informants. The discussion that follows presents the theoretical model and empirical approach, research design, and primary data collection method. Included is a description of the data source, the data collection process, and coding and construction of each variable.

3.3.1 Theoretical Model and Empirical Approach

The discussion that follows presents the underlying theoretical model and the empirical approach used to explore the relationship between socioeconomic, farm-level, and institutional characteristics and rice yields.

3.3.1.1 Theoretical Model

There are many factors that influence crop agricultural productivity. In terms of rice production, the existence of these factors individually or collectively may affect the quantity of rice harvested. In studying climate impacts, different authors have used different classifications or some variant of exogenous variables. Van Passel, Massetti, and Mendelsohn (2017) grouped exogenous variables into climate variables, exogenous control variables, and socioeconomic variables. Climate variables include temperature and rainfall; exogenous controls variables include distance from urban areas, distance from ports, mean elevation, and soil; and socioeconomic variables include market access, agricultural land value, and ownership of land (Van Passel, Massetti, and Mendelsohn 2017).

Similarly, Antle and Stöckle (2017) classified exogenous variables into three subgroups: physical, biological, and socioeconomic. Physical factors include temperature and precipitation, humidity and solar radiation, social conditions and atmospheric CO₂; biological factors include crop genetics and pest and diseases; and socioeconomic factors include technology, prices, policy, and institutions (Antle and Stöckle 2017).

Other classifications of exogenous variables used in assessing the impacts of climate change on agricultural productivity include climate, soil, and socioeconomic variables (Mendelsohn and Nordhaus 1994); purchase inputs, climate variables, labor characteristics, and farm characteristics (Kurukulasuriya and Ajwad 2007); and purchased input prices, climate, soil, hydrological, economic variables (Charles 2009). While the general classification of exogenous variables is different, there are many similarities in the actual variables. Additionally, the variables included under each grouping may be unique to a particular crop, location, and/or agricultural system. Sarker (2012) introduced three categories of non-climatic variables that affect rice yields in Bangladesh: socioeconomic, farm-level, and institutional characteristics.

The model used in this study explores the relationship between socioeconomic, farm-level, and institutional characteristics and rice yields at the farm-level. Although climatic conditions, insects and pests, diseases, and weeds also affect yields, these variables are not considered specifically since such data are not available at the farm-level for Guyana. Instead, perceptions of these variables are included in the model and are treated as farm-level characteristics. In Equation [3.5], rice yield is a function of socioeconomic, farm-level, and institutional characteristics.

Rice Yields = f (Socioeconomic Characteristics, Farm-level Characteristics, Institutional Characteristics) [3.5]

3.3.1.1.1 Socioeconomic Characteristics

Socioeconomic characteristics include farmers' age, gender, education level, and sources of income. The age of a farmer serves as a proxy for experience (Kurukulasuriya and Ajwad 2007; Charles 2009; Sarker 2012; Bello and Maman 2015; Closset, Dhehibi, and Aw-Hassan 2015; Huong, Bo, and Fahad 2018. Older farmers may possess greater farming knowledge and experience regarding climate change and agronomic practices and thus are able to obtain higher yields by making changes to their farming practices over time. In developing countries, traditional gender inequalities often mean that male farmers are more productive. Women often lack the same access to productive resources such as land, technology, financial services, education, and markets (FAO 2011). In Guyana, male farmers usually congregate in the evening and discuss farming practices while female farmers are seldom present during these conversations because of cultural norms. Kurukulasuriya and Ajwad (2007) suggest that the female head of farm households may face discrimination and do not benefit from the household having two adults to contribute to farm labor. Other studies that considered gender in their analysis include Sarker 2012; Ochieng, Kirimi, and Mathenge 2016; Huong, Bo, and Fahad 2018.

Education level is a proxy measure of farmers' intellectual ability to adopt new technologies, and optimize inputs and marketing practices (Kurukulasuriya and Ajwad 2007; Nyuor et al. 2016). Farmers with higher levels of education or more years of schooling are expected to produce higher yields under a changing climate (Kurukulasuriya and Ajwad 2007; Nyuor et al. 2016; Kabubo-Mariara and Karanja 2007; Sarker 2012; Bello and Maman 2015; Ochieng, Kirimi, and Mathenge 2016; and Huong, Bo, and Fahad 2018.

According to Janvry and Sadoulet (2016), small farmers usually derive approximately 50 percent of their income from off-farm employment and self-employment in rural non-farm economy. Such income diversification plays an important role in risk management and income stabilization (Janvry and Sadoulet 2016). Under a changing climate, the availability of non-agricultural income is expected to help small farmers pay for the costs of adaptation that would otherwise be out of their reach. Examples of such practices may include investing in equipment, purchasing additional fertilizer and chemicals, paying for additional land preparation, and purchasing new seeds each season.

3.3.1.1.2 Farm-level Characteristics

Farm-level characteristics include perceived changes in rainfall, temperature, extreme weather events, insects and pests, diseases and weeds. Other farm-level characteristics include primary occupation is rice farming, the size of farm, ownership of farmland, experience of farmer, farmer owned a tractor, farmer owned livestock, type of soil on the land, soil test was done, household member(s) help with rice farming, farmer fully adopted the six-point practice, seasonal labor was employed, and seed variety planted.

In the absence of farm-level climate data, farmers' perceptions of changes in rainfall, temperature, and extreme weather events are used as explanatory variables. Changes in rainfall, temperature, and extreme weather events are also expected to have a positive effect on the presence of insects, diseases, and weeds (Rosenzweig et al. 2001), which are assumed to negatively impact yields. It is assumed that farmers that perceived changes in these variables are also likely to experience changes in their yields and subsequently change their farm management practices leading to better yields.

Farmers whose primary occupation is rice farming is expected to produce higher yields than those that plant rice on the side (Kabubo-Mariara and Karanja 2007; Sarker 2012). This is because these farmers have a vested interest in rice cultivation since it is their main source of income. As such, they are expected to be deeply involved in the day-to-day management of their fields and well-informed about new developments in farm management practices under a changing climate.

The size of a farm is an important factor in determining yields. Large farmers can spread the risk of adverse climatic conditions (J. Benhin 2006) by diversifying crops and livestock (Charles 2009). They can also apply better production technologies (Sarker 2012). KabuboMariara and Karanja (2007) noted that large farm size may be associated with higher productivity. Large farm sizes have also been found to correlate with higher farm net revenue (Wood and Mendelsohn 2014; Nyuor et al. 2016; Huong, Bo, and Fahad 2018). However, other studies (Sarker 2012; Closset, Dhehibi, and Aw-Hassan 2015) have found a negative relationship between large farms and net revenues. In other words, small farms may be more efficient and hence more productive than large farms.

In developing countries, farm households may have incomplete or no property rights. According to Besley and Ghatak (2010), property rights refer to an owners right to use or transfer a good or asset. As such, property rights encourage farm households to invest in their land and/or allow them to gain access to financial resources such as credit. Kurukulasuriya and Ajwad (2007) argued that land tenure reduces the uncertainty of reaping the benefits of investment in time, inputs, and capital. Hence, farmers with secured property rights can make decisions that optimize their yields.

While age is often used as a proxy for experience, it may not always translate into actual experience. For instance, a farmer may have started planting rice later in life. As a result, the number of years farming would be lower. Therefore, the actual experience of a farmer is a more accurate measure. Sarker (2012) found that although farming experience had a positive impact on net revenues, it was not statistically significant.

Ownership or access to machinery and equipment enables farmers to attend to their fields in a timely manner. For instance, ownership of a tractor allows farmers to become more independent thus avoiding delays in land preparation. They can take advantage of ideal conditions and begin dry land preparation almost immediately. Early land preparation ensures farmers sow and harvest on time which reduces overall losses due to potential flooding and lodging. By

77

preparing the land themselves, farmers are also ensuring that it is done properly. Proper land preparation helps with water management and weed control. Access to tractors was found to have a positive influence on net farm revenues (Charles 2009; Mishra, Sahu, and Sahoo 2016).

The ownership of livestock provides farmers with an additional source of on-farm income. Income from the sale of livestock may help pay for adaptation practices that contribute to better yields. However, livestock ownership could also negatively impact on yields through increase completion for limited on-farm resources such as farmers' time and grazing space. Kabubo-Mariara and Karanja (2007) found that livestock ownership negatively impacted crop agriculture.

Nyuor et al. (2016) noted that there is a direct relationship between household size and agricultural output. This is because household size is often used as a proxy for farm labor (Kabubo-Mariara and Karanja, 2007). Wood and Mendelsohn (2015) found that larger households contribute to higher agricultural net revenues. In Guyana, household member(s) active participation in rice farming may help to ensure timely completion of farming activities and lower overall labor costs. Household members are also likely to take a greater interest in the quality of work performed.

In 2008, the Guyana Rice Development Board (GRDB) Extension Department introduced six improved management practices known locally as the six-point practice. The improved practices include advice on planting in-season (time of sowing), reducing plant densities (seed rate), treating seeds before sowing (seed treatment), controlling weeds, using balance nutrition fertilizer, and managing water in the field. Although the initial focus of the six-point practice was to improve yields, its adoption is viewed as a key response to changes in environmental conditions facing rice farmers. As such, farmers that fully implement the six-point practice would benefit from higher yields.

Since rice is direct seeded in Guyana, other household members' substitution for paid labor are usually not sufficient during peak sowing, fertilizing, and spraying times. As such, seasonal labor is employed to help ensure that specific farming activities are completed in a timely manner. This is especially the case for farmers that maintain off-farm employment and need the extra hands to help with putting the crop in on time. Having worked on multiple farmers, seasonal labor usually boasts good skills, knowledge, and experience broadcasting seedlings, applying fertilizer, and spraying for insects and diseases. As such, the use of seasonal labor may help boost yields. However, the quality of work performed by seasonal labor may lead to lower yields. For example, seasonal labor may do a poor job applying fertilizer or spraying for insects.

Since 1997, there have been 15 rice varieties released through the rice breeding program in Guyana (GRDB 2015b). Each variety differs across many characteristics including days to maturity, yield potential, disease resistance, and grain length, width, and shape. While some older varieties are susceptible to disease, some newer varieties are high yielding. Yet other newer varieties can withstand lodging even though their yields may be lower. Under a changing climate, planting a high yielding variety does not guarantee higher yields at harvest. For instance, heavy rainfall leading up to harvesting can cause a high yielding variety to lodge resulting in greater losses. Regardless, farmers are planting the varieties of their choice. Therefore, the variety planted may influence the yields of farmers.

3.3.1.1.3 Institutional Characteristics

Institutional characteristics include farmer participated in rice extension training(s); access to advance weather information; access to input credit, membership in an agricultural organization(s), access to adequate irrigation, the source of seed, and farmer pumped water.

79

Agricultural extension plays a crucial role in climate change adaptation. Extension agents transfer knowledge from researchers to farmers, advise farmers in decision-making, educate farmers to make similar decisions in the future, enable farmers to clarify their own goals and possibilities and to realize them, and stimulate desirable agricultural developments (Van den Ban and Hawkins 1996). As such, farmers' access to rice extension training sessions help ensure that they are receiving up-to-date information on adaptation strategies and practices which can help improve farm management practices and by extension yields. The influence of extension services on net farm revenues have been explored by Charles (2009); Sarker (2012); Nyuor et al. (2016); and Huong, Bo, and Fahad (2018).

Successful adaptation to climate change calls for timely and accurate weather information. Sarker (2012) argued that access to advance weather information could augment responses to and lessen the adverse effects of climate change. Farmers that have access to advance weather information can effectively and efficiently adapt to climate change thereby improving yields.

Farm household access to input credit is an important institutional support. Nyuor et al. (2016) noted that credit allows farmers to purchase basic agricultural inputs such as labor, fertilizer, seeds, and herbicides. Under a changing climate, credit also eases the financial burden of engaging in costly adaptation practices. For example, credit may assist a farmer to purchase new seeds each season or purchase a tractor to help with early land preparation. Several studies have explored the relationship between access to credit, adaptation, and farm household net revenues (Sarker 2012; Bello and Maman 2015; Nyuor et al. 2016); Huong, Bo, and Fahad 2018).

Membership in an agricultural organization is advantageous to farmers (Wang et al. 2009). As members, farmers are privy to information on regulations and policies that is likely to influence the rice industry. They may also readily access information on the impacts of climate change, new adaptation strategies, and training opportunities. Access to this information may guide farmers in their response to climate change which can translate into higher yields.

Given that rice cultivation in Guyana is under irrigation, access to adequate water remains paramount to realizing good quality and quantity yields. Moreover, timely release of water under a changing climate is even more crucial to water management practices, especially during dry spells. Given that irrigation is an effective adaptation measure (Gbetibouo and Hassan 2005; Kabubo-Mariara and Karanja 2007), farmers that benefited from adequate irrigation will boast higher yields.

Farmers that pump water to flood their fields may benefit from better yields in that they are able to engage in better water management. Alternatively, pumping water may be a sign of limited water availability in the irrigation system which may negatively affect yields. The definitions, source, and anticipated impact of the socioeconomic, farm-level, and institutional variables are highlighted in Table 3.1.

Non-climatic Variables	Variable	Variable Description	Measure	Expected Sign	Source
Socioeconomic	AGE	Age of the farmer	No. of years	+	Kurukulasuriya and Ajwad (2007); Charles (2009); Sarker (2012); Bello and Maman (2015); Closset, Dhehibi, and Aw-Hassan (2015); Huong, Bo, and Fahad (2018)
Socioeconomic	GENDER	Gender of the farmer	1=male and 0=female	+	Kurukulasuriya and Ajwad (2007); Sarker (2012); Ochieng, Kirimi, and Mathenge (2016); Huong, Bo, and Fahad (2018)
Socioeconomic	EDUC	Education level of the farmer	0=no school 1=primary 2=communit y high 3=high school 4=technical/ vocational 5=university	+	Kurukulasuriya and Ajwad (2007); Kabubo- Mariara and Karanja (2007); Sarker (2012); Bello and Maman (2015); Ochieng, Kirimi, and Mathenge (2016); Nyuor et al. (2016); Huong, Bo, and Fahad (2018)
Socioeconomic	SECNAG	Non-agricultural income	1=yes and 0=no	+	Author
Farm-level	RAINCHNG	Perceived changes in rainfall	1=yes and 0=no	+	Author
Farm-level	TEMPCHNG	Perceived changes in temperature	1=yes and 0=no	+	Author
Farm-level	EXTREWEAT	Perceived changes in extreme weather events	1=yes and 0=no	+	Author
Farm-level	INSECCHNG	Perceived changes in insects and pests	1=yes and 0=no	+	Author
Farm-level	DISEACHNG	Perceived changes in diseases	1=yes and 0=no	+	Author
Farm-level	WEEDSCHNG	Perceived changes in weeds	1=yes and 0=no	+	Author
Farm-level	PRIMOCCU	Rice farming is the primary occupation	1=yes and 0=no	+	Kabubo-Mariara and Karanja (2007); Sarker (2012)

Table 3.1 Definitions, Expected Signs and Source of Socioeconomic, Farm-level, and Institutional Variables

Farm-level	FARMSIZE	Size of farm	No. of hectares	+	Kabubo-Mariara and Karanja (2007); Sarker (2012); Wood and Mendelsohn (2015); Closset, Dhehibi, and Aw-Hassan (2015); Nyuor et al. (2016); Huong, Bo, and Fahad (2018)
			1=yes and		
Farm-level	TENURE	Ownership of farm land	0=no	+	Kurukulasuriya and Ajwad (2007); Sarker (2012)
Farm-level	FARMEXP	Experience of farmer	No. of years	+	Kurukulasuriya and Ajwad (2007); Sarker (2012)
			1=yes and		
Farm-level	TRACTOR	Owned a tractor	0=no	+	Charles (2009); Mishra, Sahu, and Sahoo (2016)
			1=yes and		
Farm-level	OWNLIVE	Owned livestock	0=no	-	Kabubo-Mariara and Karanja (2007)
			1=yes and		
Farm-level	SOILTEST	Soil test was done	0=no	+	Author
		Household member(s) help with	1=yes and		
Farm-level	HOUSEPART	rice farming	0=no	+	Author
			1=ves and		
Farm-level	SIXPOINT	Fully adopted the six-point practice	0=no	+	Author
		Tany adopted the bin point practice	1 1		
E 11		S	I=yes and	. /	Chatileans and Harry (2005)
Farm-level		Seasonal labor was employed	0-no	+/-	Gbetibouo and Hassan (2003) Cl = 1 (2000) $Cl = 1$ (2012) Normal 1 (2010)
Institutional	EXTIKAIN	Participated in rice extension	1=yes and	+	Charles (2009); Sarker (2012); Nyuor et al. (2016); Human D_{2} and E_{2} had (2018)
		training(s)	0=no	1	$\frac{1}{2} = \frac{1}{2} = \frac{1}$
T		Access to advance weather	I=yes and	+	Sarker (2012)
Institutional	ADVWEATH	information	0=no		C 1 (2012) D 11 114 (2015) N (
Tu atituti a u al	CDEDIT	A 4- innert 114	1	1	Sarker (2012); Bello and Maman (2015); Nyuor et
Institutional	CREDIT	Access to input credit	1=yes and	+	al. (2016); Huong, Bo, and Fanad (2018)
		Mamhanshin in an a miaultural	1-was and		
Institutional	AGMEM	organization(s)	1-yes and	+	Sorker 2012: Wong et al. 2000
Institutional	AOMENI	organization(s)	0-110	1	Chatibous and Hassen (2005); Kabuba Mariara
Institutional	ADEOIRRIG	Access to adequate irrigation	1=ves and	+	and Karania (2007): Charles (2009): Sarker (2012):
mstitutional	ADLQINNO	Access to adequate inigation	0 = no	'	Huong Bo and Fahad (2018)
			1=ves and		11001 <u>2</u> , <u>b</u> 0, and <u>r</u> and <u>(2010)</u>
Institutional	PUMPWAT	Pumped water	0=no	+/-	Author
Regional	REGION2	Farm located in region 2	Region=?	+/-	Author
Regional	REGION3	Farm located in region 3	Region=3	+/-	Author
Regional	REGION4	Farm located in region 4	Region=4	+/-	Author
Regional	REGION5	Farm located in region 5	Region=5	+/-	Author
Regional	REGION6	Farm located in region 6	Region=6	+/-	Author
regional	1010101	i unit toeuteu in tegton o	itegion o	• • •	1 10/11/1

3.4 Research Design

The survey part of this research was aimed at collecting primary data from small farmers and key informants to answer research questions two, three, four, and five. The sections that follow describe the research design.

3.4.1 Study Area

The study area for this research is the five primary rice-producing regions in Guyana: Pomeroon-Supenaam (Region 2), Essequibo Islands-West Demerara (Region 3), Demerara-Mahaica (Region 4), Mahaica-Berbice (Region 5), and East Berbice Corentyne (region 6)¹⁵. These administrative regions are located on the low coastal plain natural region which stretches 285 miles along the Atlantic coast and is primarily below sea-level. This low coastal plain flat terrain and fertile soil is the main crop agriculture area in the country. Figure 3.1 and 3.2 illustrate the natural regions and primary rice producing administrative regions, respectively.

¹⁵ Rice is also grown at Moco Moco in Upper Takutu-Upper Essequibo (Region 9).



Figure 3.1 Map of natural regions in Guyana Source: GLSC (2006)



Figure 3.2 Map of primary rice producing regions in Guyana Credit: Author

In terms of rice production, region two boasts the largest number of small farmers. Regions four and five have the smallest and largest number of acres under cultivation, respectively. Table 3.2 presents a summary of rice farmers by acreage and region as of spring 2018. Figure 3.3 shows the areas rice is grown.

1000000000000000000000000000000000000	Table 3.2 Summa	ıry of Farmer	rs by Acreage	and Region
---------------------------------------	-----------------	---------------	---------------	------------

Region/	2		3		4		:	5	6		Tot	al
Acreage	# of farmers	Acreage	# of farmers	Acreage	# of farmers	Acreage	# of farmers	Acreage	# of farmers	Acreage	# of farmers	Acreage
1-10	1261	6562	462	2570.75	267	1378.9	554	3650.25	170	1173.25	2714	15335.2
11-20	299	4540	186	2812.5	77	1149.5	367	5962.23	295	4602.25	1224	19066.5
21-30	107	2695	86	2189.7	37	930	244	6614.19	251	6939.5	725	19368.4
31-40	89	3168	43	1527	20	686.2	137	5044.6	87	3239	376	13664.8
41-50	41	1866	31	1393	13	602.5	115	5408.04	143	6566	343	15835.5
50 & above	127	14473	94	9473	38	3801.9	459	70792.86	294	34112	1012	132652.8
Total	1924	33304	902	19965.95	452	8549	1876	97472.17	1240	56632	6394	215923.1

Source: GRDB (2018)



Figure 3.3 Map of Guyana showing rice producing areas Source: USDA-FAS (2018)

3.4.2 Study Population

The population of this study is all small farmers located in the five main rice-producing administrative regions in Guyana. As of spring 2017, there were 2,896 small farmers across Guyana (GRDB 2017). Table 3.3 presents the number of small farmers in each rice-producing region.

|--|

Region	# of small farmers	%
Region 2 - Pomeroon-Supenaam	1,366	47.2
Region 3 - Essequibo Islands-West Demerara	495	17.1
Region 4 - Demerara-Mahaica	289	9.9
Region 5 - Mahaica-Berbice	526	18.2
Region 6 - East Berbice Corentyne	220	7.6
Total	2,896	100.0

Source: GRDB (2017)

3.4.3 Sample Size and Selection

Conventional wisdom suggests larger samples are better. However, access, costs, time, the overall size of the population, and the number of variables also influence sample size. According to Creswell (2012), a sample size of approximately 350 is appropriate for a survey study. In homogenous populations, however, smaller samples can be equally effective (Neuman 2011). Since farmers in each region in Guyana are likely to face similar socioeconomic, environmental, and climatic conditions, a smaller sample can be representative of the population (Blaikie 2000). Following Bartlett, Kotrlik, and Higgins (2001), a five percent sample size was considered sufficiently large.

To select the sample for each region, a copy of the rice farmers' register as of Spring 2017 was obtained from the Guyana Rice Development Board (GRDB). The register is maintained in Microsoft Excel and is organized by regions. Each region contains the following information:
farmer's first and last name, farmer's address (name of the village), farmer's call name¹⁶, and acreage sown. A table summarizing the number of farmers by acreage categories and across regions is also included in the farm register.

Given that the farm register was already stratified by region, a list of small farmers was subsequently created for each region to facilitate the sample selection. This was done by sorting the list of farmers by acreage planted. Farmers that planted more than 4.45 hectares (11 acres) were discarded since they did not meet the definition of small farmer established for this research. Once the list of small farmers was created for each region, the random function (RAND) in Microsoft Excel was used to generate and assign random numbers greater than or equal to 0 and less than 1 for each small farmer in a separate column. Since new random numbers are returned each time the worksheet is calculated, the list of random numbers generated was subsequently hardcoded by copying and pasting the values to prevent the numbers from changing. The list of small farmers for each region was then sorted by the random numbers' column from lowest to highest and the first five percent of small farmers were selected for interviews.

3.4.4 Data Source and Description of Survey Instrument

Two structured questionnaires comprising of both qualitative and quantitative questions were used to conduct face-to-face interviews with small farmers and key informants in Guyana. The main purpose of the questionnaires was to collect data on perceptions, impacts, and adaptation as it relates to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds.

¹⁶ In Guyana, persons usually go by a call name which can be a shortened version of their legal name or a nickname. The call name was very important in locating farmers since this is the name they are usually known by in their respective villages.

At the farm-level, a cross-sectional household survey comprising of 49 questions was designed and used for interviewing small farmers. Question types varied, and several questions contained multiple parts. Open-ended questions solicited a narrative response from small farmers and key informants. This approach was used because it provides a voice for those normally unheard. It also emphasizes the sequence of events from perceptions to adaptation.

Respondent ID, the name of the respondent, village, telephone number, GPS coordinates of interview location and administrative region aside, the questionnaire was divided into four parts as follows: socioeconomic characteristics; farm structure and characteristics; farmers' perceptions, impacts, and adaptation; and institutional accessibility.

The socioeconomic section collected data on the farmers' age, gender, education level, primary and secondary agricultural and non-agricultural occupations/sources of income. The farm structure and characteristics section captured data on household participation in rice farming, the use of part-time labor, acreage planted and yields, land tenure, farming experience, type of farming equipment owned, shared, and/or rented, livestock owned, soil type, and soil testing.

The perceptions, impacts, and adaptation section elicited responses from farmers regarding perceived changes, impacts, and responses to shifts in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. This section also collected data on the cropping season most affected by the observed changes, directional change of yields, crop failure, coping strategies, and changes in farming practice in response to observed changes. The sources of and farmers' ability to pay for adaptation, barriers to adaptation, small farmers' understanding of climate change, and farmers' familiarity with and implementation of six improved management practices introduced by the Guyana Rice Development Board (GRDB) were also collected.

The institutional accessibility section of the questionnaire collected data on farmers' access to and participation in extension training and related activities, access to and source of advance weather information, agricultural credit and insurance, and irrigation facilities. The source of and seed variety planted, the quality of inputs and equipment and farmers' membership in the Rice Producer Association (RPA) or other agricultural related association or institution were also collected. A copy of the questionnaire used at the farm-level is included as Appendix A.

The farm-level questionnaire was modified for interviews with key informants. Key informants comprise of district rice extension officers, an experienced farmer (+25 years of experience), a rice miller, and senior staff (Chief Scientist/Plant Breeder, Plant Pathologist, Agronomist, Entomologist, and the Rice Extension Manager) at the Burma Rice Research Station in Guyana. Apart from the respondent ID, name of the respondent, position, district, telephone number, GPS coordinates of interview location, and administrative region, the questionnaire collected data on the age, gender, education level, and experience of key informants. Key informants were also asked to describe perceived changes and impacts of rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds and how farmers are responding to these changes. In addition, key informants were asked to describe their current efforts to help farmers adapt to observed changes, specific resources needed now and/or in the future to effectively support adaptation, and the barriers or challenges that currently plague the rice sector. A copy of the questionnaire used for key informants is included as Appendix B.

3.4.5 Validity and Reliability of the Survey Instrument

Content validity and measurement reliability are important considerations in research involving primary data collection through the use of a questionnaire. Content validity is the extent to which survey questions measure the full content of a conceptual definition (Neuman 2011). In other words, do the survey questions capture the complete meaning? Given that the questions included in the survey are theoretically defined and based on a review of the literature on the impact of climate change at the farm-level, the questions are considered content valid. Validity was also confirmed by the fact that open-ended elements provided some opportunities for farmers to share information in their own words. Measurement reliability refers to the dependability or consistency of the measure (Neuman 2011). In other words, the results of the survey instrument do not change each time the instrument is used. Measurement reliability was improved through a pre-pilot test of the survey instrument and updating accordingly. This was followed by a pilot test which led to further refinements of the questions.

3.4.6 Pre-Pilot Testing

Although the draft questionnaire had undergone multiple revisions, a pre-pilot testing exercise was conducted in order to enhance the validity and reliability of the instrument. Pre-pilot testing helps to identify problem areas, reduce measurement errors, reduce respondent burden, and determine whether or not respondents are interpreting questions correctly (Ruel, Wagner, and Gillespie 2015). The pre-pilot testing exercise involved rice extension officers and a large farmer who provided valuable feedback which was incorporated into the questionnaire before the commencement of the pilot test.

3.4.7 Pilot Testing

After the pre-pilot testing was completed and the questionnaire updated, the instrument was pilot tested in early July 2017. Five small farmers in Region 3 - Essequibo Islands-West

Demerara were interviewed at their homes. Based on the responses received, further changes were made to refine the questionnaire. These included the re-wording of some questions and the creation of several new questions from existing questions. After the updates were made to the questionnaire, updated approval was sought and received from the Institutional Review Board (IRB) before the first interviews were conducted.

3.4.8 Interviews

Farm-level interviews were conducted in two phases between July and September 2017 and in May 2018. District Rice Extension Officers assisted in locating small farmers randomly selected for interviews. Interviews lasted between 30 and 45 minutes and were conducted primarily at the homes of small farmers. However, other interview locations included by the roadside, on the farm, and off-farm employment sites.

During the initial interviewing phase, four primary difficulties were encountered: farmers refused to participate, farmers could not be located, farmers planted more than 4.45 hectares (11 acres), and farmers no longer planted rice. Two farmers refused to participate. Further discussions with extension officers alluded to a general lack of trust as being the primary reason for non-participation. Difficulties in locating farmers include farmer passed away, farmer migrated overseas, or farmer was not at home. Farmers not at home were traveling, on their farm which was not located nearby or were engaged in off-farm employment. A second attempt was made to interview these farmers but often proved futile.

In several cases, farmers randomly selected for interviews indicated that they were planting more than 4.45 hectares (11 acres). As a result, interviews were discontinued because these farmers did not meet the definition of small farmers established for this research. The GRDB

93

extension manager explained that the difference between the farm register and what farmers reported was due in part to the failure of some extension officers to properly update the data for their respective district each season. It is common for the acreage to change from season to season since farmers may rent additional land thus increasing their acreage. Additionally, there was a compilation error where the acreage of farmers that planted in more than one districts was improperly summed and included in the final register. For example, a farmer that planted eight acres in one district and six acres in another district was incorrectly included in the register as planting eight or six acres instead of 14 acres. Farmers that did not plant rice in 2016 usually rented the land to a large farmer.

Of the 145 small farmers randomly selected for interviews in phase 1, 98 (67.6%) were replaced. Table 3.4 summarizes the reasons for replacing farmers. Replacement of farmers was done in two ways. Replacements were first selected from the next five percent of randomly selected small farmers. However, only 22 (15.2%) small farmers were located and subsequently interviewed. Given the limited success in locating small farmers coupled with time and cost considerations, extension officers were called upon to identify small farmers in the same village or neighboring villages. In some instances, replacement farmers were identified and interviewed from villages that were not previously captured in the random samples taken for each region.

Region	Refused to	Farmer	Farmer	Not at	Large Former	Did not Plant in	Total
	1 articipate	Away	Overseas	Home	Parmer	2016	
Region 2 - Pomeroon-							
Supenaam	-	1	-	38	-	-	39
Region 3 - Essequibo							
Islands-West Demerara	1	3	1	5	3	3	16
Region 4 - Demerara-							
Mahaica	1	1	2	7	1	1	13
Region 5 - Mahaica-							
Berbice	-	1	-	15	3	2	21
Region 6 - East Berbice							
Corentyne	-	-	1	7	-	1	9
Total	2	6	4	72	7	7	98

Table 3.4 Reasons for Replacing Farmers

It must be noted that in regions 3, 4, 5, and 6, the population of small farmers are relatively small in comparison to region 2. As a result, a five percent sample amounted to less than 30 small farmers in each of these regions. In order to satisfy the statistical convention of a large sample, the sample size was increased to a minimum of 30 small farmers in each region and the second phase of interviews were conducted in May 2018. During the second phase of interviews, extension officers were relied upon exclusively to identify and arrange interviews with additional small farmers in order to increase the sample size to 30 in regions 3, 4, 5, and 6. This approach helped eliminate the difficulties encountered in phase 1 thus reducing the time and cost involved in trying to locate farmers for an interview. Table 3.5 presents the sample of small farmers selected for each region. Figure 3.4 illustrates the GPS locations where interviews were conducted across the study area.

Region	# of small farmers	Sample Size	%
Region 2 - Pomeroon-Supenaam	1,366	68	5.0
Region 3 - Essequibo Islands-West Demerara	495	31	6.3
Region 4 - Demerara-Mahaica	289	30	10.4
Region 5 - Mahaica-Berbice	526	30	5.7
Region 6 - East Berbice Corentyne	220	30	13.6
Total	2,896	189	6.5

Table 3.5 Sample Size by Region

Locations of Farm-level Interviews





Figure 3.4 Map of locations of farm-level interviews Credit: Author

To supplement and corroborate data collected from small farmers, interviews were also conducted with key informants. Key informants comprise of district rice extension officers, an experienced farmer (+25 years of experience), a rice miller, and senior staff (Chief Scientist/Plant Breeder, Plant Pathologist, Agronomist, Entomologist, and the Rice Extension Manager) at the Burma Rice Research Station in Guyana. These interviews were conducted at the Guyana Rice Development Board (GRDB) regional offices in each of the five administrative regions and at the homes of the rice miller and seed paddy producer.

3.4.9 Coding and Tabulation Process

The completed questionnaires were reviewed by the researcher before data entry commenced. The purpose of the review was to identify any missing information or inconsistencies in the responses. For the most part, responses were complete. However, in a few instances missing information and/or inconsistencies were resolved by reviewing field notes and/or audio recordings of the specific interview if it was available since not all interviews were recorded.

A codebook was subsequently created to help with data entry. The codebook contained the variable name, variable description, variable type (e.g., text, continuous, categorical, ordinal, binary), and variable codes. The codes were primarily derived from the questionnaire which contained preassigned codes for most questions. In a few cases, however, codes were expanded to capture additional responses. For example, while codes were created and assigned for different rice varieties, no code was preassigned for small farmers that planted multiple varieties on the same field or on separate fields. Using the codebook as a reference, the survey data were entered into a Microsoft Excel and coded accordingly. The list of codes used for analyzing the responses of small farmers and key informants are presented in Appendix C. Microsoft Excel is useful for

tabulating survey data and allows for the dataset to be easily imported into Stata statistical software for multivariate analyses.

Upon completion of the data entry, the data was cleaned by reverse tracing the entry from the spreadsheet to individual questionnaires and making corrections where necessary. Once the data was cleaned, summary statistics for each question was created in separate tabs. For example, in the master spreadsheet, the column containing farmers' age was copied to a new worksheet, sorted from low to high, and tabulated using different age ranges. For questions that generated qualitative data (e.g., how did rainfall patterns change?), descriptive statistics were used to analyze each narrative response. A more in-depth discussion on the analysis of the qualitative data collected is presented in section 3.5 - Method for analyzing small farmers' and key informants' perceptions, impacts, and adaptation.

3.5 Method of Analyzing Rice Yield and Non-climatic Data at the Farm-level

The following sections introduce the theoretical model used for statistical estimation and the empirical approach taken. The contents of the multiple regression equations are introduced and described. Stata version 14 was used to generate the results presented in Chapter 5.

3.5.1 Multiple Regression

Climatic conditions aside, socioeconomic, farm-level, and institutional characteristics (non-climatic factors) also influence rice yields. As such, multiple regression is used to determine what non-climatic factors influence small farmers' yields. The regression model is presented in equation [3.6] where β_i are estimated coefficients, μ is the error term, which is assumed to be independent and identically distributed.

$$Y/ha = \alpha + \beta_1 F + \beta_2 G + \beta_3 H + \mu$$
[3.6]

where,

Y/ha = rice yields per hectare F = set of socioeconomic variables G = set of farm-level variables H = set of institutional variables μ = error

3.5.2 Multiple Regression Equation

Equation [3.7] represents the baseline version of the multiple regression equation. It comprises of those non-climatic independent variables that are expected to affect rice yields based on the literature review and feedback received from key informants. Two additional scenarios are presented in equations [3.8] and [3.9]. The estimates of these equations are presented in Tables 5.15 and 5.16 in Chapter 5.

 $\begin{aligned} &Yields/Ha = \lambda_{0} + \lambda_{1}RAINCHNG + \lambda_{2}TEMPCHNG + \lambda_{3}EXTREWEAT + \lambda_{4}INSECCHNG + \\ &\lambda_{5}DISEACHNG + \lambda_{6}WEEDSCHNG + \lambda_{7}AGE + \lambda_{8}GENDER + \lambda_{9}EDUC + \lambda_{10}PRIMOCCU + \\ &\lambda_{11}FARMSIZE + \lambda_{12}TENURE + \lambda_{13}FARMEXP + \lambda_{14}TRACTOR + \lambda_{15}OWNLIVE + \\ &\lambda_{16}EXTTRAIN + \lambda_{17}ADVWEATH + \lambda_{18}CREDIT + \lambda_{19}AGMEM + \lambda_{20}REGION2 + \\ &\lambda_{21}REGION3 + \lambda_{22}REGION4 + \lambda_{23}REGION5 + \lambda_{24}REGION6 + v \end{aligned}$ $\begin{aligned} &[3.7] \end{aligned}$

The dependent variable, *Yields/Ha*, is rice yields per hectare. The independent variables include: perceived changes in rainfall (*RAINCHNG*); perceived changes in temperature (*TEMPCHNG*); perceived changes in extreme weather events (*EXTREWEAT*); perceived changes in insects and pests (*INSECCHNG*); perceived changes in diseases (*DISEACHNG*); perceived

changes in weeds (*WEEDSCHNG*); age of the farmer (*AGE*); gender of farmer (*GENDER*); education level of the farmer (*EDUC*); rice farming is the primary occupation (*PRIMOCCU*); size of farm (*FARMSIZE*); ownership of farm land (*TENURE*); experience of farmer (*FARMEXP*); owned a tractor (*TRACTOR*); owned livestock (*OWNLIVE*); participated in rice extension training(s) (*EXTTRAIN*); access to advance weather information (*ADVWEATH*); access to input credit (*CREDIT*); membership in an agricultural organization(s) (*AGMEM*); non-agricultural income (*SECNAG*); soil test was done (*SOILTEST*); household member(s) help with rice farming (*HOUSEPART*); access to adequate irrigation (*ADEQIRRIG*); fully adopted the six-point practice (*SIXPOINT*); seasonal labor was employed (*LABOR*); pumped water (*PUMPWAT*); farm located in region 2 (*REGION2*); farm located in region 3 (*REGION3*); farm located in region 4 (*REGION4*); farm located in region 5 (*REGION5*); and farm located in region 6 (*REGION6*).

 $\begin{aligned} &Yields/Ha = \lambda_{0} + \lambda_{1}RAINCHNG + \lambda_{2}TEMPCHNG + \lambda_{3}EXTREWEAT + \lambda_{4}INSECCHNG + \\ &\lambda_{5}DISEACHNG + \lambda_{6}WEEDSCHNG + \lambda_{7}AGE + \lambda_{8}GENDER + \lambda_{9}EDUC + \lambda_{10}PRIMOCCU + \\ &\lambda_{11}FARMSIZE + \lambda_{12}TENURE + \lambda_{13}FARMEXP + \lambda_{14}TRACTOR + \lambda_{15}OWNLIVE + \\ &\lambda_{16}EXTTRAIN + \lambda_{17}ADVWEATH + \lambda_{18}CREDIT + \lambda_{19}AGMEM + \lambda_{20}SECNAG + \\ &\lambda_{21}SOILTEST + \lambda_{22}HOUSEPART + \lambda_{23}ADEQIRRIG + \lambda_{24}SIXPOINT + \lambda_{25}REGION2 + \\ &\lambda_{26}REGION3 + \lambda_{27}REGION4 + \lambda_{28}REGION5 + \lambda_{29}REGION6 + v \end{aligned}$

Scenario 2 is captured by equation [3.8]. In this scenario, equation [3.7] is expanded to include non-agricultural income (*SECNAG*); soil test was done (*SOILTEST*); household member(s) help with rice farming (*HOUSEPART*); access to adequate irrigation (*ADEQIRRIG*); and fully adopted the six-point practice (*SIXPOINT*).

 $\begin{aligned} &Yields/Ha = \lambda_0 + \lambda_1 RAINCHNG + \lambda_2 TEMPCHNG + \lambda_3 EXTREWEAT + \lambda_4 INSECCHNG + \\ &\lambda_5 DISEACHNG + \lambda_6 WEEDSCHNG + \lambda_7 AGE + \lambda_8 GENDER + \lambda_9 EDUC + \lambda_{10} PRIMOCCU + \\ &\lambda_{11} FARMSIZE + \lambda_{12} TENURE + \lambda_{13} FARMEXP + \lambda_{14} TRACTOR + \lambda_{15} OWNLIVE + \\ &\lambda_{16} EXTTRAIN + \lambda_{17} ADVWEATH + \lambda_{18} CREDIT + \lambda_{19} AGMEM + \lambda_{20} SECNAG + \\ &\lambda_{21} SOILTEST + \lambda_{22} HOUSEPART + \lambda_{23} ADEQIRRIG + \lambda_{24} SIXPOINT + \lambda_{25} LABOR + \\ &\lambda_{26} PUMPWAT + \lambda_{27} REGION2 + \lambda_{28} REGION3 + \lambda_{29} REGION4 + \lambda_{30} REGION5 + \\ &\lambda_{31} REGION6 + v \end{aligned}$

In scenario 3, equation [3.8] is expanded to include seasonal labor was employed (*LABOR*) and farmer pumped water (*PUMPWAT*).

3.6 Method for Analyzing Small Farmers' and Key Informants' Perceptions, Impacts, and Adaptation

Small farmers and key informants provided a narrative (i.e. qualitative) description of their perceptions, impacts, and adaptation to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. Qualitative data provide a detailed account of a phenomenon thus allowing for a more in-depth analysis. Several recent studies have successfully employed qualitative data in part or as a whole to better understand farmers' perception of climate change and their adaptive capacity (Gandure, Walker, and Botha 2013; Amdu, Ayehu, and Deressa 2013; Bryan et al. 2013; Takahashi et al. 2016; Zamasiya, Nyikahadzoi, and Mukamuri 2017; Hitayezu, Wale, and Ortmann 2017; Tripathi and Mishra 2017; Ayanlade, Radeny, and Morton 2017; Appiah et al. 2018).

A narrative approach using descriptive statistics was subsequently used to analyze the data collected. A narrative analysis involves organizing life experiences into themes or patterns that brings order and understanding in a meaningful way (Denzin and Lincoln 2011; Schutt 2014). Interview responses were first transcribed into Microsoft Excel. Here, each response was carefully read to identify keywords and phrases. Each identified keyword or phrase was subsequently coded to produce descriptive statistics of each response. At the end of the coding exercise, some codes were combined to create categories that added depth and insight. The frequency and percentage of each keyword or phrase were then tabulated. To validate the authenticity of the data collected and coded, quotes from small farmers and key informants were embedded in the analysis. In addition, photos were included to strengthen responses.

3.7 Conclusion

This chapter introduced the various data and methods used to answer the research questions posed. Climate data were obtained from the Climate Research Unit (CRU) country dataset (CRU CY v.3.24.01) at East Anglia University. The CRU CY v.3.24.01 is a gridded dataset that contains 115 years of minimum and maximum temperature data measured in degrees Celsius and precipitation data measured in millimeters (mm). The dataset has a spatial resolution of 0.5 X 0.5 degrees and is compiled in monthly, seasonal, and annual timeframes.

Farmer-level data were collected through face-to-face interviews with small farmers and key informants. Two structured questionnaires were used to collect the data. The data collected include socioeconomic, farm-level, and institutional characteristics of small farm households. Data on small farmers' and key informants' perceptions, impacts and adaptation as it relates to changes

in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds were also collected. Collected data were coded and entered into Microsoft Excel for further analysis.

To analyze climate variability in Guyana, three statistical methods were introduced: descriptive statistics, a simple trend model, and a two-sample t-test. These methods are employed to analyze the 115 years of rainfall and temperature data obtained from the CRU. The results are presented in Chapter 4. Multiple regression is used to assess the impacts of socioeconomic, farm-level, and institutional characteristics on rice yields at the farm-level. The results generated by this method are presented in Chapter 5. Descriptive statistics are used to analyze small farmers' and key informants' perceptions, impacts, and adaptation. Photos and quotes from farmers and key informants are included to strengthen the analysis. The results are presented in Chapter 6.

Chapter 4 An Overview of Climate Variability and Climate Change in Guyana

4.1 Introduction

Climate varies across countries and regions of the world. This is because geographic factors such as latitude, vegetation, altitude, and proximity to the ocean influence regional and local climate regimes (Baede et al. 2001). Although climate change is global in scale, regional variations have also been observed (Hewitson et al. 2014). Additionally, projected rainfall changes show spatial variation where some regions will observe an increase, others a decrease, and yet others little change. Similarly, temperature changes will not be regionally uniform (Collins et al. 2013). Given Guyana's geographic location and diverse land mass, analyzing temperature and rainfall data for the country will provide a better understanding of how the climate has changed locally. It will also provide insights into how rice cultivation may have been affected.

With this in mind, this chapter commences by defining and distinguishing between weather and climate as well as climate variability and climate change. A working definition of climate change is subsequently advanced. A description of the current climatic conditions in Guyana is presented. This is followed by analysis and discussion of rainfall, as well as minimum and maximum temperature over the last 115 years. Future climate change projections for the country are presented followed by some concluding remarks.

4.2 Weather and Climate

Heinlein (1973) wrote, "climate is what you expect, weather is what you get." Weather is what we experience each day. It is the state of the atmosphere at any given point in time (Fry et al. 2010) and consists of short-term variations in atmospheric conditions (NOAA, 2017). For example, air temperature, rainfall, humidity, and wind speed can vary by location and also change within hours or days at a specific location. Thus, weather is highly variable and often unpredictable

despite improvements in weather forecasting (Mann and Kump, 2015). On the other hand, climate is more predictable since it encompasses the long-term trends of weather at a specific location. In other words, climate is the statistical expression of weather conditions, including the standard deviation, mean, and extremes over an extended period of time (Fry et al. 2010).

4.3 Climate Variability and Climate Change

The basic difference between climate variability and climate change is the timescale during which each occurs. Climate variability is defined as the deviation in climate statistics (e.g., mean and standard deviation) over all temporal¹⁷ and spatial¹⁸ scales (WMO, 2017). It usually occurs over months, years, or decades (Fry et al. 2010) and is beyond that of individual weather events (Allwood et al. 2014). Rosenzweig and Hillel (2008) note, Earth's climate has always been intrinsically variable with the rates of variation also changing.

In contrast, climate change refers to the overall shift in the average climate conditions over long timescales. Such deviation usually occurs over multiple decades and/or centuries. The Intergovernmental Panel on Climate Change (IPCC) defines climate change as any change in the Earth's climate that continues for a prolonged period, usually decades or longer (Allwood et al. 2014). Changes in climate may occur naturally from internal processes such as the interaction between the oceans, atmosphere, and land masses (Rosenzweig and Hillel 2008) or external forcings such as volcanic eruptions, solar variations, anthropogenic changes in the atmosphere's composition, and land use change (Allwood et al. 2014).

Alternatively, Article 1 of the United Nations Framework Convention on Climate Change

¹⁷ Changes over time ranging from seasonal to geological (up to hundreds of millions of years) (Planton 2013).

¹⁸ Changes over geographic regions ranging from local (less than 100,000 km²), through regional (100,000 to 10 million km²) to continental (10 to 100 million km²) (Planton 2013).

(UNFCCC) defines climate change as a change in climate attributed to human activities which directly or indirectly alters the composition of the global atmosphere (Sands 1992) While the IPCC definition considers both natural and external forcings, the UNFCCC definition places greater emphasis on the anthropogenic changes that have been credited with increasing Earth's average surface temperature over many decades.

While the distinction between weather, climate, climate variability, and climate change are evident in terms of timescale, the difference between climate variability and climate change is not absolute. The very fact that the climate is changing may induce a change in variability around a shifting mean (Rosenzweig and Hillel 2008). In other words, climate change may facilitate climate variability. For the purpose of this research, climate change is defined as a statistically significant shift in average total rainfall, as well as minimum and maximum temperature for a given region over multiple decades or longer. Figure 4.1 illustrates the temporal differences among weather, climate variability, and climate change.



Figure 4.1 Timescale of weather, climate variability, and climate change Source: Author (as adapted from PCF, 2017).

4.4 Climate of Guyana

The primary controller of weather and climate in Guyana is the Inter-tropical Convergence Zone (ITCZ)¹⁹, which influences rainfall patterns resulting in two wet and two dry seasons (NATCOM1, 2002). As a result, the country experiences a moist tropical climate characterized by high temperatures and ample rainfall (NATCOM2, 2012). The El Niño-Southern Oscillation (ENSO) coupled ocean-atmosphere phenomenon also impacts the country by changing the intensity and duration of the traditional wet and dry seasons (NATCOM1, 2002). Analyzing time series data from 1916 to 2007 for Georgetown, Rama Rao et al. (2012) found that El Niño/ La Niña is directly related to monthly mean rainfall with more profound impacts during the secondary rainy season [December-January].

4.4.1 Rainfall and Climate Types

Guyana receives annual average rainfall of between 1,600 mm and 3,000 mm (NATCOM2, 2012). However, spatial variations in rainfall create three sub-climate zones: the tropical savannah, where annual rainfall is less than 1,778 mm; very wet tropical rainforest, where annual rainfall exceeds 2,728 mm; and wet/dry tropical rainforest, where annual rainfall ranges between 1,778 mm and 2,728 mm (NATCOM2, 2012). The primary rainy season extends from mid-April to the end of July, and the secondary rainy season runs from mid-November to the end of January. However, the Rupununi Savannahs, located in the southwest of the country, only receive rainfall from mid-April to August (NATCOM2, 2012). Figure 4.2 illustrates the three climate types in Guyana based on the Köppen-Geiger classification of climate: Af – tropical rainforest, Am – tropical monsoon and Aw – tropical savannah. Figure 4.3 illustrates the rainfall regimes of Guyana.

¹⁹ The areas near the equator where the northeast and southeast trade winds converge.



Figure 4.2 Köppen-Geiger climate type map of Guyana Source: (Peel, Finlayson, and McMahon 2007)



Figure 4.3 Climatic regions of Guyana Source: Velasco (2014)

4.4.2 Temperature, Wind, Sunshine, and Humidity

Annual average air temperature ranges between 16 and 34° C across the country and between 22 and 31° C along the coastal plain due to the cooling effect of the Atlantic Ocean and a northeastern trade wind of typically about 5 meters per second. (NATCOM2 2012). In the highland regions, the mean annual air temperature ranges between 20 and 23° C (McSweeney, New, and Lizcano, 2010). During the dry and wet seasons, the duration of sunshine averages seven and five hours per day, respectively (NATCOM2, 2012). Relative humidity (RH) averages at least 80% in the coastal regions, 70% in the savannah regions, and 100% in the rainforest regions (NATCOM2 2012). Utilizing the average annual rainfall and temperature data obtained from the Climate Research Unit (CRU) country dataset (CRU CY v.3.24.01) at East Anglia University, Figure 4.4 shows the mean monthly temperature and total monthly rainfall for Guyana over the last 115 years. The CRU CY dataset is derived directly from the CRU time-series (CRU TS v. 3.21), which is a gridded time-series dataset covering all land areas (excluding Antarctica) at 0.5° resolution. The spatial averages are calculated using area-weighted means (Harris et al. 2014).



Figure 4.4 Mean temperature and rainfall for Guyana (1901-2015) Source: Author

4.5 Characterizing Climate Variability in Guyana

To characterize climate variability in Guyana, three statistical methods are employed. The first method utilizes the descriptive statistics mean, standard deviation, and coefficient of variation (CV). The second method uses a linear trend model to assess the changes in rainfall, and minimum and maximum temperature over time. The third method uses a two-sample t-test to examine the change in climate variables. Livezey et al. (2007) note that numerous empirical studies and simulations indicate that the latest period of modern global warming began in the mid-1970s. As such, a two-sample t-test is used to compare average annual total rainfall (mm), average annual maximum temperature (⁰C), and average annual minimum temperature (⁰C) between 1934-1974 and 1975-2015. Although Figure 4.4 indicates 115 years of rainfall and temperature data for

Guyana, only 41 years have elapsed since modern global warming began. As such, these two periods allow for the comparison of the 41 years before and after modern global warming began.

In observing century-long data records, one must be cognizant that it is likely that more recent temperature and precipitation data are more reliable, given the increase in station density and the definite improvement in weather instrument technology. Over the last 20 years, the number rainfall stations in Guyana has increased and they have become more widely distributed. Of the 145 rainfall stations currently in existence, 116 have been in existence for less than 20 years. Figure 4.5 illustrates the density of rainfall stations in Guyana.



Figure 4.5 Rainfall Stations in Guyana Source: (Hydromet, n.d.)

4.6 Empirical Results of Climate Change and Climate Variability in Guyana

4.6.1 Evidence from Descriptive Statistics

Based on the World Metrological Organization (WMO) Climate Normals approach, the Climate Research Unit (CRU) country dataset (CRU CY v.3.24.01) is used to compute the long term (30-year) mean, standard deviation, and coefficient of variation (CV) of the average annual total rainfall, average annual minimum temperature, and average annual maximum temperature for Guyana. Climate Normals are three-decade averages of climatological variables such as rainfall and temperature (NOAA, 2018). It should be noted that the annual rainfall, minimum and maximum temperature data are aggregated for the entire country. Table 4.1 summarizes climate in Guyana over three historical 30-year periods and a more recent 25-year period.

Major Climate						
Variable	Statistical Tool	1901 - 1930	1931 -1960	1961 -1990	1991-2015	
Average Annual	Mean	2,385.31	2,558.10	2,398.01	2,439.63	
Total Rainfall	Standard Deviation	266.15	427.27	369.14	404.92	
(mm)	Coefficient of Variation (%)	11.16	16.70	15.39	16.60	
Average Annual	Mean	21.41	21.53	21.63	21.97	
Minimum	Standard Deviation	0.20	0.28	0.39	0.38	
Temperature (°C)	Coefficient of Variation (%)	0.94	1.29	1.80	1.71	
Average Annual	Mean	30.17	30.29	30.40	30.73	
Maximum	Standard Deviation	0.20	0.27	0.39	0.36	
Temperature (°C)	Coefficient of Variation (%)	0.66	0.88	1.29	1.17	
^a Computations are based on data obtained from the Climate Research Unit (CRU) country dataset (CRU CY						
v.3.24.01) at East Anglia University for the period 1901-2015.						

Table 4.1 Climate Variability in Guyana (1901-2015)^a

Table 4.1 indicates that the mean average annual total rainfall increased over the first two 30-year periods from 2,385mm (1901-1930) to 2,558mm (1931-1960). The 30-year period from 1961 to 1990 saw the mean average annual total rainfall decrease to 2,398mm. However, the most recent 25 years shows an increase in the mean average annual total rainfall over the previous 30 years. Analyzing rainfall patterns along the coastal plain of Guyana, Ramraj (1996) found that rainfall was heavier during 1941-1980 than the preceding 40 years. The standard deviation which

measures the absolute variability and coefficient of variation (CV) which measures the relative variability followed a similar pattern.

In terms of temperature, there is evidence of change over three historical 30-year periods and a more recent 25-year period. The mean for both average annual minimum temperature and average annual maximum temperature increased gradually over the four periods. However, the standard deviation and CV increased in the first three 30-year periods but decreased slightly in the most recent 25 years. It is important to note that the relative variability in average annual minimum temperature is greater than that of average annual maximum temperature over the four periods. This indicates there is greater variation associated with average annual minimum temperature.

Although the Climate Normals (three-decade averages) of average annual total rainfall, average annual minimum temperature, and annual maximum temperature provide evidence of a changing climate in Guyana over the last 115 years, a closer examination of these changes over the years is warranted. To smooth-out any short-term fluctuations and highlight long-term trends, a 5-year moving average is used to compute the mean, standard deviation, and CV from Climate Research Unit (CRU) gridded time-series country dataset (CRU CY v.3.24.01).

The moving average is computed by taking the arithmetic mean of series of 5-year subsets starting with an initial subset and shifting forward by excluding the first value in the series and including the next value in the subset. For example, the 5-year moving average of average annual total rainfall for 1905 is computed by taking the mean of average annual total rainfall for 1901, 1902, 1903, 1904, and 1905 while the 5-year moving average of average annual total rainfall for 1905 is computed by taking the mean of average of average annual total rainfall for 1905, 1906 is computed by taking the mean of average annual total rainfall for 1905, 1906 is computed by taking the mean of average annual total rainfall for 1902, 1903, 1904, 1905, and 1906. This process is continued until the last 5-year subset.

By computing the 5-year moving average, the full dataset of 115 years is reduced to 111 years. Based on the 5-year moving average, the mean, standard deviation, and CV of average annual total rainfall, average annual minimum temperature, and annual maximum temperature for the last 111 years is presented in Appendix D. The absolute and relative variability of each climate variable is more apparent when these observations are plotted over time. This is accomplished in Figures 4.6 - 4.14.

Figure 4.6 provides a visual illustration of the changes in the average annual rainfall for Guyana. Although the average annual rainfall fluctuated over the last 111 years, the overall pattern indicates a slight upward trend. There are two distinct periods of changes in rainfall patterns. The average annual rainfall increased from 1927, peaked in 1956 before experiencing a steep decline that ended in 1961. From 1965 onward, the average annual rainfall began increasing once again despite several peaks and troughs along the way.



Figure 4.6 Moving average of mean annual rainfall for Guyana Source: Author

The absolute variability in rainfall, as measured by the standard deviation, is shown in Figure 4.7. Given the increase in average annual rainfall, the standard deviation shows an

upward trend over the entire period. In particular, the period 1975-2015 shows an increase in the dispersion of annual rainfall relative to the mean. This indicates that on average, annual rainfall has fluctuated more in the last four decades.



Figure 4.7 Standard deviation of mean annual rainfall for Guyana Source: Author

Figure 4.8 depicts the coefficient of variation of annual rainfall. This measure of relative variability follows a similar pattern as that of the absolute variability reflected in Figure 4.7. Apart from a strong upward movement over the last 111 years, the degree of variation in annual rainfall has increased in the last four decades.



Figure 4.8 Coefficient of variation of annual rainfall for Guyana Source: Author

Figure 4.9 illustrates the mean annual minimum temperature over the last 111 years. Despite several peaks and troughs, a strong upward trend is evident. Of significance is the increase in the mean annual minimum temperature observed over the period 1975-2001. While the last 15-year period has seen a downturn, on average the mean annual minimum temperature has been rising steadily in the last four decades, more so than at any other period of time.



Figure 4.9 Moving average of mean annual minimum temperature for Guyana Source: Author

The behavior of the absolute variability of the annual minimum temperature is depicted in Figure 4.10. Overall, there is an upward movement across the entire period. However, the period 1970-2015 indicates that there is greater variability in minimum temperature characterized by the increase in spread observed.



Figure 4.10 Standard deviation of mean annual minimum temperature for Guyana Source: Author

The relative variability of the annual minimum temperature depicted in Figure 4.11 also indicates an increasing trend over the entire period. Between 1905-1970, the increase in the coefficient of variation of annual minimum temperature appears to be relatively smoother in comparison to later years. Thus, although increasing, the period 1970-2015 appears to be more dynamic.



Figure 4.11 Coefficient of variation of annual minimum temperature for Guyana Source: Author

The mean annual maximum temperature over the last 111 years is portrayed in Figure 4.12. The period 1905-1975 fluctuated significantly while exhibiting a slight increase in the overall average. However, from 1975 onwards, the mean annual maximum temperature shows a distinct upward trajectory coupled with a great spread. Analyzing data for Georgetown, Stephenson et al. (2014) found a significant increase in the diurnal temperature range from 1961 to 2010.



Figure 4.12 Moving average of mean annual maximum temperature for Guyana Source: Author

The absolute variability of the annual maximum temperature is depicted in Figure 4.13. While there is an overall increasing trend, the period 1905-1970 reflects a smoother upward movement in comparison to the most recent 45 years. Thus, annual maximum temperature appears to be more dynamic variable between 1970-2015.



Figure 4.13 Standard deviation of mean annual maximum temperature for Guyana Source: Author

The relative variability of annual maximum temperature illustrated in Figure 4.14 also exhibits similar characteristics as that of the absolute variability captured in Figure 4.13. The gradual increase in annual maximum temperature observed between 1905-1970 is followed by more erratic observations in the last four decades. Hence, significant variability is evident in the last 45 years despite the upward trajectory observed.



Figure 4.14 Coefficient of variation of annual maximum temperature for Guyana Source: Author

4.6.2 Evidence from Linear Trend Model

Simple linear regression is subsequently employed to show any time trend in the mean, standard deviation, and coefficient of variation (CV) of annual rainfall, minimum and maximum temperature over the last 111 years. While changes through the full historical record may not be linear in nature, linear regression does provide some insight into the general tendency of the climate variables over the record.

4.6.2.1 Annual Rainfall

In equation [4.1], the dependent variable (Y_{Rain}) is the moving average of annual rainfall while in equation [4.2], the dependent variable (Y_{SDRain}) is the standard deviation of annual rainfall. In equation [4.3], the dependent variable (Y_{CVRain}) is the coefficient of variation of annual rainfall. In each equation, the time variable (t) is the year the variable was observed and is referred to as the trend variable. Its coefficient (β_1) indicates the direction of the trend.

$$Y_{Rain} = \beta_0 + \beta_1 t + \varepsilon_t$$
[4.1]
where,

$$Y_{Rain} = \text{moving average of annual rainfall}$$

$$t = \text{year}$$

$$\varepsilon_t = \text{error term}$$
[4.2]
where,

$$Y_{SDRain} = \text{standard deviation of annual rainfall}$$

$$t = \text{year}$$

$$\varepsilon_t = \text{error term}$$

$$Y_{CVRain} = \beta_0 + \beta_1 t + \varepsilon_t \tag{4.3}$$

where, Y_{CVRain} = coefficient of variation of annual rainfall t = year ε_t = error term

The results of the linear regression for annual rainfall are presented in Table 4.2. There is no statistically significant time trend in the mean annual rainfall. However, there is a noticeable time trend in the standard deviation and coefficient of variation, both of which are statistically significant at the 95 percent probability level. Specifically, for each additional decade, the standard deviation and coefficient of variation of annual rainfall increases by 13.83mm and 0.55 percent, respectively.
		Dependent Variables	
Independent Variable			Coefficient of Variation
	Mean	Standard Deviation	(Standard Errors)
	(Standard Errors)	(Standard Errors)	
Intercept	1,418.652	-2,375.316***	-93.809***
_	(1,110.726)	(679.124)	(27.622)
Year	0.527	1.383***	0.055***
	(0.567)	(0.346)	(0.014)
Observations	111	111	111
R ²	0.008	0.1275	0.1220
F-value	0.86	15.93	15.15
Prob > F	0.355	0.0001	0.0002
*** Significant at the 1 pe	rcent level.		
** Significant at the 5 per	cent level.		
* Significant at the 10 per	cent level.		

 Table 4.2 Trend Analysis of the Variability in Annual Rainfall

4.6.2.2 Annual Minimum Temperature

In equation [4.4], the dependent variable (Y_{MinT}) is the moving average of annual minimum temperature while in equation [4.5], the dependent variable (Y_{SDMinT}) is the standard deviation of annual minimum temperature. In equation [4.6], the dependent variable (Y_{CVMinT}) is the coefficient of variation of annual minimum temperature. In each equation, the time variable (t) is the year the variable was observed and is referred to as the trend variable. Its coefficient (β_1) indicates the direction of the trend.

$$Y_{MinT} = \beta_0 + \beta_1 t + \varepsilon_t \tag{4.4}$$

where, Y_{MinT} = moving average of annual minimum temperature t = year ε_t = error term

$Y_{SDMinT} = \beta_0$	$\beta_0 + \beta_1 t + \varepsilon_t$	[4.5]	
00110101 1 0			

where, Y_{SDMinT} = standard deviation of minimum temperature t = year ε_t = error term

$$Y_{CVMinT} = \beta_0 + \beta_1 t + \varepsilon_t$$

where, Y_{CVMinT} = coefficient of variation of minimum temperature t = year ε_t = error term

Table 4.3 summarizes the results of the linear regression for annual minimum temperature. For each additional decade, the mean, standard deviation, coefficient of variation of annual minimum temperature increases by 0.06 (°C), 0.02 (°C), and 0.1 percent, respectively. Thus, there is a noticeable time trend observed for the mean, standard deviation, and coefficient of variation all of which are statistically significant at the 5 percent level.

		Dependent Variables	
Independent Variable			Coefficient of Variation
	Mean	Standard Deviation	(Standard Errors)
	(Standard Errors)	(Standard Errors)	
Intercept	10.104***	-3.937***	-17.592***
	(1.195)	(0.539)	(2.489)
Year	0.006***	0.002***	0.010 ***
	(0.0006)	(0.0003)	(0.001)
Observations	111	111	111
\mathbb{R}^2	0.460	0.355	0.342
F-value	92.84	59.99	56.58
Prob > F	0.000	0.000	0.000
*** Significant at the 1 pe	rcent level.		
** Significant at the 5 per	cent level.		
* Significant at the 10 per	cent level.		

Table 4.3 Trend Analysis of the Variability in Annual Minimum Temperature

4.6.2.3 Annual Maximum Temperature

In equation [4.7], the dependent variable (Y_{MaxT}) is the moving average of annual maximum temperature while in equation [4.8], the dependent variable (Y_{SDMaxT}) is the standard deviation of annual maximum temperature. In equation [4.9], the dependent variable (Y_{CVMaxT}) is the coefficient of variation of annual maximum temperature. In each equation, the time variable (t) is the year the variable was observed and is referred to as the trend variable. Its coefficient (β_1) indicates the direction of the trend.

$$Y_{MaxT} = \beta_0 + \beta_1 t + \varepsilon_t \tag{4.7}$$

where, Y_{MaxT} = moving average of mean annual maximum temperature t= year ε_t = error term

$$Y_{SDMaxT} = \beta_0 + \beta_1 t + \varepsilon_t$$
[4.8]

where, Y_{SDMaxT} = standard deviation of mean maximum temperature t= year ε_t = error term

$$Y_{CVMaxT} = \beta_0 + \beta_1 t + \varepsilon_t \tag{4.9}$$

where, Y_{CVMaxT} = coefficient of variation of maximum temperature t = year ε_t = error term

Table 4.4 summarizes the results of the linear regression for annual maximum temperature. There is statistically significant time trend in the mean, standard deviation, and coefficient of variation of annual maximum temperature. For each additional decade, the mean, standard deviation, coefficient of variation of annual maximum temperature increases by 0.06 (°C), 0.02 (°C), and 0.07 percent, respectively.

		Dependent Variables	
Independent Variable			Coefficient of Variation
	Mean	Standard Deviation	(Standard Errors)
	(Standard Errors)	(Standard Errors)	
Intercept	18.577***	-3.955***	-12.7***
_	(1.189)	(0.523)	(1.715)
Year	0.006***	0.002***	0.007***
	(0.001)	(0.0002)	(0.001)
Observations	111	111	111
\mathbb{R}^2	0.475	0.371	0.362
F-value	98.56	64.19	61.80
Prob > F	0.000	0.000	0.000
*** Significant at the 1 pe	ercent level.		
** Significant at the 5 per	cent level.		
* Significant at the 10 per	cent level.		

Table 4.4 Trend Analysis of the Variability in Annual Maximum Temperature

4.6.3 Evidence from Two Sample T-test

A two-sample t-test was used to compare the mean, standard deviation, and coefficient of variation for the average annual total rainfall (mm), average annual minimum temperature (0 C), and average annual maximum temperature (0 C) between 1934-1974 and 1975-2015.

In terms of rainfall, there was no statistically significant evidence that mean annual rainfall changed between the periods 1934-1974 and 1975-2015 (t= 0.88, df = 40, p>0.05). There is no statistically significant evidence that the standard deviation of the mean annual rainfall changed between the periods 1934-1974 and 1975-2015 (t= 0.37, df = 40, p>0.05). Similarly, statistically significant evidence does not exist that the coefficient of variation of the annual rainfall changed between the periods 1934-1974 and 1975-2015 (t= 0.14, df = 40, p>0.05).

For minimum temperature, there is statistically significant evidence that the mean annual minimum temperature changed between the periods 1934-1974 and 1975-2015 (t= -5.72, df = 40, p<0.05). There is also statistically significant evidence that the standard deviation of the annual minimum temperature changed between the periods 1934-1974 and 1975-2015 (t= -2.78, df = 40, p<0.05). Similarly, statistically significant evidence exists that the coefficient of variation of the

annual minimum temperature changed between the periods 1934-1974 and 1975-2015 (t= -2.64, df = 40, p<0.05).

With regards to maximum temperature, there is statistically significant evidence that the mean annual maximum temperature changed between the periods 1934-1974 and 1975-2015 (t= - 5.73, df = 40, p<0.05). Furthermore, statistically significant evidence exists that the standard deviation of the annual maximum temperature changed between the periods 1934-1974 and 1975-2015 (t= -2.92, df = 40, p<0.05). As expected, there is also statistically significant evidence that the coefficient of variation of the annual maximum temperature changed between the periods 1934-1974 and 1975-2015 (t= -2.82, df = 40, p<0.05).

4.7 Future Climate Change Projections

Using an Atmosphere-Ocean General Circulation Model (AOGCM), McSweeney et al. (2010) projected for Guyana that the mean annual rainfall is expected to decrease by 4% to 8% by 2060s and 4% to 5% by 2090s with a proportionate increase in heavy precipitation events. The mean annual temperature is likely to increase by 0.4°C to 2.0°C by 2030s, 0.9°C to 3.3°C by 2060s, and 1.4°C to 5.0°C by 2090 with a significant increase in the number of hot days and nights projected (McSweeney et al. 2010). Table 4.5 summarizes the projected future rainfall and temperature for Guyana.

Climate Variable	2030s	2060s	2090s
Mean Annual	Median: 0 to -4%	Median: -4% to -8%	Median: -4% to -5%
Precipitation change	Min-Max: -29% to 14%	Min-Max: -41% to 13%	Min-Max: -63% to 20%
% of rainfall that falls as		Median: 1% to 2%	Median: 2 to 3%
heavy precipitation events	No Data	Min-Max: -3% to 10%	Min-Max: -8% to 12%
Mean Annual temperature change	0.4°C to 2.0°C	0.9°C to 3.3°C	1.4°C to 5.0°C
Frequency of Hot Days	No Data	Median: 30% to 44% Min-Max: 18% to 56%	Median: 43% to 61% Min-Max: 19% to 79%
Frequency of Hot Nights	No Data	Median: 47% to 69% Min-Max: 33% to 90%	Median: 63% to 94% Min-Max: 49% to 99%

Table 4.5 Future Climate Scenarios for Guyana

Source: McSweeney et al. 2010.

While McSweeney et al. (2010) provide insights into the future climate of Guyana, such projections were based on the earlier Special Report on Emission Scenarios (SRES) use by the Inter-governmental Panel on Climate Change (IPCC). To see how rainfall and temperature will change under the Inter-governmental Panel on Climate Change (IPCC) Representative Concentration Pathways (RCPs)²⁰ introduced in the fifth Assessment Report (AR5), projected rainfall and near surface temperature were plotted using the Climate Change Atlas function that forms part of the Royal Netherlands Meteorological Institute (KNMI) Climate Explorer web application. The projected changes in rainfall and near surface temperature are based on the General Circulation Model (GCM) Coupled Model Intercomparison Project (CMIP5) dataset which incorporates course-resolution models and time series averaged over specific regions or countries. The dataset uses only a single realization of each climate model and weighs all models equally with the only differing characteristic being the model parameter settings (KNMI 2018). Figure 4.15 and 4.16 illustrate projected changes in rainfall and near surface temperature for three future timeframes under each RCP. The base period is 1986-2005 and the hatching represents areas

²⁰ Representative Concentration Pathways (RCPs) is the most recent scenarios used by IPCC to reflect projections of atmospheric greenhouse gas (GHG) concentrations.

where the signal is smaller than one standard deviation of natural variability (KNMI 2018). While there appears to be little projected change in rainfall, projected changes in temperature are more apparent for Guyana.



^a "RCP8.5 is a business-as-usual scenario with increasing greenhouse gas emissions over time, leading to high greenhouse gas concentration levels; RCP6.0 is a stabilization scenario in which emissions rise quickly up to 2060 and then decrease; RCP4.5 assumes quicker action to limit greenhouse gas emissions with emissions peaking in 2040 and declining strongly until 2080; and RCP2.6 describes an all-out effort to limit global warming to below 2°C with emissions decreasing sharply after 2020 and zero from 2080 onward" (Collins et al. 2013).



Figure 4.15 Projected change in precipitation (mm/day) relative to 1986–2005 Source: Author (as created through the KNMI website)



^a "RCP8.5 is a business-as-usual scenario with increasing greenhouse gas emissions over time, leading to high greenhouse gas concentration levels; RCP6.0 is a stabilization scenario in which emissions rise quickly up to 2060 and then decrease; RCP4.5 assumes quicker action to limit greenhouse gas emissions with emissions peaking in 2040 and declining strongly until 2080; and RCP2.6 describes an all-out effort to limit global warming to below 2°C with emissions decreasing sharply after 2020 and zero from 2080 onward" (Collins et al. 2013).



*Figure 4.16 Projected change in near surface temperature (*₀*C) relative to 1986–2005* Source: Author (as created through the KNMI website)

4.8 Conclusion

Three statistical methods were employed to show variability in average annual total rainfall and minimum and maximum temperature for Guyana over the last 115 years. Descriptive statistics, linear trend models, and two-sample t-tests provide strong evidence of shifts in minimum and maximum temperature. However, evidence supporting changes in rainfall over the same period remains tenuous and is supported by the projected change in precipitation for the remainder of the century as illustrated in Figure 4.15. Although descriptive statistics and linear trend models provide some evidence of changes in rainfall, the two-sample t-test failed to provide statistically significant evidence that rainfall changed. Regardless, even relatively minor changes in rainfall can have significant implications for rice cultivation.

While this is an interesting finding, one must be cognizant that the analysis was based on available aggregated rainfall data. As such, the distribution, intensity, and duration of rainfall are not captured through aggregated data. However, these variables are as important to agriculture and rice cultivation specifically, as the amount of rainfall itself. Given that the spatial variations in rainfall create three sub-climate zones in Guyana, rainfall changes in one sub-climate zone may offset rainfall changes in another sub-climate zone without drastically changing the aggregated annual rainfall for the country. For example, rainfall distribution may have shifted such that areas along the coastal sub-climate zone where rice is primarily cultivated received more rainfall over a short period of time) may have also changed across sub-climate zones. As a result, more intense rainfall events may have led to excess water that inundate rice fields and overwhelm existing drainage infrastructure. Similarly, the duration of the traditional rainy season may have also changed. An extended wet season where rainfall starts early and finishes late may affect both sowing and harvesting.

Considering the importance of these three climatological variables to rice cultivation, Chapter 6 employs farm-level data to address the limitations of the aggregated data. It will also enhance our understanding of the changes in rainfall patterns perceived by small farmers and key informants across the five primary rice-producing regions in Guyana.

Chapter 5 Multivariate Analysis of Farm-level Data

5.1 Introduction

Climate averages and aggregated yield data provide important insights into the impacts of climate change on rice yields at the national and regional levels. However, climate averages mask individual extreme weather events and microclimates which can have important effects on yields (Lobell, Cahill, and Field 2007). Similarly, aggregated yield data conceals how individual farm households are affected by climate change. Moreover, climate variables alone are not the only determinant of agricultural yields. The heterogeneity in both farm-specific biophysical and socio-economic conditions across farm households (Antle, Stoorvogel, and Valdivia 2014; Claessens et al. 2012) signals the importance of controlling for other explanatory variables (Ochieng, Kirimi, and Mathenge 2016; De Salvo, Raffaelli, and Moser 2013).

Sarker (2012) notes that the impacts of climate change on crop production varies depending on non-climatic factors. In other words, explanatory variables such as soils, infrastructure, agricultural services, and other socio-economic variables (Mendelsohn and Dinar 2005) play a pivotal role in determining yields under a changing climate. As such, this chapter uses multiple regression to explore the effects of perceived changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds together with socio-economic, farm-level, and institutional factors on rice yields. Descriptive statistics of farmers' perception of changes in rainfall, temperature, extreme weather events, insects and pests, diseases, meeds, and the various socio-economic, farm-level, and institutional factors are presented. This is followed by a summary of farmers' responses regarding the quality of inputs and equipment. Results and discussion of the multivariate analyses are presented next. The chapter concludes with a summary of the key findings.

5.2 Small Farmers' Perceptions: Rainfall, Temperature and Extreme Weather, Insects and Pests, Diseases, and Weeds

Among the 189 small farmers interviewed, 182 (96.3%) perceived changes in rainfall; 170 (89.9%) perceived changes in temperature; 169 (89.4%) perceived changes in extreme weather events; 185 (97.9%) perceived changes in insects and pests; 73 (38.6%) perceived changes in diseases; and 168 (88.9%) perceived changes in weeds.

5.3 Socio-economic Characteristics of Small Farmers

Socio-economic characteristics include farmers' age, gender, education level, and sources of income. Based on the sampling methods applied (see Chapter 3) we assume these demographics are representative of the wider population of small farmers.

5.3.1 Age, Gender, and Education

The average age of small farmers in the sample population is 51 years with the youngest and oldest farmers being 20 and 89 years, respectively. Table 5.1 summarizes the age distribution of farmers interviewed.

Age	No. of Farmers	%
Under 30 years	25	13.2
31 - 40 Years	24	12.7
41 - 50 years	37	19.6
51 - 60 years	53	28.0
61 - 70 years	38	20.1
Above 70 years	12	6.3
Total	189	100.0

Table 5.1 Age of Small Farmers

Males accounted for approximately 97 percent of the sample population. Approximately 55 percent of the farmers attended primary school while 32 percent attended high school. Less

than 10 percent of the farmers obtained a tertiary education while three farmers never attended school. Table 5.2 presents the education level reported by small farmers.

Table 5.2 Eaucation Level of Small Farmers			
Education Level	No. of Farmers	%	
Primary	103	54.5	
Community High School	8	4.2	
High School	61	32.3	
Technical/vocational	10	5.3	
University	4	2.1	
No schooling	3	1.6	
Total	189	100.0	

5 2 5 1 ... 1.00

5.3.2 Primary and Secondary Sources of Income

Rice farming is the primary source of income for 93 (49.2%) farmers. However, 181 (95.8%) farmers reported other agricultural and/or non-agricultural sources of income. Table 5.3 captures the sources of income of farmers.

Source of Income No. of Farmers % Rice only 4.2 8 Rice and other agricultural 69 36.5 Rice and non-agricultural 60 31.7 Rice, other agricultural, and non-agricultural 52 27.5 Total 189 100.0

Table 5.3 Income Source of Small Farmers

Other agricultural sources of income include cash crops²¹, livestock, and agricultural labor while old age and National Insurance Scheme (NIS) pension, services (e.g., taxi and bus driver), and other off farm employment (e.g., mechanic and welder) were the main non-agricultural sources of income. Other agricultural and non-agricultural sources of income are reflected in Table 5.4 and 5.5, respectively.

²¹ Vegetables including eggplants, bora (a type of long beans), spinach, peppers, green onions, etc.

Other Agricultural Sources of Income	No. of Farmers	%
Cash crops (e.g. peppers, spinach)	58	30.7
Livestock	57	30.2
Fruits	11	5.8
Fishery	3	1.6
Ground provisions	7	3.7
Other (e.g. laborer, tractor operator)	57	30.2

Table 5.4 Other Agricultural Sources of Income of Small Farmers

Table 5.5 Non-Agricultural Sources of Income of Small Farmers

Non-Agricultural Sources of Income	No. of Farmers	%
Pension (old age and NIS)	40	21.2
Service (e.g. taxi and bus driver)	21	11.1
Carpentry	8	4.2
Small business	8	4.2
Wholesale/ retail	5	2.6
Remittances	2	1.1
Other (e.g. mechanic, welder)	45	23.8

5.4 Farm-level Characteristics

Farm-level characteristics include the farm size, tenure status, farming experience, household participation, seasonal labor, ownership and rental of farming equipment, livestock, and soil type.

5.4.1 Farm Size and Tenure Status

The average farm size in the sample is 2.38 hectares. Farming plots ranged from 0.4 to 4.5 hectares with 59 (31.2%) farmers cultivating plots of three or more hectares. Table 5.6 provides a breakdown of farm sizes in the sample. While 146 (77.2%) farmers reported that the land was owned personally or by a family member, only 85 (44.9%) farmers reported having legal title to the land.

Hectare	No. of Farmers	%
0 - 0.9 Hectare	17	8.9
1 - 1.9 Hectares	48	25.4
2-2.9 Hectares	65	34.4
3 - 3.9 Hectares	33	17.5
4 – 4.9 Hectares	26	13.8
Total	189	100.0

Table 5.6 Farm Size by Hectare

5.4.2 Farming Experience, Household Participation, and Seasonal Labor

The average farmer had approximately 20 years of experience planting rice with 1.5 years being the least and 70 years being the most experience. Table 5.7 provides a breakdown of farmers' experience.

Table 5.7 Farming Experience	ce of Small Farmers	
Farming Experience	No. of Farmers	%
1 - 5 years	32	16.9
6 - 10 years	41	21.7
11 - 15 years	15	7.9
16 - 20 years	25	13.2
21 - 25 years	18	9.5
26 - 30 years	15	7.9
31 - 35 years	11	5.8
36 - 40 years	12	6.3
Above 40 years	20	10.6
Total	189	100.0%

ing English and Complete **T** 11 **F F**

Approximately one-third of the farmers reported that a family member helps with rice farming. The most common household members that help with rice farming are son(s), brother, and father. Despite household members helping out in the field, seasonal labor remains an important part of rice cultivation in Guyana. While land preparation and harvesting are done mechanically (e.g., tractors and combines), sowing, applying fertilizer, and spraying for weeds, insects and pests, and diseases are primarily done by seasonal laborers. As such, 162 (85.7%) farmers reported employing seasonal laborers to help with these tasks. For the other 27 (14.3%) farmers, employing part-time laborer is an additional cost that they avoid by doing the work themselves. As one farmer relate, "I do the work myself....if I had to pay [laborers], I would have nothing [profit] for myself". Table 5.8 presents the tasks farmers employ laborers to perform.

Task	No. of Farmers	%
Sowing seedlings	161	99.4
Fertilizer application	141	87.0
Spraying chemicals	121	74.7
Sowing, fertilizer, spraying	120	74.1
Sowing and fertilizer	115	71.0
Drains and water tracks	9	5.6
Rogueing of weeds	8	4.9
Weeding	5	3.1
Drying	1	0.6

Table 5.8 Tasks Performed by Seasonal Labor

5.4.3 Equipment Ownership and Rental

Among the 189 small farmers interviewed, 161 (85.2%) indicated owning farming equipment. Of those that owned farming equipment, 145 (90.1%) own a manual spray can, 87 (54.0%) own a motor blower, 33 (20.5%) own a tractor, and 23 (14.3%) own a small pump. However, only a third of these farmers reported sharing their equipment, specifically the motor blower and spray can. In terms of equipment rental, 186 (98.4%) farmers rented a combine harvester while 162 (87.1%) rented a tractor.

5.4.4 Livestock Ownership

Eighty-six (45.5%) farmers owned one or more livestock. Of these farmers, 58 (67.4%) own cows, 37 (43.0%) own sheep, 12 (14.0%) own ducks, and 10 (11.6%) own goats. Other livestock own include fowls, broilers/layers, and pigs. Table 5.9 summarizes farmers that own livestock.

Livestock	No. of Farmers	%
Cows	58	67.4
Sheep	37	43.0
Ducks	12	14.0
Goats	10	11.6
Fowls	7	8.1
Broilers/layers	5	5.8
Pigs	3	3.5

Table 5.9 Type of Livestock Owned

5.5 Institutional Accessibility

Institutional accessibility refers to farmers access to extension services, weather information, credit, insurance, and adequate irrigation.

5.5.1 Farmers' Participation in Extension Training

Ninety-two (48.7%) farmers acknowledged participating in one or more rice extension training courses. Of these, 75 (39.7%) farmers indicated that the training they participated in was based on the six-point practice. In 2008, the extension department of the Guyana Rice Development Board introduced six improved management practices for rice cultivation which are primarily based on two earlier writings on rice cultivation in Guyana: Hints for Profitable Rice Farming by Madramootoo (1969) and Some Notes on Rice Cultivation by the Research and Extension Division of the Guyana Rice Board (GRB 1976). The improved practices include advice on planting in-season (time of sowing), reducing plant densities (seed rate), treating seeds before sowing (seed treatment), controlling weeds, using balance nutrition fertilizer, and managing water in the field. Farmers were also asked separately if they had heard of the six-point practice with 110 (58%) responding in the affirmative. However, 21 (19.1%) of these farmers had only heard of the six-point practice and could not elaborate on the specific practices involved. While some farmers may have heard of the six-point practice as a whole, they may not be practicing it or fully aware of the various components. As one farmer admitted, "I heard of it but don't know the details." On

the other hand, 35 (31.8%) farmers stated that the six-point practice educates farmers on how to get better yields with 10 (9.1%) farmers declaring that the six-point practice works in that their yields had increased. Farmers also demonstrated their knowledge of the six-point practice by highlighting the specific components. Table 5.10 captures farmers' knowledge of the six-point practice.

Response	No. of Farmers	%
How to plant rice to get better yields	35	31.8
Balance nutrition (fertilizer)	28	25.5
Don't know	21	19.1
Treatment of seeds	17	15.5
Time of sowing	16	14.5
Seed rate	14	12.7
Increased yields	10	9.1
Weed control	8	7.3
Water management	8	7.3
Land preparation	6	5.5
Other	11	10.0

Table 5.10 Small Farmers' Knowledge of the Six-Point Practice

Using the list of six improve management practices, farmers were further asked to indicate which of the practices they had adopted. About half of the farmers fully adopted the six-point practice. In terms of specific practices being followed, all farmers reported that they controlled for weeds and approximately 97 percent engaged in water management and balance nutrition fertilizer. Almost 95 percent sowed in-season and 92 percent treated seeds or applied chemicals to the water in the field. Less than 60 percent of farmers reduced their seed rates. Some farmers noted that it is difficult to reduce seed rate because the land is low, and they are catering for potential losses due to flooding.

Other training courses farmers attended include paddy bug control, land preparation, fertilizer application, paddy grading, and markets for rice. Most of these training courses were

done as part of farmer field schools, field visits, and field days. The Guyana Rice Development Board (GRDB) extension department was the primary source of the training courses.

5.5.2 Weather Information in Advance

One hundred and forty (74.1%) farmers indicated that they received weather information in advanced. Television was the most popular source of weather information which accounted for 99 (52.4%) farmers. Twelve (6.3%) farmers received weather information through the radio and 17 (9.0%) farmers reported multiple sources including the internet and farming almanac. Access to accurate and timely weather information can help farmers adjust farm management practices to mitigate the impacts of climate change.

5.5.3 Input Credit and Agricultural Insurance

One hundred (52.9%) farmers received fertilizer credit. Eighty-one (42.9%) farmers indicated that the miller was the source of the credit. Supplier/distributors and large farmers provided credit to nine (4.8%) and seven (3.7%) farmers, respectively. Farmers responded that they do not have agricultural insurance and suggested that such products may not even exist in Guyana. According to the World Bank (2010) there is currently no agricultural insurance provision in Guyana and no clear national policy framework to this effect. As a result, farmers have no knowledge, experience, or awareness of agricultural insurance products and the related costs, benefits, and constraints (WB 2010a).

5.5.4 Access to Irrigation and Pumping of Water

One hundred and fifty-eight (83.6%) farmers indicated that they had proper irrigation facilities. However, 74 (39.2%) farmers reported that they had to pump water to flood their fields. This was because there is no irrigation system in their area or the water levels in the irrigation canals were not high enough to accommodate gravity flow. In a few cases, some fields were too high to allow for gravity flow from the nearby irrigation canals. While irrigation was up to standard, drainage was a major problem according to these farmers.

5.5.5 Membership in an Agricultural Organization

Thirty-three (17.5%) farmers reported that they were members of the Rice Producer Association (RPA). The RPA is generally responsible for "the protection, promotion, and advancement of the interests of rice producers"(GRPAA, 1946). Sixteen (8.5%) farmers also noted that they were members of other agricultural organizations such as the water user association and credit society. As members of an agricultural organization, farmers may have greater access to information on climate change and farm management practices that can help them successfully adapt.

5.6 Inputs and Equipment

Inputs include seeds, fertilizers, insecticides, pesticides and labor while equipment include tractor and combine used for land preparation and harvesting, respectively. The quality of inputs and equipment is also important in determining yields.

5.6.1 Seed Variety

One hundred and twenty-five (66.1%) farmers planted GRDB 10 rice variety. GRDB 10 is popular because it is a high yielding variety with a shorter time to mature. Nineteen (10.0%) farmers planted multiple varieties. This is because they had more than one plot or they rotated varieties between the two seasons in order to hedge against expected climatic conditions. For instance, they planted GRDB 10 in the traditionally short rainy season (November-December/ spring crop) given this variety's tendency to lodge because of heavy rainfall and related flooding. In the traditionally long rainy season (May-June/ autumn crop) they would plant a sturdier variety such as the GRDB 12 or 14. Hence, the choice of seed variety is an adaptation strategy employed by small farmers which is further explored in Chapter 6.

In a few cases, farmers reported mixing varieties on the same field. One farmer contends that the mixing of varieties on the same field helps to prevent lodging since a sturdier variety (e.g., GRDB 12) serves as a brace for a weaker variety (e.g., GRDB 10) in the event of rainfall induced lodging. However, this practice is ill advised since different varieties have different maturity times. If farmers harvest before the longer of the two varieties mature, the harvested yields will contain green grains which will result in a lower grade received at the mill. On the other hand, if the farmer harvest after both varieties have matured, the variety with the shorter maturity time will likely be too ripe leading to moisture loss and brittle grains once it goes through the milling process. It is also important to note that 16 (13.8%) farmers are still planting old varieties such as G98-22-4, G98-196, Rustic, G98-135 and 33-3. Although these old varieties are not disease resistant, older farmers continue to plant them because they are reluctant to change and/or because some millers still request these varieties. Table 5.11 presents the rice varieties planted.

Response	No. of Farmers	%
GRDB 10	125	66.1
Multiple	19	10.1
GRDB 14	11	5.8
G98 - 22-4	9	4.8
G98 - 196	7	3.7
Rustic	6	3.2
GRDB 12	3	1.6
G98 - 135	2	1.1
33-3	2	1.1
Other (Diwani and Brazilian)	5	2.6
Total	189	100.0

Table 5.11 Seed Variety Planted

5.6.2 Quality of Inputs and Equipment

Farmers were also asked to rate the quality of inputs and equipment. A total of 166 (87.8%) farmers rated the quality of seeds as being good or higher. This may be attributed to the fact that 182 (96.2%) farmers indicated that they bought new seeds each or every other season. In comparison, farmers were more circumspect when it came to the quality of fertilizers, insecticides, and pesticides. Fertilizers received a rating of good or higher by 148 (78.3%) farmers while insecticides and pesticides received the same rating by 147 (77.8%) farmers. In terms of labor, 158 (83.6%) farmers responded that the quality of labor was good or higher with 27 (14.3%) farmers reporting that they did the work themselves. Equipment rental received a rating of good or higher by 159 (84.1%) farmers. Table 5.12 presents farmers rating of the main inputs involved in rice farming.

	No. of Farmers										
Inputs and	Very								Very		
Equipment	Good	%	Good	%	Acceptable	%	Poor	%	Poor	%	Total
Seeds	55	29.1	111	58.7	16	8.5	6	3.2	1	0.5	189
Fertilizers	40	21.2	108	57.1	24	12.7	15	7.9	2	1.1	189
Pesticide and											
Herbicide	35	18.5	112	59.3	26	13.8	15	7.9	1	0.5	189
Labor	35	18.5	123	65.1	23	12.2	5	2.6	3	1.6	189
Equipment:											
Combine and	29	15.3	130	68.8	25	13.2	4	2.1	1	0.5	189
Tractor											

Table 5.12 Quality of Inputs and Equipment

5.7 Yields and its Determinants: Multivariate Analyses

Ordinary Least Square (OLS) is used for statistical estimation. This method is employed to determine which socio-economic, farm-level, and institutional factors influence small farmers' yields. In the absence of farm-level climate data, farmers' perception of changes in rainfall, temperature, and extreme weather events are used as explanatory variables. In addition, changes in rainfall, temperature, and extreme weather events are expected to have a positive effect on the presence of insects, diseases, and weeds (Rosenzweig et al. 2001), which are assumed to negatively impact yields. It is assumed that farmers that perceived changes in these climate variables are also likely to experience changes in their yields and subsequently change their farm management practices. Some socio-economic, farm-level, and institutional factors included in the model are identified from the literature and described in detail in Chapter 3. Other socio-economic, farm-level, and institutional explanatory variables are added based on farming practices and conditions prevalent in Guyana. Furthermore, independent variables that control for regional differences in yield are included.

5.7.1 Estimation Procedure: Heteroskedasticity and Multicollinearity

In analyzing cross-sectional data, two issues usually arise: heteroskedasticity and multicollinearity (Greene 2012; Benhin 2008; Cameron and Trivedi 2005). Heteroskedasticity occurs when the error term exhibit non-constant variance (Wooldridge 2013). In other words, the variance of the error term changes depending on the value of one or more independent variables. This results in biased standard errors which leads to bias test statistics and confidence intervals. The Breusch-Pagan test was used for checking heteroskedasticity and robust standard errors are computed to deal with this issue.

Multicollinearity refers to the successive inclusion of additional variables that increases correlation among independent variables in a multiple regression model (Wooldridge 2013). The variance inflation factor (VIF) was used to detect the problem of multicollinearity. It shows how the variance of an estimator is inflated by the presence of multicollinearity (Gujarati, Porter, and Gunasekar 2012). The higher VIF, the greater the signs of multicollinearity.

5.7.2 Results and Discussion of the Multivariate Analysis

Equation [3.7] was estimated using multiple regression. Three specification scenarios are estimated. In scenario 1, the equation contains farmers' perceptions and those characteristics that are expected to influence the small farmers' yields based on the review of literature. The variables contained in the estimated equation for Scenario 1 are described under Chapter 3.3. Keeping the equation from Scenario 1, Scenario 2 introduces additional independent variables to the equation. It was reasoned that including farmer has secondary non-agricultural sources of income (SECNAG), farmer conducted a soil test (SOILTEST), household members participated in rice

farming (HOUSEPART), farmer received adequate irrigation (ADEQIRRIG), and farmer fully implemented the six-point practice (SIXPOINT) would influence the amount of grains harvested.

According to Wooldridge (2013), while it may be appropriate to exclude certain independent variables from a model, including irrelevant independent variables in the equation has no effect on the unbiasedness of the intercept and other slope estimators. However, incorrectly excluding relevant independent variables can lead to inconsistency in the estimates (Wooldridge 2013). Building on Scenario 2, Scenario 3 included farmer employed seasonal labor (SEALAB), and farmer pumped water to flood field (PUMPWAT). These variables were added with the expectation that they would further enhance the model. Using the same scenarios, analyses are also conducted by season. Stata version 14 was used to generate the results. Table 5.13 present the definitions, mean, standard deviation, and frequency distribution for the independent variables used in the analysis; the results of the multivariate analyses are presented in Table 5.14 and 5.15.

Variable	Variable Description	Measure	Expected Sign	Mean (Std. Dev.)/ Frequency
	F F		1	n = 189
				Yes = 96.3%
RAINCHNG	Perceived changes in rainfall	1=yes and 0=no	+	No = 3.7%
				Yes = 89.9%
TEMPCHNG	Perceived changes in temperature	1=yes and 0=no	+	No = 10.1%
				Yes = 89.4%
EXTREWEAT	Perceived changes in extreme weather events	1=yes and 0=no	+	No = 10.6%
				Yes = 97.9%
INSECCHNG	Perceived changes in insects and pests	1=yes and 0=no	+	No = 2.1%
				Yes = 38.6%
DISEACHNG	Perceived changes in diseases	1=yes and 0=no	+	No = 61.4%
				Yes = 88.9%
WEEDSCHNG	Perceived changes in weeds	1=yes and 0=no	+	No = 11.1%
AGE	Age of the farmer	No. of years	+	50.6 (14.6)
				Yes = 97.9%
GENDER	Gender of the farmer	1=male and 0=female	+	No = 2.1%
		0=no school		No school = 1.6%
		1=primary		Primary = 54.5%
EDUC	Education level of the farmer	2=high school		High school = 36.5%
		3=technical/vocational	+	Technical/vocational = 4.8%
		4=university		University $= 2.6\%$
				Yes = 50.8%
PRIMOCCU	Rice farming is the primary occupation	1=yes and 0=no	+	No = 49.2%
				2 20 (1 07)
FARMSIZE	Size of farm	No. of hectares	+	2.38 (1.07)
TENHIDE		1		Y es = 44.9%
TENURE	Ownership of farm land	1=yes and 0=no	+	NO = 55.1%
EADMEND	E	Na sfaaraa	1	20.2(14.0)
FARMEAP	Experience of farmer	No. of years	Ŧ	20.3(14.9)
TDACTOD	Owned a treator	1-was and 0-no	1	1 es - 17.370 No - 92 59/
IKACIUK	Owned a tractor	1-yes and 0-no	Т	N0 = 62.376 $V_{20} = 45.5\%$
OWNI IVE	Owned livestock	1=ves and 0=no		$N_0 = 54.5\%$
OWNERVE	Owned IIVESIOCK	1-yes and 0-no	-	$V_{es} = 48.7\%$
F X T T R Δ I N	Particinated in rice extension training(s)	1=ves and 0=no	+	$N_0 = 51.3\%$
	r articipated in rice extension training(s)	i yes and 0 no	1	110 J1.J/0

Table 5.13 Definitions, Expected Signs, and Descriptive Statistics of Independent Variables

				Yes = 74.1%
ADVWEATH	Access to advance weather information	1=yes and 0=no	+	No = 25.9%
				Yes = 52.9%
CREDIT	Access to input credit	1=yes and 0=no	+	No = 47.1%
				Yes = 22.8%
AGMEM	Membership in an agricultural organization(s)	1=yes and 0=no	+	No = 77.2%
				Yes = 59.8%
SECNAG	Non-agricultural income	1=yes and 0=no	+	No = 40.2%
				Yes = 28.0%
SOILTEST	Soil test was done	1=yes and 0=no	+	No = 72.0%
				Yes = 34.9%
HOUSEPART	Household member(s) help with rice farming	1=yes and 0=no	+	No = 65.1%
				Yes = 83.6%
ADEQIRRIG	Access to adequate irrigation	1=yes and 0=no	+	No = 16.4%
				Yes = 49.7%
SIXPOINT	Fully adopted the six-point practice	1=yes and 0=no	+	No = 50.3%
				Yes = 85.7%
LABOR	Seasonal labor was employed	1=yes and 0=no	+/-	No = 14.3%
				Yes = 39.2%
PUMPWAT	Pumped water	1=yes and 0=no	+/-	No = 60.8%
REGION2	Farm located in region 2	Region=2	+/-	0.36 (0.48)
REGION3	Farm located in region 3	Region=3	+/-	0.16 (0.37)
REGION4	Farm located in region 4	Region=4	+/-	0.16 (0.37)
REGION5	Farm located in region 5	Region=5	+/-	0.16 (0.37)
REGION6	Farm located in region 6	Region=6	+/-	0.16 (0.37)

Independent	Predicted	Saanamia 1	Saanania 2	Samaria 3
variable	Relationship	Scenario I Violda (MT/ILa)	Scenario 2 Violda (MT/ILa)	Scenario 5
		(Standard Errors)	(Standard Errors)	(Standard Errors)
			(Standard Enois)	
INITEDCEDT		(1 502)	(1 590)	(1.652)
INTERCETT		(1.392)	1 207*	(1.032)
DAINCHNG	+	-1.300	(0.603)	-1.199
KAINCIINO	I	0.349	0.164	(0.700)
TEMPCHNG	+	(0.34)	(0.430)	(0.130)
	1	-0 444	-0.306	-0.316
FXTRFWFAT	+	(0.454)	(0.445)	(0.452)
		0.672	0.385	0 346
INSECCHNG	+	(0.955)	(0.940)	(0.945)
IIIIIIIIII		-0.229	-0.223	-0.234
DISEACHNG	+	(0.259)	(0.253)	(0.254)
DISERTCHING		-0.256	-0.335	-0.340
WEEDSCHNG	+	(0.387)	(0.379)	(0.381)
		-0.003	-0.005	-0.006
AGE	+	(0.012)	(0.012)	(0.012)
		1.384*	1.346*	1.376*
GENDER	+	(0.826)	(0.809)	(0.818)
		0.162	0.059	0.050
EDUC	+	(0.115)	(0.116)	(0.119)
		-0.053	-0.064	-0.072
PRIMOCCU	+	(0.267)	(0.264)	(0.270)
		-0.076	-0.078	-0.087*
FARMSIZE	+	(0.047)	(0.048)	(0.049)
		0.599**	0.536**	0.524**
TENURE	+	(0.252)	(0.249)	(0.250)
		0.009	0.003	0.004
FARMEXP	+	(0.012)	(0.012)	(0.012)
		1.114***	1.078***	1.048***
TRACTOR	+	(0.334)	(0.334)	(0.337)
		-0.579**	-0.581**	-0.572**
OWNLIVE	-	(0.246)	(0.242)	(0.243)
		-0.447*	-0.510*	-0.512*
EXTTRAIN	+	(0.257)	(0.267)	(0.268)
		-0.194	-0.071	-0.049
ADVWEATH	+	(0.286)	(0.282)	(0.284)
		0.216	0.204	0.212
CREDIT	+	(0.262)	(0.257)	(0.258)
		0.628**	0.531*	0.525*
AGMEM	+	(0.305)	(0.312)	(0.313)
			0.570**	0.547**
SECNAG	+		(0.265)	(0.268)
			0.461	0.458
SOILTEST	+		(0.292)	(0.293)
			-0.424*	-0.433*
HOUSEPART	+		(0.253)	(0.255)
			-0.518	-0.556
ADEQIRRIG	+		(0.341)	(0.357)
			0.262	0.278
SIXPOINT	+		(0.248)	(0.250)
				0.313

Table 5.14 Multivariate Analysis of Small Farmers' Yields^aIndependentPredicted

LABOR	+/-			(0.348)			
				-0.022			
PUMPWAT	+/-			(0.319)			
		1.619***	1.477***	1.424***			
REGION2	+/-	(0.389)	(0.393)	(0.484)			
		0.517	0.382	0.337			
REGION3	+/-	(0.437)	(0.448)	(0.490)			
		2.183***	2.198***	2.185***			
REGION4	+/-	(0.422)	(0.423)	(0.497)			
		0.471	0.346	0.305			
REGION5	+/-	(0.421)	(0.434)	(0.469)			
No. of Observation ^b		188	188	188			
\mathbb{R}^2		0.359	0.410	0.413			
F-value		3.99	3.95	3.69			
Prob > F		0.000	0.000	0.000			
Breusch-Pagan prob.		0.257	0.187	0.123			
Variance Inflation							
Factor (VIF)		1.54	1.57	1.69			
^a REGION6 has been dropped from the analysis.							
^b One farmer could not recall his yields for 2016.							
*** Significant at the 1 percent level.							

^{**} Significant at the 5 percent level.

In the baseline model specified (Scenario 1), several independent variables were found to have an important relationship with total rice yield measured in metric ton per hectare (TRYLDMTHA). It should be noted that this relationship suggests correlation and not causation. The statistically significant variables include small farmers perceived changes in rainfall (RAINCHNG), gender of farmer (GENDER), land tenure status (TENURE), tractor ownership (TRACTOR), livestock ownership (OWNLIVE), farmer participate in rice extension training (EXTTRAIN), and membership in an agricultural organization (AGMEM).

Perceived changes in rainfall (RAINCHNG) was negatively correlated with rice yields. In other words, rice yields were likely to be lower if a farmer observed changes in rainfall patterns. This may be due to the lack of resources to successfully adapt and/or

^{*} Significant at the 10 percent level.

ineffective adaptation practices under changing rainfall patterns. Additionally, farmers may not be able to adapt due to biophysical limits. For instance, farmers may not be able to pump out excess water if the drainage infrastructure is overwhelmed by excess rainfall.

There was a positive relationship between the gender of the farmer (GENDER) and yields. Specifically, higher yields were associated with male farmers. While this regression is not meant to explain causality, there are various possible reasons for this correlation. For one, male farmers are likely to have better access to inputs and information. Land tenure (TENURE) was positively correlated with rice yields. Farmers are more inclined to invest in their own land because property rights enhance the certainty of benefiting from capital and labor investments. This positive relationship was also found by other studies (e.g., Kurukulasuriya and Ajwad 2007; Charles 2009; Sarker 2012). Tractor ownership (TRACTOR) also has a positive relationship with rice yields. One possible explanation for this relationship, supported by qualitative responses during interviews, is that small farmers that own a tractor do not have to wait in line for their land to be prepared by a large farmer that owns machinery and equipment. As a result, small farmers can begin land preparation early thus allowing them the time to do additional dry land preparation, if needed. Dry land preparation can improve soil quality through the removal of straw and other vegetation, reduce infestation of pest and diseases and increase soil nutrient (GRB 1976). It also consumes less fuel and is important for controlling red rice, the major rice weed in Guyana. Also, being able to prepare their own land helps ensure that they sow within the suggested planting windows of May-June and December-January. Charles (2009) and Mishra, Sahu, and Sahoo (2016) found that tractor ownership had a strong positive relationship with net farm revenues.

Livestock ownership (OWNLIVE) has a negative relationship with rice yields. Livestock ownership among small rice farmers appears to be their main farming activity; in other words, farmers may be classified as small rice farmers but large livestock farmers. As noted in Table 5.9 above, 58 farmers own cows, 37 own sheep, 10 own goats. Of those that own cows, sheep and goats, 14 (24.1%) own ten or more cows, 18 (48.6%) own ten or more sheep, and 8 (80.0%) own ten or more goats. In Guyana, these are considered large livestock farmers. Kabubo-Mariara and Karanja (2007) found similar results, implying competition rather than complementarity between livestock and rice farming.

It was expected that small farmers that had access to and took advantage of training opportunities (e.g., field schools, field day, plot demonstrations, etc.) facilitated by district rice extension officers would benefit from the transfer of new information, technologies, and management practices that would help with adaptation under a changing climate. However, participation in rice extension training (EXTTRAIN) was negatively correlated with rice yields. This is an extremely curious and unexpected finding. One probable explanation is that the advice received by farmers that attended training sessions is not appropriate under current biophysical and climatic conditions. For example, farmers whose land is low and thus prone to flooding may suffer a decline in yields by reducing seed rates. Flooding may wash away seeds leading to even less seedlings remaining in the field to grow. On the other hand, the advice received may be poorly implemented thus causing more harm than good. For example, the use of balance nutrition fertilizer may be counterproductive if farmers fail to control weeds or engage in proper water management in the field. In short, farmers may need to apply extension advice holistically otherwise risk further losses. It is noteworthy that other studies (e.g., Charles 2009; Sarker 2012; Nyuor et al. (2016); Huong, Bo, and Fahad (2018) found the opposite—i.e., a positive relationship—for this variable.

In contrast to extension training, membership in an agricultural organization (AGMEM) has a positive relationship with rice yields. Small farmers that are members of one or more agricultural organizations may benefit from information sharing and farmer-to-farmer extension as it relates to new farming practices, agricultural technologies, and weather. Previous studies have reported mixed results for this variable. Wang et al. (2009) found that participation in a production association was advantageous to farmers while Sarker (2012) found it to be negatively correlated with net revenues.

Alternative model specifications were explored in scenarios two and three. Under scenario two, the equation was expanded to include secondary non-agricultural income (SECNAG), soil test (SOILTEST), household member(s) help with rice farming (HOUSEPART), access to adequate irrigation (ADEQIRRIG), and full adoption of the sixpoint practice (SIXPOINT). It was reasoned that including a variable measuring off-farm income would enhance farm household income levels and hence, the capacity to engage in additional farm management practices. A soil test would heighten awareness of soil health and quality which would help farmers make better farm management decisions at is relates to fertilizer application and water management which could increase yields. Similarly, other household members helping with rice farming were thought to enhance the quality of work performed in terms of sowing, spraying, and fertilizer application and continuous monitoring of the field for insects and pests, diseases, weeds. In addition, farmers that received adequate irrigation are more likely to benefit from better yields since water is readily available to start and finish the crop. Furthermore, farmers that fully adopted the

six improved management practices were likely to experience higher yields. Secondary non-agricultural income has a positive relationship with yields. Sarker (2012) found that higher income levels were positively correlated with net revenue from rice farming. Household member(s) participation in rice farming has a negative relationship with yields. It is likely that household members that help with rice farming may not boast the skills and experience of seasonal laborers.

Building on Scenario 2, in Scenario 3 the equation was expanded to include the use of seasonal labor (LABOR) and farmer pumped water (PUMPWAT). It was thought that yields would increase through the use of seasonal labor to sow, apply fertilizer, and spray chemicals. In comparison to household participation, seasonal laborers have broad knowledge and experience from working on different plots. However, the quality of work performed by seasonal labor may lead to lower yields. For example, seasonal laborers may do a poor job applying fertilizer or spraying for insects and pests. Farmers that pump water to flood their fields may benefit from better yields in that they are able to engage in better water management. Alternatively, pumping water may be a sign of limited water availability in the irrigation canals which may affect yields. Although neither variable added were statistically significant, one previously included variable became significant. Farm size was negatively correlated with yields. One probable explanation is that larger farms are more difficult to manage especially as it relates to controlling insects and weeds. Similar findings were reported by Kabubo-Mariara and Karanja (2007) and Closset, Dhehibi, and Aw-Hassan (2015) while contrasting findings were reported by Charles (2009), Sarker (2012), Wood and Mendelsohn (2015), Nyuor et al. (2016), and Huong, Bo, and Fahad (2018).

In the three scenarios considered, two regional variables were positively correlated with yields. Region two (REGION2) and region four (REGION4) were both statistically significant at the 1% level. As such, farms located in these regions on average produce higher yields when compared to region six, which was dropped from the analysis. There are two possible explanations for this finding. In the sample, the average farm size in region two is 2.1 hectares which is representative of the population. In general, smaller farms are considered more efficient. Also, farmers are able to better manage water and control weeds and insects on small plots. Region four is the smallest in area under cultivation. Thus, the extension officer is able to provide good coverage of the area in terms of providing advice and guidance to farmers. In fact, the extension officer for region four was recently recognized as the first Extension Officer of Excellence in the Caribbean.

The F-value for all three scenarios implies that the models were statistically significant. The goodness of fit measured by the R^2 for scenario 1, 2 and 3 were 0.36, 0.41, and 0.41, respectively. For farm-level data, this is considered reasonable (Sarker 2012). Among the considered model, Scenario 3 is the preferred specification. Apart from no evidence of multicollinearity and heteroskedasticity, this specification encompasses the most complete list of independent variables and as such is likely to result in more consistent estimates.
Variable	Relationshin	Scena	rio 1	Scena	ario 2	Scenario 3			
v al labic	iterationship	Spring	Autumn	Snring	Autumn	Spring	Autumn		
		Vields	Vields	Vields	Vields	Vields	Vields		
		(MT/Ha)	(MT/Ha)	(MT/Ha)	(MT/Ha)	(MT/Ha)	(MT/Ha)		
		(Standard	(Robust	(Standard	(Robust	(Standard	(Robust		
		Errors)	Standard	Errors)	Standard	Errors)	Standard		
		Enoiby	Errors)	Lifeib)	Errors)	Liioib)	Errors)		
		3.228***	2.050**	3.462***	2.576***	3.552***	2.422**		
INTERCEPT		(0.875)	(0.803)	(0.891)	(0.776)	(0.928)	(0.781)		
IIII DITODI I		-0.856**	-0.532	-0.773**	-0.434	-0.756*	-0.443		
RAINCHNG	+	(0.386)	(0.405)	(0.389)	(0.362)	(0.393)	(0.357)		
		0.208	0.141	0.108	0.056	0.104	0.076		
TEMPCHNG	+	(0.238)	(0.394)	(0.241)	(0.380)	(0.243)	(0.380)		
		-0.332	-0.112	-0.278	-0.028	-0.263	-0.053		
EXTREWEAT	+	(0.250)	(0.220)	(0.249)	(0.216)	(0.254)	(0.220)		
		0.211	0.462	0.095	0.290	0.096	0.250		
INSECCHNG	+	(0.525)	(0.497)	(0.527)	(0.428)	(0.531)	(0.442)		
		-0.113	-0.117	-0.108	-0.114	-0.106	-0.128		
DISEACHNG	+	(0.142)	(0.154)	(0.142)	(0.151)	(0.143)	(0.151)		
		-0.093	-0.162	-0.125	-0.209	-0.124	-0.217		
WEEDSCHNG	+	(0.213)	(0.245)	(0.212)	(0.227)	(0.214)	(0.223)		
		0.001	-0.003	0.000	-0.005	0.000	-0.006		
AGE	+	(0.007)	(0.008)	(0.007)	(0.008)	(0.007)	(0.008)		
		0.549	0.835*	0.564	0.782*	0.545	0.831*		
GENDER	+	(0.454)	(0.459)	(0.453)	(0.448)	(0.459)	(0.437)		
		0.096	0.067	0.053	0.005	0.049	-0.001		
EDUC	+	(0.063)	(0.067)	(0.065)	(0.066)	(0.067)	(0.068)		
		0.104	-0.157	0.099	-0.163	0.089	-0.161		
PRIMOCCU	+	(0.147)	(0.157)	(0.148)	(0.149)	(0.152)	(0.156)		
		-0.053**	-0.023	-0.054**	-0.024	-0.054*	-0.034		
FARMSIZE	+	(0.026)	(0.033)	(0.027)	(0.032)	(0.028)	(0.032)		
		0.309**	0.290*	0.297**	0.239	0.299**	0.225		
TENURE	+	(0.138)	(0.156)	(0.140)	(0.155)	(0.141)	(0.153)		
		0.001	0.008	0.001	0.005	-0.002	0.006		
FARMEXP	+	(0.006)	(0.007)	(0.006)	(0.007)	(0.007)	(0.008)		
		0.659***	0.455**	0.678***	0.400*	0.678***	0.369*		
TRACTOR	+	(0.183)	(0.228)	(0.187)	(0.218)	(0.189)	(0.222)		
		-0.236*	-0.344**	-0.222	-0.359**	-0.221	-0.351**		
OWNLIVE	-	(0.135)	(0.143)	(0.136)	(0.141)	(0.137)	(0.140)		
		-0.333**	-0.113	-0.329**	-0.181	-0.328**	-0.184		
EXTTRAIN	+	(0.141)	(0.159)	(0.149)	(0.168)	(0.150)	(0.165)		
		-0.016	-0.178	0.030	-0.102	0.029	-0.078		
ADVWEATH	+	(0.157)	(0.185)	(0.158)	(0.171)	(0.159)	(0.168)		
		-0.016	0.232	-0.015	0.219	-0.018	0.229		
CREDIT	+	(0.144)	(0.164)	(0.144)	(0.161)	(0.145)	(0.164)		
		0.242	0.386**	0.226	0.304**	0.226	0.299*		
AGMEM	+	(0.168)	(0.155)	(0.175)	(0.155)	(0.176)	(0.161)		
				0.243	0.327*	0.249*	0.298*		
SECNAG	+			(0.148)	(0.179)	(0.151)	(0.174)		
				0.168	0.294**	0.172	0.286*		
SOILTEST	+			(0.163)	(0.150)	(0.165)	(0.151)		
				-0.244*	-0.180	-0.246*	-0.188		
HOUSEPART	+			(0.142)	0.143	(0.143)	(0.143)		

Table 5.15 Multivariate Analysis of Small Farmers' Yield by Season^aIndependentPredicted

				-0.131	-0.387**	-0.151	-0.405**
ADEQIRRIG	+			(0.191)	(0.182)	(0.200)	(0.196)
				0.042	0.220	0.047	0.231
SIXPOINT	+			(0.139)	(0.157)	(0.141)	(0.157)
						0.003	0.309
LABOR	+/-					(0.195)	(0.270)
						-0.068	0.046
PUMPWAT	+/-					(0.179)	(0.172)
		0.619**	0.999***	0.558**	0.918***	0.499*	0.924***
REGION2	+/-	(0.214)	(0.250)	(0.220)	(0.271)	(0.272)	(0.298)
		0.114	0.403	0.067	0.315	-0.026	0.310
REGION3	+/-	(0.240)	(0.299)	(0.251)	(0.334)	(0.275)	0.325)
		1.249***	0.934***	1.249***	0.948***	1.194***	0.991***
REGION4	+/-	(0.232)	(0.265)	(0.237)	(0.289)	(0.279)	(0.312)
		0.251	0.219	0.205	0.141	0.169	0.136
REGION5	+/-	(0.231)	(0.290)	(0.243)	(0.282)	(0.263)	(0.292)
No. of							
Observation ^b		188	188	188	188	188	188
R ²		0.326	0.290	0.336	0.348	0.356	0.357
F-value		3.45	3.51	3.13	3.52	2.90	3.30
Prob > F		0.000	0.000	0.000	0.000	0.000	0.000
Breusch-Pagan							
prob.		0.766	-	0.693	-	0.734	-
Mean Variance							
Inflation Factor		1.54	1.54	1.57	1.57	1.69	1.69
(VIF)							
& DECION6 has been	drannad from	the analyzig					

^a REGION6 has been dropped from the analysis.

^b One farmer could not recall his yields for 2016.

*** Significant at the 1 percent level.

** Significant at the 5 percent level.

* Significant at the 10 percent level.

Several variables were statistically significant in the baseline model (Scenario 1) for the spring season. Land tenure status (TENURE) and tractor ownership (TRACTOR) were positively correlated with yields while perceived changes in rainfall (RAINCHNG), size of farm (FARMSIZE), livestock ownership (OWNLIVE), and farmer participate in rice extension training (EXTTRAIN) were negatively associated with yields. For the autumn season, male farmers (GENDER), land tenure status (TENURE), tractor ownership (TRACTOR), and membership in an agricultural organization(s) (AGEM) had a positive

relationship with yields. However, livestock ownership (OWNLIVE) was negatively correlated with yields.

Under scenario 2, spring yields were positively correlated with land tenure status (TENURE) and tractor ownership (TRACTOR). The size of farm (FARMSIZE), perceived changes in rainfall (RAINCHNG), farmer participate in rice extension training (EXTTRAIN), and household member(s) help with rice farming (HOUSEPART) had a negative relationship with yields. For the autumn season, male farmers (GENDER), tractor ownership (TRACTOR), and membership in an agricultural organization(s) (AGEM), secondary non-agricultural income (SECNAG), and soil test (SOILTEST) were positively correlated with yields. Livestock ownership (OWNLIVE) and adequate irrigation (ADEQIRRIG) were negatively associated with yields.

For scenario 3, spring yields were positively correlated with land tenure status (TENURE), tractor ownership (TRACTOR), and secondary non-agricultural income (SECNAG). Perceived changes in rainfall (RAINCHNG), size of farm (FARMSIZE), and farmer participate in rice extension training (EXTTRAIN) were negatively associated with yields. Autumn yields were positively correlated with male farmers (GENDER), tractor ownership (TRACTOR), membership in an agricultural organization(s) (AGEM), and soil test (SOILTEST) while livestock ownership (OWNLIVE) and adequate irrigation (ADEQIRRIG) were negatively associated with yields. For both seasons under the three scenarios considered, region two (REGION2) and region four (REGION4) were both statistically significant regional variables and had a positive relationship with yields when compared to region 6.

The F-value for both seasons under each of the three scenarios implies that the models are statistically significant. The goodness of fit measured by the R^2 under each scenario is considered reasonable. While there were no evidence of multicollinearity and heteroskedasticity for the spring season, the Breusch-Pagan test showed signs of heteroskedasticity in the autumn season. As such, robust standard errors were computed to ensure unbiased test statistics and confidence intervals.

5.8 Conclusion

The chapter employed multiple regression to determine the impacts of various socio-economic, farm-level, and institutional factors on rice yields. Three model specifications were used to estimate annual and seasonal yields. Overall, the models estimated were statistically significant. The variance inflation factor (VIF) showed little evidence of multicollinearity. However, robust standard errors were computed to address concerns of heteroskedasticity in the autumn season.

Across the three models specified, several variables were found to be statistically significant for annual and seasonal yields. Gender of farmer, land tenure status, tractor ownership, membership in an agricultural organization(s), secondary non-agricultural income, and farms located in regions two and four were found to be positively correlated with yields. Perceived changes in rainfall, farm size, livestock ownership, farmer participate in rice extension training, and household members help with rice farming were found to have a negative relationship with yields.

In the spring season, land tenure status, tractor ownership, and secondary nonagricultural income were positively correlated with yields. Perceived changes in rainfall, farm size, livestock ownership, farmer participate in rice extension training, and household member(s) help with rice farming were found to have a negative relationship with yields. In comparison, male farmers, land tenure status, tractor ownership, membership in an agricultural organization(s), secondary non-agricultural income, and soil test have a positive relationship with autumn yields. Livestock ownership and adequate irrigation were found to have a negative relationship with yields. Farms located in regions two and four also have a positive association with yields for both seasons. The spring season appears to have more factors that are negatively correlated with yields while the autumn season have more factors that are positively associated with yields. This finding is important because it highlights important differences between the two seasons which can be used to inform policy decisions.

The results presented are based on small farmers across the five main riceproducing regions in Guyana. As such, caution needs to be taken in generalizing the results across all farmers and for specific regions. Regardless, the results support several of the expected relationships and findings in the literature.

Chapter 6 Farmers' Perceptions, Impacts, and Adaptation

6.1 Introduction

Farmers have been coping with year-to-year fluctuation in climate since time immemorial. Yet, climate change poses a different kind of threat as it is expected to hasten the need for and scale of adaptation (Parry et al. 2009). Any response to climate change, however, begins with farmers' perception of the environmental changes and the observed impacts occurring in and around their farms. Perceptions (cognitive processes) of climate change are important factors in determining whether farmers are likely to undertake adaptive measures (Niles, Lubell, and Haden, 2013; Haden et al. 2012) Additionally, a better understanding of farmers' perceptions, impacts, and ongoing adaptation measures is crucial for informing policies aimed at promoting successful adaptation strategies among small farmers²².

To learn about farm-level perceptions, impacts, and adaptation, responses were solicited from both small farmers and key informants²³. It was valuable to talk to both small farmers and key informants given the different perspectives each group offers. While small farmers provide an own-farm perspective, key informants are better positioned to comment on larger patterns seen across many farms. In addition, responses from key informants may help validate information gather through small farmers. A detailed discussion of the method used for collecting this data is presented in Chapter 3.

This chapter begins with an overview of small farmers' understanding of climate change or global warming. An analysis of small farmers' perceptions, impacts, and

²² Farmers that cultivate 4.45 hectares (11 acres) or less in rice each season.

²³ Key informants comprise of district rice extension officers, an experience farmer (+25 years of experience), a rice miller, and senior staff (Chief Scientist/Plant Breeder, Plant Pathologist, Agronomist, Entomologist, and the Rice Extension Manager) at the Burma Rice Research Station in Guyana. Please refer to Chapter 3 – Data and Methods for additional information.

responses in relation to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds is presented next. This is supported by responses from key informants. The adaptation practices implemented and small farmers' ability to pay for these changes in farming practices are explored. The planting season affected, and the coping strategies employed by small farmers to mitigate the impact of losses attributed to crop failure and the high cost of inputs, low market prices, poor grades, and/or late payments by millers are discussed. Barriers and limits to adaptation are examined next. The chapter concludes with an inquiry of the service and support provided to farmers by district rice extension officers employed by the Guyana Rice Development Board (GRDB).

6.2 Small Farmers' Understanding of Climate Change

Earth's climate is changing at an unprecedented rate (Jay et al. 2018). Yet, for many people across the world, an understanding of climate change may still be a novelty. For developing countries like Guyana, emphasis on climate change education may be low on the list of priorities. However, knowledge and awareness of climate change is an important step in understanding if and/or how people are likely to respond to changes in environmental conditions attributed to climate change. As such, small farmers were asked to describe their understanding of climate change and what it meant to them. It should be noted that this was an open-ended question; farmers were not given an explicit list of possible impacts to choose from among, but rather simply asked to describe their understanding.

Of the 189 small farmers interviewed, 168 (88.9%) noted that they have heard of climate change or global warming. When asked what climate change means to them, 64

(38.1%) responded that climate change refers to a change in weather patterns. Sixty-seven (39.9%) specifically stated that climate change meant a change in rainfall or more heavy rainfall events while a similar number stated a change in temperature. A further 21 (12.5%) small farmers acknowledged that climate change meant more disasters, including drought and flooding while 20 (11.9%) referenced sea level rise or ice melting. Eleven (6.5%) small farmers responded that climate change has a negative effect on farming and livestock. Other responses by small farmers include heavy winds, increase humidity, greenhouse effect, pollution, and colder climate. Despite hearing about climate change, 18 (10.7%) small farmers did not know what it really meant. Table 6.1 summarizes small farmers' understanding of climate change.

Table 6.1 Meaning of Climate Change to Small Farmers

Meaning of Climate Change	No. of Farmers	%
Change in rainfall/ heavy rain	67	39.9
Change in temperature (hot and cold)	67	39.9
Weather pattern change	64	38.1
Disaster (including drought and flooding)	21	12.5
Sea level rise/ ice melting	20	11.9
No idea	18	10.7
Affects farming/ livestock	11	6.5
Other (heavy wind, greenhouse effect, pollution, etc.)	26	15.5

Although small farmers may not fully understand how the climate is changing, the majority had a good understanding of the changes in atmospheric conditions related to climate change. This is not surprising given that farmers have access to newspapers, radio, and/or television where national and international information and trends about climate change are frequently reported and/or discussed. The establishment of Guyana REDD+ Investment Fund (GRIF) in 2010 which supported Guyana's Low Carbon Development Strategy (LCDS), would have also caused information on climate change to be circulated in the local media.

It is also likely that district rice extension officers attached to the Guyana Rice Development Board (GRDB) have shared information with farmers regarding weather pattern changes in an effort to help them adapt. In short, small farmers' ability to identify specific environmental conditions attributed climate change was both affirming and expected given the likely multiple sources of information available in Guyana. The next several sections explore small farmers' and key informants' awareness, impacts, and responses to perceive changes in rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds.

6.3 Changes in Rainfall

Changes in rainfall patterns may pose serious challenges for small farmers. As the main source of soil moisture, rainfall is probably the most important factor determining agricultural yields (Motha 2011). Thus, the frequency, intensity, timing, and duration of rainfall (Guan et al. 2015), may have direct impacts on yields, field conditions, irrigation systems, and transportation infrastructure. As such, responses were solicited from small farmers and key informants regarding their perceptions of changes in rainfall patterns. Specific changes in rainfall, the related impacts, and changes in farm management practices in response to observed changes are discussed below.

6.3.1 Perceptions of Changes in Rainfall

Changes in rainfall patterns were perceived by 182 (96.3%) small farmers and 28 (100.0%) key informants. When asked about specific changes observed²⁴, 106 (58.2%)

²⁴ This was an open-ended question; participants were not given a specific list of changes to choose from.

small farmers and 12 (42.9%) key informants reported an increase in total rainfall in and out of season. Additionally, 80 (44.0%) small farmers and 12 (42.9%) key informants observed an increase in out-of-season rainfall alone. However, only nine (4.9%) small farmers observed rainfall increased during the traditional rainy seasons.

In a related response, small farmers and key informants also observed that weather patterns had shifted. Specifically, 43 (23.6%) small farmers and 14 (50%) key informants reported that there are "no more seasons" due in part to extended rainy seasons. The distinction between the wet and dry seasons seems to have faded since rain starts early and ends late as reported by 19 (10.4%) small farmers and 13 (46.4%) key informants. One small farmer related, "we normally expect rain in May-June but now you don't get rain....rain is out of season as rainfall patterns have shifted....the total amount of rain you get is similar but out of season."

While increases in rainfall in and/or out of season were the most popular responses by small farmers, changes in rainfall intensity was the number one response by key informants. Twenty-three (82.1%) key informants and 56 (30.8%) small farmers reported an increase in rainfall intensity. An important consideration regarding changes in rainfall relates to its intensity. That is, the amount of rainfall over a specific time period. More rainfall over a shorter time span has greater intensity and usually spells disaster for rice farmers. This is because farmlands are below sea level and draft²⁵ relies on the change in tidal flow to facilitate the movement of excess water from inundated fields via drainage canals and eventually into a river and/or the Atlantic Ocean. Given that tides change every six hours, kokers [sluices] cannot be opened if intense rainfall occurs during high tide.

²⁵ Draining of the land via the trenches and canals.

Factoring in existing drainage capacity and a poorly maintained drainage infrastructure, more rainfall over a short period of time would easily overwhelm the current drainage system even during the low tide.

Small farmers and key informants also reported that unlike years past, rainfall has become more unpredictable. As evidence of the mercurial nature of rainfall patterns in recent years, 25 (13.7%) small farmers and eight (28.6%) key informants noted that it is difficult to predict the rain. In the words of one small farmer, "yo kyant predict de weather any mo" [you can't predict the weather anymore] while another mentioned, "it is not that rain has stopped falling, it is you don't know when it will fall." In other words, rain falls when it is least expected. This may explain why the analysis of precipitation patterns discussed in Chapter 4 shows little aggregate change, yet farmers are reporting consequential shifts - it is not the average precipitation that is shifting substantially, but rather the patterns of when and with what intensity.

Furthermore, 15 (8.2%) small farmers indicated that in recent years, rainfall fluctuated between seasons and/or years. That is, in some seasons and/or years, rainfall appeared to follow historical patterns while in other seasons and/or years it increased or decreased in and out of season. One small farmer noted, "Sometimes yo get mo rain in de small crop than de big crop." In other words, rainfall during the spring crop (November – April) or short rainy season sometimes exceeds that of autumn crop [May-October] or long rainy season. This is a key observation in that historically, more rainfall occurred during the long rainy season (May-June). Rama Rao et al. (2012) found that rainfall variability was greater during the secondary rainy season [December-January] than the primary rainy season [May-July].

Excess rainfall (in comparison to the historical norm) was reported by 12 (6.6%) small farmers while another six (3.3%) mentioned that they had experienced flooding in recent years. It must be noted that excess rainfall, poorly maintained drainage, and the tidal flow collectively impact the magnitude of flooding. However, small farmers in some areas noted that even with well-maintained drainage, flooding still occurs because of the intensity of rainfall events. The sheer volume of water coupled with inadequate infrastructure to discharge excess water leads to an overflow of the waterways. As a result, there is no outlet for water in inundated rice fields to flow since often times the water in the field and in the nearby trenches and canals are at the same level.

While the majority of the respondents reported some degree of change in rainfall, 15 (8.2%) small farmers observed a decrease in rainfall while six (3.3%) reported a lack of rainfall due to drought. This is not surprising given that Guyana experienced drought as recent as 2015-2016 (NAPG, 2016). Other notable observations by small farmers and key informants include the existence of microclimates where rainfall patterns in one area differed from adjacent areas; the number of rainy days had increased; and there were cloudier/overcast conditions. Table 6.2 summarizes small farmers' and key informants' perception of changes in rainfall.

Perceptions	Small Farmers			Key Informants			
	No.	%	Rank	No.	%	Rank	
Rainfall increased (in and out of season)	106	58.2	1	12	42.9	4	
Rainfall increased (out of season)	80	44.0	2	10	35.7	5	
Rainfall intensity increased	56	30.8	3	23	81.2	1	
No more seasons/ weather patterns shifted	43	23.6	4	14	50.0	2	
Unpredictable rainfall	25	13.7	5	8	28.6	6	
Early/late rainfall	19	10.4	6	13	46.4	3	
Rainfall fluctuated between seasons	15	8.2	7	-	-	-	
Less/no rainfall	15	8.2	8	-	-	-	
Excess rainfall	12	6.6	9	1	3.6	7	
Rainfall increased (in-season)	9	4.9	10	-	-	-	
Drought (lack of rainfall)	6	3.3	11	-	-	-	
Flooding	6	3.3	11	1	3.6	7	
Other	4	2.2		5	17.9		

Table 6.2 Changes in Rainfall Reported by Small Farmers and Key Informants

Although the responses between small farmers and key informants are not exactly alike, both groups have perceived similar changes in rainfall patterns. An important observation is the increase in rainfall intensity observed by both small farmers and key informants. It provides a different perspective on the changes in rainfall patterns not captured through the analysis of aggregated rainfall data presented in Chapter 4. In Chapter 4 it was reported that although there was some evidence that rainfall has changed, such evidence was not statistically significant.

However, the increase in rainfall intensity captured through farm-level and key informant interviews shed new light on the changes in rainfall patterns. Changes in the rate and timing of heavy downpours may be masked when looking at average annual rainfall. As such, it is plausible that despite the lack of statistical significance in the changes of average annual rainfall, changes in the intensity on an hourly, daily or weekly timeframe still pose a significant risk to small farmers. Hence, this finding underscored the importance of talking to small farmers and key informants since such details are usually lost in aggregated data analyzed in Chapter 4.

6.3.2 Perceived Impacts of Changes in Rainfall

Small farmers and key informants were also asked how the observed changes in rainfall affected the quality and quantity of rice yields. Among the 182 (96.3%) small farmers that observed changes in rainfall patterns, 37 (20.3%) specifically stated that yields decreased. In terms of specific impacts, both small farmers and key informants stated that changes in rainfall affected harvesting. Ideal conditions for harvesting call for dry dams [access roads] and fields. However, 89 (48.9%) small farmers and 24 (85.7%) key informants reported that poor dams, wet fields [during harvesting], and lodged plants [plants that fell down] hindered harvesting leaves access dams in poor condition. This is because the dams are made primarily of mud, which makes traversing especially difficult for heavy-duty machinery such as tractors and combine harvesters. Even if some farmers are able to access their fields, the machinery often leaves large potholes in the aftermath, which makes access difficult for farmers that harvest their fields later in the season. Figure 6.1 illustrates the condition of an access dam in Region 6.



Figure 6.1 Poor condition of access dam Photo credit: Author

Wet fields make it difficult for combine harvesters to maneuver seamlessly. This leads to combine operators (usually large farmers) often refusing to harvest parts of the field that are too wet for fear that the machine will become stuck in the mud. As a result, rice plants/grains in wet parts of the field are left un-harvested, thereby lowering the yields of small farmers. In addition to wet fields, out of season rainfall also causes the plants to lodge or fall down in the field. As rice grains mature and approach harvesting, the plants become drier and weaker. The weakened plant stacks absorb water more easily causing the plants to fall down. This makes it difficult for the combine harvester to collect the grains as it passes through parts of the field that contains fallen plants. Thus, lodged plants are left un-harvested thereby reducing the overall yields. Figure 6.2 illustrates harvesting under wet and dry field conditions.



Figure 6.2 Harvesting under wet (left) and dry (right) field conditions Photo credit: Davindra Singh – District Rice Extension Officer

Restricted or delayed access to fields causes the rice grains to over ripen. Over ripen rice grains contain lower moisture content thus reducing the yield weight received by small farmers at the mills. Millers also penalize farmers in terms of the price they receive for paddy that are too dry. This is because the grains become brittle when it is too dry and as it passes through the various stages of milling results in broken or damaged grain that is less valuable. Even when small farmers are able to successfully harvest the rice in the fields, damaged dams can make it difficult to transport the harvested grains from the fields to the mills.

Poor farm to market roads makes access to fields and transport of yields more difficult and costlier. Farmers are forced to use grain carts which is an additional cost to bring the harvested yields out of the field. Grain carts limited capacity means that they are only used to collect and transport the grains from fields and empty into a larger capacity tractor-trailer which then transports the grains to the mill. As a result, farmers incur additional cost associated with renting a grain cart and the increase fuel required to make multiple trips to and from the field and for off-loading the grains. Figure 6.3 depicts harvesting and direct transport of grains from farm to market and a grain cart that is required to act as an intermediary between harvesting and final transport to the mill under poor farm to market road conditions.



Figure 6.3 Combine harvester and transport truck (left) and grain cart (right) Photo credit: Author

In some cases, extremely poor farm-to-market roads force farmers to use boats to bring their yield out from the fields. This entails additional time and costs since the capacity of boats is typically much smaller than trucks or grain carts and requires more labor to load and off-load the boats. Access roads that are destroyed during harvesting also causes more difficulty in terms of access to the fields when sowing begins in the following season.

A total of 80 (44.0%) small farmers and 19 (67.9%) key informants reported that rainfall during the reproductive or flowering stage "beats-off" [knocks-off] the flowers and/or pollens of the plants. This results in lower pollination rates and at harvest, more "wind paddy". Wind paddy refers to a husk with little or no grain material inside (Madramootoo 1974). From a physiological perspective, cloudy or overcast conditions during this stage may also cause the flowers to remain close further limiting pollination rates. Lower pollination rates result in more wind paddy or unfilled grains, hence lower yields.

Excess rainfall during sowing or early stages of growth also affects yields. A total of 58 (31.9%) small farmers and five (17.9%) key informants reported that excess rainfall led to flooding that negatively impacted seedlings and young rice plants. During the sowing stage, flooding causes some seedlings to float-off with the receding water. On the other hand, seedlings that remain submerged for an extended period eventually rot. Flooding also kills young rice plants that remain submerged for an extended period. One small farmer related, "Floods caused the young rice plants to melt [dissolve]." All three cases lead to a reduction in plant population in parts of the field. The end result is lower yields at harvest time.

The change in rainfall pattern also affected the timely application of fertilizer, herbicides, and pesticides as reported by 13 (7.1%) small farmers and 14 (50%) key informants. The inability to apply fertilizer at the prescribed time means that the plants are deprived of necessary nutrients needed for healthy growth. In cases where fertilizer was applied on time, sudden rainfall led to increase run-off thus depriving the plants of nutrients. On the other hand, the untimely control of weeds means increased competition for nutrients and space while delayed spraying for insects result in damaged plants and/or grains. In cases where small farmers were able to apply fertilizer and spray for weeds and insects on time, unexpected downpours wash away the fertilizer and chemicals thus leaving

the plants malnourished and unprotected from insects and weeds. Consequently, the quality and quantity of rice yields decreased.

Land preparation was also reportedly affected by changes in rainfall patterns. As such, 12 (6.6%) small farmers and 13 (46.4%) key informants reported that the changes in rainfall patterns affected proper land preparation. Rainfall leading up to sowing prevents farmers from performing a dry plow as part of proper land preparation. Farmers that decide to wait in order to perform the dry plow usually end up planting out of season which causes them to harvest in the rain. As such, the land is usually prepared under wet conditions which is costlier in terms of fuel consumption by tractors. Regardless, land preparation under such conditions is considered less effective.

Given that rice fields are lightly flooded during wet land preparation, limited rainfall affects final land preparation which in turn impacts yields because the land is not properly irrigated prior to sowing. If the field is wet during harvesting, the combine harvester usually destroys the land by leaving deep ditches as it traverses the field. This makes leveling the land costlier and more difficult for the next season. Inability to properly level the land means that some parts of the field develop furrows resulting in uneven water distribution. Uneven water levels across the field makes it difficult for seedlings to grow through parts of the field that contain deeper water. As a result, plants grow sparsely in and around these troughs resulting in lower plant population and by extension, lower yields. Figure 6.4 illustrates an unleveled field, which resulted in uneven water distribution.



Figure 6.4 Unlevelled field resulting in uneven water distribution Photo credit: Author

Despite the various impacts discussed above, 15 (8.2%) small farmers were unsure or noted that they were not affected. These farmers noted that their land is higher and thus the changes in rainfall did not really affect them because they were able to get the water out of the field. Other impacts of changes in rainfall patterns reported by small farmers and key informants include an increase in pests, diseases and weeds, lack of sunshine or more cloud cover, fertilizer run-off, grain discoloration due to wet grains at harvest, and early harvest to avoid the onset of rain led to green grains and poor quality. Table 6.3 summarizes the impact of changes in rainfall patterns reported by small farmers and key informants.

Impacts	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Affected harvesting (poor dams, wet field, plant lodges)	89	48.9	1	24	85.7	1
Knocked-off flower/pollen; lower pollination/ more						
unfilled grains	80	44.0	2	19	67.9	2
Flooded field/ affected seedling/ young plant	58	31.9	3	5	17.9	5
Reduced yields	37	20.3	4	-	-	-
Unsure/ no impact	15	8.2	5	-	-	-
Affected timely application of fertilizer and chemicals	13	7.1	6	14	50.0	3
Affected land preparation	12	6.6	7	13	46.4	4
Pests/ weeds increased	5	2.7	8	4	14.3	6
Lack of sunshine/more cloud cover	-	-	-	2	7.1	7
Fertilizer run-off	-	-	-	2	7.1	7
Other	14	7.7		5	17.9	

Table 6.3 Impacts of Changes in Rainfall Reported by Small Farmers and Key Informants

While the impacts of perceived changes in rainfall reported by both small farmers and key informants appear similar, a few notable differences exist in the responses reported. Impacts regarding timely application of fertilizer and chemicals were emphasized among key informants. Given their training, knowledge, and experience, key informants may have a better understanding of the importance of timely application of fertilizer and chemicals since applying fertilizer and chemicals outside of the prescribed windows may be ineffective and/or counter-productive to plant development. Key informants also see the impacts of untimely application of fertilizer and chemicals across different fields.

Impacts on land preparation were also highlighted by key informants as a major concern. The agronomic importance of good land preparation as it relates to water and weed management may also be better understood by key informants. Additionally, small farmers often rely on large farmers to prepare their lands. As a result, they may not be well informed regarding how changes in rainfall patterns affect land preparation. An important impact that was raised by key informants relates to fertilizer runoff due to flooding. Fertilizer run-off not only deprives plants of nutrients but also results in the pollution of waterways and the fertilization of weeds in these waterways. Again, small farmers may not be aware of these environmental impacts.

On the other hand, the flooding of young seedlings/plants was underscored by small farmers. Key informants may not view this as a very important impact since damages caused by flooding during the early stages can be overcome by farmers re-sowing their fields. However, the additional costs associated with re-sowing fields coupled with the risk of out of season harvesting are important considerations for small farmers. Despite these differences, it is apparent that both small farmers and key informants are observing similar impacts due to the perceived changes in rainfall.

6.3.3 Responses to Perceived Changes in Rainfall

Given the observed changes in rainfall patterns and the related impacts, small farmers and key informants were asked to describe any changes in farm management practices undertaken. Responses depended on the growth stage of the crop. Of the 182 (96.3%) small farmers that observed changes in rainfall patterns, 57 (31.3%) noted that they adjusted the planting dates based on water availability in the irrigation canals. In comparison, only two (7.1%) key informants noted that farmers adjust planting dates. Since rice is irrigated, the planting dates of farmers in most parts of the country depend on the release of water into the irrigation canals from major conservancies that store rainwater for agricultural use. The release of water is based on a time run system where each farming section receives water for a short period (e.g., two days) before water is made available to the next section. A culvert system is used to restrict the flow of water beyond the receiving section thus allowing water levels to build up so that farmers can open inlet pipes and allow

gravity flow to flood their fields. This is also the reason why six (3.3%) small farmers and four (14.3%) key informants mentioned that planting is done in blocks within the season. That is, farmers in each section plant together based on water availability within the traditional sowing season of May-June and December-January.

Regardless of the existing water management system however, 24 (13.2%) small farmers and 13 (46.4%) key informants noted that planting was done early in each season. These farmers noted that they paid for water to be pumped into their fields instead of waiting on the release of water through the irrigation system. Early planting provides two benefits in response to changes in rainfall patterns. It helps ensure that rice plants reach a certain height by the time rainfall steps in. This gives plants a better chance of survival under inundated conditions. It also helps farmers avoid harvesting in the rain since early sowing would lead to early harvest. Figure 6.5 shows early sowing in region 2 - Pomeroon Supenaam in May 2018.



Figure 6.5 Early sowing in region 2 - Pomeroon Supenaam Photo credit: Author

Despite the benefit of planting early, six (3.3%) small farmers and three (10.7%) key informants reported planting late because water was not available early and/or they were not able to pay for pumping. In response to excess rainfall that led to flooding, 14 (7.7%) small farmers reported that they maintained drainage often at their own expense and time and/or added more outlets to their fields to facilitate the faster draining of flood waters. This was supported by eight (28.6%) key informants. Similarly, 11 (6.0%) small farmers indicated that they opened existing outlets to drain their fields and/or waited for the water to recede. This practice was also mentioned by two (7.1%) key informants. Figure 6.6 shows a farmer cleaning a drainage canal of vegetation and an example of an outlet pipe that is used for draining fields.



Figure 6.6 A farmer cleaning a drainage canal (left) and drainage outlet pipe (right) Photo credit: Author

Pumping out excess water from the field was reported by 11 (6.0%) small farmers and eight (28.6%) key informants. It is important to note that opening of outlets, maintaining drainage, and pumping-out water are only effective when the water level in the drainage canals is lower than that of the field. Furthermore, nine (4.9%) small farmers replanted parts of the field that were damaged because of flooding.

Fourteen (50.0%) key informants stated that farmers planted different varieties. In particular, they rotated varieties between seasons. For instance, farmers may plant a high yielding but lodge-prone variety (e.g., GRDB 10) in the spring crop because less rainfall is expected and a sturdier variety (e.g., GRDB 14) in the autumn crop when more rainfall is expected based on historical norms. However, only five (2.7%) small farmers reported planting different varieties between the seasons.

Six (21.4%) key informants noted that farmers are also preparing and/or maintaining farm-to-market roads to ensure easy access to their fields during sowing and harvesting. In order to counteract flooding due to excess rainfall, farmers also build-up the embankments around their fields as reported by four (14.3%) key informants. This helps to keep overflowing irrigation and drainage canals from inundating their fields while also allowing them to pump out water effectively.

While many small farmers reported making changes to their farming practices, 27 (14.8%) reported that they did nothing. As one small farmer opined, "what can we do? That's God's work." In other words, he was emphasizing that there is nothing farmers can do about excess rainfall especially when the crop is already planted. Other responses by farmers include leveling the land, partition of land to improve water management, increasing seed density, using stickers when spraying chemicals, spraying chemicals more frequently, applying more fertilizer, drying the field earlier, farmer-to-farmer extension, investing in equipment and pooling resources, using the Internet to check weather forecast,

using a boat to access field, and monitoring the field closely. Table 6.4 presents the various responses to perceive changes in rainfall undertaken by farmers.

Responses	Small Farmers			Key Informants			
	No.	%	Rank	No.	%	Rank	
Adjusted planting date (based on water availability)	57	31.3	1	2	7.1	9	
Did nothing	27	14.8	2	-	-	-	
Planted early	24	13.2	3	13	46.4	2	
Maintained drainage and added drains	14	7.7	4	8	28.6	3	
Drained the field/ waited for water to recede	11	6.0	5	2	7.1	9	
Pumped-out water	11	6.0	6	8	28.6	3	
Re-planted field	9	4.9	7	1	3.6	11	
Pumped-in water	8	4.4	8	-	-	-	
Planted late	6	3.3	9	3	10.7	8	
Planted in blocks	6	3.3	9	4	14.3	6	
Planted different rice variety	5	2.7	11	14	50.0	1	
Maintained access roads	-	-	-	6	21.4	5	
Raised embankment/meres	-	-	-	4	14.3	6	
Other	7	3.8		8	28.6		

 Table 6.4 Responses to Perceived Changes in Rainfall

Although similarities between the responses of small farmers and key informants are evident, there are notable differences. One possible explanation is that key informants are reporting on the responses by farmers in general and not small farmers specifically. For instance, the maintenance of farm-to-market roads, maintenance of drainage, raising of embankments, and pumping out of water are most likely undertaken by large farmers that own equipment and/or could afford to pay out of pocket. Due to economies of scale, the additional effort and/or costs is a worthwhile investment for these farmers. Regardless, small farmers still benefit since they also use the same farm-to-market roads and drainage canals.

Key informants also stressed the planting of different varieties as being a major response by farmers. However, this response appears to be less popular among small farmers. Small farmers may be more resistant to change. As a result, they may continue to plant older varieties that they know about regardless of advice received from district rice extension officers. Additionally, small farmers may have accepted the risk of planting older high yielding varieties even though such varieties may be more sensitive to climate change. Furthermore, unlike large farmers that own multiple plots, most small farmers have a single plot of land. Thus, they are not able to hedge against risk by planting different varieties on different plots, which prevents them from seeing first-hand the potential of other varieties under a changing climate.

Another observable difference is the emphasis on planting early reported by key informants. This may be more appropriate for farmers with their own equipment. Small farmers' ability to plant early depends on whether their land is prepared early. Given that most small farmers rely on large farmers for land preparation, it is very difficult for small farmers to plant early since large farmers prioritize their own land preparation. Even if small farmers are able to sow early, their harvest will be delayed if their neighbors' fields are not ready for harvest. This is because combine operators will not enter an area to harvest a single field. This delays the harvest which could have negative consequences on the quality and quantity of yields because of over-ripen grains.

It is also important to note that while some small farmers responded that they did nothing, this may not be an accurate reflection of farm-level behavior. Perhaps, small farmers may have viewed their actions as being routine and ongoing even as they do gradually shift practices overtime. Actions such as maintaining drains in and around the field, draining the field, and planting in blocks can be considered regular farm management practices. However, the intensification of such actions may be even more important under a changing climate.

6.4 Changes in Temperature

Changes in temperature may also impact small farmers' yields. Temperature affects the rate of plant development (Hatfield and Prueger 2015) with temperature tolerance also varying by development stage (R. Wassmann et al. 2009). As such, small farmers and key informants were asked whether they had observed any changes in temperature in the last five years. Specific changes in temperature, the related impacts, and changes in farm management practices in response to observed changes are discussed below.

6.4.1 Perceptions of Changes in Temperature

Changes in temperature were perceived by 170 (89.9%) small farmers and 27 (96.4%) key informants interviewed. When asked how temperature has changed²⁶, 167 (98.2%) small farmers and all key informants indicated that the days have become hotter or more heated [sic]. As one key informant recalled, "field work used to be quite pleasant. You can go out without an umbrella in the sun and you won't feel the heat or you won't feel your skin burning. But for the last five years, that has changed; it is not the same. You have to go with a hat, an umbrella, a long sleeve shirt. The kind of sweating [perspiration] you do now is far more that you use to do 10, 15 years ago."

²⁶ This was an open-ended question; participants were not given a specific list of changes to choose from.

While seven (4.1%) small farmers mentioned that there has been an increase humidity, 14 (51.9%) key informants revealed that the days in particular were more humid. Five (2.9%) small farmers indicated that the temperature has been hot and cold. That is, days have gotten hotter while some nights and early mornings have become colder. Another three (1.8%) small farmers indicated that the atmosphere has become breezier and colder. Other responses by small farmers include hot days followed by rainfall and more dew in the mornings.

In addition, 12 (44.4%) key informants indicated that the temperature was hotter during the day and parts of the night while five (18.5%) reported colder nights. Hot days and cold nights resulted in more morning dew as reported by eight (29.6%) respondents. According to one respondent, "dew in the morning creates ideal conditions for fungal diseases." Table 6.5 summarizes the observation of small farmers and key informants with regards to changes in temperature.

Perceptions	Small Farmers			Key Informants					
	No.	%	Rank	No.	%	Rank			
Hotter/more heated	167	98.2	1	27	100.0	1			
Higher humidity	7	4.1	2	14	51.9	2			
Hot days and cold nights	5	2.9	3	-	-	-			
Breezy and cold	3	1.8	4	-	-	-			
Hotter days and partial nights	-	-	-	12	44.4	3			
More dew	-	-	-	8	29.6	4			
Cold nights	-	-	-	5	18.5	5			
Other	6	3.5		-	-				

Table 6.5 Changes in Temperature Reported by Small Farmers and Key Informants

In Chapter 4, descriptive statistics, a linear trend model, and a two-sample t-test used to analyze gridded temperature data for Guyana all showed increases in both maximum and minimum temperature over the last 111 years. In particular, the trend appeared to be increasing in the last two decades. Small farmers' and key informants' observations of hotter or more heated days and nights are consistent with the results and trends presented in Chapter 4. In addition to hotter days and nights, higher temperatures also increase humidity since more water vapor is needed to achieve saturation in the atmosphere. Thus, the more humid conditions that were perceived by both small farmers and key informants would be expected.

General temperature trends aside, small farmers alluded to some days being hot and nights cold. Such observations may be attributed to sunny days followed by rain in the evening and/or throughout the night and early morning. More so, such conditions are most likely in areas further inland or where homes benefit from significant tree canopy. A few small farmers also perceived that some days are breezy and cold. Such conditions are usually perceived during rainy days which are sometimes accompanied by heavy winds. While both small farmers and key informants agreed that the days have been hotter, key informants reported that nights were also hotter. This occurs primarily in the dry months and is exacerbated by traditional building materials.

6.4.2 Perceived Impacts of Changes in Temperature

The perceived impacts of changes in temperature on the quality and quantity of yields was also solicited from both small farmers and key informants. Among small farmers, 62 (36.5%) reported that the increase in temperature resulted in their fields drying out faster. Fields drying out faster was confirmed by 21 (77.8%) key informants. Higher temperature during the day causes the water in the fields to evaporate at a faster rate. In

uneven fields, high areas are the first to suffer from evaporation since the water levels are usually lower.

Regardless of faster evaporation of water, 39 (22.9%) small farmers and four (14.8%) key informants indicated that the change in temperature had no impact on yields. As one small farmer related, "as long as you get water to add back to the field, yields are not affected." In fact, eight (4.7%) small farmers reported that they harvested higher yields under hotter conditions. In addition, seven (4.1%) small farmers and three (11.1%) key informants noted that the heat was better for rice plants. These respondents link higher temperature to more sunlight and less rainy and/or overcast conditions. They argue that more sunlight is better especially during the reproductive or flowering stage since it facilitates higher pollination rates and deters the arrival of insects and pests. This results in more filled grains and hence higher quality and quantity yields. Welch et al. (2010) found that higher maximum (minimum) temperature raised (reduced) rice yields.

The temperature-sunlight nexus also enhances harvesting conditions through the timely preparation of farm-to-market roads which allows for easier and timely access to fields and transport of yields to the mills. Thus, yields increase because harvesting is expedited thereby limiting opportunities for losses due to lodging of plants. Additionally, the temperature-sunlight link helps the fields to dry faster in preparation for harvesting. Dried fields ensure that the combine harvester can traverse the entire field thus harvesting the majority of the grains. Fields that are completely harvested due of ideal conditions produces higher yields.

Despite some farmers reporting favorable yields under higher temperature conditions, 20 (11.8%) small farmers and four (14.8%) key informants reported that yields

decreased. In addition, the quality of grains under increasingly hot conditions were highlighted. Chalky²⁷ and unfilled grains were reported by six (3.5%) and five (2.9%) small farmers, respectively. In general, small farmers argued that the increase in temperature hastened the evaporation of water in the field. This lack of water slowed plant growth and as a result plants did not produce to their potential. The lack of adequate water during the grain filling stage produced chalkier and/or unfilled grains. According to Qiu et al. (2015), chalky grains are more prone to breakage during milling and thus reduces the amount of whole grains. As a result, the overall quality and quantity of yields are lower. In cases where water was available in the field, 16 (9.4%) small farmers noted that plant growth was negatively affected because the water became too hot. Hotter water in the field served as a form of heat stress on the plants.

Twenty-eight (16.5%) small farmers and 13 (48.1%) key informants reported that the increase in temperature contributed to more pest and/or diseases. Blast²⁸ and brown spot²⁹ were two diseases referenced. It is contended that hot days followed by cold nights created ideal conditions for funguses like blast and brown spot which diminish the quality and quantity of grains harvested. Even though the majority of small farmers reported one or more impacts, 18 (10.6%) did not know or had little idea of how the perceived changes in temperature affected yields.

Three (11.1%) key informants also perceived that the increase in daytime temperature made working in the field tougher. They asserted that farmers are not able to

²⁷ Grains that contain loosely packed starch granules with air spaces between them (Qiu et al. 2015).

²⁸ Blast is a fungal disease that attacks all stages of plant growth and all parts of the rice plant located above the ground (Cheaney and Jennings 1975).

²⁹ Brown spot is a fungal disease that attacks seedlings, leaves, and developing grains (Cheaney and Jennings 1975).

spend extended periods of time tending to their fields because of higher temperatures during the day. As such, farmers have to settle with visiting their fields for a few hours after dawn and a few hours before dusk each day. Similarly, the hotter climate affected the availability of field labor and the quality of work they performed since working in the sweltering heat is not an undertaking field labor are eager to perform. Other impacts observed by key informants include surfacing of salt in the land and lower weight of paddy grains. Table 6.6 presents the impacts of changes in temperature as reported by small farmers and key informants.

Impacts	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Dried out water in field	62	36.5	1	21	77.8	1
No impact on yields	39	22.9	2	4	14.8	3
Pests and/or diseases increased	28	16.5	3	13	48.1	2
Reduced yields	20	11.8	4	4	14.8	3
Don't know/ no idea	18	10.6	5	-	-	-
Water in field hotter/ affected plant growth	16	9.4	6	-	-	-
Higher yields	8	4.7	7	-	-	-
Heat was better for rice plants	7	4.1	8	3	11.1	5
Chalky and damaged grains/ lower quality	6	3.5	9	-	-	-
Unfilled grains/ wind paddy	5	2.9	10	2	7.4	7
Killed plants	3	1.8	11	-	-	-
Makes working on the farm harder	-	-	-	3	11.1	5
Other	8	4.7		5	18.5	

Table 6.6 Impacts of Changes in Temperature

The top four impacts of perceived changes in temperature reported by small farmers and key informants are similar; large proportions in both groups reported the water in the filed dried out, no impact on yields, pest and diseases increased, and reduced yields. However, differences in the responses between the two groups exist. Some small farmers may not know or understand how temperature increases are affecting rice yields. Assuming adequate water is available to replenish fields, the effects of heat stress may be minimal and/or go unnoticed. In addition, key informants made no reference to heat stress caused by water in the field becoming hotter. Assuming small farmers are able to add water to help cool the field and plants, this may not be as impactful as reported by some small farmers.

Although not highlighted by key informants, small farmers reported that higher temperatures are responsible for chalky and damaged grains. High temperatures have been found to impact grain yield and quality in rice (Shi et al. 2016; Nevame et al. 2018). It should be noted that the association of higher yields and increased temperature alluded to by small farmers is much more complex. While it is possible that temperature increase resulted in higher yields, this depends on the development stage of the plants; the heat tolerance of plants varies depending on the growth stage (Wahid et al. 2007). While the threshold temperature for grain yield in rice is approximately 34 degrees Celsius (Morita et al. 2004), Lanning et al. (2011) found that high temperatures during the grain filling stage can cause severe damage to grain quality.

6.4.3 Responses to Perceived Changes in Temperature

Data collected through the farm-level surveys suggest that small farmers' response to changes in temperature has been limited. Assuming water was available in the irrigation system, 92 (54.1%) small farmers responded by managing the water in the field. This entailed replenishing the water in the field primarily through mechanical pumping and/or keeping extra water in the field as a hedge against water loss due to evaporation. Fields that are low however are better placed to respond to increased temperature since water available in the irrigation system can be easily accessed. Adding additional water to the field was also confirmed by 22 (81.5%) key informants. Forty (23.5%) small farmers reportedly did nothing. The general sentiment among these farmers is that you cannot do anything about the heat. In the absence of a heat tolerant rice variety, there is very little farmers can do since low land rice requires adequate water supply all the way through the grain filling stage.

Interviews with key informants provided additional insights regarding responses to change in temperature. Hot days and cold nights create ideal conditions for fungal diseases. While only five (2.9%) small farmers reported spraying fungicide, eight (29.6%) key informants reported that farmers increase preventative spraying of fungicide to neutralize the threat of diseases like blast and brown spot. Furthermore, five (18.5%) key informants related that farmers are planting different rice varieties and using balance nutrition fertilizer. The use of short duration varieties ensures less water is needed while balance nutrition fertilizer provides plants with more nutrients so that they can fend off the threat of diseases. The risk of fields drying out faster also resulted in farmers monitoring their fields more frequently as reported by four (14.8%) key informants. Given that farmers do not usually reside in close proximity to their farms, increase field visits help ensure faster response to water loss in the fields.

Other responses by small farmers and key informants include lowering the seed rate, leveling the field, and using stickers³⁰. A lower seed rate at sowing ensures that the plants in the field is less clustered thus allowing air to past through especially during hot days. The more aerated the field, the less likelihood of fungal diseases. Another response by farmers involved the leveling of their fields. A level field allows for better water management since farmers can maintain an even water level across the field at sowing.

³⁰ A sticking agent that helps synthetic chemicals to remain on the plants after application.
This prevents some areas of the field from drying out faster than others. Farmers are also using stickers to apply chemicals. Since hot days and cold nights result in more morning dew, the plants are usually covered with water in the morning. Thus, the use of stickers to apply insecticide and fungicide helps ensure greater effectiveness against insects and diseases. Table 6.7 summarizes farmers' response to changes in temperature.

Responses	Small Farmers				Key Informants		
	No.	%	Rank	No.	%	Rank	
Added water to field	92	54.1	1	22	81.5	1	
Did nothing	40	23.5	2	-	-	-	
Increase preventative spraying	5	2.9	3	8	29.6	2	
Planted different varieties	2	1.2	4	5	18.5	3	
Balanced nutrition fertilizer	-	-	-	5	18.5	3	
Increase monitoring of fields	-	-	-	4	14.8	5	
Other	4	2.9		7	25.9		

Table 6.7 Responses to Perceived Changes in Temperature

For the most part, small farmers and key informants reported similar responses to perceived changes in temperatures. Depending on the soil type, location, and elevation of their fields, it is likely that some small farmers made no changes. Farms that contain pure clay soil holds water longer while those located close to the source of irrigation water and/or that are relatively low may not face difficulties in accessing water.

However, doing nothing may be exceptional since it is likely that all farmers are doing something even if it is increasing the otherwise routine work in and around their fields. This was evident in the responses from some key informants that farmers have increased monitoring of their fields. The routine nature of field visits may not seem important or worth mentioning by small farmers. In reality however, increased monitoring provides multiple benefits such as early detection with regards to water deficits and insect infestation. Key informants also indicated that more farmers are using balanced nutrient fertilizer. While some small farmers may be using balanced fertilizer, this may be more prevalent among larger farmers considering the costs. For small farmers that barely make ends meet, the immediate costs of balanced fertilizer may not outweigh the benefits at harvest. Additionally, fertilizer credit provided by some millers may be too costly and oftentimes restricts small farmers from selling their grains elsewhere.

6.5 Changes in Extreme Weather Events

Extreme events are typified by precipitation, temperature, and/or wind speed phenomena above (or below) the upper (lower) observed threshold values of what is considered normal weather (IPCC 2012). In crop agriculture, such events can cause physical damage and affect the timing and conditions of field operations (Powell and Reinhard 2015). In Guyana, extreme weather events such as excess rainfall, flooding, drought, and wind storms, pose unique challenges for small farmers. As such, small farmers and key informants were asked to describe any observed changes in extreme weather events in the last five years. Specific changes in extreme weather events, the related impacts, and changes in farm management practices in response to the observed changes are discussed below.

6.5.1 Perceived Changes in Extreme Weather Events

Observed changes in extreme weather events were reported by 169 (89.4%) small farmers and 27 (96.4%) key informants. When asked to describe how extreme weather

events have changed³¹, 136 (80.5%) small farmers and 23 (85.2%) key informants noted that the occurrence of excess rainfall led to more flooding. In addition, 56 (33.1%) small farmers and 16 (59.3%) key informants reported an increase in heavy winds. Some small farmers observed that heavy winds either occurred independent of or in concert with heavy rainfall. While the agricultural belt along the Atlantic coast of Guyana normally experience a northeastern trade wind, it appears that the combination of rainfall and heavy winds have become more noticeable.

Moreover, 110 (65.1%) small farmers experienced a drought in the last five years. One small farmer stated, "meh had to pay people fa pump wata because ah drought and den when de drought done de rain tek off and duck out the whole place" [he paid for water to be pumped into his field because of drought only to experience flooding soon after]. In comparison, 20 (74.1%) key informants reported an increase in drought. Table 6.8 summarizes extreme weather events observed by small farmers and key informants.

ingor manus						
Perceptions	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Excess rainfall/ flooding	136	80.5	1	23	85.2	1
Drought	110	65.1	2	20	74.1	2
Heavy winds	56	33.1	3	16	59.3	3
Other (high temperature)	1	0.6		-	-	

 Table 6.8 Changes in Extreme Weather Events Reported by Small Farmers and Key

 Informants

As Table 6.8 suggests, the perceived changes in extreme weather events reported by small farmers and key informants are very similar. These observations are also supported by independent observations. For instance, in the last two decades Guyana has

³¹ This was an open-ended question; participants were not given a specific list of changes to choose from.

experienced floods (2005, 2006, 2008, 2010, 2011, 2013, 2014, and 2015) and droughts (1997-1998, 2009-2010, and 2015-2016) (NAPG 2016). Therefore, there is little doubt that the observations by small farmers and key informants with regards to flooding and drought are not misplaced.

While greater emphasis is placed on reports of flooding and drought, heavy winds especially as it relates to agriculture in general and rice specifically is often overlooked and/or ignored. This is because heavy winds are not likely seen as a separate threat but as part of heavy downpours where flooding due to excess water is of greater concern. Heavy winds are often only acknowledged if there is a human toll and/or property damage. For example, in April 2018 a "freak storm" comprising of heavy winds and rainfall resulted in five injuries, and damage to four homes and power lines in region 2 (Seulall 2018). As such, the observation of heavy winds is a key finding of this research and its impacts will be explored in the next section.

6.5.2 Perceived Impacts of Changes in Extreme Weather Events

Given their observations, small farmers and key informants described the various impacts of extreme weather events on the quality and quantity of yields. Twenty-four (14.2%) small farmers specifically mentioned that extreme weather events as a whole led to lower yields. During the early growth stage, flooding hinders plant establishment and/or submerges young plants. However, farmers are able to re-sow the field albeit at an additional cost and at the risk of reaping outside the traditional harvesting window. Flooding during the later growth stages, however, is more detrimental especially after the maturation stage where grains are fully developed and curve towards the ground. During this stage flood waters make plants weak thus resulting in lodging. In parts of the field where lodging occurs, grains oftentimes go unharvested because it is difficult for a combine harvester to reach the plants especially if the field is wet.

As it relates to drought, the lack of adequate water diminishes grain filling potential which leads to less solid grains at harvest and overall lower weight at the mill. In extreme drought conditions, the lack of water in the irrigation or drainage canals mean that farmers are unable to pump water into their fields to save the crop. Regardless, seven (4.1%) small farmers reported that there was little or no impact on yields and five (3.0%) small farmers reported that they received better yields during drought. It is likely that these small farmers were located in areas close to the irrigation source and thus were able to access adequate water to pump into their fields despite experiencing drought.

In terms of specific impacts, 57 (33.7%) small farmers and seven (25.9%) key informants reported that excess rainfall and associated flooding that occurs during the early vegetative growth stage uproots and/or kills young submerged plants in lower parts of the field. In unleveled fields, lower parts are more affected by flooding since the water is deeper. Although some rice varieties are more water tolerant than others, younger plants cannot survive being waterlogged for an extended period of time (usually less than a week). As a result, farmers are often faced with the prospects of lower plant population or a thinning of plants in deeper sections of the field that remain submerged for extended periods. Flooding also delays sowing and tillering³² which usually extends the growing period resulting in harvesting occurring out of season in wet conditions. In extreme cases, the entire field maybe loss because flood waters took longer to recede. Figure 6.7 illustrates

³² Establishment of the vegetative branch of the rice plant composed of roots, culm, and leaves (GRDB 2009).

the aftermath of a field that was under water for several days which resulted in the entire field being loss.



Figure 6.7 A rice field after flood waters receded Photo credit: Author

Thirty-four (20.1%) small farmers and eight (29.6%) key informants reported that during the grain filling (milk and dough) stages, the lack of water due to drought resulted in more unfilled and/or chalky grains. This is usually uncovered when paddy is taken to the mill for sale. At the mill, samples are taken to first determine the extent of dockage for unfilled grains and other contents that are not purely rice paddy. Thus, harvested fields that contain higher quantities of unfilled grains result in a higher percent being deducted which results in lower yields.

Twenty-two (13.0%) small farmers and 13 (48.1%) key informants stated that heavy winds caused rice plants to lodge in the field and depending on the cultivated variety results in re-germination. As noted above, lodge plants are difficult to harvest and are often left in the field thus reducing the quantity of grains harvested. Furthermore, some rice varieties tend to re-germinate on the plant if it becomes wet thus resulting in lower overall quality if such grains are included in the harvest. Figure 6.8 illustrates lodging caused by heavy winds.



Figure 6.8 Heavy winds caused lodging of rice plants Photo credit: Author

Excess rainfall and related flooding also affected timely harvesting as mentioned by 13 (7.7%) small farmers. Favorable harvesting conditions necessitate dry grains and field. Any attempt to harvest under wet conditions has a direct impact of the amount of grains harvested since wet parts of the field are left unharvested. Additionally, wet grains contain more moisture and farmers are forced to engage in an additional step of drying the grains which increases the cost of production. As such, farmers prefer to allow the grains and field to dry before attempting to harvest. However, delay harvesting means that grains are not taken out at their optimal level of maturity which may also result in lower quality grains.

Apart from delaying the harvest, excess rainfall causes farm-to-market roads to deteriorate which makes timely access to the fields and transporting the harvest more difficult and costlier. As noted above, harvesting under wet conditions is less efficient since combine harvesters are usually unable to harvest all parts of the field. Combine owners/operators are also reluctant to traverse their machinery in parts of the field that is extremely wet for fear that the machinery will become stuck in the mud leading to damages and/or high repair and maintenance costs. Even if combine operators are willing to maneuver in wet fields, the mud unearthed by the tracks are inadvertently thrown on unharvested grains along the outside of the tracks. This result in these grains being loss.

Nine (5.3%) small farmers reported that drought and flooding resulted in crop failure or loss of part of field while four (2.4%) small farmers highlighted that excess rainfall and/or flooding affected the timely application of and effectiveness of fertilizer and pesticides. Other impacts reported by small farmers include lower pollination rates due to excess rainfall; seedlings float off with receding flood waters; more weeds entering the field because of overflowing irrigation and drainage canals during flooding.

Key informants provided additional insights regarding the impacts of extreme weather events. As such, five (18.5%) key informants disclosed that drought led to saltwater infiltration in the irrigation system. In some areas, the irrigation system flows directly into the Atlantic Ocean and thus is susceptible to saltwater entering the system which is exacerbated by the absence of freshwater due to drought. Other impacts reported include soil erosion due to flooding; increase insects and pests because of excess rainfall; and lower pollination rates because of excess rainfall. Table 6.9 presents the impacts reported by small farmers and key informants.

Impacts	Small Farmers			Key Informa		
	No.	%	Rank	No.	%	Rank
Submerged/ uprooted/ killed young plants - Excess rainfall/flooding	57	33.7	1	7	25.9	3
Stunted growth/ wind/chalky grains - Drought	34	20.1	2	8	29.6	2
Reduced yields	24	14.2	3	-	-	-
Lodged plants - Heavy winds/flooding	22	13.0	4	13	48.1	1
Affected harvesting - Excess rainfall/flooding	13	7.7	5	-	-	-
Crop failure/ Loss partial field - Drought and flooding	9	5.3	6	-	-	-
No/ slight impact on yields	7	4.1	7	-	-	-
Better yields - Drought	5	3.0	8	-	-	-
Affected fertilizer and pesticides application and effectiveness - Excess rainfall/flooding	4	2.4	9	-	-	-
Saltwater infiltration - Drought	-	-	-	5	18.5	4
Delayed sowing/ delayed tillering - Flooding	-	-	-	2	7.4	5
Other	6	3.6		4	14.8	

Table 6.9 Impacts of Changes in Extreme Weather Events

While the impacts of extreme weather events reported by small farmers and key informants are similar, there are some notable differences in the ranking and specific impacts. According to small farmers, the primary impact is the damage of young plants due to excess rainfall and associated flooding. However, key informants felt that lodging attributed to heavy winds and flooding are more noteworthy. This may be due to farmers' inability to effectively respond and/or recover once mature plants are lodged. As noted above, farmers at least have the option of re-sowing if damages occur in the early stages.

An important impact of extreme weather events is soil erosion. However, only one key informant reported this impact. A plausible explanation is that because farmers are increasing the use of fertilizer, the true impact of soil erosion may not be felt. Additionally, only key informants mentioned saltwater infiltration due to drought. In general, small farmers are aware of the risk of saltwater and in farming areas where there are greater risks, they often call upon the district rice extension officer to test the salinity level of the water in the irrigation canals before irrigating their fields.

6.5.3 Responses to Perceived Changes in Extreme Weather Events

Over the last five years, farmers engaged in several responses in relation to extreme weather events. Under drought conditions, 58 (34.3%) small farmers and 13 (48.1%) key informants reported that pumping water into the field was a major response. This is assuming water was available in the irrigation canals. It must be noted that farmers whose fields are located along the main irrigation canals and close to the source of the water are at an advantage during dry spells. This is because they have first access to pump the limited water available in the system. Farms located further down the line are less likely to benefit.

In response to flooding, 31 (18.3%) small farmers and ten (37.0%) key informants noted that excess water was pumped out of the field. Another response reported by 23 (13.6%) small farmers and five (18.5%) key informants is the creation of additional or clearing existing drainage to allow for the smooth flow of flood waters based on the tidal change. Farmers are also maintaining drainage through self-help or at their own expense. However, making drainage and pumping excess water out are only effective if the particular field is high and/or drainage system can accommodate the excess water.

Additionally, ten (5.3%) small farmers and three (11.1%) key informants noted that the replanting of fields and/or sowing more seeds is being done in response to flooding. Nine (5.3%) small farmers and four (14.8%) key informants reported that farmers just waited for the tide change to drain the field. In other instances, farmers are coordinating with the koker [sluice] attendants to ensure timely opening of the sluice doors to allow excess water to recede.

Despite the different adaptation responses, 43 (25.4%) small farmers and two (7.4%) key informants noted that nothing could be done about extreme weather events. In

particular, these respondents disclosed that during the drought, there is no water to pump into the fields or only saltwater was available. With regards to flooding, pumping out water is difficult because everywhere was flooded and the drainage canals were overflowing. As such, there was nowhere for the water to retreat. Figure 6.9 illustrates a flooded rice field and overflowing drainage canal almost at the same water level.



Figure 6.9 Drainage canal (right) overflowing rice field (left) Photo Credit: Davindra Singh – District Rice Extension Officer

Although small farmers maintained that there was nothing they could do about heavy winds, two (7.4%) key informants noted that some farmers are planting trees to serve as windbreakers to reduce the risk of plants lodging. While the planting of trees is a good idea in theory, the mere size and layout of farms would require significant investment in trees to be effective. As such, this response is likely to be impractical and/or prohibitively costly for small farmers.

Key informants also provided additional insights regarding adaptation to extreme weather events. Ten (37.0%) key informants noted that farmers planted different varieties and improved water management. Specifically, farmers plant short duration varieties to avoid harvesting in the rain or sturdier varieties that can withstand lodging caused by excess rainfall and heavy winds. In terms of water management, farmers are dividing their field into smaller plots to help ensure an even distribution of water across the field. This is also facilitated by leveling their lands during land preparation. They are also building up meres and embankments to help keep flood waters out.

Other responses to extreme weather events alluded to by small farmers and key informants include using boats to access fields because access roads were in a poor state; increase seed rate to compensate for losses due to flooding; abandoning the field; planting early to avoid bad weather; adding more outlets to drain inundated fields faster; and using balance nutrition fertilizer so that plants become sturdier to withstand flooding and lodging. Table 6.10 presents farmers' responses to extreme weather events.

Responses	Small Farmers			Key Informan		
	No.	%	Rank	No.	%	Rank
Pumped water into field - Drought	58	34.3	1	13	48.1	1
Did nothing	43	25.4	2	2	7.4	8
Pumped water out of field - Flooding	31	18.3	3	10	37.0	2
Created/cleared drainage - Flooding	23	13.6	4	5	18.5	5
Replanted/ sowed more seeds - Flooding	10	5.9	5	3	11.1	7
Co-ordinate with sluice attendant/ waited on tide to drain						
- Flooding	9	5.3	6	4	14.8	6
Planted different variety - Flooding and heavy winds	-	-	-	10	37.0	2
Improved water management - Drought	-	-	-	10	37.0	2
Planted trees – Heavy winds	-	-	-	2	7.4	8
Other	14	8.3		8	29.6	

Table 6.10 Responses to Perceived Changes in Extreme Weather Events

The responses to extreme weather reported by key informants mesh well with those reported by small farmers. However, two noticeable differences are evident. Key informants noted that farmers are planting different rice varieties to combat flooding and heavy winds. While small farmers have indicated planting different varieties, they may not view this as a direct response to flooding or heavy winds. From a small farmer's perspective, planting different varieties may be more in response to the promise of higher yields. Key informants also mentioned farmers are improving water management in response to drought. These farmers are likely to be large farmers with the resources and equipment to undertake water management. Unlike small farmers, large farmers often own pumps used for irrigation, and tractors and implements needed for leveling the field and building embankments and meres.

6.6 Changes in Insects and Pests

Climatic conditions play an important role in the distribution, development, and population dynamics of insects (Lamichhane et al. 2015). According to Walthall et al. (2012), higher temperatures and changes in rainfall patterns will alter environmental conditions that will affect insect population and distribution. Deutsch, Tewksbury, and Tigchelaar (2018) note that warming increases the population growth and metabolic rates of insects. As such, small farmers and key informants were asked whether they had observed any changes in insect and pest populations and impacts over the last five years. Specific changes, the related impacts, and changes in farm management practices in response to the observed changes are discussed below.

6.6.1 Perceived Changes in Insect and Pest Populations and Impacts

One hundred and eighty-five (97.9%) small farmers and all key informants interviewed indicated that early, mid, and/or late season pests have changed. Early season pests include the water weevil (*Helodytes foveolatus*), leaf miner (*Hydrellia sp.*), caterpillar

(Spodoptera frugiperda), and snail (Pomacea sp.) while mid-season pests include the caterpillar (Spodoptera frugiperda) and stem borer (Rupela albinella). The major late season pests are the paddy bug³³ (Oebalus poecilus) and plant hopper (Tagosodes orizicolus).

When asked to describe the insects and/or pests that have changed³⁴, 51 (27.6%) small farmers and ten (35.7%) key informants noted that water weevil increased while 12 (6.5%) small farmers and six (21.4%) key informants indicated an increase in root worm. The root worm is the larvae of the water weevil. Increases in snail was reported by 49 (26.5%) small farmers and 15 (53.6%) key informants while caterpillar increase was reported by 48 (25.9%) small farmers and nine (32.1%) key informants.

An increase in the leaf miner or heart worm was reported by 42 (22.7%) small farmers and nine (32.1%) key informants. It is interesting to note that 20 (10.8%) small farmers and five (17.9%) key informants reported seeing slugs in their fields for the first time in the last five years. Key informants noted that the appearance of slugs is relatively new, and much is not yet known about this pest. Increases in the stem borer moth was reported by 15 (8.1%) small farmers and four (14.3%) key informants.

The major rice pest in Guyana is the paddy bug. As such, 181 (97.8%) small farmers and 23 (82.1%) key informants noted increase presence of paddy bugs rice fields. One small farmer bellowed, "in meh whole life of 61 years, I neva see paddy bugs like dis ...paddy bug increased 1,000 fold" [in his entire life he had never observed so many paddy bugs like he did in the 2018 spring season]. However, three (10.7%) key informants noted

³³ Also referred to as bush bug, gandhi, or stink bug.

³⁴ This was an open-ended question; participants were not given a specific list of changes to choose from.

that paddy bug infestation fluctuates from season to season depending on the prevailing climatic conditions.

Small farmers also stated that flies, some of which they could not identify are appearing for the first time in their fields. Other pests reported by small farmers and key informants include birds, wild ducks, grasshoppers, rats, plant hopper, and beetles. According to small farmers, these insects and pests are becoming more common in their rice field than ever before. Figure 6.10 illustrates the major pests observed by farmers. Despite an overwhelming majority of small farmers indicating that insects and pests have increased, five (2.6%) small farmers and two (7.1%) key informants reported a decrease in insects and pests. It is contended that increase chemical control of pests over the years have resulted is less pests being observed.



Figure 6.10 Paddy bug (top left), snail (top right), adult water weevil (center left), water weevil larvae (center right), caterpillar (bottom left) and leaf miner (bottom right) Source: Rice Farmer's Manual (GRDB 2009)

Apart from the insects and pests noted above, according to one key informant, two new pests have shown up in rice fields in Guyana. In the last two seasons, the brown planthopper (*Nilaparvata lugens*) was observed in rice fields where there was 80 and in one case 90 percent damage recorded. The other pest is the rice leaffolder (*Cnaphalocrocis medinalis*) which was previously never recorded in Guyana. Figure 6.11 depicts the brown planthopper and rice leaffolder. Table 6.11 summarizes the changes in insects and pests reported by small farmers and key informants.





Figure 6.11 Brown planthopper (left) and leaffolder (right) Source: International Rice Research Institute (IRRI)

Perceptions	Small Farmers			Key Informan		
	No.	%	Rank	No.	%	Rank
Paddy bug increased	181	97.8	1	23	82.1	1
Water weevil increased	51	27.6	2	10	35.7	3
Snail increased	49	26.5	3	15	53.6	2
Caterpillar increased	48	25.9	4	9	32.1	4
Leaf miner/ heart worm increased	42	22.7	5	9	32.1	4
Slug increased	20	10.8	6	5	17.9	8
Flies increased	16	8.6	7	-	-	-
Stem borer moth increased	15	8.1	8	4	14.3	9
Root worm increased	12	6.5	9	6	21.4	6
Insects and pests decreased	5	2.7	10	2	7.1	11
Grasshopper increased	-	-	-	6	21.4	6
Paddy bug fluctuated	-	-	-	3	10.7	10
Birds and wild ducks increased	-	-	-	2	7.1	11
Rat/ rodent increased	-	-	-	2	7.1	11
Other insects/pests increased	14	7.6		7	25.0	

 Table 6.11 Changes in Insects and Pests Reported by Small Farmers and Key Informants

Changes in insect and pest populations observed by small farmers and key informants are similar. However, an important observation by a few key informants is that paddy bug infestations are correlated with changes in rainfall patterns. They asserted that in seasons and/or years with heavy rainfall, paddy bug infestation is more severe. One possible explanation is that wetter conditions disturb paddy bugs' natural habitat and/or food supply which leads to the invasion of rice fields.

While changes in weather patterns may be indirectly responsible for the observed changes in insects and pests, consideration should also be given to a few other factors. The cultivation of uncertified rice varieties originating from neighboring Brazil and Suriname may also be playing a role in the spread of insects and pests. Guyana's porous national borders are conducive to smuggling, which has facilitated the introduction of uncertified varieties into the rice sector; insects and diseases may be accompanying these foreign varieties. The increased use of insecticides and pesticides may also be playing a role in the proliferation of some insects and pests. Continuous and intensive use of chemicals may lead to some insects becoming immune to synthetic control. As a result, the use chemicals may have become less effective in controlling insects in recent years. Additionally, the increased use of chemicals may also be harming beneficial insects. Thus, natural control of harmful insects may have been weakened.

An increase in weeds may also be contributing to the proliferation of insects and pests. Serving as temporary hosts, it is plausible that weeds may have helped sustain insects and pests until the rice plants approaches the graining filling stage. Given the observations of small farmers and key informants, climate change alone may not be responsible for the changes in insects and pests. It is likely that a combination of these factors may have also played an important role.

6.6.2 Perceived Impacts of Changes in Insects and Pests

Small farmers and key informants were also asked to describe how the observed changes in insects and pests affected the quality and quantity of yields harvested. Eighty-seven (47.1%) small farmers noted that the increase in insects and pests led to lower yields. In terms of specific impacts, 116 (62.7%) small farmers and 22 (78.6%) key informants stated that the increase in paddy bug infestations resulted in damaged grains which reduced the quality and quantity of grains harvested. During the milk³⁵ and dough³⁶ stages of plant growth, paddy bugs suck the sap from the developing grains which results in malformation

 ³⁵ The stage in grain development when the grains yield a milk-white substance (Madramootoo 1974).
 ³⁶ The stage in grain development when crushed grains yield a white, floury substance (Madramootoo 1974).

and discoloration of the grains. In severe cases there are more wind or unfilled grains as reported by 18 (9.7%) small farmers. The damages caused by paddy bugs result in lower yields, reduce quality, and brittleness that lead to increased breakage during milling (GRDB, 2009).

According to 17 (9.2%) small farmers and 13 (46.4%) key informants, the increase in snail and water weevil resulted in damage seedlings and/or young plants. The adult water weevil feeds on the radicle³⁷ of the sprouting seeds and on the leaves of young plants while the root worm (water weevil larvae) feeds on the young roots and root tips (Madramootoo 1974). Snails feed on young and emerging rice plants thus affecting plant establishment in the field (GRDB, 2009). According to one key informant, snails, water weevil and root worm "reduces plant population per square foot." Lower plant population across the field leads to lower yields at harvest.

Eleven (5.9%) small farmers and nine (32.1%) key informants alluded to the damages caused by caterpillar and slugs. While caterpillars consume plant leaves, slugs strip the leaves of chlorophyll which reduces the plants ability to generate food through photosynthesis. The inability to generate food slows the growth of the plants and in some cases causes the plants to die. Ten (5.4%) small farmers and four (14.3%) key informants reported that the increase in leaf miner affected plant growth and in severe cases, killed the plants. Leaf miners burrow into leaves and feed on the plant tissues (Madramootoo 1974). This affects the establishment of tillers³⁸ which causes stunted growth and in severe cases, delays panicle³⁹ initiation. The overall impact is lower yields and uneven plant growth

³⁷ Embryotic root of the plant.

³⁸ Secondary shoots that surround the main stem (Madramootoo 1974).

³⁹ The terminal shoot of the rice plant that produces the grain (GRDB 2009).

across the field. Uneven plant growth also contribute to lower quality as some grains will mature later which results in increase green grains in the harvest.

Five (17.9%) key informants also pointed out that root worm feeds on the roots while four (14.3%) noted that the increase in stem borer damaged the plant stems. By tunneling into the plant stems, stem borers restrict the passage of nutrients to the grains. Key informants also described the impacts of other pests. At sowing, birds and wild ducks consume exposed seedlings while birds shell the paddy grains closer to harvesting. Rats damage the plants and feed on the grains. Table 6.12 presents the various impacts caused by changes in insects and pests.

Impacts	Small Farmers			Key	[,] Inform	ants
	No.	%	Rank	No.	%	Rank
Damaged grains - Paddy bug	116	62.7	1	22	78.6	1
Reduced yields	87	47.0	2	-	-	-
Unfilled grains/ wind paddy - Paddy bug	18	9.7	3	-	-	-
Damaged seedlings/ young plants - Snail and water weevil	17	9.2	4	13	46.4	2
Consumed plant leaves/ reduced photosynthesis - Caterpillar and Slug	11	5.9	5	9	32.1	3
Affected plant growth/ killed plants - Leafminer	10	5.4	6	4	14.3	5
Damaged plant roots - Root worm	-	-	-	5	17.9	4
Damaged the stem – Stem borer	-	-	-	4	14.3	5
Other	9	4.9		5	17.9	

Table 6.12 Impacts of Changes in Insects and Pests

While similarities between the responses of small farmers and key informants are evident, two notable exceptions are noted. Specifically, key informants made no reference to reduced yields. The failure to control insects and pests in a timely manner may have led to reduced yields among small farmers. However, the extent of the losses may not be serious enough to warrant the attention of key informants. In addition, key informants made no mention of unfilled grains due to paddy bug. It is likely that these farmers may have detected the infestation problem late and as a result the damage was already done.

6.6.3 Responses to Perceived Changes in Insects and Pests

The use of synthetic chemicals was the primary mechanism reported by small farmers in response to the observed changes in insects and pests. As such, 168 (90.8%) small farmers and 21 (75%) key informants noted that in recent years there have been an increase in preventative spraying to address the increase in insects and pests population and prevalence. As one small farmer relates, "before, I use to spray two or three times each season, now I am spraying five or six times and sometimes more depending on the infestation of paddy bugs." Another small farmer declared "if you don't spray, you get nothing" [he has to spray for paddy bugs, otherwise there is no grains to harvest].

The use of different chemicals and rotation of chemicals were also reported by seven (3.8%) small farmers and eight (28.6%) key informants. Farmers rotate chemicals in order to prevent insects and pests from becoming immune to a particular insecticide or pesticide. They are not only rotating among different brands of chemicals but also between systemic and contact treatment approaches. Systemic chemicals attack the different development stages while contact chemicals only affects individual insects and pests that come into contact. Figure 6.12 illustrates farmers checking and spraying for paddy bugs, respectively.



Figure 6.12 Farmer sweeping (left) and spraying for paddy bugs (right) Photo credit: Author

Twenty-two (11.9%) small farmers and 18 (64.3%) key informants noted that farmers engage in seed treatment while five (2.7%) small farmers and 11 (39.3%) key informants stated that farmers are treating the water in the field. Seed treatment protects against early season pests such as snails and water weevil. However, since seed treatment is usually done at home, some farmers prefer treating the water in the field because it is safer in terms of preventing their livestock from accidentally consuming chemically treated seeds. Key informants noted that if farmers treat seeds, there is no need to also treat the water. Regardless, more conservative farmers engage in both seed and water treatment.

Four (2.2%) small farmers reported that they dried the field to expose root worms to the elements. This practice was confirmed by three (10.7%) key informants. In response to early season insects and/or pests that went undetected and caused extensive damage, farmers usually re-sow the field or transplant plants from other parts of the field. This was reported by two (1.1%) small farmers and three (10.7%) key informants. While only three (1.6%) small farmers reported engaging in better sanitation of their field and its surroundings, 16 (57.1%) key informants noted farmers are engaging in this practice. Sanitation involves removing the host plants (e.g., wild rice) of insects and pests from the field and spraying the weeds and bushes that grow on the meres, dams, and embankments that surrounds their respective plots. Figure 6.13 depicts weeds on an embankment of a rice field that was sanitized with a contact chemical.



Figure 6.13 Weeds on the embankment of a rice field sanitized Photo credit: Author

Key informants also offered two additional insights into farmers' behavior. Seven (25.0%) key informants noted that farmers are engaging in block sowing and spraying in order to effectively address paddy bug infestation. Block sowing entails farmers in a specific farming section planting together so that all the fields are at a similar growth stage. This makes controlling for paddy bugs more effective in that all the farmers in a particular block would also apply insecticide at the same time. The uniform application of pesticide across the block reduces the risk of paddy bugs swapping between neighboring fields to take refuge.

Five (17.9%) key informants also noted that farmers are monitoring their fields more closely, especially during grain formation (milk and dough stage). Early detection of insects and pests in the field allows farmers to respond sooner in that failure to do so may result in the entire field being loss in a matter of days. Only one small farmer reported increase monitoring of his field.

Other responses by farmers reported by key informants include mechanized spraying and reduce walking distance while applying chemicals to ensure better coverage; use of scarecrows and air guns to keep birds and wild ducks away; and applying chemicals early in the morning and late in the afternoon when insects and pests are more active. Farmers are also engaging in some cultural practices such as burning tires, boiling and spraying neem leaves⁴⁰, and planting neem trees around the fields. In addition, farmers are doing more research and learning about the beneficial insects in an effort to tackle insects and pests more effectively and efficiently. Table 6.13 presents responses to changes in insects and pests as reported by small farmers and key informants.

Responses	Small Farmers			Key Informa		
	No.	%	Rank	No.	%	Rank
Preventative spraying/ sprayed more	168	90.8	1	21	75.0	1
Treated seeds	22	11.9	2	18	64.3	2
Rotate/ use different chemicals (contact and systemic)	7	3.8	3	8	28.6	5
Treated water in field	5	2.7	4	11	39.3	4
Drain field/ fresh water	4	2.2	5	3	10.7	8
Remove host plants/ sanitize field; surroundings (meres,						
dams, embankments, etc.)	3	1.6	6	16	57.1	3
Re-sow/ transplant from other fields	2	1.1	7	3	10.7	8
Block sowing/ spraying	-	-	-	7	25.0	6
Increase monitoring	1	0.5	8	5	17.9	7
Other	2	1.1		11	39.3	

 Table 6.13 Responses to Perceived Changes in Insects and Pests

⁴⁰ The neem tree's (*Azadirachta indica*) bitter leaves and bark is used as an insect repellant.

Farmers are engaging in a myriad of adaptation practices in reaction to perceived changes in insects and pests. Preventative spraying and seed treatment were the most popular adaptation responses reported by small farmers and key informants. While preventative spraying of chemicals and seed treatment may be effective, they also pose a serious threat to the environment. In addition to killing insects and pests, synthetic chemicals contaminate soil, water, turf, and other vegetation (Aktar, Sengupta, and Chowdhury 2009). As such, other non-targeted organisms such as birds, fish, and beneficial insects are harmed, resulting in loss of biodiversity.

It is interesting to note that key informants highlighted that improved sanitation in and around the field is a major response to insects and pests. It is likely that this practice is very popular among small farmers. However, the routine nature of this practice may have caused small farmers to downplay its importance. Similarly, block sowing and increased monitoring reported by key informants may be so common among small farmers that they did not bother to mention.

6.7 Changes in Diseases

Increased concentrations of CO₂ in the atmosphere coupled with changes in temperature and rainfall patterns are predicted to have a direct impact on the incidence and severity of diseases in agricultural crops (Gautam, Bhardwaj, and Kumar 2013). As such, small farmers and key informants were asked to describe any noticeable changes in diseases affecting their fields. Specific changes in diseases, the related impacts, and changes in farm management practices in response to the observed changes are discussed below.

6.7.1 Perceived Changes in Diseases

Seventy-three (38.6%) small farmers and 23 (82.1%) key informants observed changes in diseases. When asked how diseases have changed⁴¹, 34 (46.6%) small farmers reported an increase in brown spot (*Cochilobolus miyabeans*). In comparison, 22 (95.7%) key informants reported an increase in brown spot. Key informants may have a better idea of this shift as they see many fields. Also, the difference in observation may be due to small farmers referring to brown spot as leaf blast and thus reported observe changes in brown spot as a form of blast.

Twenty-eight (28.8%) small farmers highlighted an increase in blast (*Pyricularia grisea*). However, only four (17.4%) key informants highlighted an increase in blast associated with older rice varieties such as the Rustic and 22-4. Despite the newer varieties of rice being blast resistant, some farmers continue to plant the older varieties that are susceptible to blast. In some districts, farmers continue to plant older varieties that are not blast resistant because they are reluctant to change and/or because some millers still request the older varieties. Withstanding this observation, 12 (16.4%) small farmers related that blast has decreased. Figure 6.14 illustrates brown spot and blast, respectively.

⁴¹ This was an open-ended question; participants were not given a specific list of changes to choose from.



Figure 6.14 Brown spot (left) and blast disease (right) Source: International Rice Research Institute (IRRI) and Rice Farmer's Manual (GRDB 2009)

Thirteen (17.8%) small farmers also reported an increase in brown or red tip. However, this observation may actually be attributed to iron toxicity rather than disease. Iron toxicity occurs in acidic, flooded soils with a pH general below 5.5 (Cheaney and Jennings 1975). Only three (4.1%) small farmers noticed an increase in sheath blight (*Rhizoctonia solani*) and sheath rot (*Sarocladium oryzae*). In comparison, 14 (60.9%) and 11 (47.8) key informants noticed an increase in sheath blight and sheath rot, respectively. Sheath blight and sheath rot are fungal diseases that causes lesions on the lower and uppermost leaf sheaths, respectively (GRDB 2009). Figure 6.15 depicts sheath blight and sheath rot.



Figure 6.15 Sheath blight (left) and sheath rot (right) Source: International Rice Research Institute (IRRI)

Interestingly, five (21.7%) key informants highlighted an increase in false smut which was not mentioned by small farmers perhaps because this disease does not pose a major threat to their yields at present. Other observations by small farmers and key informants include an increase in black tip on the grains and kernel smut. Table 6.14 presents changes in diseases seen by small farmers and key informants.

Perceptions	Small Farmers			Key Informants			
	No.	%	Rank	No.	%	Rank	
Brown spot increased	34	46.6	1	22	95.7	1	
Blast increased	21	28.8	2	4	17.4	5	
Brown/ red tip increased	13	17.8	3	-	-	-	
Blast decreased	12	16.4	4	-	-	-	
Sheath blight increased	3	4.1	5	14	60.9	2	
Sheath rot increased	3	4.1	5	11	47.8	3	
False smut	-	-	-	5	21.7	4	
Other	4	5.5		3	13.0		

Table 6.14 Changes in Diseases Reported by Small Farmers and Key Informants

Although small farmers and key informants appear to perceive changes in diseases similarly, observations by key informants as it relates to sheath blight and sheath rot are different. Contextualizing the perceived increase in sheath blight and sheath rot, one key informant hypothesized that the appearance of these funguses may be linked to the predominant cultivation of blast resistant varieties by farmers. The cultivation of blast resistant varieties eases the application of fungicide. However, the same fungicide that treats blast may also control other fungal diseases like sheath blight and sheath rot.

6.7.2 Impacts of Perceived Changes in Diseases

To understand how grain quality and quantity were affected by the observe changes in diseases, both small farmers and key informants were asked to describe the various impacts observed. Diseases usually restrict the flow of nutrients and water to the grains and if untreated it is likely to kill the plants as reported by 11 (15.1%) small farmers and six (26.1%) key informants.

In terms of specific impacts, 35 (47.9%) small farmers and four (17.4%) key informants reported that the observed changes in diseases affected plant growth either through damaging the plants and/or stunting the growth. In particular, sheath blight reduces yields by restricting the uptake of nutrients thus leading to stunted growth while sheath rot deforms the panicle. This delays the maturity of the plants and/or diminished the yield potential.

The restriction on nutrient uptake also contributed to the quality and quantity of yields. Eleven (15.1%) small farmers and nine (39.1%) key informants noted that there were more wind paddy or unfilled grains while poor grain quality or damaged grains were reported by seven (9.6%) small farmers and seven (30.4%) key informants.

Although not mentioned specifically by small farmers, 14 (60.9%) key informants pointed out that the increase in brown spot, in particular, reduced the plants ability to effectively manufacture food through photosynthesis. Since plants depends on its leaves to manufacture food, damaged chlorophyll caused by brown spot impedes the plants ability to absorb energy from sunlight. Six (8.2%) small farmers noted that they were not sure or did not know how the change in diseases affected their yields while another three (4.1%) small farmers claimed that yields had increase due to the reduction in blast. Table 6.15 summaries the impacts reported by small farmers and key informants.

Impacts	Small Farmers			Key	Inform	ants
	No.	%	Rank	No.	%	Rank
Damaged plant/ stunted growth	35	47.9	1	4	17.4	5
Wind paddy/ sterile grains	11	15.1	2	9	39.1	2
Reduce plant population/ rot plants	11	15.1	2	6	26.1	4
Poor grain quality/ damage grains	7	9.6	4	7	30.4	3
Not sure/ don't know	6	8.2	5	-	-	-
Yields increased	3	4.1	6	-	-	-
Reduce food manufacture through photosynthesis	-	-	-	14	60.9	1
Deform panicle	-	-	-	3	13	6

Table 6.15 Impacts of Changes in Diseases

While the impacts of perceived changes in diseases reported by both small farmers and key informants are similar, two responses by key informants standout. Reference to the role of photosynthesis in the manufacture of plant food and deformation of the panicle should come as no surprise given the agricultural education and experience of the majority of key informants. Although some farmers may understand these impacts, articulating them may have been difficult at the time of interview.

6.7.3 Responses to Perceived Changes in Diseases

Given the observed changes and related impacts reported, farmers reportedly engaged in several adaptation practices. To mitigate the impact of diseases, 49 (67.1%) small farmers and 21 (91.3%) key informants reported preventative spraying of fungicide and/or rotating chemicals as the major response. That is, farmers are spraying multiple times for fungus regardless of whether they exist in the fields. They are also rotating among different chemicals. Although, some farmers planted blast disease resistant varieties, they still apply fungicide as a precaution. As one small farmer noted, "prevention is better than cure."

Four (5.5%) small farmers and 11 (47.8%) key informants reported increased use of balance nutrition fertilizer and/or less nitrogen-based fertilizer. It is argued that balance nutrition fertilizer and less nitrogen-based fertilizer enable the plants to become healthier in order to fend off the threat of diseases. According to one key informant, nitrogen-based fertilizer weakens the plants making them more susceptible to diseases.

Eight (11.0%) small farmers interviewed did nothing and/or accepted the losses. Considering the cost of purchasing fungicides and labor to spray the field, these farmers may have felt it is not worth the cost especially if the disease was already widespread and/or detected late. Other responses by small farmers included drying the field, using a blast resistant variety, and burning the forage left behind in the field after harvest. By burning the forage, any pathogens that is still lingering is destroyed before the start of land preparation for the next season.

Interviews with key informants provided additional insights regarding farmers' adaptation to changes in diseases. Farmers are also reducing the seed rate as highlighted by 14 (60.9%) key informants. From a disease mitigation perspective, less plants per square foot enhances the uptake of nutrients, water and sunlight making plants healthier and more resistant to diseases. It also enhances air flow through the field which disrupts ideal conditions for diseases to develop and spread. Farmers are also planting blast resistant varieties as indicated by five (21.7%) key informants. However, blast resistant varieties are still susceptible to other diseases such as brown spot, sheath blight, and sheath rot.

Additionally, farmers are improving sanitation in and around their fields as mentioned by four (17.4%) key informants. By removing host plants such as wild rice from within the field and clearing meres and embankments of weeds and bushes, farmers are disrupting the environment that is conducive to diseases. Other responses by farmers reported by key informants include reducing water level in the field; dry land preparation; sow better quality seeds; and burning forage after harvesting. Table 6.16 presents farmers' responses to observed changes in diseases as reported by small farmers and key informants.

Responses	Small Farmers			ners Key Inforn		
	No.	%	Rank	No.	%	Rank
Preventative spraying of fungicide / rotate chemicals	49	67.1	1	21	91.3	1
Did nothing/ accepted loss	8	11.0	2	-	-	-
Applied balance nutrition fertilizer/ reduce urea	4	5.5	3	11	47.8	3
Reduce seed rate	-	-	-	14	60.9	2
Plant different varieties	-	-	-	5	21.7	4
Improved sanitation of field	-	-	-	4	17.4	5
Other	6	8.2		8	34.8	

Table 6.16 Responses to Perceived Changes in Diseases

Preventative spraying of fungicide and the application of balanced nutrient fertilizer were the most popular answers by both small farmers and key informants. However, key informants offered three additional responses that warrant a closer examination. Historically, farmers have sown more seeds. However, one of the six improved management practices introduced by the Guyana Rice Development Board (GRDB) calls for farmers to sow less seeds. Small farmers specifically may be less inclined to reduce seed rate because of the belief that more is better. In other words, they may be trying to maximize yields by sowing more seeds with the expectation that more plants will result in more grains. In addition, the elevation of fields may warrant sowing more seeds; fields that are low are more susceptible to flooding which causes some seedlings to float away once flood waters recede. As such, sowing more seeds may help to mitigate potential losses arising from the risk of flooding.

While planting different varieties may help stem the proliferation and spread of diseases, emphasis on maximizing yields may influence small farmers' decisions on what variety to plant. Small farmers are more likely to continuously plant a high yielding variety (e.g. GRDB 10) instead of shifting to varieties that limit the occurrence and spread of diseases but yield less grains. Additionally, small farmers do not have the means to diversify. Unlike large farmers with multiple plots, which allows for the diversification of risk through lower seed rates and planting different varieties, most small farmers have a single plot. It is also possible that small farmers did not mention planting different varieties because they do not consider planting different varieties as a means of responding to the threat of diseases. It is also likely that small farmers did not mention improved sanitation of fields since this may be seen as routine ongoing work.

6.8 Changes in Weeds

Changes in atmospheric CO₂, temperature, and rainfall also impact the spatial and temporal distribution and proliferation of weeds (Rosenzweig et al. 2001; Peters, Breitsameter, and Gerowitt 2014). Their genetic diversity and physiological plasticity allows weeds to respond and adapt quickly to environmental changes (Varanasi, Prasad, and Jugulam 2016). According to Rosenzweig et al. (2001), humid conditions increase the proliferation of weeds which compete with crops for soil nutrients, light, and space. As a result, the quality and quantity of crop agricultural yields may also be affected leading to economic loss. As such, small farmers and key informants were asked whether they had observed any changes in weeds in the last five years. Specific changes in weeds, the related impacts, and changes in farm management practices in response to the observed changes are discussed below.

6.8.1 Perceived Changes in Weeds

Of the 189 small farmers interviewed, 168 (88.9%) observed changes in grass, sedge, and broadleaf weeds present in their rice fields. Similarly, 25 (89.3%) key informants acknowledged that weeds found in rice fields have changed. In Guyana, grasses include schoonord (*Echinochloa sp.*), muraina ⁴² (*Ischaemum rugosum*), birdseed (*Echinochloa colonum*), and monkeytail (*Echinochloa crus-galli*); sedges include jhussia⁴³ (*Fimbristylis miliacea*), and water sedge (*Cyperus difformis*); and broadleaf weeds include soapbush (*Sphenoclea zeylanica*), wild clove (*Ludwigia spp.*), and duckweed (*Sagittaria*)

⁴² Also known as rock-steady.

⁴³ Also known as masala and matchstick grass.

guyanensis) (GRDB, 2009). Additionally, red rice⁴⁴ (*Oryza sativa L*.) is a major weed that impacts the quantity and quality of rice yields.

When asked which weeds have changed⁴⁵, 114 (67.9%) small farmers and 18 (72.0%) key informants noted that red rice has increased. Red rice is believed to have originated from the continual crossing between wild species or as a result of breeding within domestic varieties (Mackill, Coffman, and Garrity 1996; Holm et al. 1997). As one key informant noted, "the shift in weather pattern affects proper land preparation which is one of the main reasons for the red rice problem." However, four (16.0%) key informants acknowledged that red rice was on the decline. This is primarily due to various adaptation measures undertaken by farmers over the years. Thus, it is not abnormal that some farmers have been able to control red rice.

One hundred and four (61.9%) small farmers and 17 (68.0%) key informants indicated an increase in duckweed. Duckweed is an aquatic weed with submerged and floating heart-shaped leaves (GRDB 2009). Figure 6.16 depicts red rice rising above the cultivated variety and duckweed displacing rice plants as it spreads.

⁴⁴ Also known as overhead, wild rice, and jharanga.

⁴⁵ This was an open-ended question; participants were not given a specific list of changes to choose from.


Figure 6.16 Red rice (left) and duckweed (right) Photo credit: Author

Increases in jhussia was also reported by 60 (35.7%) small farmers and six (24.0%) key informants. A prolific seed producer that germinates year round, jhussia serves as an alternate host for diseases, insects, and nematodes (Galinato, Moody, and Piggin 1999). Thirty-two (19.0%) small farmers and five (20.0%) key informants also noticed an increase in monkeytail. In lowland direct-seeded rice like in Guyana, monkeytail is detrimental because of its rapid growth, competitiveness, and ability to multiply quickly (Holm et al. 1977). Jhussia and monkeytail are illustrated in Figure 6.17.



Figure 6.17 Jhussia (left) and monkeytail (right) Source: Rice Farmers' Manual (GRDB 2009)

Additionally, 29 (17.3%) small farmers and two (8.0%) key informants reported an increase in muraina grass. Highly competitive with rice, muriana is an annual grass that emerges later than other weeds and thrives in lowland direct-seeded rice cultivation (Holm et al. 1977) It also serves as an alternate host of diseases, insects, and nematodes (Galinato, Moody, and Piggin 1999). The late emergence often means that this weed avoids weed control which is usually done 16-25 days after sowing or before the first application of urea fertilizer.

Twenty-one (12.5%) small farmers and three (12.0%) key informants observed an increase in sedges. Water and umbrella sedges high plant densities forms a thick vegetative cover in fields containing young plants (Holm et al. 1977). Figure 6.18 illustrates muraina and water sedge. Nineteen (11.3%) small farmers and two (8.0%) key informants reported an increase in hasser string grass. Eight (4.8%) small farmers and three (12.0%) key informants also noticed an increase in soap bush.



Figure 6.18 Muraina (left) and water sedge (right) Source: Rice Farmer's Manual (GRDB 2009)

Other weeds that small farmers noticed an increase include tanner grass, antelope grass, wild clove, turkey claw, and busy-busy. Despite the increase in different weeds reported, 19 (11.3%) small farmers and four (16.0%) key informants stated that weeds have decreased while nine (4.8%) small farmers mentioned that weeds varied from season to season. Table 6.17 summarizes the changes in weeds observed by small farmers and key informants.

Perceptions	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Red rice increased	114	67.9	1	18	72.0	1
Duckweed increased	104	61.9	2	17	68.0	2
Jhussia increased	60	35.7	3	6	24.0	4
Monkey tail increased	32	19.0	4	5	20.0	5
Muraina increased	29	17.3	5	2	8.0	8
Water /umbrella sedge increased	21	12.5	6	3	12.0	6
Hasser string grass increased	19	11.3	7	2	8.0	8
Soap bush increased	8	4.8	8	3	12.0	6
Schoonord grass increased	3	1.8	9	7	28.0	3
Weeds decreased	19	11.3		4	16.0	
Weeds varied by season	9	5.4		-	-	
Other	13	7.7		3	12.0	

Table 6.17 Changes in Weeds Reported by Small Farmers and Key Informants

The changes in weeds observed by small farmers and key informants are similar especially as it relates to red rice and duckweed. However, there are a couple observable differences in the responses reported. For instance, the increase in schoonord grass was observed by more key informants than small farmers. One explanation is that the effect of schoonord grass on small farmers maybe marginal at present. As such, small farmers may have placed greater emphasis on major weeds such as red rice and duckweed that have greater impacts on the quality and quantity of yields. Alternatively, small farmers may have been proactive in managing schoonord grass.

The observed decrease in weeds may be attributed to an increase and/or continuous control with herbicides. Additionally, overall improvement in agronomic practices may have eventually led to the suppression of weeds over time. Changes in weather patterns from year-to-year may also influence the presence of weeds. For instance, duckweed proliferates under wet conditions so in seasons and/or years where there is increase rainfall, it is likely that duckweed will increase and vice versa. Flooding due to excess rainfall may have also contributed to the dissemination of seeds of different weeds. Given that combine harvesters work multiple fields, it is also likely that weeds may have been carried from one field to another.

6.8.2 Impacts of Perceived Changes in Weeds

Small farmers and key informants were also asked to describe how the observed changes in weeds affected the quality and quantity of yields. As such, 79 (47.0%) small farmers indicated that grasses, sedges, and broadleaf weeds reduced the plant population in the field because they compete directly with rice plants. Specifically, 30 (17.9%) small

farmers and 22 (88.0%) key informants noted that weeds compete for space, water, nutrients, and sunlight. In the words of one small farmer, "weeds grow fast...tek ova de field and choke de rice" [weeds grow vigorously, spread quickly to all parts of the field, and prevent the rice plants from growing]. Lower plant population leads to lower yields as reported by 63 (37.5%) small farmers.

Thirty-eight (22.6%) small farmers and eight (32.0%) key informants related that red rice (Figure 6.19) reduced the quality of the harvest which results in poor grades and prices received at the mills. Red rice grains carry a red pericarp or pigmentation which usually requires additional milling to remove thus resulting in more broken grains (Holm et al. 1977). As a result, farmers whose harvest contains more red rice receives a lower overall grade for their grains at the mill. In addition, four (2.4%) small farmers and seven (28.0%) key informants related that because most weeds grow taller, they canopy the rice plants which leads to lodging. That is, the taller weeds fall over on the rice plants causing the plants to fall down.



Figure 6.19 Red rice grains after milling Photo credit: Author

Although not mentioned by small farmers, three (12.0%) key informants highlighted that the increase weeds observed in recent years served as a host for insects and diseases. For example, since red rice matures earlier, it serves as a surrogate for paddy bugs to feed on until the pure variety matures. Table 6.18 presents the impacts of changes on weeds reported by small farmers and key informants.

Impacts	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Reduced plant population	79	47.0	1	-	-	-
Lowered yields	63	37.5	2	-	-	-
Reduce quality of yields	38	22.6	3	8	32.0	2
Compete for nutrients, space, fertilizer, and sunlight	30	17.9	4	22	88.0	1
Canopied/lodged rice plants	4	2.4	5	7	28.0	3
Hosts for insects and diseases	-	-	-	3	12.0	4
Other	6	3.6		-	-	

Table 6.18 Impacts of Changes in Weeds

In general, impacts reported by small farmers and key informants are similar with two main exceptions. While small farmers noted that weeds reduced plant population and lowered yields, key informants did not mention either impact. One possible explanation is that the true impacts of weeds on yields may be marginal given that farmers are likely taking actions to control weeds before irreversible damage occurs.

6.8.3 Responses to Perceived Changes in Weeds

The observed changes in weed pressure saw several responses from farmers to prevent losses in yields, higher cost of production, and preservation of grain quality. The primary response by farmers involves the increase application of herbicides. As such, 128 (76.2%) small farmers and 22 (88.0%) key informants reported an increased in the

application of herbicide to control weeds in recent years. Although herbicides are expensive, it is widely available to farmers. Farmers also mix and/or rotate among different herbicides available on the market or based on the recommendation of the district rice extension officer.

In addition, 11 (6.5%) small farmers and 13 (52.0%) key informants reported that farmers are improving water management in an effort to suppress weeds. Maintaining a lightly flooded field until the cultivated rice covers the field helps to suppress weeds. Lands that are high in some areas and low in others make it difficult to maintain equal water depth across the entire field. As a result, farmers are splitting larger fields into smaller plots so that they can maintain uniform water levels across each plot and better manage water resources as a whole. Although maintaining water in the field helps to suppress most weeds, this response actually helps duckweed to spread. This is because duckweed thrives under lightly flooded field conditions. Hence, five (3.0%) small farmers noted that they drained the field in response to duckweed.

Given that red rice shares similar genetic characteristics as cultivated varieties, conventional application of synthetic chemicals is not appropriate since these chemicals will also control the cultivated varieties. As such, both small farmers and key informants have indicated a couple novel strategies that are being used to control of red rice. Since red rice matures earlier and are usually taller than cultivated rice varieties, 23 (13.7%) small farmers and 17 (68.0%) key informants reported the use of the rope and stick method to burn red rice plants with a contact chemical.

The stick method involves wrapping a piece of cloth to the end of a stick and moistening it with the chemical. Farmers then walk through the field bringing the

239

dampened end into contact with the red rice plants. In fields where red rice infestation is widespread, the rope method is more efficient. This involves two persons pulling a tight rope soaked in the chemical across parts of the field that contains red rice. Figure 6.20 shows red rice plants chemically burnt using the rope and stick method.



Figure 6.20 Red rice plants controlled using the rope and stick method Photo credit: Author

Another method used by farmers to control red rice involves rogueing. Rogueing involves clearing the field of all plants other than the cultivated variety (Madramootoo 1974). As such, 17 (10.1%) small farmers and 15 (60.0%) key informants reported the use of water⁴⁶ and/or dry⁴⁷ rogueing to control red rice. Since red rice usually germinates before cultivated varieties, it is easier to identify and distinguish from cultivated varieties. Hence, farmers walk through the field and physically cut and/or remove the red rice plants. Figure 6.21 shows a field that was water rogued and free from weeds while Figure 6.22 captures a farmer using a grass knife [sickle] to dry rogue non-cultivated varieties from his

⁴⁶ Rogueing that takes place during the vegetative but before the flowering stage.

⁴⁷ Rogueing that takes place before harvesting.

field. In both cases, the individual farmers were seed paddy producers. Thus, producing good quality seeds entails sanitizing fields of weeds.



Figure 6.21 Water rogued rice field Photo credit: Author



Figure 6.22 A farmer engaged in dry rogueing Photo credit: Author

Key informants offered additional insights into farmers' responses to the observed changes in weeds. Although not mentioned by small farmers, 12 (48.0%) key informants noted that farmers in general are engaging in the flood and grow-out method. This method

of red rice control entails flooding the field after harvesting to encourage red rice plants to germinate. Once red rice plants have established, farmers either spray with a chemical and/or plows the weeds back into the ground.

As part of land preparation, farmers are ensuring that their fields are level. As such, eight (32.0%) key informants reported that farmers are engaging in better land preparation. This entails both dry and wet land preparation and leveling the field to allow for better water management. Figure 6.23 shows a farmer leveling his field by dragging a log across the puddled surface. A level land helps ensure consistent water depth across the field. This allows farmers to grow the seedlings through water as reported by four (16.0%) key informants.



Figure 6.23 A farmer leveling the land prior to sowing Photo credit: Author

Moreover, four (16.0%) key informants stated that farmers are using better quality seeds. Clean and certified seeds are usually free from red rice and other weeds. Other responses by farmers in general include using a higher seed rate to help suppress weeds; using different varieties; and burning the field after harvest to destroy any red rice seeds

on the surface or in seed banks just below the surface. Table 6.19 presents farmers' response to changes in weeds as reported by small farmers and key informants.

Responses	Small Farmers			Key Informants		
	No.	%	Rank	No.	%	Rank
Sprayed more herbicide	128	76.2	1	22	88.0	1
Burnt with chemical	23	13.7	2	17	68.0	2
Dry and water rogued	17	10.1	3	15	60.0	3
Improved water management	11	6.5	4	13	52.0	4
Dried field	5	3.0	5	-	-	-
Flood/ grow-out method	-	-	-	12	48.0	5
Better/ additional land preparation	-	-	-	8	32.0	6
Used better quality seed	-	-	-	4	16.0	7
Grow through water	-	-	-	4	16.0	7
Other	14	8.3		6	24.0	

Table 6.19 Responses to Perceived Changes in Weeds

The top four responses reported by both small farmers and key informants are similar. However, additional insights offered by key informants may be beyond the reach of small farmers. While the flood and grow-out method may be an effective control against red rice, it may be prohibitively costly for small farmers that lack financial resources and/or equipment to engage in this additional land preparation activity. Hence, they are less likely to engage in this practice. Similarly, better land preparation means additional dry land preparation. Since small farmers rely on large farmers for land preparation, large farmers may not be available when needed. And even if they are available, small farmers may be reluctant to incur the additional costs.

While good quality seeds produced by the rice research center are available to all farmers, preference is often shown to large farmers since they buy in larger quantities. As such, small farmers whose seed needs are much smaller may not be able to access them. Additionally, the cost to transport a few bags of seeds from the research center to different parts of the country is prohibitively high. While some small farmers may be able to secure seeds through large farmers, the great majority of them purchase seeds from local farmers in the area which may not consistently be of good quality.

6.9 Changes to Farming Practices

In order to capture any additional responses to changes in climatic and non-climatic conditions, small farmers were read a general list of practices and asked to identify which practices they have implemented in the last five years. All small farmers in the sample population reported making changes to their farming practices in the last five years. Considering the increase in insects and weeds observed, it is not surprising that 183 (96.8%) small farmers increased the quantity and frequency of herbicides and pesticides used. One hundred and eighty-two (96.3%) small farmers reported using different rice varieties and/or rotate varieties between seasons depending on the expected weather conditions. For instance, the majority of small farmers are planting the newer high yielding and blast resistant varieties. As noted above, this is one of the reasons for increases in yields. Additionally, in the autumn season when excess rainfall is usually expected, some farmers plant a variety that can withstand lodging.

One hundred and eighty-two (96.3%) small farmers reported buying new seeds each or every other season. This is in contrast to the historic practice where many farmers retained seeds from their harvest to replant. Good quality seeds help to ensure better germination. They are also likely to be sanitized of weeds, insects and pests, and diseases all of which facilitates better yields. One hundred and seventy (89.9%) small farmers also treat their seeds for early season pests such as water weevil and snail. Seed treatment help to ensure that seedlings remain viable and plant population remain high. Ninety-seven (51.3%) small farmers also applied insecticide to the water in the field instead of or in addition to seed treatment. In general, farmers with livestock are more likely to apply the chemicals to the water in the field in order to prevent their livestock from accidentally consuming harmful treated seeds.

Given the lack of predictability in rainfall patterns, 169 (89.4%) small farmers reported adjusting their planting dates while 137 (72.5%) pumped water. Although rice in Guyana is irrigated, the release of water into the irrigation canals depend of rainfall that replenishes the water conservancies or reservoirs. As such, farmers adjust their planting dates based on the early or late release of water. However, the lack of in-season rainfall usually results in the water level in the irrigation canals not being high enough to allow for gravity flow into the fields. As such, farmers resort to pumping water to start land preparation and subsequent sowing. It must be noted that in some areas of the country where there is no irrigation system, farmers are forced to pump water. However, if there is adequate rainfall at the time of sowing, less water is required to be pumped into the field.

One hundred and sixty-four (86.8%) small farmers increased the quantity and frequency of fertilizer application. In an effort to increase yields and combat the adverse weather conditions, more farmers are using balance nutrition fertilizer on their fields. Farmers that had a soil test done also adjusted the type and quantity of fertilizer used. Ninety-seven (51.3%) small farmers also reduced the plant density in their field. By sowing less seeds, fewer plants compete for water and nutrients. This allows for increase yields since plants are able to absorb the necessary nutrients.

Although reduce plant densities are expected to generate better yields, 32 (16.9%) small farmers indicated that they sowed more seeds. Farmers in flood prone areas caters that some seedlings will die or float off. As such, they sow more seeds to compensate for potential loss of seedlings. Twenty-three (12.2%) small farmers reported leaving their land unplanted for one of the seasons. This was due the lack of water to start the crop (drought) or late harvesting which did not allow enough time for re-sowing. As one small farmer related, it was better to skip the crop and plant the next crop on time than to waste resources planting late and risk harvesting in the rain. Table 6.20 presents the various adaptation practices small farmers adopted.

Adaptation	No. of Farmers	%
Increase herbicide and pesticide application (quantity and frequency)	183	96.8
Use different rice varieties	182	96.3
Buy new seeds each season	182	96.3
Treat seeds for insects	170	89.9
Changing sowing/ planting dates	169	89.4
Increase fertilizer application (quantity and frequency)	164	86.8
Pump water	137	72.5
Reduce plant densities	97	51.3
Sprayed water in field	97	51.3
Sow more seeds	32	16.9
Leave land idle	23	12.2
Rent out land to large farmers	4	2.1
Rice variety rotation	3	1.6
Exited rice cultivation and seek off-farm employment	1	0.5
Other (block planting, patching, etc.)	14	7.4

Table 6.20 Adaptation by Small Farmers

As Table 6.20 suggests, small farmers are engaging in a many different adaptation practices. One can reasonably expect this to continue in the near future. However, it is also likely that some small farmers will need to accelerate the adoption of these adaptive measures under a changing climate. For instance, increase drought will force more small farmers to pump water. If adequate water is not available, a greater number of small farmers may need to leave their land idle, rent out their land, or stop planting rice altogether.

6.10 Seasonal Changes and Impacts on Yields

Given the observed changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds, 89 (47.1%) small farmers indicated that the autumn season was most affected while 68 (36.0%) small farmers referenced the spring season. Twenty-seven (14.3%) small farmers reported that both seasons were affected by the observed changes. Regardless, 135 (71.4%) small farmers reported an increase in yields while 24 (12.7%) reported lower yields in the last five years. Despite changes in climatic conditions, increases in yields may be attributed to farmers engaging in better agronomical practices because of climatic changes and planting new high yielding varieties. Since farmers are engaging in a wide range of adaptation practices, this would also help mitigate the negative externalities created by climate change. Furthermore, 16 (8.5%) small farmers noted that there was no change in yields while 14 (7.4%) stated that yields fluctuated from season to season.

6.11 Crop Failure and Coping Strategies

Climate shocks such as drought, flooding, poor or excess rainfall, and high temperature increase stress on agricultural crops, oftentimes lead to crop failure. Mendelsohn (2007) defines crop failure as the complete loss of crops on a farm. However, crop failure may also encompass poor yields relative to expectation. Adverse weather conditions reduce soil quality, drain soil nutrients, and/or limit the access to soil nutrients during critical growth stages (Coulibaly et al. 2015). Non-climatic factors such as weed infestation and pest and disease outbreaks may also cause crop failure and are inextricably influenced by climatic conditions.

Given the link between crop failure and changes in weather patterns, small farmers were asked whether they experienced crop failure in the last five years. Ninety-eight (51.8%) small farmers stated that they suffered losses to their entire or partial field due to excess rainfall that led to flooding, lack of water due to drought and/or paddy bug infestation that was detected too late. Of these small farmers, 52 (53.1%) reported that losses occurred in the spring season while 33 (33.7%) suffered crop failure in the autumn season. Four (4.1%) small farmers noted that they have experience crop failure in both seasons while 9 (9.2%) small farmers could not remember the season the losses occurred.

The banes of crop failure coupled with high cost of production, low prices, poor grades received at the mills, and late payments by millers force small farmers to deploy various coping strategies to manage income shocks. Coping strategies are usually short term responses to unexpected or abnormal events (van der Geest and Warner 2014). In other words, specific ex post risk management options employed to minimize livelihood impacts of adverse climatic shocks (Cooper et al. 2008). Eighty-six (45.5%) small farmers stated that they relied on money from other sources to cover income shortfall from rice farming and/or to cover the costs of replanting the following season. This is not surprising given that 181 (95.8%) small farmers interviewed reported engaging in some form of offfarm employment and/or receive a government pension. Janvry and Sadoulet (2016) note that the poor usually have multiple sources of income while small farmers typically derives

approximately fifty percent of their income from off-farm employment and selfemployment.

Thirty-eight (20.2%) small farmers reported that because of the losses sustained, they had to adjust their spending on food and other household expenses or use savings to cover the shortfall in income. Consumption smoothing is commonly practiced among farmers in developing countries (Rosenzweig and Wolpin 1993; Townsend 1994; Morduch 2002; Hoddinott 2006; Pandey et al. 2007). Mehar, Mittal, and Prasad (2016) found that reduce food intake is a general response among farmers coping with climate shocks. Another study showed that farmers reduced the number of meals taken and reduce expensive food items to cope with income loss (Ashraf, Routray, and Saeed 2014). However, adverse risk coping strategies such as reduce food consumption may have long-term irreversible health consequences (Janvry and Sadoulet 2016).

It is interesting to note that small farmers utilize their savings as a coping strategy. Good production years and strong prices coupled with off-farm employment allow them to engage in formal savings. The use of savings to help with agricultural inputs and avoid complete crop failure is a common practice (Ashraf, Routray, and Saeed 2014). Furthermore, 15 (7.9%) small farmers noted that they were able to use the money received from one season to re-plant in the next season while just 11 (5.8%) relied on input credits. While rice millers are mandated under the law to pay 50 percent of the amount due within two weeks of receipt of the grains and the remainder within 42 days (RFA 1998), this is not always the case. Additionally, millers usually offer input credits at high interest rates.

Nine (4.8%) small farmers engaged in self-help practices where they rotated helping each other sow, apply fertilizer, and spray chemicals. This form of solidarity

249

network helps to reduce input costs and minimizes the risk of poor quality work perform by unsupervised paid laborers. Input adjustment, sale of livestock, and borrowing are also important coping strategies (Ashraf, Routray, and Saeed, 2014). Eight (4.2%) small farmers indicated that they adjusted or deferred rice input expenses. For example, instead of buying new seeds, they retained seeds to replant, reduced the recommended amount of fertilizer and/or sprayed less chemicals. While these coping strategies are cost saving in nature, the risk of lower yields at harvest is likely to be higher. Other coping strategies include the sale of livestock and borrowing from family to help pay for inputs. Table 6.21 summarizes the various coping strategies employed by small farmers.

Table 6.21 Coping Strategies Employed by Small Farmers

Coping Strategies	No. of Farmers	%
Use money from other sources	86	45.5
Adjust food/household expenses	19	10.1
Use savings	19	10.1
Rice turnover	15	7.9
Input credit	11	5.8
Self-help to reduce cost	9	4.8
Adjust/ defer input expenses	8	4.2
Sell livestock	4	2.1
Borrowed	3	1.6

6.12 Paying for Changes in Farming Practice

Paying for adaptation practices may cause small farmers to become more vulnerable (Mertz et al. 2009). That is, losses or complete crop failure may still occur leaving farmers out of pocket and/or in debt. For example, a small farmer may purchase a tractor on credit to ensure timely land preparation and hence sowing within the season to avoid early season floods and/or harvesting in the rain. However, erratic rainfall patterns may still occur leading to losses due to flooding and lodging of plants. At this point, the farmer has not only loss the crop but also falls into debt. Kelly and Adger (2000) argue that

vulnerability can only be accurately assessed after adaptation has taken place. As such, understanding how small farmers pay for adaptation practices is vital to our understanding of net vulnerability related to climate change.

Small farmers reported using various means to help pay for the changes in farming practice that they have undertaken in response to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. One hundred and eighty-one (95.8%) small farmers used savings and/or money received from the sale of the grains to pay for adaptation measures. The use of savings in many cases is warranted given that farmers do not always receive payment from the millers in a timely manner. The Rice Factories (Amendment) Act of 2009 requires millers to pay farmers 50 percent of the amount due within two weeks of receipt of the grains and the remainder within 42 days (RFA 1998). However, some millers seldom abide by this law and their actions often go unpunished by the Guyana Rice Development Board (GRDB). In some instances, farmers are offered postdated checks which restricts them from cashing it until some future date while in other cases piece meal payments are made. In both cases, farmers do not receive interest on late payments. As one small farmer related, "I am almost ready to harvest the current crop [September 2017] and I have not received payment for the spring crop which was harvested in April [2017]." Such situations are not uncommon and small farmers often feel the brunt of late payments since they are forced to go into their savings or rely on other sources of income until they receive payment from millers.

One hundred and twenty-four (65.6%) small farmers rely on income from off-farm employment to help with rice cultivation while 109 (57.7%) reported taking input credit, primarily in the form of fertilizer credit from the millers. However, some farmers avoid

251

taking credit from the millers because of the high interest rate. Fifty-five (29.1%) small farmers noted that they sold livestock to help with rice farming while 46 (24.3%) borrowed from family members. Although, borrowing from family members is usually interest free, farmers try to avoid doing so because it may lead to strain relations.

Thirty-nine (20.6%) small farmers reported using their pension to help pay for additional inputs. Guyanese citizens become eligible for old age pension at age sixty-five while those that contributed to the National Insurance Scheme (NIS) becomes eligible at age sixty. Given that over a fifth of small farmers interviewed stated they receive a pension, this is an important source of adaptation funding.

Nineteen (10.1%) small farmers reported borrowing from other farmers to help cover the costs of inputs. In some cases, small farmers borrow actual inputs (e.g., fertilizer) from large farmers and replace after they harvest. Fifteen (7.9%) small farmers reported receiving government support in the form of seed paddy and/or fertilizer. While this was in relation to helping farmers recover from crop failure and not directly related to adaptation, it may have allowed farmers to repurpose their own funds towards adaptation. Furthermore, 13 (6.9%) small farmers obtained microcredit while six (3.2%) borrowed from a commercial bank. Other sources of funds used for adaptation include cash crops, remittances, sale of productive assets, credit from tractor/combine operator, small business and shop keeping. It is worth pointing out that it is customary practice for tractor and combine operators (usually large farmers) to do land preparation and harvesting on credit. However, this was not mentioned by the majority of small farmers interviewed. Perhaps, this is such a traditional practice that small farmers felt that it was not something new worth

mentioning. Table 6.22 summarizes the various ways small farmers paid for the adaptation practices implemented.

Source of Funding	No. of Farmers	%
Use savings/ crop turnover	181	95.8
Off-farm employment	124	65.6
Input credit (fertilizer, herbicide, pesticide, seeds)	109	57.7
Sell livestock	55	29.1
Borrow from family members	46	24.3
Pension	39	20.6
Borrow from other farmers	19	10.1
Government support/ subsidy	15	7.9
Obtain institutional microcredit	13	6.9
Cash crop	10	5.3
Borrow from commercial bank	6	3.2
Remittances	4	2.1
Sell productive assets	2	1.1
Other (credit from tractor/combine operator, small business/shop keeping)	6	3.2

Table 6.22 Paying for Adaptation

As Table 6.22 suggests, the majority of small farmers used their savings or income from the harvest and income from off-farm employment to pay for adaptation. Borrowing from family members and other farmers is usually interest free with no fixed payback period, although the aim is usually to clear debts after harvest is completed and payment is received from the mill. Interestingly, only 19 (10.1%) small farmers borrowed from microfinance institutions and commercial banks to help pay for adaptation. This may be due to several factors including the lack of actual need to borrow, the lack of collateral or other means to assure payback and/or small farmers aversion to borrowing or being in debt. Given small farmers propensity to engage in limited borrowing from formal lending institutions, their net vulnerability may be lower post adaptation.

6.13 Rice Extension Services and Small Farmers' Adaptation

Farmers ability to successfully adapt to climate change rest on information, technologies, and education on how to cope with adverse changes in weather patterns (Davis 2009). In this regard, agricultural extension plays a crucial role in climate change adaptation. Extension agents transfer knowledge from researchers to farmers, advise farmers in decision-making, educate farmers to make similar decisions in the future, enable farmers to clarify their own goals and possibilities and to realize them, and stimulate desirable agricultural developments (Van den Ban and Hawkins 1996). According to Julie, Amungwa, and Manu (2017), the importance of agricultural extension cannot be overstated in that extension agents often initiate changes in knowledge, attitudes, resilience capacities, and skills of farmers.

To learn what agricultural extension services and support are being provided to build the adaptive capacity of farmers, responses were solicited from key informants employed by the Guyana Rice Development Board (GRDB) (district rice extension officers, chief scientist/plant breeder, plant pathologist, agronomist, entomologist, and the rice extension manager). Advice on better agronomic practices was the main service provided to farmers. Specifically, farmers are trained on the six-point practice. The sixpoint practice consists of six improved management practices for rice cultivation introduced by the GRDB extension department. This include advice on planting in-season (time of sowing), reducing plant densities (seed rate), treating seeds before sowing, controlling weeds, using balance nutrition fertilizer, and managing water in the field. The six-point is presented primarily through on-farm demonstrations and trials. Here district rice extension officers demonstrate to farmers' best practices for handling chemicals, treating seeds for early season insects and pests, and the correct timing and application of insecticides, pesticides, herbicides, fungicides, and balance nutrition fertilizer.

In addition, the use of the correct dosage of insecticide, pesticide, herbicide, fungicide and fertilizer are demonstrated and emphasized. Referring the proclivity of some farmers to exceed dosage recommendations, one key informant noted that "too much of one thing is good for nothing." Carter, Zhong, and Zhu (2012) note that productivity from increase fertilizer usage declines while Guo et al. (2010) points to reduce soil quality from the use of excess fertilizer. Ramli et al. (2018) notes that the brown planthopper – the most serious rice pest in Asia - developed resistance to insecticides leading to its resurgence. Hence, farmers are encouraged to follow usage directions since exceeding recommended quantities may cause more harm than good. Farmers are also trained to manage paddy bug and red rice infestations in an integrative and holistic manner. Demonstration plots illustrate appropriate plant densities, proper land preparation, and water management techniques.

District rice extension officers also serve as facilitators for farmer field schools. Originally developed by the Food and Agricultural Organization (FAO) to promote integrated pest management among Indonesian rice farmers in the late 1980s (Berg and Jiggins 2007), farmer field school is a bottom-up approach that focuses on an innovative, participatory, interactive, and experiential learning approach that seeks to improve the problem solving capacity of farmers (Anderson and Feder 2007). Farmer field schools allow participants to share their knowledge regarding various adaptation practices. As a result, farmers are provided with a basket of adaptation practices to choose from, adopting the practices that is most suitable to their individual farm characteristics and conditions. Interagency coordination is also a major support provided by district rice extension officers. Concerns of farmers are often communicated to different government departments calling for action to be taken. For instance, issues regarding access to adequate and timely water supply, maintenance of canals, trenches, and access dams (farm-to-market roads) are relayed to the respective water user association, National Drainage and Irrigation Agency (NDIA), regional authorities, and local government organ responsible. Extension officers also advocate for access to timely, easily understood, and accurate weather information from the Hydrometeorological Service on behalf of farmers. In some areas, districts rice extension officers make reports to the relevant authorities when drainage pumps are not working and co-ordinate with koker (sluice) attendants to ensure timely draft of excess rainfall to mitigate the risk of flooding.

The diffusion of new technology is a primary role played by agricultural extension (Agarwal 1983). According to Davis (2009), extension has traditionally entailed sharing information and promoting new technologies. Information communicated on improved and new technologies to prospective users through communication networks such as extension agents impact the likelihood of the innovation being adopted (Rodgers 2003). As such, district rice extension officers noted that they also promote and market seed paddy. That is, they share information with farmers regarding the yield potential and variety characteristics of new seed technologies introduced by the rice research center through the plant breeding program. Based on specific conditions such as soil type, flood risk, drainage and irrigation system, and the general history of the area with regards to floods, they advise farmers on adoption of the new variety or the variety that is best suited. They also serve as

a conduit between farmers and the rice research center to ensure that farmers are able to purchase good quality seeds that are free from weeds and diseases in a timely manner.

Another important avenue of support provided by GRDB is the hosting of periodic outreach meetings across each region to address issues facing farmers and the industry as a whole. Outreach meetings provide farmers with the opportunity to share their concerns regarding issues facing their district and/or region. This may include farm-level matters such as drainage and irrigation and conditions of access roads or marketing concerns such as the price received and on-time payment by millers. In addition, experts such as the plant pathologist, agronomist, and entomologist are usually present to help ease concerns and provide immediate feedback to farmers.

District rice extension officers also coordinate field and farmer exchange visits, distribute brochures on seed varieties and different agronomic practices, conduct onsite water salinity tests, and circulate weather forecasts and calendars. Furthermore, they are providing general advice on soil health management, integrated pest management (IMP), and climate smart agriculture. In the future, district rice extension officers could provide support for further climate change adaptation.

6.14 Barriers to Adaptation

Successful adaptation to environmental changes attributed to climate change begins with perception of the changes and an understanding of the corresponding impacts. However, barriers to adaptation often prevent and/or restrict effective and efficient response to the adverse effects of climate change. Barriers to adaptation are those conditions, factors, or obstacles that impede, divert, or block the process of developing and implementing climate change adaptation strategies (Moser and Ekstrom 2010; Biesbroek et al. 2013; Barnett et al. 2015). These may include technological, financial, cognitive and behavioral, and social and cultural constraints (Adger, Agrawala, and Mirza 2007).

The literature highlights many barriers to adaptation, but the list of possible barriers is seemingly infinite (Biesbroek et al. 2013). Specific barriers explored by previous studies include social (Stage 2010; Jones and Boyd 2011; Vulturius and Gerger Swartling 2015), economic (Biesbroek et al. 2011), cultural (Nielsen and Reenberg 2010), informational (Deressa et al. 2009; Amdu, Ayehu, and Deressa 2013), policy and institutional environment (Eakin 2005; Sietz, Boschütz, and Klein 2011; Oberlack 2017), and cognitive, affective and behavioral (Lorenzoni, Nicholson-Cole, and Whitmarsh 2007) among others.

While there are a number of factors that impede farmers' ability to respond to climate change, such factors are highly context specific and difficult to compare (Biesbroek et al. 2011)). Demographic, farm-level, socio-economic, and institutional characteristics may differ from country-to-country. Thus, barriers to adaptation may also differ. In the context of Guyana, small farmers were asked to rank the importance of potential socio-economic, informational, and institutional barriers to adaptation⁴⁸.

One hundred and seventy-five (92.6%) small farmers indicated that the availability of information regarding the changes in weather patterns observed was important or very important in making adaptation decisions. Effective response to climate change requires accurate, accessible, and useful climate information (Roncoli, Ingram, and Kirshen 2002). Although daily weather forecasts by the Hydrometeorological Service are available via newspapers, radio, and television, the accuracy of such information is often lacking

⁴⁸ A list of potential barriers to adaptation was read to small farmers; they were asked to rank each barrier in terms of importance (very important, important, slightly important, not important).

according to farmers. Farmers further noted that forecast information is usually granular in nature and thus often fails to convey the specific weather conditions expected in a particular area or district. Gbetibouo (2009) found that the lack of information about long term climate change was an important barrier to adaptation in the Limpopo Basin, South Africa while Gadédjisso-Tossou (2015) noted that the exchange of climate information through climate change communication would enable small farmers to change planting dates in three regions of Togo. Furthermore, the existence of microclimates across different regions often render general forecast for a particular region less valuable to some farmers. For instance, a forecast of persistent rainfall in a specific region may be limited to only the northern part of that region. However, farmers throughout the region may take precaution which limits the adaptive behavior of farmers in areas not affected.

One hundred and seventy-one (90.5%) small farmers ranked access to adequate irrigation and drainage infrastructure as important or higher when making adaptation decisions. Since rice in Guyana is irrigated, the importance of a well-maintained irrigation system remains paramount to farmers' response to changes in climatic conditions. Timely and adequate access to water for land preparation helps farmers to sow in-season in an attempt to avoid harvesting in wet out-of-season conditions. Early sowing also helps ensure that plants are well established in the fields and can survive temporary flooding if or when sudden and/or heavy rainfall arrives. Adequate irrigation facilities also assist farmers to engage in better water management in order to control weeds and replenish evaporated water. In areas where gravity flow is customary, access to adequate and timely water help farmers save on input costs related to mechanically pumping water into their fields. Such savings can be subsequently redirected towards other adaptation measures, as needed. Access to water (Gbetibouo 2009) and water scarcity (Amdu, Ayehu, and Deressa 2013) were found to be major barriers to climate change adaptation. Access to proper drainage is equally important as it relates to excess rainfall that leads to flooding. Poorly maintained drainage infrastructure impedes farmers' ability to successfully adapt. For instance, adding more outlets to the field to facilitate faster drainage during intense rainfall is futile if drainage canals are not properly maintained to allow excess water to recede smoothly.

One hundred and eighty-three (96.8%) small farmers noted that access to knowledge regarding adaptation practices is important or very important as it relates to making changes to their farming practices. Lack of or poor access to extension services (Amdu, Ayehu, and Deressa 2013; Bryan et al. 2013), and lack of information or knowledge concerning appropriate adaptations (Gbetibouo 2009; Deressa et al. 2009) were important barriers to adaptation found in the literature. However, in Guyana extension officers have been communicating and sharing information on improved management practices through farmer field schools, field days, field visits, and seminars. However, attendance at these events have dwindled in recent years. This is due in part to the lack of new information or new farm management practices that would encourage more farmers to participate. For instance, the six improvement management practices (six-point) for higher yields was introduced a decade ago. As such, farmers are wary of re-cycled information being presented. Gadédjisso-Tossou (2015) argued that intensifying extension activities would enhance farmers' adaptive capacity.

Farmers' ability to make changes to their farming practices also depend on their ability to pay for the changes. The socio-economic position of households (Ziervogel et al. 2006) or financial constraints (Deressa et al. 2009) may affect farmers' capacity to adapt.

As such, 51 (27.0%) and 93 (49.2%) small farmers indicated that access to money, savings and/or credit is very important and important, respectively. The majority of farmers have a secondary source of income while some were able to receive input credit, albeit at a steep interest rate. In the absence of a dedicated agricultural lending institution, access to agricultural finance is very difficult since commercial banks are reluctant to lend to small farmers or do so at high interest rates. Rural finance in Guyana is very limited due to the banking sector risk-averse attitude towards the risk faced by the agricultural sector, the lack of agricultural expertise in the banking sector, and difficulties to realize such securities (WB 2010a). Additionally, small farmers often lack legal title to the land which could otherwise be used to secure a loan. It must be noted also that several small farmers were averse to taking credit for the fear of being in debt. Instead, they are forced to appropriate whatever little savings and/or income from other sources, to cover adaptation costs. Access to credit (Bryan et al. 2009; Gbetibouo 2009) and affordable credit schemes (Gadédjisso-Tossou 2015) are important barriers to adaptation found in the literature. Despite difficulty in securing agricultural loans, 39 (20.6%) small farmers noted that access to money, savings and/or credit is not important. These farmers have multiple sources of income and thus can afford to pay for the changes needed to successfully adapt.

Insecure property rights is also a key barrier to adaptation (Gbetibouo 2009; Sarker 2012). As such, 71 (37.6%) small farmers indicated that ownership of land is very important while 108 (57.1%) indicated that it is important in guiding their decision to adapt. Land tenure allows farmers to engage lending institutions to obtain credit which can be used for adaptation purposes. Ownership of land also incentivize farmers to invest in their land and make improvements where necessary. For instance, farmers could invest in

leveling the land, build-up levees, and/or add more drainage knowing that the land could not be repossessed. Ownership also means lower production costs since farmers do not have to make rental payments. Such savings can be put towards land improvement and adaptation as a whole. While access to land (Bryan et al. 2009), lack of suitable land (Bryan et al. 2013) and land scarcity (Amdu, Ayehu, and Deressa 2013) were found to be major barriers to adaptation elsewhere in the world, this was not the case in Guyana.

Even if farmers have the knowledge and financial resources to adapt, they still need labor to help with adaptation. Fifty-two (27.5%) small farmers indicated that the availability of labor is very important while 102 (54.0%) stated it is important when it comes to adapting to climate change. Seasonal labor help with sowing, applying fertilizer and spraying for weeds and insects. However, changes in rainfall, temperature, insects and pests, diseases, weeds, and extreme weather events increases the overall demand for labor. For example, additional labor is needed to help make and clear drains, spray more for paddy bugs, load and off load grains from boats, and dry wet grains. The demand for labor to help respond to climate change is also thwarted by the decrease in the labor pool. This is because the number of young people involved in rice farming is declining. With the increase demand and a smaller labor pool, the cost of labor has increased which creates another barrier to adaptation. Regardless, 28 (14.8%) small farmers indicated that labor was not important. These farmers usually engaged in rice farming on a full-time basis and do the work themselves or are a part of a solidarity network that help each other. Studies by Sarker (2012) and Amdu, Ayehu, and Deressa (2013) found labor shortage to be a major barrier to adaptation.

All small farmers interviewed indicated that the price received for paddy grains were important or very important to their decisions to adapt. In Guyana, the price received by farmers is determined by the market. Small farmers contend that the high costs of production coupled with potentially low prices received at the mills result in small profit margins. As such, they are hesitant to invest in costly adaptation practices given the uncertainty surrounding the price they will receive at harvest.

One hundred and eighty-seven (98.9%) small farmers indicated that the absence of subsidized inputs and the access to tax exemption on inputs and equipment play a pivotal role in their decisions to adapt. Similarly, 185 (97.9%) small farmers noted that the lack of government assistance limits their ability to successfully adapt to the changes observed. Subsidized inputs and tax exemptions leave more money in farmers' pockets which can be used towards adaptation. In addition, tax exemption provides small farmers with the opportunity to acquire their own equipment and machinery to help with adaptation. Government assistance such as free seed paddy and other safety nets can help small farmers recover from crop failure and other losses attributed to climate change.

Gbetibouo (2009) found that the lack of access to markets was an important barrier to adaptation. As such, 183 (96.8%) small farmers reasoned that the availability of rice markets is important or very important to their adaptation efforts. They contend that the availability or access to markets would increase the demand for their grains which would increase the price received at the mills. Thus, increase access to markets would incentivize farmers to invest in adaptation practices with the expectation that they will receive a higher return on their investment. In addition, some farmers note that the lack of timely payment from millers constrain their ability to effectively adapt. Poor farm-to-market roads limit farmers' ability to easily monitor their fields and respond in a timely manner. Some farmers also conveyed that the absence of experts at outreach meetings hosted by the Guyana Rice Development Board (GRDB) do not allow for timely responses to their questions and/or concerns. Table 6.23 summarizes the importance of various barriers to adaptation to farmers.

Table 6.23 Barriers to Adaptation

	No. of Farmers							
Barriers	Very	%	Important	%	Slightly	%	Not	%
	Important				Important		Important	
Availability of information about extreme weather events, changes in	85	45.0	90	47.6	7	3.7	7	3.7
rainfall and temperature								
Access to adequate irrigation and drainage facilities	89	47.1	82	43.4	7	3.7	11	5.8
Access to knowledge regarding adaptation practices	87	46.0	96	50.8	1	0.5	5	2.6
Access to money/savings/credit	51	27.0	93	49.2	6	3.2	39	20.6
Ownership of land	71	37.6	108	57.1	1	0.5	9	4.8
Availability of labor	52	27.5	102	54.0	7	3.7	28	14.8
Price received for rice paddy	155	82.0	34	18.0	0	0.0	0	0.0
Subsidized inputs	130	68.8	57	30.2	0	0.0	2	1.1
Tax exemption on inputs and equipment	114	60.3	73	38.6	1	0.5	1	0.5
Availability of markets for rice	106	56.1	77	40.7	1	0.5	5	2.6
Government assistance	115	60.8	70	37.0	1	0.5	3	1.6
Other (payment on time, access to dam, outreach)	26	100.0	0	0.0	0	0.0	0	0.0

Given the various difficulties faced, 163 (86.2%) small farmers noted that the government should help while 26 (13.8%) specifically mentioned the Guyana Rice Development Board (GRDB). As one small farmer relates, "the government should intervene and see that small farmers get a better deal". Despite this expectation however, farmers are wary of government assistance. Another small farmer noted, "you have to help yourself, [there is] no expectation from government."

6.15 Co-benefits, Maladaptation, and Limits to Adaptation

As the preceding sections highlight, adaptation strategies undertaken by farmers in response to perceived changes and related impacts of rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds are numerous. While efforts to adapt in one area are likely to generate co-benefits in other areas, some adaptation practices may also conflict and/or become maladaptive in other areas. In addition, adaptation practices may be finite in nature. Although not specifically part of the survey data collected, the ensuing discussion examines possible co-benefits, maladaptation, and limits to adaptation.

6.15.1 Co-benefits

Co-benefits or ancillary benefits refer to the multiple positive effects of one policy, strategy or action across different objectives (Mach, Planton, and von Stechow 2014). In adaptation terms, more than one benefit can be derived from a single practice. Weed control reduces competition for water, space, and nutrients. As a result, rice plants are able to thrive freely and make full use of the natural and artificial resources in the field. This will likely result in better quality and quantity of yields. By controlling weeds, farmers are simultaneously removing potential host plants for insects and pests. This co-benefit helps ensure that the quality and quantity of yields are not affected by insects such as the paddy bug.

Leveling the field enables better water management. This allows for improved management of excess water which reduces the number of patches⁴⁹ in the field. As a result, uniform growth of plants is maintained across the field which increases the quality and quantity of yields. A co-benefit of better water management is that it can also be used to suppress some weeds. By growing the rice through water, some early season weeds are unable to grow through the shallow flood in the field.

Another adaptation practice that fosters co-benefits is the rotation of rice varieties. Some farmers reported rotating varieties between the two seasons and sometimes from year to year. In the autumn season when there is a greater chance of excess rainfall and flooding in certain areas, some farmers plant a variety that are sturdier and can withstand lodging. In the spring crop where there is greater likelihood of dry spells, farmers plant a short duration variety which requires less water. Rotating varieties also help to prevent diseases.

6.15.2 Maladaptation

According to the Intergovernmental Panel on Climate Change (IPCC) Fifth Assessment Report, "maladaptation refers to actions, or inaction that may lead to increased risk of adverse climate-related outcomes, increased vulnerability to climate change, or diminished welfare, now or in the future" (Noble et al. 2014). Several current practices by farmers may be beneficial in one area and harmful in another area.

⁴⁹ Patches are areas in the field where plants take longer to establish and/or fail to establish.

As part of their adaptation to changes in temperature and diseases, small farmers indicated that they increased the quantity of fertilizer applied to their fields. While the increase in fertilizer benefits the plants up to a certain point, it also benefits weeds in the field. Also, the final release of water from the field leading up to harvesting increases nutrient cycling. That is, more nutrients flow into the drainage canals which fertilizes grasses along the sides leading to increase overgrowth that eventually obstruct the smooth flow of water especially during heavy rainfall. Additionally, as a form of environmental pollution, fertilizer run-off affects soil and water quality and threatens other life forms thus leading to biodiversity loss.

While increased preventative spraying for insects like the paddy bug may help protect the crop in the short run, it is maladaptive in terms enhancing insect resistance and its control of beneficial insects. This short-term solution may turn into a long-term problem where farmers will need to continuously rely on synthetic chemicals to control insects instead of natural enemies. The continuous and/or accelerated use of synthetic chemicals also negatively impacts air quality and biodiversity.

6.15.3 Limits to Adaptation

While the intent of adaptation is to reduce climate related risk now and/or in the future (Adger et al. 2012), interactions among climate change, biophysical, and socioeconomic constraints can lead to the emergence of limits to adaptation (Klein et al. 2014). That is, at some point or outside a certain range, adaptation may no longer be feasible or deemed effective. Dow, Berkhout, and Preston (2013) note that exceeding adaptation limits will result in rising losses or require transformational change. As such,
understanding the adaptation range and limits to small farmers' capacity to effectively respond to climate change remains paramount (Yohe and Tol 2002).

Limits to adaptation are thresholds or tipping points beyond which adaptive actions cannot safeguard from intolerable risks (Adger et al. 2009; Moser and Ekstrom 2010; Dow et al. 2013; Monirul Islam et al. 2014; Barnett et al. 2015). Limits include the inability of the ecological systems to adapt to the rate and magnitude of climate change (Adger, Agrawala, and Mirza 2007). For instance, the biophysical environment may be limited by changes in temperature, precipitation, salinity, acidity, and intensity and frequency of extreme events including storms, drought, and wind (Klein et al. 2014). Economic limits such as "existing livelihoods, economic structures, and economic mobility" and socio-cultural limits such as "social norms, identity, place attachment, beliefs, worldviews, values, awareness, education, social justice, and social support" may also limit adaptation (Klein et al. 2014). Additionally, limits to adaptation may be further categorized as being hard or soft. Hard adaptation limits mean that no adaptation action is foreseeable while soft adaptation limits mean that options are not currently available but could become available in the future (Klein et al. 2014).

Considering the biophysical environment in Guyana, several limits to adaptation are likely to currently exist now and/or in the future. Increase high temperature limits the amount of time farmers can tend to their fields without suffering from heat exhaustion or other health issues. Even if farmers wear protective clothing (e.g., long sleeve shirts and a hat), the increase in humidity makes working in the fields for extended periods more uncomfortable. With temperature projected to rise further in the foreseeable future, rice cultivation will become more difficult. It will further reduce the amount of time farmers can spend in their fields since protective clothing may no longer be effective. Furthermore, once the heat tolerance of the plant is reached and in the absence of heat tolerant varieties, rice cultivation may no longer be viable. Even if a heat tolerant variety existed, there will still be a heat tolerance threshold to contend with.

Shifts in rainfall patterns, low lands, and current drainage and irrigation systems limit farmers' ability to respond to flooding. In some sections and/or districts across each region, excess rainfall usually overwhelms existing drainage systems resulting in overflows across the land. Thus, pumping out water from inundated fields is physically impossible. As a result, farmers are left with little choice but to wait for the water to recede. Physical environmental thresholds aside, financial limits relating to the cost of pumping water may preclude small farmers from artificially draining their fields. Despite having off-farm income, small farmers are also limited by finite financial resources. Additionally, the cost of maintaining and/or raising levies and dredging silted canals and outfalls on a regular basis are prohibitively high for developing countries like Guyana with endless development priorities. Furthermore, upgrading infrastructure such as levies and canals based on present norms may be not suffice under future conditions. Yohe and Tol (2002) argue that levies can still be overwhelmed, and silt will re-deposit over time.

The intensity and frequency of extreme events such as drought and heavy winds also places limits on adaptation. The lack of water available in the irrigation system during dry spells and drought leaves farmers with no response. In cases where some water is available in the system, small farmers incur additional cost to pump the water into their fields. Kusters and Wangdi (2013) found that water available to poor farmers became increasingly expensive during periods of irregular rainfall. At some point, the costs of pumping water may outweigh the benefits. Drought also lead to saltwater infiltration which limits adaptation in the absence of drought and salt-tolerant rice varieties in Guyana. Farmers in general are also limited in their adaptation to heavy winds. Given the layout of rice fields, the planting of trees to serve as wind breakers is impractical given the number of trees required and the associated costs of planting. While the most recent variety (GRDB 15) released in April 2018 promises better performance against lodging, this may not be the case under future environmental conditions.

The topography, elevation and soil type also limit the type of crops that can be planted. Given that the great majority of the land is clay base, flat, and below sea level, planting other crops that are far less water tolerant presents greater flood related risk. While small farmers could in theory reshape their fields to plant row crops, there are several limits to consider. It is prohibitively expensive to transform the field. The lack of expensive specialized equipment means that harvesting will need to be done manually. However, there is likely to be a labor shortage given Guyana's small population. Still there are likely limits relating to farmers' knowledge of other crops and problems with finding markets.

There are also socio-cultural limits related to social norms, identity, place attachment, age, and education. For most small farmers, rice farming is a social norm of rural life. Having grown up in rice farming, it is part of who they are and what they know; a way of life handed down from their fore parents. It is their legacy and they are intimately attached to the land and rice. Replacing rice with another crop is borderline sacrilegious for many of these farmers regardless of the promised economic benefit. However, at some rainfall and temperature threshold point, they may no longer be able to adapt thus forcing them to give up rice production altogether. The age and education level of small farmers also induce limits to adaptation. Older farmers are reluctant to change their farming practices. There is also a cultural component that encumbers their way of thinking in that the older generation "knows best". As one extension officer related, "there are certain farmers that are so bent in their ways, that they would never take advice from a young extension officer." The lack of education and knowledge may also limit some small farmers from undertaking adaptation practices.

Given that opportunities and resources may be finite for many small farm households, there are also economic limits at the farm-level. Coase (1960) suggested that in a perfectly competitive marketplace where transaction costs are low, efficient outcomes would prevail. High costs of inputs such as fertilizer and chemicals create disincentives for small farmers to adapt. Thus, the lack of economic incentives will do little to encourage farmers to engage in optimal adaptation practices. Similarly, if the price received at the mill is not profitable, then farmers will not incur additional expenses related to adaptation. The lack of government subsidies and tax exemptions would also limit small farmers' ability to acquire farming equipment to help with adaptation while the absence of safety nets often means farmers have little to fall back on or look forward to in the event of losses due to climate change.

There are also limits to adaptation regarding the application of fertilizer and chemicals such as insecticides and fungicides. At some point, the law of diminishing marginal returns steps in where the returns from applying additional fertilizer, insecticides and/or fungicides results in relatively smaller increase in yields. Additionally, increasing insecticide use may result in insects developing resistance thus rendering the control less effective. Increase use of fungicide also increases the risk of higher traces of chemical residues on grains which could affect quality. In either case, such behavior can be maladaptive.

The lack of ownership of farming equipment also limits small farmers' ability to effectively adapt. For instance, if good weather prevails, small farmers need to wait until large farmers finish preparing their own lands and perhaps other small farmers' lands. Thus, they may likely miss the ideal sowing window. Alternatively, even if small farmers are able to sow early, their harvesting may be delayed if other farmers in the planting section fields are not ready to harvest. This is because, combine operators will not enter a section until multiple fields are ready to harvest.

6.16 Conclusion

This chapter provides an in-depth analysis of small farmers' perceptions, impacts and adaptations to changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds. Small farmers in general had a good understanding of climate change or global warming and were able to reflect on common changes in atmospheric conditions attributed to climate change and how they relate to them. An overwhelming majority of small farmers perceived changes in rainfall, temperature, extreme weather events, insects and pests, and weeds. However, perceived changes in diseases were less noticeable perhaps because most small farmers are planting newer blast resistant varieties. These observations were also supported by key informants.

The specific changes concerning rainfall provide additional insights not captured through the analysis of aggregated precipitation data for Guyana. Small farmers noted that rainfall increased in and out of season while key informants alluded to an increase in

273

rainfall intensity. Although total annual rainfall across the country may not have changed significantly (see Chapter 4), changes in the distribution and intensity of rainfall still pose considerable risks, which was highlighted through the farm-level data.

Among the multiple impacts of the perceived changes in rainfall, both small farmers and key informants indicated that harvesting was most affected due to poor dams, wet fields, and lodging of plants. In response to the various impacts reported, farmers engaged in different adaptation practices. The adjustment of planting dates based on water availability and cultivation of different rice varieties were the main response alluded to by small farmers and key informants, respectively.

Both small farmers and key informants agreed that the observed changes in temperature caused the days to be hotter. This coincides with the findings of aggregated temperature data analyzed which showed that both minimum and maximum temperatures have increased over the last 111 years. Hotter days accelerated the evaporation of water in the field thus requiring farmers to replenish the water to allow the grains to mature.

An increase in paddy bugs was the main observation regarding changes in insects and pests. Paddy bugs damaged the grains leading to lower quality and quantity at harvest. In response, farmers are engaging in more preventative spraying. While this is a more conservative approach, it appears farmers prefer to be safe rather than sorry. However, the increase in preventative spraying may also lead to insects in general and paddy bugs specifically becoming immune and/or the destruction of beneficial insects.

Small farmers and key informants also agreed that brown spot disease has increased. Although newer varieties are blast resistant, they are still susceptible to brown spot. Damage to plants or stunted growth was the major impact of diseases reported by small farmers while key informants felt that changes in disease in general and brown spot in particular, reduced the amount of food the plants manufactured through photosynthesis. In both cases, the yield potential of the plants is negatively affected. Preventative spraying and rotation of fungicides were the primary responses by farmers.

An increase in red rice and duckweed were the major change in weeds observed by both small farmers and key informants. The primary impact reported by small farmers was a reduction in plant population due to displacement. However, key informants provided a more detailed account, noting that the observed changes in weeds increased competition for nutrients, space, fertilizer, and sunlight. The primary response alluded to by both groups involved the spraying of more herbicide and the use of a contact chemical to burn red rice in particular.

Excess rainfall that led to flooding, drought, and heavy winds were the primary extreme weather events observed by small farmers and key informants. In terms of specific impacts, small farmers emphasized the impact of flooding during the early stages of growth. They reported that excess rainfall and associated flooding submerged, uprooted, and/or killed young plants. Key informants however, mentioned that lodging of plants due to heavy winds and flooding was the major impact. Matured fields are more susceptible to lodging since the combination of rainfall and flooding saturates the plant stack making it less sturdy to withstand heavy winds. In response to flooding, farmers pumped out water from their fields while they pumped water into the field during dry spells and drought. There is very little that farmers can do in response to heavy winds.

Some adaptation practices farmers engaged in created synergies while others resulted in conflict. For instance, leveling fields facilitates better water management and also helps farmers suppress weeds by growing the rice through water. Similarly, controlling weeds not only reduces competition for water, space, and nutrients, but also helps to remove potential host plants for insects. On the other hand, increased use of fertilizer helps with plant nutrition but also help weeds to proliferate. Fertilizer leaching and/or run-off also provides nutrients for weeds that grow in drainage canals leading to overgrowth that impedes the flow of excess water.

Given the various changes observed and impacts experienced, all small farmers interviewed acknowledged making changes to their farming practice in the last five years. With a list of adaptation practices read to them, small farmers confirmed engaging in practices that were not previously mentioned. This data meshed well with adaptation practices that were reported and/or ranked higher by key informants.

While both seasons were reportedly affected by the changes observed, small farmers noted that the autumn season has been more affected. Yet, more small farmers reported suffering crop failure in the spring season. This is because the spring season is usually at risk of both drought and flooding. Despite the impacts experienced including crop failure, most small farmers reported higher yields on average. This may be credited to better agronomic practices, ongoing adaptation efforts, and the cultivation of newer high yielding rice varieties. It is worth mentioning that perhaps yields could have been higher had it not been for the negative impacts associated with climate change. Additionally, cost of production would be lower if farmers did not have to pay for various adaptation practices to secure yield quality and quantity.

Most small farmers were able to use money from other sources to cope with low prices, late payments by millers, and the losses suffered due to changes in weather patterns. This is because most small farmers had at least one secondary source of income to help them cope with any shortfall in income from rice farming. Regardless, there is a loss in welfare since such income could have been saved or used elsewhere.

It is important to note that the rice extension service is also playing a vital role in helping farmers to adapt. Not only are extension officers disseminating new information and technologies, they are also actively involved in farmers' education through farmer field schools, field visits, field demonstrations, and outreaching events. They are also involved in interagency co-ordination as it relates to access to irrigation, and the maintenance of drainage and farm-to-market roads.

Despite the best efforts to adapt, barriers and limits to adaptation do exist. Access to accurate and timely weather information, access to knowledge regarding adaptation practices, access to markets and credit, and government assistance and subsidies prevent farmers from effectively and efficiently adapting. While these barriers may be overcome at some point in the future, biophysical environment limits, and some economic and sociocultural thresholds cannot be overcome now and/or in the future.

Chapter 7 Conclusion and Recommendations

7.1 Introduction

This research primarily adopts a descriptive approach to study climate change perceptions, impacts, and adaptation among small rice farmers in Guyana. The findings provide important insights that serve to enhance our understanding of climate change at the farm-level and can be used to further bolster the incentive farmers have to adapt to shifts in weather patterns while providing empirical evidence for policymakers and other key stakeholders to act upon. Section 7.2 provides a summary of the key findings, while section 7.3 presents the policy implications and recommendations. Section 7.4 suggests research areas for future work.

7.2 Summary of Key Findings

The major findings organized by research questions are presented below.

7.2.1 Research Question 1: How has the climate in Guyana changed?

Evidence from descriptive statistics, linear trend model, and a two-sample t-test show that minimum and maximum temperatures have increased over the last 111 years. The absolute variability measured by the standard deviation and the relative variability measured by the coefficient of variation (CV) suggests greater shifts in minimum and maximum temperatures over the last 45 years. A strong time trend for the mean, standard deviation, and CV of minimum and maximum temperatures is also evident. Hence, these variables are time-dependent. There is also statistically significant evidence that the mean annual minimum and maximum temperatures are different between the periods 1934-1974 and 1975-2015. Whether or not the amount of rainfall has changed over the last 111 years remains less clear, although longer-term models suggest that precipitation patterns are changing and will further change over the remainder of the century. Evidence from the descriptive statistics indicates a slight upward trend in the mean annual rainfall while there is a steeper trend in the standard deviation and CV. While there is no evidence of a statistically significant time trend in the mean annual rainfall, there is statistically significant evidence of a time trend in the standard deviation and coefficient of variation. This would suggest that, while the overall volume of precipitation may be relatively constant, variability is increasing. There is no statistically significant evidence that the mean annual rainfall is different between the periods 1934-1974 and 1975-2015.

7.2.2 Research Question 2: What non-climatic factors influence rice yields of small farmers?

Land tenure status, tractor ownership, membership in an agricultural organization(s), secondary non-agricultural income, and farms located in regions two and four were found to have a positive relationship with annual yields. These relationships do not indicate causality, but possible explanations emerged from interviews with farmers and other key interviewees. Small farmers that have property rights are more likely to invest in the development of their land thereby increasing their yields. Owning a tractor enables farmers to take advantage of suitable weather conditions and engage in proper land preparation. This helps ensure that they are able to plant within the season and avoid potential disaster of harvesting out of season under wet conditions thus contributing to higher yields. Membership in an agricultural organization(s) allows farmers to gain access

to knowledge and information on new technologies and adaptation practices that can be used to improve their farm management practices. Off-farm income provides farmers with the financial means to afford costly adaptation practices such as additional control measures for insects and pests, pumping of water and the purchase of new seeds. Small farmers in regions two and four appear to be more involved in the day-to-day management of their farms. They also benefit from a close working relationship with extension officers who often go above and beyond. In 2017, the extension officer for region four was recognized as the Extension Officer of Excellence in the Caribbean by Caribbean Agricultural Extension Providers Network (CAEPNET).

Perceived changes in rainfall, farm size, livestock ownership, farmer participate in rice extension training, and household members help with rice farming were found to have a negative relationship with annual yields. Again here, correlation does not imply causality; however, there are noteworthy possible explanations. It is likely that farmers that perceived changes in rainfall were unable to adapt in a timely manner or the adaptive efforts undertaken were ineffective. The adaptation practices applied on larger small farms are usually more difficult and costly to manage. For instance, more labor and larger quantities of chemicals to control insects and weeds may be outside the financial means of small farmers with larger plots. Livestock ownership among small farmers appears to be the main farming activity. Thus, it is likely that these are competing interests where greater emphasis is placed on rearing livestock instead of planting rice.

Farmers that participate in rice extension training may not be able to implement such knowledge because of biophysical and climatic conditions that limit their ability to act. For example, farmers whose land is already low and thus prone to flooding may suffer a decline in yields by reducing seed rates. Alternatively, there may be poor implementation of advice received. Household members that help with rice farming may not possess the skills and experience of seasonal laborers. As such, their involvement with rice farming may be ineffective and/or counterproductive.

In the spring season, land tenure status, tractor ownership, secondary nonagricultural income, and farms located in regions two and four were found to have a positive relationship with rice yields while perceived changes in rainfall, farm size, livestock ownership, farmer participate in rice extension training, and household member(s) help with rice farming were found to have a negative relationship with yields. In the autumn season, male farmers, land tenure status, tractor ownership, membership in an agricultural organization(s), secondary non-agricultural income, soil test, and farms located in regions two and four have a positive relationship with yields. Male farmers are more likely to access new information on adaptation best practices from their male counterparts. A soil test informs farmers of the health and quality of the soil. This information helps them to make better farm management decisions at is relates to fertilizer application and water management which could increase yields. Livestock ownership and adequate irrigation were found to have a negative relationship with yields. Farmers that receive adequate irrigation may not be engaging in proper water management which negatively affects yields.

7.2.3 Research Question 3: How do small rice farmers and key informants perceive changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?

In general, small farmers have a good understanding of climate change or global warming and are able to reflect on common changes in atmospheric conditions attributed to climate change. However, they do not quite understand how or why the climate is changing. The great majority of small farmers and key informants perceived changes in rainfall, temperature, extreme weather events, insects and pests, and weeds. However, perceived changes in diseases were less noticeable perhaps because most farmers are planting newer disease resistant varieties.

Rainfall increasing in and out of season was the key change reported by small farmers while key informants noted an increase in rainfall intensity. This is an important finding since the analysis of aggregate data suggests total rainfall has not changed. Other observations of changes in rainfall include an increase in out of season rainfall, no more seasons, unpredictable rainfall, early or late rainfall, and rainfall fluctuates between seasons. Small farmers and key informants perceived changes in temperature by referring to the increase in hotter days. This coincides with the findings of aggregated temperature data analyzed which showed that both minimum and maximum temperatures have increased over the last 111 years.

An increase in paddy bugs was the main change in insects reported. Small farmers and key informants also contend that paddy bug infestations are correlated with changes in rainfall patterns. An important observation relates to the appearance of the brown planthopper and the leaffolder in the last two seasons. These two insects were previously never recorded in Guyana. Other insects and pests that have increased include water weevil, snail, caterpillar, leaf miner, and slug. Small farmers and key informants noted that brown spot disease has increased. This may be attributed to the decline in spraying for blast disease since the same fungicide also controls brown spot. While the newer rice varieties are blast resistant, some farmers still reported the presence of blast. These are likely to be farmers that are planting the older varieties.

An increase in red rice and duckweed were the major change in weeds observed. Both weeds compete with the rice for space, sunlight, nutrients, and water. However, a higher percentage of red rice grains in the harvest reduces the quality and by extension the price farmers received. Other weeds that have increased include jhussia, monkey tail, muraina, and sedges. Some farmers and key informants did report a decrease in weeds due to continual control efforts. Excess rainfall that led to flooding, drought, and heavy winds were the primary extreme weather events observed.

7.2.4 Research Question 4: What impacts are farmers and key informants seeing due to the observed changes in rainfall, temperature, extreme weather events, insects and pests, diseases, and weeds?

Changes in rainfall had the greatest impact on harvesting due to poor dams [farmto-market roads], wet fields, and lodged plants. Impassable dams prevented timely access to the fields and increased the time and costs associated with transporting the harvested grains from the field to the mills. Wet fields delayed harvesting which reduced the quality of grains while lodged plants often go unharvested. The major impact of higher temperatures was the acceleration of water evaporation in the field. This often resulted in farmers needing to replace loss water. However, farmers in general noted that higher temperatures resulted in better yields.

Paddy bugs damage the grains leading to lower quality and quantity at harvest. Infestations detected too late almost certainly results in damages to the entire field. Damage to plants and stunted growth were the major impacts of diseases reported by small farmers while key informants noted that changes in disease in general and brown spot in particular, reduced the amount of food the plants manufactured through photosynthesis. In both cases, the yield potential of the plants is negatively affected.

Small farmers noted that weeds reduced plant population due to displacement. Key informants noted that the observed changes in weeds increased competition for nutrients, space, fertilizer, and sunlight. As such, the amount and quality of grains harvested are negatively impacted. According to small farmers, excess rainfall and associated flooding submerges, uproots, and/or kills young plants. However, key informants mentioned that lodging of plants due to heavy winds and flooding was the main impact.

7.2.5 Research Question 5: What adaptive measures are they adopting in response to these impacts?

The two main responses by farmers in relation to the changes in rainfall involved adjusting planting dates based on water availability and cultivating different rice varieties. Some farmers also planted early and/or ensured drainage in and around their fields were properly maintained. They also added drainage outlets to their fields and/or used mechanical pumps to hasten the removal of excess water from the field. It is interesting to note that some farmers reported not engaging in any adaptation in response to changes in rainfall. In response to changes in temperature that accelerated the evaporation of water from the field, over 50 percent of farmers replenish the water loss from their field. However, about a quarter of farmers did nothing in response. This is perhaps due to low water levels in the irrigation canals or the lack of water as a whole.

Farmers responded to the threat of paddy bug infestation by engaging in more preventative spraying. They are also engaged in seed treatment in an effort to control early seasons insects and pests. Preventative spraying and the rotation of fungicides were the primary responses by farmers in relation to changes in diseases. As it relates to changes in weeds, the primary response by farmers involved spraying of more herbicide and the use of a contact chemical to eliminate red rice. Some farmers, especially seed paddy producers are engaging in dry and/or water rogueing while some are improving water management in an effort to suppress early season weeds.

Farmers are engaging in several responses with regards to extreme weather events. In response to drought, farmers are pumping water into their fields. This is assuming water is available in the irrigation canals. On the other hand, farmers are pumping out water from their fields in response to excess rainfall that leads to flooding. They are also creating new drainage outlets or maintaining existing drainage in anticipation of excess rainfall and associated flooding. Farmers are limited in their response to heavy winds.

7.2.6 Other Findings

Apart from the findings related to the specific research questions, other findings are worth mentioning:

All small farmers interviewed indicated that they have been engaging in multiple adaptation practices over the last five years. The primary practices include increased herbicide and pesticide application (quantity and frequency); planting different rice varieties; buying new seeds each season; treating seeds for insects; changing sowing/ planting dates; increased fertilizer application (quantity and frequency); pump water; reduce plant densities; spray water in field; sow more seeds; leave land idle; rent out land to large farmers; rotate rice variety; exited rice cultivation and seek off-farm employment; block planting; and patching.

While both seasons were reportedly affected by the changes observed, small farmers noted that the autumn season was more affected. However, more small farmers reported suffering crop failure in the spring season. The autumn season coincides with the primary rainy season [May-July] and is historically susceptible to flooding. However, the spring season is at risk of both drought-like conditions and flooding.

Despite the changes observed, most small farmers reported higher yields on average which may be due to better agronomic practices, ongoing adaptation efforts, and the cultivation of newer high yielding rice varieties. Since 2009, six new varieties have been released through the Guyana Rice Development Board plant breeding research program.

The majority of small farmers had at least one secondary source of income and were able to utilize money from such sources to cope with low prices, late payments by millers, and the losses suffered due to changes in weather patterns. Thus, off-farm employment is a key coping strategy. Rice extension service is helping farmers to adapt by disseminating information and technologies and facilitating farmers' education through farmer field schools, field visits, field demonstrations, and outreach events. However, the lack of new information to share with farmers has resulted in low attendance in recent times.

Extension officers are also involved in interagency coordination as it relates to access to irrigation, the maintenance of drainage and farm-to-market roads. Access to accurate and timely weather information, access to knowledge regarding adaptation practices, access to markets and credit, and government assistance and subsidies are major barriers to effective adaptation reported by farmers.

No form of crop insurance currently exists in Guyana. A report by the World Bank found institutional, technical, financial, and operational challenges related to the provision of crop insurance in Guyana. There is currently no drought, flood or submergence-tolerant, or salt-tolerant rice varieties in Guyana. Drainage and farm-to-market roads are neglected and/or poorly maintained. Small farmers and key informants emphasized the poor state of farm-to-market roads and the drainage system despite repeated calls for action to be taken.

7.3 Policy Implications and Recommendations

The findings of this research have far-reaching policy implications for the rice industry. Given the observed changes in rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds, policy recommendations to improve rice production and farmer's resilience under a changing climate are organized into four general areas: crop development and farming practices; irrigation and water resource management; crop and income loss risk management; and disaster risk management (Bockel 2009).

7.3.1 Crop Development and Farming Practices

Policies in this area focus on research and development capacity, including the development of new rice varieties that are tolerant and suitable under the current and projected future climatic conditions. Additionally, support policies such as tax reduction on agricultural inputs and equipment and improving rice extension service capacity in terms of resources, supplies, and equipment can help encourage and/or increase farmers' adaptation and resilience.

7.3.1.1 Research and Development

Changes in rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds are only expected to increase over time, necessitating greater emphasis on research and development. As such, the GRDB should invest heavily in its research facility to bring it up to international standards. This includes purchasing laboratory equipment and supplies to improve research in areas of plant breeding, agronomy, pathology, and entomology. Examples of equipment and supplies needed include a large oven to break dormancy, a screen house and insects to conduct biological control research, and laboratory equipment to conduct grain analysis. A fully equipped research facility will help enhance the efficiency of research and development activities that focus on addressing ongoing and future risks posed by climate change.

While there are ongoing efforts by the GRDB on varietal development, there should be greater emphasis and urgency in developing drought, flood or submergence-tolerant, and salt tolerant rice varieties. Resources from the International Rice Research Institute (IRRI), the International Center for Tropical Agriculture (CIAT), and the Latin American Fund for Irrigated Rice (FLAR) should be leveraged to help fast track the development of these varieties in order to combat extreme weather events. Alternatively, the GRDB should approve the use of high yielding varieties from other countries that are compatible with the agronomic conditions in Guyana.

7.3.1.2 Taxes Exemption

Changes in rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds have precipitated various adaptation practices among small farmers. However, these practices are costly and sometimes fail to offer relief under extreme conditions. As such, the government (Ministry of Finance (MoF) and the Guyana Revenue Authority (GRA)) should remove value-added tax (VAT) from agricultural inputs (e.g., fertilizer, insecticides, fuel, spare parts, and seeds) and taxes and import duty on machinery and equipment (e.g., motor blower, tractors and implements, combine harvesters) in order to reduce cost of production and enhance farmers' ability to adapt to climate change. By making agricultural inputs zero-rated items under the VAT regime, farmers will have additional disposable income to spend on adaptation practices. It should be noted however, that lower prices for fertilizer and insecticides could be maladaptive. Lower prices can certainly precipitate inefficiencies leading to higher greenhouse gas emissions and other environmental impacts per metric ton of rice produced. Duty-free concessions on farm machinery and equipment will significantly lower the acquisition costs, making it more affordable for farmers in general and small farmers, specifically.

7.3.1.3 Resources and Capacity Development for Extension Officers

The negative correlation between extension training and yields suggests there is room for improvement in their services; this may be particularly true as they are pushed to help farmers deal with uncertain climatic changes. Hence, the GRDB should provide additional professional development and training opportunities for extension officers. This may include both local and overseas training courses and seminars. Areas of training may include climate change impacts and adaptation, weather forecasting, integrated pest management (IPM), rice production and quality control, rice marketing, and training across different agricultural agencies. The knowledge and resources acquired by extension officers will be used to enhance the quality of extension service provided to farmers under a changing climate. Several authors have highlighted the importance of extension service as it relates to climate change adaptation (Charles 2009; Sarker 2012; Nyuor et al. 2016; Huong, Bo, and Fahad 2018).

The GRDB should also ensure that extension officers are properly equipped to effectively perform their duties and responsibilities to support farmers. This includes providing personal protective gear (e.g. raincoats), first aid kits, saltwater meters, projectors, and cameras for field visits and visits with farmers. These resources will help protect extension officers in the field while enabling them to effectively and efficiently discharge their duties and responsibilities. To enhance training activities and the experience of farmers, GRDB should ensure that chemicals and equipment are provided for field demonstrations, and printed brochures and weather calendars are available for distribution. The GRDB should also incentivize farmers for cooperating with the board as it relates to plot demonstration and other activities that require significant farmer involvement.

7.3.1.4 Integrated Pest Management (IMP) and Improved Management Practices

Increases in paddy bug infestations and other insects and pests suggest the need for swifter uptake of IPM. Although IMP currently exists in Guyana, uptake is still relatively nascent. Therefore, the MoA and the GRDB should emphasize the importance of IPM and encourage its adoption through various farmers' education programs and activities. Given that the six 'improved management practices' were introduced over a decade ago, the GRDB should improve upon these practices. They should also develop video programs on various agronomic practices that farmers can easily access.

7.3.2 Irrigation and Water Resource Management

Policies in this area focus on irrigation system management to allow for better water management and adaptation practices in order to address the increasing frequency of droughts experienced in Guyana.

7.3.2.1 Improve Maintenance and Management of Irrigation System

Poor irrigation system management is exacerbated by the changes in precipitation patterns identified by small farmers. The MoA through the National Drainage and Irrigation Authority (NDIA) and the respective water user associations should ensure that irrigation systems are consistently and properly maintained and/or improved. Proper maintenance of irrigation systems helps ensure that water flow is unimpeded thereby ensuring easy access to water, especially during sowing. Additionally, improve demand management and water allocation is crucial during drought. Thus, rationing of available water is necessary to enhance the efficiency of water use and ensure that farmers further away from the source also benefits.

7.3.2.2 Improve Water Management Innovation

During droughts, the irrigation systems are often affected by saltwater intrusion. With the expected future increase in droughts, improvement in water management innovations such as small-scale water capture, storage, and use can help to mitigate the risk posed by saltwater. This is even more important in areas where there are no water conservancy (e.g., Leguan) and/or irrigation infrastructure in place.

7.3.3 Crop and Income Loss Risk Management

Policies in this area are both *ex ante* and *ex post* in nature. Access to stable rice markets and price floors incentivize farmers to grow rice while access to credit helps farmers pay for expensive adaptation practices. The introduction of crop insurance programs helps to spread risk while the provision of subsidies and assistance help farmers recover from crop failure.

7.3.3.1 Rice Markets and Price Support

The high costs of various adaptation practices farmers are currently engaging in are unsustainable under climate change. The increased costs of production coupled with the risk of lower yields and prices pose a major challenge for small farmers. It is only under profitable market conditions that farmers are incentivized to engage in costly adaptation practices with the expectation that their returns on investment will be worthwhile. In addition, price stabilization policies such as a minimum price can help farmers decide in advance their farm management strategies under a changing climate. Thus, the MoA and the GRDB should intensify the promotion and marketing of Guyana's rice in an effort to further diversify rice exports and secure stable and more lucrative markets. Diversification of rice markets would help to reduce risk associated with price fluctuations resulting from changes in regional and global demand. As a means of supporting farmers, the government should consider providing price support to help mitigate low prices and/or high cost of production that is evident in the rice industry. Stable markets and a minimum guaranteed price will help farmers to make better farm management decisions given the current and expected future changes in climate.

7.3.3.2 Access to Credit

The impacts of changes in rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds coupled with ongoing adaptation efforts reinforce the importance of credit markets under a changing climate. Apart from providing funds for farm investment (e.g., land improvement and machinery and equipment purchases), improving post-harvest practices, smoothing household cash flow, enabling better access to markets, and promoting better risk management, access to credit also plays a crucial role in climate adaptation and resilience of agriculture to climate change (IFC 2014). For instance, access to credit enables farmers to purchase better quality inputs (e.g., seeds, fertilizer, and pesticides), implement irrigation systems, and engage in crop rotation and

agroforestry in response to climate change. Thus, the MoA and the GRDB should engage commercial lending institutions to increase the availability of credit at reasonable interest rates to farmers in general and small farmers specifically. Additionally, the government should explore the possibility of re-establishing an agricultural development bank to help support farmers as a whole.

7.3.3.3 Access to Crop Insurance

One of the most significant problems associated with climate change is that it can magnify uncertainty. As such, agricultural insurance can help farmers to transfer or mitigate some risk under a changing climate. The MoA and GRDB should incentivize insurance agencies to develop and provide crop insurance products to help farmers mitigate the risk posed by adverse weather, insects and pests, and disease. For instance, the introduction of Area-yield Index insurance products can prove instrumental in helping farmers to cope with the losses in yields attributed to weather (WB 2010a). In addition, the MoA and GRDB should educate farmers on the importance of crop insurance under a changing climate in order to encourage uptake.

7.3.3.4 Provision of Subsidies and Assistance

The increase in extreme weather events (e.g., flooding, drought, heavy winds) suggests the need to insulate farmers from crop failure. The MoA and GRDB should develop and institute safety net programs that provide subsidies and assistance to farmers, especially those that suffer catastrophic losses due to extreme weather events. Subsidies on inputs can help mitigate the high cost of production. Assistance to farmers who have

suffered crop failure and income loss due to extreme weather events can include seed paddy and/or fertilizer so that they can re-plant their fields in the subsequent season.

7.3.4 Disaster Risk Management

Policies in this area emphasize the importance of developing early warning systems to protect against loss from flood damage. In addition, support policies such as developing a framework for better interagency coordination can help rehabilitate and/or maintain drainage and other infrastructure (e.g., farm-to-market roads) systems in a timely manner, helping to mitigate the impacts of excess rainfall and associated flooding currently experienced in Guyana.

7.3.4.1 Improve Weather Forecasts and Climate Monitoring

Shifts in rainfall patterns such as out of season rain and increases in rainfall intensity precipitate the need for more robust weather forecasting and reporting. As such, the capacity of the Hydrometeorological Service (Hydromet) should be strengthened in order to improve the monitoring and collection of climate data. This should include collecting more detailed data (e.g., rainfall intensity) and improving the weather forecast information available to farmers. Specifically, weather reports should be tailored to each region and areas within each region in an effort to develop community-based early warning systems. In addition, the weather forecasts should be presented in a manner that is easy to understand for farmers, many of whom lack formal education. Sarker (2012) stressed the importance of farmers accessing advance weather information to help with climate adaptation. Accurate and timely weather forecast information would also enable extension

agents to give appropriate agronomic advice to farmers so that they can make better farm management decisions under the current and expected future climatic conditions.

7.3.4.2 Improved Management of Drainage System

More intense rainfall events, including extreme rainfall that leads to flooding, can easily overwhelm the current drainage infrastructure. The MoA through the National Drainage and Irrigation Authority (NDIA) should ensure that drainage systems are consistently and properly maintained and/or improved in order to help mitigate the risk posed by extreme weather events such as excess rainfall and related flooding. This may include larger appropriations for the dredging of the drainage canals and outfalls, the installment of additional pumps at key locations, and the timely provision of resources (e.g., fuel) to maintain and operate existing pumps. Such a policy can help stabilize the farming environment thus making it less shock prone during extreme weather events such as excess rainfall.

7.3.4.3 Better Interagency Coordination

Severe impacts from both extreme rainfall and drought conditions can be abated with improved communication and collaboration between the various agencies under the MoA. For instance, the prediction of extreme weather events by the Hydromet office could be relayed in a timelier manner to NDIA and the GRDB. The NDIA can ensure that major drainage and irrigation canals are properly functioning to accommodate excess water in the case of flood risks or the storage of water in the case of drought. The GRDB through the extension officers can ensure that farmers are aware of the impending extreme weather events so that they can make better farm management decisions. Thus, the MoA and GRDB should ensure that there is better interagency coordination and collaboration among the various national, regional, and local government agencies and organs that play a role in the rice industry. Improved coordination and collaboration will help ensure better information sharing and that timely action as it relates to getting farm-to-market roads and district-specific drainage canals properly maintained and operationally functional. This may entail increased allocation and/or the timely release of financial resources, dredging of new drainage canals, and/or the construction of all-weather roads in key locations.

7.3.5 Other Policy Recommendations

The findings of this research also highlight the need for policy improvement in several other areas. Improvements in these areas can individually and/or collectively enhance farmers' ability to effectively respond to the observed changes and corresponding impacts of climate change now and in the future.

7.3.5.1 Timely Payment of Farmers

In some areas, timely payment of farmers is a chronic problem, which is further exacerbated by climate change. Farmers whose payments are significantly delayed lack the resources needed to effectively and efficiently respond to climate change. As a result, they may be forced to abandon a specific adaptation strategy or cut corners. Thus, the Ministry of Agriculture (MoA) through the Guyana Rice Development Board (GRDB) should strictly enforce the payment requirements stipulated in the Rice Factories Act of 1998. The law requires rice millers to pay 50 percent of the amount due to farmers within two weeks of receipt of the grains and the remainder within 42 days. As an example, the GRDB can tie the renewal of milling licenses to full compliance with this law. Timely payment will allow farmers in general and small farmers specifically to have the necessary operating capital to effectively and efficiently engage in costly adaptation practices in response to climate change.

7.3.5.2 Independent Graders

Farmers engaging in costly adaptation practices are usually discouraged if the price they received for their grains is not competitive. A major concern is distrust between farmers and millers as it relates to the grading of grains and by extension the price farmers receive. Currently, the grading of grains is done by graders employed by the rice millers. Such an arrangement raises concerns over the integrity of the grading process since millers are in a position to influence graders under their employment. As such, the MoA through the GRDB should introduce independent graders to ensure that farmers are treated fairly by millers. In addition, the GRDB should mandate that all rice mills follow the established grading system. Independent graders and fair grading practices will help ensure that farmers receive optimal prices for their grains thus incentivizing adaptation practices.

7.3.5.3 Centrally Located Drying Facilities

Shifts in rainfall patterns affect harvesting. Grains harvested under wet conditions need to be dried in order to preserve milling characteristics and quality. As such, the MoA through the GRDB should construct additional drying facilities at strategic locations in each rice-producing region to facilitate the drying of paddy that is harvested under wet conditions. This will help farmers to prevent spoilage and/or maintain grain quality and hence price received at the mill. Additional, drying facilities will also reduce the dangers associated with the common practice of drying paddy on the roadways.

7.3.5.4 Consumer Protection

Although the majority of small farmers have increased the use of fertilizer and insecticides as part of their adaptation strategies, issues of quality remain a major concern. Thus, the Guyana National Bureau of Standards (GNBS) should ramp-up its efforts to ensure that fertilizer and chemicals available on the local markets are consistently of high quality. This will protect farmers from unscrupulous distributors and suppliers thus ensuring that they receive value for money spent on farm inputs. Similarly, the Pesticides and Toxic Chemicals Control Board (PTCCB) should ensure that imported chemicals are legal and registered for use in Guyana. This will help promote the safety of farmers and the environment under a changing climate.

7.3.5.5 Agricultural Association Membership

The positive relationship between rice yields and membership in an agricultural association emphasizes the importance of farmers joining agricultural groups. Although the Rice Producer Association (RPA) is expected to play an important role in protecting, promoting, and advancing the interest of rice producers, it is often viewed as politically driven and large farmer-focused. As a result, small farmers do not see the potential benefits of membership. As such, farmers in general and small farmers, in particular, should create or join other agricultural organizations or farmers self-help groups in order to strengthen

their voices and gain access to information on new technology, adaptation practices, and climate change. Agricultural organizations can also actively engage farmers in developing positions on issues and connecting legislators whose viewpoints are in sync with that of farmers.

7.3.5.6 Soil Testing

The positive correlation between rice yields and a soil test emphasizes the importance of farmers conducting soil tests. However, the limited access to a soil testing facility in Guyana has been a historical barrier. With the recent construction of a soil testing facility at the University of Guyana's Berbice Campus, the MoA and the GRDB should ensure that small farmers have equal access to the services of the facility and at reasonable costs. These tests provide farmers with important information on soil quality and nutrient needs. Given the observed changes in rainfall, temperature, and extreme weather events, this information can prove crucial in helping farmers better adapt as it relates to water management and quantity and type of fertilizer to apply.

7.3.5.7 Internal Communication and Collaboration

The need for efficient and effective information dissemination is critical under a changing climate. The timely release of new information helps farmers to respond faster and make better farm management decisions. As such, greater attention should be given to communication and collaboration between the research and extension departments at the GRDB. Improved communication will help ensure that farmers are receiving timely advice on agronomic practices and new technologies to help combat the impacts of changes in

rainfall, temperature, extreme weather events, diseases, insects and pests, and weeds. In addition, experts in specific research areas should be more involved in extension and outreach activities so that farmers can receive firsthand advice and answers to their questions directly from the respective experts.

7.4 Future Work

Given the strong descriptive nature of this research, future studies can explore the causal effect between rice yields and climate variables. For example, panel data studies based on secondary production and climate data can be used to explore the fixed or random effects of changes in rainfall and temperature on rice yields.

Although this research concentrates on small farmers across the five primary riceproducing regions of Guyana, future research could extend the analysis by including medium and large-scale farmers. This would allow for comparison across groups at the national and regional levels regarding climate change perceptions, impacts, and adaptation. Additionally, research could focus exclusively on specific areas that have historically suffered from floods, droughts, and saltwater infiltration. Such research findings could help policymakers to devise specific policies that target differences at the district and/or regional levels. Future research could also focus on differences across farmers that plant specific varieties. The literature suggests that climate change has heterogeneous effects on different varieties. Yields on older and newer high yielding varieties can be compared to ascertain which varieties are more vulnerable under a changing climate. Future research could also apply similar methodologies in other countries. The multifaceted nature of this topic also encourages further research in related areas. As such, future studies could examine the impacts of climate change on other agricultural areas. While rice and sugarcane have historically been the bedrock of Guyana's agricultural economy, research can focus on cash crops such as vegetables which forms an integral part of the local agricultural output.

Another appealing research avenue would be to examine the effects of climate change on livestock production. The literature indicates that livestock is affected by floods, drought, and heatwaves. Research in this area can offer interesting insights from a noncrop agriculture perspective which would further strengthen thin empirical evidence that currently exist in Guyana.

Given ongoing concerns regarding the current state of the drainage and irrigation infrastructure, research can focus on ways to improve the current drainage and irrigation systems to better accommodate and/or withstand the additional stress created by a changing climate. Thus, research can explore ways to channel or divert water to prevent flooding during heavy rainfall events or ways to store water in the event of drought.

Future research can also focus on co-benefits, maladaptation, and limits to adaptation in the rice sector and beyond. Adaptation practices that result in multiple benefits being realized can be further explored to ascertain how best to replicate such practices across farmers, districts, and/or regions. In addition, research on the spillover effects of various adaptation practices can be pursued. As noted, some practices are maladaptive. Therefore, research can focus on the effects on maladaptation as it relates to rice production and environmental degradation as a whole. Research can also examine the limits to adaptation by studying the threshold at which the marginal costs of adaptation practices exceed the marginal benefits. In short, Guyana's agricultural sector provides ample research opportunities that would generate greater information and awareness of current and future impacts of climate change.
References

- Ackerman, Frank, and Elizabeth A. Stanton. 2013. "Climate Impacts on Agriculture: A Challenge to Complacency?" 13-01. Medford, MA.
- Adams, Richard M. 1989. "Global Climate Change and Agriculture: An Economic Perspective." *American Journal of Agricultural Economics* 71 (5): 1272–79.
- Adams, Richard M., Ronald A. Fleming, Ching-Chang Chang, Bruce A. McCarl, and Cynthia Rosenzweig. 1995. "A Reassessment of the Economic Effects of Global Climate Change on U.S. Agriculture." *Climate Change* 30 (2): 147–67.
- Adams, Richard M., Brian H. Hurd, Stephanie Lenhart, and Neil Leary. 1998. "Effects of Global Climate Change on Agriculture: An Interpretative Review." *Climate Research* 11 (1): 19–30. https://doi.org/10.3354/cr011019.
- Adger, W. Neil, Shardul Agrawala, and M. Monirul Qader Mirza. 2007. "Assessment of Adaptation Practices, Options, Constraints and Capacity." *Climate Change 2007: Working Group III: Mitigation of Climate Change*. Cambridge, UK. https://doi.org/10.1017/CBO9781107415324.004.
- Adger, W. Neil, Jon Barnett, Katrina Brown, Nadine Marshall, and Karen O'Brien. 2012. "Cultural Dimensions of Climate Change Impacts and Adaptation." *Nature Climate Change* 3: 112–17. https://doi.org/10.1038/nclimate1666.
- Adger, W. Neil, Suraje Dessai, Marisa Goulden, Mike Hulme, Irene Lorenzoni, Donald R. Nelson, Lars Otto Naess, Johanna Wolf, and Anita Wreford. 2009. "Are There Social Limits to Adaptation to Climate Change?" *Climatic Change* 93 (3–4): 335– 54. https://doi.org/10.1007/s10584-008-9520-z.
- Agarwal, Bina. 1983. "Diffusion of Rural Innovations: Some Analytical Issues and the Case of Wood-Burning Stoves." *World Development* 11 (4): 359–376.
- Agresti, Alan, and Barbara Finlay. 2014. *Statistical Methods for the Social Sciences*. 4th ed. USA: Pearson Education Ltd.
- Aktar, Md. Wasim, Dwaipayan Sengupta, and Ashim Chowdhury. 2009. "Impact of Pesticides Use in Agriculture: Their Benefits and Hazards." *Interdisciplinary Toxicology* 2 (1): 1–12.
- Alam, Monirul G. M., Khorshed Alam, and Shahbaz Mushtaq. 2017. "Climate Change Perceptions and Local Adaptation Strategies of Hazard-Prone Rural Households in Bangladesh." *Climate Risk Management* 17: 52–63. https://doi.org/10.1016/j.crm.2017.06.006.
- Allwood, Julian M., Valentina Bosetti, Navroz K. Dubash, Luis Gómez-Echeverri, and Christoph von Stechow. 2014. "Glossary. In: Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Edenhofer, O., R. Pichs-Madruga, Y. Sokona, E. Farahani, S. Kadner, K. Seyb." Cambridge, United Kingdom and New York, NY, USA.
- Amdu, Bewket, Azemeraw Ayehu, and Andent Deressa. 2013. "Farmers' Perception and Adaptive Capacity to Climate Change and Variability in the Upper Catchment of Blue Nile, Ethiopia." 77.
- Anderson, Jock R., and Gershon Feder. 2007. "Agricultural Extension." In *In Handbook* of Agricultural Economics. Volume 3. Agricultural Development: Farmers, Farm Production and Farm Markets, 2343–78. Elsevier.

- Antle, John M., and Claudio O. Stöckle. 2017. "Climate Impacts on Agriculture: Insights from Agronomic-Economic Analysis." *Review of Environmental Economics and Policy* 11 (2): 299–318. https://doi.org/10.1093/reep/rex012.
- Antle, John M., Jetse J. Stoorvogel, and Roberto O. Valdivia. 2014. "New Parsimonious Simulation Methods and Tools to Assess Future Food and Environmental Security of Farm Populations." *Philosophical Transactions of the Royal Society B: Biological Sciences* 369 (1639): 2012.0280. https://doi.org/10.1098/rstb.2012.0280.
- Appiah, Divine Odame, Alfred C.K. Akondoh, Rhoda Kromoh Tabiri, and Amos Annan Donkor. 2018. "Smallholder Farmers' Insight on Climate Change in Rural Ghana." *Cogent Food and Agriculture* 4: 1–16.
- Araya, A., G. Hoogenboom, E. Luedeling, Kiros M. Hadgu, Isaya Kisekka, and Lucieta G. Martorano. 2015. "Agricultural and Forest Meteorology Assessment of Maize Growth and Yield Using Crop Models Under Present and Future Climate in Southwestern Ethiopia." *Agricultural and Forest Meteorology* 214–215: 252–65. https://doi.org/10.1016/j.agrformet.2015.08.259.
- Ashraf, Muhammad, Jayant K. Routray, and Muhammad Saeed. 2014. "Determinants of Farmers' Choice of Coping and Adaptation Measures to the Drought Hazard in Northwest Balochistan, Pakistan." *Natural Hazards* 73 (3): 1451–73. https://doi.org/10.1007/s11069-014-1149-9.
- Auffhammer, Maximilian, Solomon M. Hsiang, Wolfram Schlenker, and Adam Sobel. 2013. "Using Weather Data and Climate Model Output in Economic Analyses of Climate Change." *Review of Environmental Economics and Policy* 7 (2): 181–98. https://doi.org/10.1093/reep/ret016.
- Auffhammer, Maximilian, V. Ramanathan, and Jeffrey R. Vincent. 2012. "Climate Change, the Monsoon, and Rice Yield in India." *Climatic Change* 111 (2): 411–24. https://doi.org/10.1007/s10584-011-0208-4.
- Ayanlade, Ayansina, Maren Radeny, and John F. Morton. 2017. "Comparing Smallholder Farmers' Perception of Climate Change with Meteorological Data: A Case Study from Southwestern Nigeria." *Weather and Climate Extremes* 15: 24–33. https://doi.org/10.1016/j.wace.2016.12.001.
- Aziz Elbehri, Joshua Elliott, and Tim Wheeler. 2015. "Chapter 1 Climate Change, Food Security, and Trade - An Overview of Global Assessments and Policy Insights [In: Climate Change and Food Systems: Global Assessments and Implications for Food Security and Trade]." Rome, Italy.
- Baede, A.P.M., E. Ahlonsou, Y. Ding, and D. Schimel. 2001. "The Climate System: An Overview. In: Climate Change 2001: The Scientific Basis. Contribution of Working Group I to the Third Assessment Report of the Intergovernmental Panel on Climate Change [Houghton, J.T., Y. Ding, D.J. Griggs, M. Noguer, P.J. van Der." Cambridge, United Kingdom and New York, NY, USA. https://www.ipcc.ch/site/assets/uploads/2018/03/TAR-01.pdf.
- Baltagi, Badi H. 2008. *Econometric Analysis of Panel Data*. 4th ed. Chichester, West Sussex, UK: John Wiley & Sons, Inc.
- Ban, Anne Van den, and H. Stuart Hawkins. 1996. Agricultural Extension. 2nd Editio. Wiley-Blackwell.
- Barnett, Jon, Louisa S. Evans, Catherine Gross, Anthony S. Kiem, Richard T. Kingsford, Jean P. Palutikof, and M. Catherine. 2015. "From Barriers to Limits to Climate

Change Adaptation - Path Dependency and the Speed of Change." *Ecology and Society* 20 (3): 11.

- Barnett, Tony, Erik Blas, and Alan Whiteside. 1996. "AIDS Briefs: Subsistence Agriculture (USAID Health and Human Resources Analysis and Research for Africa Project)."
- Barr, Rhona, Samuel Fankhauser, and Kirk Hamilton. 2010. "Adaptation Investments: A Resource Allocation Framework." *Mitigation and Adaptation Strategies for Global Change* 15: 843–58. https://doi.org/10.1007/s11027-010-9242-1.
- Bartlett, James E., Joe W. Kotrlik, and Chadwick C. Higgins. 2001. "Organizational Research: Determining Appropriate Aample Size in Survey Research Appropriate Sample Size in Survey Research." *Information Technology, Learning, and Performance Journal* 19 (1): 43–50. https://doi.org/10.1109/LPT.2009.2020494.
- Basnayake, J., T. Inthavong, S.P. Kam, S. Fukai, J. M. Schiller, and M. Chanphengxay. 2006. Climatic Diversity within the Rice Environments in Laos. In: J. Basnayake, et Al. Eds. Rice in Laos. Los Banos, Philippine. Los Banos, Philippines: International Rice Research Institute.
- Bassu, Simona, Nadine Brisson, Jean-Louis Durand, Kenneth Boote, Jon Lizaso, James W. Jones, Cynthia Rosenzweig, et al. 2014. "How Do Various Maize Crop Models Vary in Their Responses to Climate Change Factors?" *Global Change Biology* 20 (7): 2301–20. https://doi.org/10.1111/gcb.12520.
- Bast, Joseph L. 2013. "Seven Theories of Climate Change." The Heartland Institute. Chicago. 2013. https://www.heartland.org/_templateassets/documents/publications/7theories-web.pdf.
- Battisti, R., P. C. Sentelhas, and K. J. Boote. 2018. "Sensitivity and Requirement of Improvements of Four Soybean Crop Simulation Models for Climate Change Studies in Southern Brazil." *International Journal of Biometeorology* 62 (5): 823– 32.
- Bello, Garba Hima Mamane, and Maman Nafiou Malam Maman. 2015. "A Ricardian Analysis of the Impact of Temperature and Rainfall Variability on Agriculture in Dosso and Maradi Regions of Niger Republic." *Agricultural Sciences* 6 (7): 724–33. https://doi.org/10.4236/as.2015.67070.
- Benhin, James. 2006. "Climate Change and South African Agriculture: Impacts and Adaptation Options." 21. Pretoria, South Africa.
- Benhin, James K A. 2008. "South African Crop Farming and Climate Change: An Economic Assessment of Impacts." *Global Environmental Change* 18 (4): 666–78. https://doi.org/10.1016/j.gloenvcha.2008.06.003.
- Berg, Henk Van Den, and Janice Jiggins. 2007. "Investing in Farmers The Impacts of Farmer Field Schools in Relation to Integrated Pest Management." *World Development* 35 (4): 663–686.
- Besley, Timothy, and Maitreesh Ghatak. 2010. *Handbook of Development Economics: Chapter 68 - Property Rights and Economic Development*. Edited by Dani Rodrik and Mark Rosenzweig.
- Biesbroek, G. Robbert, Judith E. M. Klostermann, Catrien J. A. M. Termeer, and Pavel Kabat. 2013. "On the Nature of Barriers to Climate Change Adaptation." *Regional Environmental Change* 13 (5): 1119–29. https://doi.org/10.1007/s10113-013-0421y.

- Biesbroek, Robbert, Judith Klostermann, Catrien Termeer, and Pavel Kabat. 2011. "Barriers to Climate Change Adaptation in the Netherlands." *Climate Law* 2 (2): 181–99. https://doi.org/10.3233/CL-2011-033.
- Blaikie, Norman. 2000. *Designing Social Research : The Logic of Anticipation*. Malden, MA: Blackwell Publishers, Inc.
- Blanc, Elodie, and John Reilly. 2017. "Approaches to Assessing Climate Change Impacts on Agriculture: An Overview of the Debate." *Review of Environmental Economics and Policy* 11 (2): 247–57. https://doi.org/10.1093/reep/rex011.
- Blanc, Elodie, and Wolfram Schlenker. 2017. "The Use of Panel Models in Assessments of Climate Impacts on Agriculture." *Review of Environmental Economics and Policy* 11 (2): 258–79. https://doi.org/10.1093/reep/rex016.
- Bockel, Louis. 2009. "Climate Change and Agricultural Policies: How to Mainstream Climate Change Adaptation and Mitigation into Agriculture Policies." Rome, Italy: Food and Agriculture Organization.
- BoG, Bank of Guyana. 2017. "Annual Report." Georgetown, Guyana. https://www.bankofguyana.org.gy/bog/economic-financialframework/publications/annual-reports.
- Bozzola, M., E. Massetti, R. Mendelsohn, and F. Capitanio. 2018. "A Ricardian Analysis of the Impact of Climate Change on Italian Agriculture." *European Review of Agricultural Economics* 45 (1): 57–79.
- Bryan, Elizabeth, Temesgen T. Deressa, Glwadys A. Gbetibouo, and Claudia Ringler. 2009. "Adaptation to Climate Change in Ethiopia and South Africa: Options and Constraints." *Environmental Science & Policy* 12 (4): 413–26. https://doi.org/10.1016/j.envsci.2008.11.002.
- Bryan, Elizabeth, Claudia Ringler, Barrack Okoba, Carla Roncoli, Silvia Silvestri, and Mario Herrero. 2013. "Adapting Agriculture to Climate Change in Kenya: Household Strategies and Determinants." *Journal of Environmental Management* 114: 26–35. https://doi.org/10.1016/j.jenvman.2012.10.036.
- Burfisher, Mary E. 2016. *Introduction to Computable General Equilibrium Models*. New York: Cambridge University Press.
- Calzadilla, Alvaro, Katrin Rehdanz, Richard Betts, Pete Falloon, Andy Wiltshire, and Richard S. J. Tol. 2013. "Climate Change Impacts on Global Agriculture." *Climate Change* 120 (1–2): 357–74.
- Cameron, A. Colin, and Pravin K. Trivedi. 2005. *Microeconometrics: Methods and Applications*. New York: Cambridge University Press.
- Carter, Colin A., Funing Zhong, and Jing Zhu. 2012. "Advances in Chinese Agriculture and Its Global Implications." *Applied Economic Perspectives and Policy* 34 (1): 1– 36. https://doi.org/10.1093/aepp/ppr047.
- Charles, Nhemachena. 2009. "Agriculture and Future Climate Dynamics in Africa: Impacts and Adaptation Options." University of Pretoria.
- Cheaney, Robert L., and Peter Randolph. Jennings. 1975. *Field Problems of Rice in Latin America*. Cali, Colombia: Centro Internacional de Agricultura Tropical (CIAT).
- Chen, Yongfu, Zhigang Wu, Katsuo Okamoto, Xinru Han, Guoying Ma, Hsiaoping Chien, and Jing Zhao. 2013. "The Impacts of Climate Change on Crops in China: A Ricardian Analysis." *Global and Planetary Change* 104: 61–74.
- Claessens, L., J. M. Antle, J. J. Stoorvogel, R. O. Valdivia, P. K. Thornton, and M.

Herrero. 2012. "A Method for Evaluating Climate Change Adaptation Strategies for Small-Scale Farmers Using Survey, Experimental and Modeled Data." *Agricultural Systems* 111: 85–95. https://doi.org/10.1016/j.agsy.2012.05.003.

Closset, Mathilde, Boubaker Ben Bechir Dhehibi, and Aden Aw-Hassan. 2015. "Measuring the Economic Impact of Climate Change on Agriculture: A Ricardian Analysis of Farmlands in Tajikistan." *Climate and Development* 7 (5): 454–68. https://doi.org/10.1080/17565529.2014.989189.

Coase, R.H. 1960. "The Problem of Social Cost." Journal of Law and Economics 3: 44.

- Collins, M., R. Knutti, J. Arblaster, J. L. Dufresne, T. Fichefet, P. Friedlingstein, X. Gao, et al. 2013. "Long-Term Climate Change: Projections, Commitments and Irreversibility. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change." Cambridge, United Kingdom and New York, NY, USA.
- Colston, Josh M, Tahmeed Ahmed, Cloupas Mahopo, Gagandeep Kang, Margaret Kosek, Francisco De Sousa, Prakash Sunder, et al. 2018. "Evaluating Meteorological Data from Weather Stations, and from Satellites and Global Models for a Multi-Site Epidemiological Study." *Environmental Research* 165: 91–109. https://doi.org/10.1016/j.envres.2018.02.027.
- Coolidge, Frederick L. 2013. *Statistics: A Gentle Introduction*. USA: Sage Publications, Inc.
- Cooper, P. J. M., J. Dimes, K. P. C. Rao, B. Shapiro, B. Shiferaw, and S. Twomlow. 2008. "Coping Better with Current Climatic Variability in the Rain-Fed Farming Systems of Sub-Saharan Africa: An Essential First Step in Sdapting to Future Climate Change?" Agriculture, Ecosystems and Environment 126 (1–2): 24–35. https://doi.org/10.1016/j.agee.2008.01.007.
- Coulibaly, Jeanne Y., Glwadys A. Gbetibouo, Godfrey Kundhlande, Gudeta W. Sileshi, and Tracy L. Beedy. 2015. "Responding to Crop Failure: Understanding Farmers' Coping Strategies in Southern Malawi." *Sustainability* 7 (2): 1620–36. https://doi.org/10.3390/su7021620.
- Creswell, John W. 2012. *Educational Research: Planning, Conducting, and Evaluating Quantitative and Qualitative Research.* 4th ed. India: Pearson Education, Inc.
- Darwin, R., M. Tsigas, J. Lewabdrowski, and A. Raneses. 1995. "World Agriculture and Climate Change: Economic Adaptations." No. 73. Washington DC. http://ageconsearch.umn.edu/bitstream/33933/1/ae950703.pdf.
- Davis, Kristin E. 2009. "Agriculture and Climate Change: An Agenda for Negotiation in Copenhagen - The Important Role of Extension Systems." 2020 Vision for Food, Agriculture and the Environment. Washington DC. www.ifpri.org.
- Dell, Melissan, Benjamin F. Jones, and Benjamin A. Olken. 2014. "What Do We Learn from the Weather? The New Climate-Economy Literature." *Journal of Economic Literature* 52 (3): 740–98.
- Denzin, Norman K., and Yvonna S. Lincoln, eds. 2011. *The Sage Handbook of Qualitative Research*. 4th ed. London.
- Deressa, Temesgen, and Rashid Hassan. 2009. "Economic Impact of Climate Change on Crop Production in Ethiopia: Evidence from Cross-Section Measures." *Journal of African Economies* 18 (4): 529–54. https://doi.org/10.1093/jae/ejp002.
- Deressa, Temesgen Tadesse, Rashid M. Hassan, Claudia Ringler, Tekie Alemu, and

Mahmud Yesuf. 2009. "Determinants of Farmers' Choice of Adaptation Methods to Climate Change in the Nile Basin of Ethiopia." *Global Environmental Change* 19 (2): 248–55. https://doi.org/10.1016/j.gloenvcha.2009.01.002.

- Deschênes, Olivier, and Michael Greenstone. 2007. "The Economic Impacts of Climate Change: Evidence from Agricultural Output and Random Fluctuations in Weather." *American Economic Review* 97 (1): 354–85.
- Deutsch, Curtis A, Joshua J Tewksbury, and Michelle Tigchelaar. 2018. "Increase in Crop Losses to Insect Pests in a Warming Climate." *Science* 361 (6405): 916–19. https://doi.org/10.1126/science.aat3466.
- Dias, Henrique Boriolo, and Paulo Cesar Sentelhas. 2017. "Evaluation of Three Sugarcane Simulation Models and Their Ensemble for Yield Estimation in Commercially Managed Fields." *Field Crops Research* 213: 174–85.
- Dinar, Ariel, and Robert Mendelsohn, eds. 2011. *Handbook on Climate Change and Agriculture*. Cheltenham, UK: Edward Elgar Publishing Limited.
- Dlugokencky, E. J., B. D. Hall, S. A. Montzka, G. Dutton, J. Mühle, and J. W. Elkins.
 2018. "Global Climate: Atmospheric Composition Long-Lived Greenhouse Gases [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society* 99 (8): S46. https://doi.org/10.1175/2018BAMSStateoftheClimate.1.
- Dow, Kirstin, Frans Berkhout, and Benjamin L. Preston. 2013. "Limits to Adaptation to Climate Change: A Risk Approach." *Current Opinion in Environmental Sustainability* 5 (3–4): 384–91. https://doi.org/10.1016/j.cosust.2013.07.005.
- Dow, Kirstin, Frans Berkhout, Benjamin L. Preston, Richard J T Klein, Guy Midgley, and M. Rebecca Shaw. 2013. "Limits to Adaptation." *Nature Climate Change* 3: 305–7. https://doi.org/10.1038/nclimate1847.
- Durand, Jean-Louis;, Kenel; Delusca, Ken; Boote, Jon; Lizaso, Remy; Manderscheid, Hans Johachim; Weigel, Alex C; Ruane, et al. 2018. "How Accurately Do Maize Crop Models Simulate the Interactions of Atmospheric CO2 Concentration Levels with Limited Water Supply on Water Use and Yield?" *European Journal of* Agronomy 100: 67–75. https://doi.org/10.1016/j.eja.2017.01.002.
- Eakin, Hallie. 2005. "Institutional Change, Climate Risk, and Rural Vulnerability: Cases from Central Mexico." *World Development* 33 (11): 1923–38. https://doi.org/10.1016/j.worlddev.2005.06.005.
- ECLAC, Economic Commission for Latin America and the Caribbean. 2011. "An Assessment of the Economic Impact of Climate Change on the Agricultural Sector in Guyana."
- Elum, Zelda A., David M. Modise, and Ana Marr. 2017. "Farmer's Perception of Climate Change and Responsive Strategies in Three Selected Provinces of South Africa." *Climate Risk Management* 16: 246–57. https://doi.org/10.1016/j.crm.2016.11.001.
- EPA, Environmental Protection Agency. 2016a. "Climate Change Indicators: Arctic Sea Ice." *Climate Change Indicators in the United States*. Washington, DC. https://www.epa.gov/climate-indicators/climate-change-indicators-arctic-sea-ice.
 - ——. 2016b. "Climate Change Indicators: Atmospheric Concentrations of Greenhouse Gases." *Climate Change Indicators in the United States*. Washington, DC.

—. 2016c. "Climate Change Indicators: Global Greenhouse Gas Emissions." *Climate Change Indicators in the United States*. Washington, DC.

https://www.epa.gov/climate-indicators/climate-change-indicators-global-

greenhouse-gas-emissions.

—. 2016d. "Climate Change Indicators: Sea Level." *Climate Change Indicators in the United States*. Washington, DC. https://www.epa.gov/climate-indicators/climate-change-indicators-sea-level.

—. 2016e. "Climate Change Indicators: Sea Surface Temperature." *Climate Change Indicators in the United States*. Washington, DC. https://www.epa.gov/climate-indicators/climate-change-indicators-sea-surface-temperature.

—. 2016f. "Climate Change Indicators: U.S. and Global Precipitation." *Climate Change Indicators in the United States*. Washington, DC. www.epa.gov/climate-indicators.

——. 2016g. "Climate Change Indicators: U.S. and Global Temperature." *Climate Change Indicators in the United States*. Washington, DC.

Fanfai, Shemuel. 2016. "Rice Sector Declined by 12.7%." *Guyana Times, Inc.*, December 30, 2016. https://guyanatimesgy.com/rice-sector-declined-by-12-7/.

- FAO, Food and Agricultural Organization. 1978. "Report on Agro-Ecological Zones." Rome, Italy.
 - ——. 1996. "Agroecological Zoning Guidelines." Rome, Italy.
 - http://www.fao.org/docrep/W2962E/w2962e00.htm#P-2.
- . 2011. "The State of Food and Agriculture: Women in Agriculture." Rome.
- ———. 2016. "The State of Food and Agriculture: Climate Change, Agriculture, and Food Security." Rome, Italy.
- ———. 2018. "The State of Agricultural Commodity Markets: Agricultural Trade, Climate Change, and Food Security." Rome, Italy.
- Fernandes, E.C.M., A. Soliman, R. Confalonieri, M. Donatelli, and F. Tubiello. 2012. "Climate Change and Agriculture in Latin America, 2020-2050: Projected Impacts and Response to Adaptation Strategies." Washington, DC.
- Fischer, G., M. Shah, F. N. Tubiello, and H. van Velthuizen. 2005. "Socio-Economic and Climate Change Impacts on Agriculture: An Integrated Assessment, 1990–2080." *Philosophical Transactions of the Royal Society B: Biological Sciences* 360: 2067– 2083.
- Fisher, Ronald A. 1925. "The Influence of Rainfall on the Yield of Wheat at Rothamsted." *Philosophical Transactions of the Royal Society of London. Series B, Containing Papers of a Biological Character* 213 (89–142).
- Fleischer, Aliza, Ivgenia Lichtman, and Robert Mendelsohn. 2007. "Climate Change, Irrigation, and Israeli Agriculture: Will Warming Be Harmful?" *Ecological Economics* 65 (3): 508–15.
- Fleisher, David H., Bruno Condori, Roberto Quiroz, Ashok Alva, Senthold Asseng, Carolina Barreda, Marco Bindi, et al. 2017. "A Potato Model Intercomparison Across Varying Climates and Productivity Levels." *Global Change Biology* 23: 1258–81. https://doi.org/10.1111/gcb.13411.
- Frank, J., and Chris Penrose Buckley. 2012. "Small-Scale Farmers and Climate Change. How Can Farmer Organisations and Fairtrade Build the Adaptive Capacity of Smallholders?" London.
- Fry, Juliane L., Hans-F Graf, Richard Grotjahn, Marilyn N. Raphael, Clive Saunders, and Richard Whitaker. 2010. *The Enclyclopedia of Weather and Climate Change - A Complete Visual Guide*. First. Berkeley and Los Angeles: University of California

Press.

- Gadédjisso-Tossou, Agossou. 2015. "Understanding Farmers' Perceptions of and Adaptations to Climate Change and Variability: The Case of the Maritime, Plateau and Savannah Regions of Togo." *Agricultural Sciences* 6: 1441–54. https://doi.org/10.4236/as.2015.612140.
- Galinato, M. I., K. Moody, and C. M. Piggin. 1999. *Upland Rice Weeds of South and Southeast Asia*. Makati City, Philippines: International Rice Research Institute.
- Gandure, S., S. Walker, and J. J. Botha. 2013. "Farmers' Perceptions of Adaptation to Climate Change and Water Stress in a South African Rural Community." *Environmental Development* 5: 39–53. https://doi.org/10.1016/j.envdev.2012.11.004.
- Garlati, Adrián. 2013. "Climate Change and Extreme Weather Events in Latin America: An Exposure Index."
- Gautam, H. R., M. L. Bhardwaj, and R. Kumar. 2013. "Climate Change and Its Impact on Plant Diseases." *Current Science* 105 (12): 1685–91. https://doi.org/10.1016/j.eswa.2010.09.036.
- Gbetibouo, G. A., and R. M. Hassan. 2005. "Measuring the Economic Impact of Climate Change on Major South African Field Crops: A Ricardian Approach." *Global and Planetary Change* 47 (2–4 SPEC. ISS.): 143–52. https://doi.org/10.1016/j.gloplacha.2004.10.009.
- Gbetibouo, Glwadys Aymone. 2009. "Understanding Farmers' Perceptions and Adaptations to Climate Change and Variability: The Case of the Limpopo Basin, South Africa." 00849. Washington, DC. https://doi.org/10.1068/a312017.
- Geest, Kees van der, and Koko Warner. 2014. Vulnerability, Coping and Loss and Damage from Climate Events. In: Collins A. (Ed). Hazards, Risks, and Disasters in Society. Elsevier. https://doi.org/10.1016/B978-0-12-396451-9.00008-1.
- Gillig, Dhazn, and Bruce A. McCarl. 2002. "Introduction to Computable General Equilibrium Models (CGE)." Department of Agricultural Economics Texas A&M University. 2002. http://agecon2.tamu.edu/people/faculty/mccarl-bruce/685/topic6trade.pdf.
- GLSC, Guyana Lands and Surveys Commission. 2006. "Administrative Map of Guyana." 2006. https://glsc.gov.gy/services/maps/#prettyPhoto.
- Gourdji, Sharon M., Adam M. Sibley, and David B. Lobell. 2013. "Global Crop Exposure to Critical High Temperatures in the Reproductive Period: Historical Trends and Future Projections." *Environmental Research Letters* 8: 10. https://doi.org/10.1088/1748-9326/8/2/024041.
- GRB, Research and Extension Division of the Guyana Rice Board. 1976. *Some Notes on Rice Cultivation*. Edited by Guyana Rice Board. Georgetown, Guyana.
- GRDB, Guyana Rice Development Board. 2008. "Guyana Celebrating 100 Years as a Rice Exporter." *Guyana Rice Development Board*. Georgetown, Guyana.
- ——. 2009. "Rice Farmer's Manual." Georgetown.
- ——. 2015a. "2015 Annual Report." Georgetown, Guyana.
- ------. 2015b. "Rice Breeding Program in Guyana." http://grdb.gy/wp-
- content/uploads/2016/09/New-Plant-Breeding-Brochure.pdf.
- ——. 2017. "Farm Registrer." Georgetown, Guyana.
- ——. 2018. "Farm Register." Georgetown, Guyana.

Greene, William H. 2011. *Econometric Analysis*. 7th Editio. New York: Pearson Education Ltd.

GRPAA. 1946. Guyana Rice Producers Association Act. Guyana: Laws of Guyana.

- Guan, Kaiyu, Benjamin Sultan, Michela Biasutti, Christian Baron, and David B. Lobell. 2015. "What Aspects of Future Rainfall Changes Matter for Crop Yields in West Africa?" *Geophysical Research Letters* 42 (19): 8001–10. https://doi.org/10.1002/2015GL063877.
- Gujarati, Damodar N., Dawn C. Porter, and Sangeetha Gunasekar. 2012. *Basic Econometrics*. 5th ed. Chennai, India: McGraw-Hill Education (India) Private Limited.
- Guo, Author J. H., X. J. Liu, Y. Zhang, J. L. Shen, W. X. Han, W. F. Zhang, P. Christie, et al. 2010. "Significant Acidification in Major Chinese Croplands." *Science* 327 (5968): 1008–10.
- Haden, Van R., Meredith T. Niles, Mark Lubell, Joshua Perlman, and Louise E. Jackson. 2012. "Global and Local Concerns: What Attitudes and Beliefs Motivate Farmers to Mitigate and Adapt to Climate Change?" *PLoS ONE* 7 (12): e52882. https://doi.org/10.1371/journal.pone.0052882.
- Haque, Md. Nazmul, Md. Hazrat Ali, and Sheikh Muhammad Masum. 2016. *Climate Change Impacts on Rice Production in Bangladesh*. LAP LAMBERT Academic Publishing.
- Harris, I., P. D. Jones, T. J. Osborn, and D. H. Lister. 2014. "Updated High-Resolution Grids of Monthly Climatic Observations - CRU TS3.10 Dataset." *International Journal of Climatology* 34: 623–42. https://doi.org/10.1002/joc.3711.
- Harvey, Celia A., Zo Lalaina Rakotobe, Nalini S. Rao, Radhika Dave, Hery Razafimahatratra, Rivo Hasinandrianina Rabarijohn, Haingo Rajaofara, and James L. MacKinnon. 2014. "Extreme Vulnerability of Smallholder Farmers to Agricultural Risks and Climate Change in Madagascar." *Philosophical Transactions* of the Royal Society B: Biological Sciences 369 (1639). https://doi.org/10.1098/rstb.2013.0089.
- Harvey, Celia A., Milagro Saborio-Rodríguez, M. Ruth Martinez-Rodríguez, Barbara Viguera, Adina Chain-Guadarrama, Raffaele Vignola, and Francisco Alpizar. 2018.
 "Climate Change Impacts and Adaptation among Smallholder Farmers in Central America." *Agriculture and Food Security* 7 (1): 1–20. https://doi.org/10.1186/s40066-018-0209-x.
- Hatfield, Jerry L., and John H. Prueger. 2015. "Temperature Extremes: Effect on Plant Growth and Development." *Weather and Climate Extremes* 10 (A): 4–10. https://doi.org/10.1016/j.wace.2015.08.001.
- Heinlein, Robert A. 1973. Time Enough for Love. New York: The Penguin Putnam Inc.
- Hertel, Thomas W. 1990. "General Equilibrium Analysis of U.S. Agriculture: What Does It Contribute?" *The Journal of Agricultural Economics Research* 42 (3): 9.
 ——. 1999. "Applied General Equilibrium Analysis of Agricultural and Resource

Hewitson, B., A.C. Janetos, T.R. Carter, F. Giorgi, R.G. Jones, W.-T. Kwon, L.O.Mearns, E.L.F. Schipper, and M. van Aalst. 2014. "Regional Context. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the

Policies." 99–2. W. Lafayette, IN.

Intergovernmental Panel on Climate Change." Cambridge, UK and New York, NY. https://doi.org/10.1017/CBO9781107415386.001.

- Hickey, Catherine, and Tony Weis. 2012. "The Challenge of Climate Change Adaptation in Guyana." *Climate and Development* 4 (1): 66–74. https://doi.org/10.1080/17565529.2012.661036.
- Hitayezu, Patrick, Edilegnaw Wale, and Gerald Ortmann. 2017. "Assessing Farmers' Perceptions about Climate Change: A Double-Hurdle Approach." *Climate Risk Management* 17: 123–38. https://doi.org/10.1016/j.crm.2017.07.001.
- Hoddinott, John. 2006. "Shocks and Their Consequences across and within Households in Rural Zimbabwe." *Journal of Development Studies* 42 (2): 301–21. https://doi.org/10.1080/00220380500405501.
- Holland, Margaret Buck, Sierra Zaid Shamer, Pablo Imbach, Juan Carlos Zamora, Claudia Medellin Moreno, Efraín J Leguía Hidalgo, Camila I Donatti, M Ruth Martínez-rodríguez, and Celia A Harvey. 2017. "Mapping Adaptive Capacity and Smallholder Agriculture : Applying Expert Knowledge at the Landscape Scale." *Climate Change* 141: 139–53. https://doi.org/10.1007/s10584-016-1810-2.
- Holm, Leroy, J. Doll, E. Holm, J. Pancho, and J. Herberger. 1997. *World Weeds: Natural Histories and Distribution*. U.S.A: John Wiley & Sons, Inc.
- Holm, Leroy G., Donald L. Plucknett, Juan V. Pancho, and James P. Herberger. 1977. *The World's Worst Weeds: Distribution and Biology*. Honolulu, Hawaii (USA): University of Hawaii Press.
- Homenauth, Oudho P. 1997. *A Comprehensive Review of the Rice Industry in Guyana*. Georgetown, Guyana: Guyana Rice Development Board.
- Houser, T., S. Hsiang, R. Kopp, and K. Larsen. 2015. *Economic Risks of Climate Change: An American Prospectus*. New York: Columbia University Press.
- Huang, B., J. Kennedy, Y. Xue, and H. M. Zhang. 2018. "Global Oceans: Sea Surface Temperatures [in 'State of the Climate in 2017']." Bulletin of the American Meteorological Society. Vol. 99.

https://doi.org/10.1175/2018BAMSStateoftheClimate.1.

Huong, Nguyen Thi Lan, Yao Shun Bo, and Shah Fahad. 2018. "Economic Impact of Climate Change on Agriculture Using Ricardian Approach: A Case of Northwest Vietnam." *Journal of the Saudi Society of Agricultural Sciences*, 1–9. https://doi.org/10.1016/j.jssas.2018.02.006.

Hydromet, Hydrometeorological Service - Guyana. n.d. "Rainfall Regimes of Guyana."

- IFC, International Finance Corporation. 2014. "Access to Finance for Smallholder Farmers: Learning from the Experiences of Microfinance Institutions in Latin America." Washington, DC.
- iNews. 2017a. "Heavy Rainfall, Floods Batter Rice Crops in Region 2." *Inews Guyana*, January 5, 2017.
 - ——. 2017b. "Minister Trotman to Seek Emergency Intervention for Region Two Rice Farmers." *Inews Guyana*, May 15, 2017.
 - ——. 2017c. "Berbice Flooded by Heavy Rainfall." *Inews Guyana*, June 24, 2017. https://www.inewsguyana.com/berbice-flooded-by-heavy-rainfall/.
 - ——. 2018. "Corentyne Rice Farmers Suffering Losses with Crop." *Inews Guyana*, March 9, 2018.
- IPCC. 2012. "Glossary of Terms. In: Managing the Risks of Extreme Events and

Disasters to Advance Climate Change Adaptation." Cambridge, UK, and New York, NY. https://www.ipcc.ch/site/assets/uploads/2018/03/SREX_Full_Report-1.pdf.

—. 2014a. "Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change." Cambridge, United Kingdom and New York, NY, USA.

—. 2014b. "Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change." Geneva, Switzerland.

https://doi.org/10.1017/CBO9781107415324.

- ——. 2018. "Summary for Policymakers. In: Global Warming of 1.5°C. An IPCC Special Report on the Impacts of Global Warming of 1.5°C above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response To." Geneva, Switzerland.
- IRRI, International Rice Research Institute. n.d. "Rice and Climate Change." Accessed December 20, 2018. http://irri.org/news/hot-topics/rice-and-climate-change.
- Jaisingh, Lloyd R. 2006. *Statistics for the Utterly Confused*. 2nd ed. New York: McGraw-Hill Education.
- Janvry, Alain De, and Elizabeth Sadoulet. 2016. *Development Economics: Theory and Practice*. New York: Routledge.
- Jay, A., D.R. Reidmiller, C.W. Avery, D. Barrie, B.J. DeAngelo, A. Dave, M. Dzaugis, et al. 2018. "Overview. In: Impacts, Risks, and Adaptation in the United States: Fourth National Climate Assessment, Volume II [Reidmiller, D.R., C.W. Avery, D.R. Easterling, K.E. Kunkel, K.L.M. Lewis, T.K. Maycock, and B.C. Stewart (Eds.)]." Washington, DC. https://doi.org/10.7930/NCA4.2018.CH1.
- Jones, Lindsey, and Emily Boyd. 2011. "Exploring Social Barriers to Adaptation: Insights from Western Nepal." *Global Environmental Change* 21: 1262–74. https://doi.org/10.1016/j.gloenvcha.2011.06.002.
- Julie, Defang Taku, F. A. Amungwa, and I. Manu. 2017. "Role of Agricultural Extension in Climate Change Adaptation in Cameroon." *International Journal of Horticulture, Agriculture and Food Science* 1 (3): 21–26.
- Kabubo-Mariara, Jane, and Fredrick K. Karanja. 2007. "The Economic Impact of Climate Change on Small Farms in Nigeria: A Ricardian Approach." 4334. Washington, DC. https://doi.org/10.1016/j.gloplacha.2007.01.002.
- Kaiser, Harry M., Susan J. Riha, Daniel S. Wilks, David G. Rossiter, and Radha Sampath. 1993. "A Farm-Level Analysis of Economic and Agronomic Impacts of Gradual Climate Warming." *American Journal of Agricultural Economics* 75 (2): 387–98.
- Kant, Kamal, Shivkumar Telkar, Meghna Gogoi, and Deepak Kumar. 2017. "Crop Simulation Models." *Biomolecule Reports*, 2017.
- Kelly, D. L., C. D. Kolstad, and G. T. Mitchell. 2005. "Adjustment Costs from Environmental Change." *Journal of Environmental Economics and Management* 50 (3): 468–95.
- Kelly, P. M., and W. N. Adger. 2000. "Theory and Practice in Assessing Vulnerability to Climate Change and Facilitating Adaptation." *Climatic Change* 47: 325–52. https://doi.org/10.1023/A:1005627828199.

- Kenney, J. F., and E. S. Keeping. 1962. *Mathematics of Statistics, Pt. 1.* 3rd ed. Princeton, NJ: Van Nostrand.
- Khadka, Nishan. 2016. "Impacts of Climate Change on Rice Yield." Helsinki Metropolia University of Applied Sciences.
- Khan, Mohamed. 2015. "Serious Paddy Bug Infestation in Essequibo." *Stabroek News*, July 23, 2015.
 - —. 2018a. "Some Region Two Rice Farmers Will Lose Crops to Rainy Weather." *Stabroek News*, June 25, 2018.
 - https://www.stabroeknews.com/2018/opinion/letters/06/25/some-region-two-rice-farmers-will-lose-crops-to-rainy-weather/.
 - —. 2018b. "Region Two Rice Farmers Who Planted Late Hit by Paddy Bug." *Stabroek News*, November 20, 2018.

https://www.stabroeknews.com/2018/opinion/letters/11/20/region-two-rice-farmers-who-planted-late-hit-by-paddy-bug/.

- Kim, Chang-Gil. n.d. "The Impact of Climate Change on the Agricultural Sector: Implications of the Agro-Industry for Low Carbon, Green Growth Strategy and Roadmap for the East Asian Region." Korea.
- Klein, Richard J. T., Guy F. Midgley, Benjamin L. Preston, Mozaharul Alam, Frans G. H. Berkhout, Kirstin Dow, and M. Rebecca Shaw. 2014. "Adaptation Opportunities, Constraints, and Limits. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Clim." Cambridge, United Kingdom and New York, NY, USA. https://doi.org/10.1017/CBO9781107415379.021.
- KN. 2018a. "E'bo Rice Farmers Halt Harvesting Due to Poor Roads." *Kaieteur News*, September 18, 2018. https://www.kaieteurnewsonline.com/2018/09/18/ebo-rice-farmers-halt-harvesting-due-to-poor-roads/.
 - ——. 2018b. "E'bo Farmers to Burn 1,000 Acres of Rice after Bug Bites Deep." *Kaieteur News*, October 24, 2018.
 - —. 2019. "El Niño-like Conditions Ravaging Region Six Agri Sector." *Kaieteur News*, January 22, 2019. https://www.kaieteurnewsonline.com/2019/01/22/el-nino-like-conditions-ravaging-region-six-agri-sector/.
- KNMI, Royal Netherlands Meteorological Institute Climate Explorer. 2018. "Climate Change Atlas." 2018.

http://climexp.knmi.nl/plot atlas form.py?id=someone@somewhere.

- Koc, B., and M. Ceylan. 2013. "Effects of Climate Change on Rice Production: The Case of Turkey." *African Journal of Agricultural Research* 8 (23): 2903–10.
- Kritee, Kritee, Drishya Nair, Daniel Zavala-araiza, Jeremy Proville, Joseph Rudek, and Tapan K Adhya. 2018. "High Nitrous Oxide Fluxes From Rice Indicate the Need to Manage Water for Both Long- and Short-Term Climate Impacts." *Proceedings of the National Academy of Sciences* 115 (39): 9720–9725. https://doi.org/10.1073/pnas.1809276115.
- Kurukulasuriya, Pradeep, and Mohamed Ihsan Ajwad. 2007. "Application of the Ricardian Technique to Estimate the Impact of Climate Change on Smallholder Farming in Sri Lanka." *Climatic Change* 81: 39–59. https://doi.org/10.1007/s10584-005-9021-2.

Kurukulasuriya, Pradeep, and Robert Mendelsohn. 2008a. "A Ricardian Analysis of the Impact of Climate Change on African Cropland." *African Journal Agriculture and Resource Economics* 2 (1): 1–23.

——. 2008b. "How Will Climate Change Shift Agro-Ecological Zones and Impact African Agriculture?" 4717. Washington, DC.

- Kurukulasuriya, Pradeep, Robert Mendelsohn, Rashid Hassan, James Benhin, Temesgen Deressa, Mbaye Diop, Helmy Mohamed Eid, et al. 2006. "Will African Agriculture Survive Climate Change?" *The World Bank Economic Review* 20 (3): 367–88. https://doi.org/https://doi.org/10.1093/wber/lhl004.
- Kusters, Koen, and Norbu Wangdi. 2013. "The Costs of Adaptation: Changes in Water Availability and Farmers' Responses in Punakha District, Bhutan." *International Journal of Global Warming* 5 (4): 387–99. https://doi.org/10.1504/IJGW.2013.057287.
- Lamichhane, Jay Ram, Marco Barzman, Kees Booij, Piet Boonekamp, Nicolas Desneux, Laurent Huber, Per Kudsk Messéan, et al. 2015. "Robust Cropping Systems to Tackle Pests under Climate Change. A Review." *Agronomy for Sustainable Development* 35 (2): 443–59. https://doi.org/DOI 10.1007/s13593-014-0275-9.
- Lanning, Sarah B., Terry J. Siebenmorgen, Paul A. Counce, Amogh A. Ambardekar, and Andy Mauromoustakos. 2011. "Extreme Nighttime Air Temperatures in 2010 Impact Rice Chalkiness and Milling Quality." *Field Crops Research* 124 (1): 132– 36. https://doi.org/10.1016/j.fcr.2011.06.012.
- Li, Tao, Toshihiro Hasegawa, Xinyou Yin, Yan Zhu, Kenneth Boote, Myriam Adam, Simone Bregaglio, et al. 2015. "Uncertainties in Predicting Rice Yield by Current Crop Models under a Wide Range of Climatic Conditions." *Global Change Biology* 21: 1328–41. https://doi.org/10.1111/gcb.12758.
- Li, Xin, Jiankang Wang, Chengsong Zhu, and Jianming Yu. 2012. "Chapter Six -Computer Simulation in Plant Breeding." *Advances in Agronomy* 116: 219–64. https://doi.org/https://doi.org/10.1016/B978-0-12-394277-7.00006-3.
- Lippert, Christian, Tatjana Krimly, and Joachim Aurbacher. 2009. "A Ricardian Analysis of the Impact of Climate Change on Agriculture in Germany." *Climatic Change* 97: 593–610. https://doi.org/10.1007/s10584-009-9652-9.
- Livezey, Robert E., Konstantin Y. Vinnikov, Marina M. Timofeyeva, Richard Tinker, and Huug M. van den Dool. 2007. "Estimation and Extrapolation of Climate Normals and Climatic Trends." *Journal of Applied Meteorology and Climatology* 46: 1759–76. https://doi.org/10.1175/2007JAMC1666.1.
- Lobell, David B., Kimberly Nicholas Cahill, and Christopher B. Field. 2007. "Historical Effects of Temperature and Precipitation on California Crop Yields." *Climatic Change* 81: 187–203. https://doi.org/10.1007/s10584-006-9141-3.
- Lorenzoni, Irene, Sophie Nicholson-Cole, and Lorraine Whitmarsh. 2007. "Barriers Perceived to Engaging with Climate Change among the UK Public and Their Policy Implications." *Global Environmental Change* 17: 445–59. https://doi.org/10.1016/j.gloenvcha.2007.01.004.
- Lowder, Sarah K, Jakob Skoet, and Terri Raney. 2016. "The Number, Size, and Distribution of Farms, Smallholder Farms, and Family Farms Worldwide Q." *World Development* 87: 16–29. https://doi.org/10.1016/j.worlddev.2015.10.041.
- Mach, Katharine J., Serge Planton, and Christoph von Stechow. 2014. "Annex II:

Glossary [Mach, K.J., S. Planton and C. von Stechow (Eds.)]. In: Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change." Geneva, Switzerland. https://doi.org/10.1017/CBO9781107415324.

- Mackill, D. J., W. R. Coffman, and D. P. Garrity. 1996. *Rainfed Lowland Rice Improvement*. Manila, Philppines: International Rice Research Intitute.
- Maddison, David. 2000. "A Hedonic Analysis of Agricultural Land Prices in England and Wales." *European Review of Agricultural Economics* 27 (4): 519–32.
- Madramootoo, Harry. 1969. *Hints for Profitable Rice Farming*. Edited by Guyana Rice Corporation. Georgetown, Guyana.
 - ——. 1973. *The Story of Rice in Guyana*. Georgetown, Guyana: Guyana Rice Corporation.

——. 1974. *A Rice Farmer's Word Book*. Georgetown: Guyana Rice Board.

- Mann, Micheal E., and Lee R. Kump. 2015. *Dire Predictions: Understanding Climate Change*. 2nd ed. New York: DK Publishing.
- Mase, Amber Saylor, Benjamin M. Gramig, and Linda Stalker Prokopy. 2017. "Climate Change Beliefs, Risk Perceptions, and Adaptation Behavior Among Midwestern U.S. Crop Farmers." *Climate Risk Management* 15: 8–17. https://doi.org/10.1016/j.crm.2016.11.004.

Mason, Robert D., Douglas A. Lind, and William G. Marchal. 1999. *Statistical Techniques in Business and Economics*. USA: Irwin/McGraw-Hill.

- Massetti, E., and R. Mendelsohn. 2011. "Estimating Ricardian Functions with Panel Data." *Climate Change Economics* 2 (4): 301–19. https://doi.org/10.1142/S2010007811000322.
- Massetti, Emanuele, and Robert Mendelsohn. 2018. "Measuring Climate Adaptation: Methods and Evidence." *Review of Environmental Economics and Policy* 12 (2): 324–41. https://doi.org/10.1093/reep/rey007.
- Matsuura, Kenji, and Cort J. Willmott. 2018. "Terrestrial Air Temperature and Precipitation: 1900-2014 Gridded Monthly Time Series (Version 4.01)." 2018. http://climate.geog.udel.edu/~climate/html pages/download.html#T2014.
- McGowan, Winston. 2008a. "The Beginnings of Rice Cultivation in Guyana (Part 1/3)." *Stabroek News*, October 30, 2008.

—. 2008b. "The Beginnings of Rice Cultivation in Guyana (Part 2)." *Stabroek News*, November 6, 2008. https://www.stabroeknews.com/2008/features/11/06/thebeginnings-of-rice-cultivation-in-guyana-part-2/.

- ——. 2008c. "The Beginnings of Rice Cultivation in Guyana (Part 3/3)." *Stabroek News*, November 13, 2008.
- Mckitrick, Ross R. 1998. "The Econometric Critique of Computable General Equilibrium Modeling: The Role of Functional Forms." *Economic Modelling* 15 (4): 543–73.
- McSweeney, C., M. New, and G. Lizcano. 2010. "UNDP Climate Change Country Profiles - Guyana." New York.
- Mehar, Mamta, Surabhi Mittal, and Narayan Prasad. 2016. "Farmers Coping Strategies for Climate Shock: Is It Differentiated by Gender?" *Journal of Rural Studies* 44: 123–31. https://doi.org/10.1016/j.jrurstud.2016.01.001.
- Meier, Walter N., Greta K. Hovelsrud, Bob E. H. van Oort, Jeffrey R. Key, KitM.Kovacs, Christine Michel, Christian Haas, et al. 2014. "Arctic Sea Ice in Transformation: A

Review of Recent Observed Changes and Impacts on Biology and Human Activity." *Reviews of Geophysics* 52 (3): 185–217.

https://doi.org/10.1002/2013RG000431.Received.

- Mendelsohn, R., and M. Reinsborough. 2007. "A Ricardian Analysis of US and Canadian Farmland." *Climate Change* 81 (1): 9–17.
- Mendelsohn, R., and D. Tiwari. 2000. "Two Essays on Climate Change and Agriculture: A Developing Country Perspective." 145. Rome, Italy: FAO Economic and Social Development.
- Mendelsohn, R, and Ariel Dinar. 2005. "Exploring Adaptation to Climate Change in Agriculture: The Potential of Cross-Sectional Analysis." *Agriculture and Rural Development Notes*. Washington, DC.
- http://siteresources.worldbank.org/INTARD/Resources/Climate_Change_3.pdf. Mendelsohn, Robert. 2007. "What Causes Crop Failure?" *Climatic Change* 81 (1): 61–
- 70. https://doi.org/10.1007/s10584-005-9009-y.
- Mendelsohn, Robert, Jesus Arellano-Gonzalez, and Perter Christensen. 2009. "A Ricardian Analysis of Mexican Farms A Ricardian Analysis of Mexican Farms." *Environment and Development Economics* 15 (2): 153–71. https://doi.org/10.1017/S1355770X09990143.
- Mendelsohn, Robert, Alan Basist, Ariel Dinar, Pradeep Kurukulasuriya, and Claude Williams. 2007. "What Explains Agricultural Performance: Climate Normals or Climate Variance?" *Climate Change* 81 (1): 85–99.
- Mendelsohn, Robert, Pradeep Kurukulasuriya, Alan Basist, Felix Kogan, and Claude Williams. 2007. "Climate Analysis with Satellite Versus Weather Station Data." *Climate Change* 81 (1): 71–83.
- Mendelsohn, Robert, William Nordhaus, and Daidee Shaw. 1994. "The Impact of Global Warming on Agriculture: A Ricardian Analysis: Reply." *American Economic Review* 84 (4): 753–71. https://doi.org/10.1257/aer.89.4.1046.
- Mendelsohn, Robert O., and Emanuele Massetti. 2017. "The Use of Cross-Sectional Analysis to Measure Climate Impacts on Agriculture: Theory and Evidence." *Review of Environmental Economics and Policy* 11 (2): 280–98. https://doi.org/10.1093/reep/rex017.
- Mertz, Ole, Kirsten Halsnæs, Jørgen E. Olesen, and Kjeld Rasmussen. 2009. "Adaptation to Climate Change in Developing Countries." *Environmental Management* 43 (5): 743–52. https://doi.org/10.1007/s00267-008-9259-3.
- Mishra, Diptimayee, Naresh Chandra Sahu, and Dukhabandhu Sahoo. 2016. "Impact of Climate Change on Agricultural Production of Odisha (India): A Ricardian Analysis." *Regional Environmental Change* 16 (2): 575–84. https://doi.org/10.1007/s10113-015-0774-5.
- Mitchell, T.D., and P.D. Jones. 2005. "An Improved Method of Constructing a Database of Monthly Climate Observations and Associated High-Resolution Grids." *International Journal of Climatology* 25: 693–712.
- MoA, Ministry of Agriculture of Guyana. 2013. "Disaster Risk Management Plan for the Agriculture Sector 2013 2018." Georgetown, Guyana.
- Molua, Ernest L. 2009. "An Empirical Assessment of the Impact of Climate Change on Smallholder Agriculture in Cameroon." *Global and Planetary Change* 67 (3–4): 205–8. https://doi.org/10.1016/j.gloplacha.2009.02.006.

- Monirul Islam, Md, Susannah Sallu, Klaus Hubacek, and Jouni Paavola. 2014. "Limits and Barriers to Adaptation to Climate Variability and Change in Bangladeshi Coastal Fishing Communities." *Marine Policy* 43: 208–16. https://doi.org/10.1016/j.marpol.2013.06.007.
- Moomaw, J. C., and B. S. Vergara. 1965. *The Environment of Tropical Rice Production. The Mineral Nutrition of the Rice Plant. Proc. of a Symposium at the International Rice Research Institute.* Baltimore, MD, USA: The Johns Hopkins Press.
- Moore, Frances, and David B. Lobell. 2014. "Adaptation Potential of European Agriculture in Response to Climate Change." *Nature Climate Change* 4: 610–614.
- MoP, Ministry of the Presidency. 2015. "Climate Resilience Strategy and Action Plan for Guyana." Georgetown.
- Morduch, Jonathan. 2002. "Consumption Smoothing Across Space: Testing Theories of Risk-Sharing in the ICRISAT Study Region of South India." 2002/55. Helsinki, Finland.
- Morita, Satoshi, Hiroyuki Shiratsuchi, Jun-ichi Takanashi, and Konosuke Fujita. 2004. "Effect of High Temperature on Grain Ripening in Rice Plants. Analysis of the Effects of High Night and High Day Temperatures Applied to the Panicle and Other Parts of the Plant." *Japanese Journal of Crop Science* 73 (1): 77–83.
- Morton, John F. 2007. "The Impact of Climate Change on Smallholder and Subsistence Agriculture." https://doi.org/10.1073/pnas.0701855104.
- Moser, S. C., and J. A. Ekstrom. 2010. "A Framework to Diagnose Barriers to Climate Change Adaptation." *Proceedings of the National Academy of Sciences* 107 (51): 22026–31. https://doi.org/10.1073/pnas.1007887107.
- Motha, Raymond P. 2011. "The Impact of Extreme Weather Events on Agriculture in the United States. In: Attri S., Rathore L., Sivakumar M., Dash S. (Eds) Challenges and Opportunities in Agrometeorology." In *Challenges and Opportunities in Agrometeorology*, 397–407. Springer, Berlin, Heidelberg. https://doi.org/10.1007/978-3-642-19360-6.
- Mulwa, Chalmers, Paswel Marenya, Dil Bahadur, and Menale Kassie. 2017. "Response to Climate Risks Among Smallholder Farmers in Malawi: A Multivariate Probit Assessment of the Role of Information, Household Demographics, and Farm Characteristics." *Climate Risk Management* 16: 208–21. https://doi.org/10.1016/j.crm.2017.01.002.
- NAPG. 2016. "National Adaptation Plan Guyana Inception Report." Georgetown, Guyana. http://journals.sagepub.com/doi/10.1177/0042098015620341.
- NATCOM1. 2002. "Initial National Communication to the UNFCCC." Georgetown, Guyana. http://unfccc.int/resource/docs/natc/guync1.pdf.
- NATCOM2. 2012. "Guyana Second National Communication to the UNFCCC." Georgetown, Guyana. https://doi.org/10.1162/LEON r 00465.
- ND-GAIN. 2016. "Notre Dame Global Adaptation Initiative Country Index." 2016.
- Nelson, Gerald C., Mark W. Rosegrant, Jawoo Koo, Richard Robertson, Timothy Sulser, Tingju Zhu, Claudia Ringler, Siwa Msangi, \ Amanda Palazzo, et al. 2009. "Climate Change: Impact on Agriculture and Costs of Adaptation." Washington DC.
- Nelson, Gerald C., Mark W. Rosegrant, Jawoo Koo, Richard Robertson, Timothy Sulser, Tingju Zhu, Claudia Ringler, Siwa Msangi, Amanda Palazzo, et al. 2009. "Impact on Agriculture and Costs of Adaptation." Washington, DC.

- Neuman, W. Lawrence. 2011. Social Research Methods: Qualitative And Quantitative Approaches. 7th ed. Boston.
- Nevame, A. Y.M., R. M. Emon, M. A. Malek, M. M. Hasan, Md Amirul Alam, Farrah Melissa Muharam, Farzad Aslani, M. Y. Rafii, and M. R. Ismail. 2018.
 "Relationship between High Temperature and Formation of Chalkiness and Their Effects on Quality of Rice." *BioMed Research International*, 18. https://doi.org/10.1155/2018/1653721.
- New, M., M. Hulme, and P.D. Jones. 1999. "Representing Twentieth-Century Space-Time Climate Variability. Part I: Development of a 1961- 90 Mean Monthly Terrestrial Climatology." *Journal of Climate* 12: 829 – 856.
 - ——. 2000. "Representing Twentieth Century Space-Time Climate Variability. Part II: Development of 1901–1996 Monthly Grids of Terrestrial Surface Climate." *Journal* of Climate 13: 2217–38.
- Nicholson, Walter, and Christopher Snyder. 2012. *Microeconomic Theory: Basic Principles and Extensions*. 11th ed. Mason, OH: South-Western, Cengage Learning.
- Nielsen, Jonas Østergaard, and Anette Reenberg. 2010. "Cultural Barriers to Climate Change Adaptation: A Case Study from Northern Burkina Faso." *Global Environmental Change* 20: 142–52.

https://doi.org/10.1016/j.gloenvcha.2009.10.002.

- Niles, Meredith T., Mark Lubell, and Van R. Haden. 2013. "Perceptions and Responses to Climate Policy Risks Among California Farmers." *Global Environmental Change* 23: 1752–60. https://doi.org/10.1016/j.gloenvcha.2013.08.005.
- NOAA, National Oceanic and Atmospheric Administration National Center for Environmnetal Information. 2017. "What's the Difference between Climate and Weather?" 2017. https://oceanservice.noaa.gov/facts/weather_climate.html.
 2018. "Defining Climate Normals in New Ways." 2018.

https://www.ncdc.noaa.gov/news/defining-climate-normals-new-ways.

- Noble, Ian R., Saleemul Huq, Yuri A. Anokhin, Jo Ann Carmin, Dieudonne Goudou, Felino P. Lansigan, Balgis Osman-Elasha, and Alicia Villamizar. 2014. "Adaptation Needs and Options. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change." Cambridge, United Kingdom and New York, NY, USA. https://doi.org/10.1017/CBO9781107415379.019.
- Nyuor, Anslem Bawayelaazaa, Emmanuel Donkor, Robert Aidoo, Samuel Saaka Buah, Jesse Naab, Stephen Nutsugah, Jules Bayala, and Robert Zougmoré. 2016. "Economic Impacts of Climate Change on Cereal Production: Implications for Sustainable Agriculture in Northern Ghana." *Sustainability* 8 (724): 1–17. https://doi.org/10.3390/su8080724.
- Oberlack, Christoph. 2017. "Diagnosing Institutional Barriers and Opportunities for Adaptation to Climate Change." *Mitigation and Adaptation Strategies for Global Change* 22: 805–38. https://doi.org/10.1007/s11027-015-9699-z.
- Ochieng, Justus, Lilian Kirimi, and Mary Mathenge. 2016. "Effects of Climate Variability and Change on Agricultural Production: The Case of Small Scale Farmers in Kenya." *NJAS Wageningen Journal of Life Sciences* 77: 71–78. https://doi.org/10.1016/j.njas.2016.03.005.

- Osborn, T. J., J. Barichivich, I. Harris, G. van der Schrier, and P. D. Jones. 2018. "Global Climate: Hydrological Cycle Drought [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society*. Vol. 99. https://doi.org/10.1175/2018BAMSStateoftheClimate.1.
- Pandey, Sushil, Humnath Bhandari, Shijun Ding, Preeda Prapertchob, Ramesh Sharan, Dibakar Naik, Sudhir K. Taunk, and Asras Sastri. 2007. "Coping with Drought in Rice Farming in Asia: Insights from a Cross-Country Comparative Study." *Agricultural Economics* 37 (s1): 213–24.
- Parry, M. L. 1990. *Climate Change and World Agriculture*. London: Earthscan Publications.
- Parry, M. L., C. Rosenzweig, A. Iglesias, G. Fischer, and M. Livermore. 1999. "Climate Change and World Food Security: A New Assessment." *Global Environmental Change* 9: 51–67.
- Parry, M L, C Rosenzweig, A Iglesias, M Livermore, and G Fischer. 2004. "Effects of Climate Change on Global Food Production Under SRES Emissions and Socio-Economic Scenarios." *Global Environmental Change* 14 (1): 53–67. https://doi.org/10.1016/j.gloenvcha.2003.10.008.
- Parry, Martin, Nigel Arnell, Pam Berry, David Dodman, Samuel Fankhauser, Chris Hope, Sari Kovats, et al. 2009. "Assessing the Costs of Adaptation to Climate Change: A Review of the UNFCCC and Other Recent Estimates." London.
- Passel, Steven Van, Emanuele Massetti, and Robert Mendelsohn. 2017. "A Ricardian Analysis of the Impact of Climate Change on European Agriculture." *Environmental* and Resource Economics 67 (4): 725–60. https://doi.org/10.1007/s10640-016-0001y.
- PCF, Pacific Climate Futures. 2017. "Understanding Climate Variability and Change." 2017. https://www.pacificclimatefutures.net/en/help/climate-projections/understanding-climate-variability-and-change/.
- Peel, M. C., B. L. Finlayson, and T. A. McMahon. 2007. "Updated World Map of the Köppen-Geiger Climate Classification." *Hydrology and Earth System Sciences* 11 (5): 1633–44. https://doi.org/10.5194/hess-11-1633-2007.
- Peng, Shaobing, Jianliang Huang, John E. Sheehy, Rebecca C. Laza, Romeo M.
 Visperas, Xuhua Zhong, Grace S. Centeno, Gurdev S. Khush, and Kenneth G.
 Cassman. 2004. "Rice Yields Decline with Higher Night Temperature from Global Warming." *Proceedings of the National Academy of Sciences* 101 (27): 9971–75.
- Perovich, D., W. Meier, M. Tschudi, S. Farrell, S. Hendricks, S. Gerland, C. Haas, et al. 2018. "The Artic: Sea Ice Cover - Sea Ice Extent [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society* 99 (8): S147-148. https://doi.org/10.1175/2018BAMSStateoftheClimate.1.
- Peters, Kristian, Laura Breitsameter, and Bärbel Gerowitt. 2014. "Impact of Climate Change on Weeds in Agriculture: A Review." *Agronomy for Sustainable Development* 34 (4): 707–21. https://doi.org/10.1007/s13593-014-0245-2.
- Pettengell, Catherine. 2010. "Enabling People Living in Poverty to Adapt."
- Pirttioja, N., T. R. Carter, S. Fronzek, M. Bindi, H. Hoffmann, T. Palosuo, M. Ruiz-Ramos, et al. 2015. "Temperature and Precipitation Effects on Wheat Yield Across a European Transect: A Crop Model Ensemble Analysis Using Impact Response Surfaces." *Climate Research* 65: 87–105.

https://doi.org/https://doi.org/10.3354/cr01322.

- Planton, Serge. 2013. "Annex III: Glossary [Planton, S. (Ed.)]. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor." United Kingdom and New York, NY, USA.
- Powell, J. P., and S. Reinhard. 2015. "Measuring the Effects of Extreme Weather Events on Yields." *Weather and Climate Extremes* 12: 69–79. https://doi.org/10.1016/j.wace.2016.02.003.
- Qiu, Xianjin, Yunlong Pang, Zhihua Yuan, Danying Xing, Jianlong Xu, Michael Dingkuhn, Zhikang Li, and Guoyou Ye. 2015. "Genome-Wide Association Study of Grain Appearance and Milling Quality in a Worldwide Collection of Indica Rice Germplasm." *PLoS ONE* 10 (12): 1–25. https://doi.org/10.1371/journal.pone.0145577.
- Ragnauth, K., D. Singh, J. Singh and B. Chintamanie. 2014. "Technology Transfer in
- Smallholder Rice Farmers in Guyana: The Success Story of the Six Improved Practices." *Proceedings of the Caribbean Food Crops Society* 50: 82–88.
- Rama Rao, Y. V., Lyndon Alves, Bhaleka Seulall, Ziona Mitchell, Kelvin Samaroo, and Garvin Cummings. 2012. "Evaluation of the Weather Research and Forecasting (WRF) Model over Guyana." *Natural Hazards* 61: 1243–61. https://doi.org/10.1007/s11069-011-9977-3.
- Ramli, Noor Hafizah, Suzana Yusup, Benjamin Wei Bin Kueh, Puteri Sarah Diba Kamarulzaman, Noridah Osman, Mardyahwati Abd. Rahim, Ramlan Aziz, Sulaiman Mokhtar, and Abu Bakar Ahmad. 2018. "Effectiveness of Biopesticides Against Brown Planthopper (Nilaparvata Lugens) in Paddy Cultivation." Sustainable Chemistry and Pharmacy 8: 16–20. https://doi.org/10.1016/j.scp.2018.01.001.
- Ramraj, Robert. 1996. "Rainfall Distribution Along Coastal Guyana, 1901-1980." *The Geographical Bulletin* 38 (1): 29–40.
- Rauff, Kazeem O., and Rasaq Bello. 2015. "A Review of Crop Growth Simulation Models as Tools for Agricultural Meteorology." *Agricultural Sciences* 6: 1098– 1105.
- Reinsborough, Michelle J. 2003. "A Ricardian Model of Climate Change in Canada." *Canadian Journal of Economics* 36: 20–40.
- RFA. 1998. *Rice Factories Act*. Guyana: Laws of Guyana.
- Ricardo, D. 1817. *On the Principles of Political Economy and Taxation (Reprinted 2004)*. London: Dover Publications.
- Rodgers, Everett M. 2003. Diffusion of Innovations. 5th ed. New York: Free Press.
- Roncoli, Carla, Keith Ingram, and Paul Kirshen. 2002. "Reading the Rains: Local Knowledge and Rainfall Forecasting in Burkina Faso." *Society and Natural Resources* 15 (5): 409–27. https://doi.org/10.1080/08941920252866774.
- Rose, Steven K. 2015. "The Inevitability of Climate Adaptation in U.S. Agriculture." *Choices* 30 (2): 1–5.
- Rosenzweig, Cynthia. 1989. "Global Climate Change: Predictions and Obselrvations." American Journal of Agricultural Economics 71 (5): 1265–71.
- Rosenzweig, Cynthia, and Daniel Hillel. 2008. *Climate Variability and the Global Harvest: Impacts of El Niño and Other Oscillations on Agroecosystems*. Ney York:

Oxford University Press.

- Rosenzweig, Cynthia, Ana Iglesias, X.B. Yang, Paul R. Epstein, and Eric Chivian. 2001. "Climate Change and Extreme Weather Events. Implication for Food Production, Plant Diseases, and Pests." *Global Change & Human Health* 2 (2): 90–104. https://doi.org/10.1007/s10584-010-9834-5.
- Rosenzweig, Cynthia, and Martin Parry. 1994. "Potential Impact of Climate Change on World Food Supply." *Nature* 367: 133–38.
- Rosenzweig, Mark R., and Kenneth I. Wolpin. 1993. "Credit Market Constriants, Consumption Smoothing, and the Accumulation of Durable Production Assets in Low-Income Countries: Investments in Bullocks in India." *Journal of Political Economy* 101 (2): 423–44. https://doi.org/10.2307/256588.
- Ruel, Erin, William E. Wagner, and Brian Joseph Gillespie. 2015. *The Practice of Survey Research: Theory and Applications*. United States: Sage Publications, Inc.
- Salvo, Maria De, Roberta Raffaelli, and Riccarda Moser. 2013. "The Impact of Climate Change on Permanent Crops in an Alpine Region: A Ricardian Analysis." *Agricultural Systems* 118: 23–32. https://doi.org/10.1016/j.agsy.2013.02.005.
- Sánchez-Lugo, A., C. Morice, P. Berrisford, and A. Argüez. 2018. "Global Climate: Temperature - Global Surface Temperature [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society*. Vol. 99. https://doi.org/10.1175/2018BAMSStateoftheClimate.1.
- Sánchez, Berta, Anton Rasmussen, and John R. Porter. 2013. "Temperatures and the Growth and Development of Maize and Rice: A Review." *Global Change Biology* 20 (2): 408–17.
- Sands, Philippe. 1992. "United Nations Framework Convention on Climate Change." *Framework for Climate Change* 1 (3): 270–77. https://doi.org/10.1111/j.1467-9388.1992.tb00046.x.
- Sanghi, Apurva, and Robert Mendelsohn. 2008. "The Impacts of Global Warming on Farmers in Brazil and India." *Global Environmental Change* 18: 655–65. https://doi.org/10.1016/j.gloenvcha.2008.06.008.
- Sarker, Mohamed A. R. 2012. "Impacts of Cimate Change on Rice Production and Farmers' Adaptation in Bangladesh." University of Southern Queensland.
- Schlenker, By Wolfram, W Michael Hanemann, and Anthony C Fisher. 2005. "Will U.S. Agriculture Really Benefit from Global Warming? Accounting for Irrigation in the Hedonic Approach." *American Economic Review* 95 (1): 395–406.
- Schutt, Russell K. 2014. Investigating the Social World: The Process and Practice of Research. 8th ed. London: Sage Publications, Inc.
- Seo, S. Niggol, and Robert Mendelsohn. 2008a. "Animal Husbandry in Africa: Climate Change Impacts and Adaptations." *African Journal Agriculture and Resource Economics* 2 (1): 65–82.
 - ——. 2008b. "Climate Change Impacts on Latin American Farmland Values: The Role of Farm Type." *Revista de Economia e Agronegocio* 6 (2): 159–76.
 - . 2008c. "Measuring Impacts and Adaptations to Climate Change: A Structural Ricardian Model of African Livestock Management." *Agricultural Economics* 38: 151–65.
- Seo, S. Niggol, Robert Mendelsohn, Ariel Dinar, Rashid Hassan, and Pradeep Kurukulasuriya. 2012. "The Impact of Climate Change on Agro-Ecological Zones:

Evidence from Africa A Ricardian Analysis of the Distribution of Climate Change Impacts on Agriculture across Agro-Ecological Zones in Africa." *Environment and Development Economics* 17 (6): 663–87.

https://doi.org/10.1017/S1355770X12000241.

- Seo, Sung-no Niggol, Robert Mendelsohn, and Mohan Munasinghe. 2005. "Climate Change and Agriculture in Sri Lanka: A Ricardian Valuation." *Environment and Development Economics* 10: 581–96. https://doi.org/10.1017/S1355770X05002044.
- Seulall, Mohanlall. 2018. "Freak Storm Hits Essequibo Coast." *Guyana Times, Inc.*, April 18, 2018. https://guyanatimesgy.com/freak-storm-hits-essequibo-coast/.
- Shi, Wanju, Xinyou Yin, Paul C. Struik, Fangming Xie, Ralf C. Schmidt, and Krishna S. V. Jagadish. 2016. "Grain Yield and Quality Responses of Tropical Hybrid Rice to High Night-Time Temperature." *Field Crops Research* 190: 18–25. https://doi.org/10.1016/j.fcr.2015.10.006.
- Shikuku, Kelvin M., Leigh Winowiecki, Jennifer Twyman, Anton Eitzinger, Juan G. Perez, Caroline Mwongera, and Peter Läderach. 2017. "Smallholder Farmers' Attitudes and Determinants of Adaptation to Climate Risks in East Africa." *Climate Risk Management* 16: 234–45. https://doi.org/10.1016/j.crm.2017.03.001.
- Sietz, Diana, Maria Boschütz, and Richard J. T. Klein. 2011. "Mainstreaming Climate Adaptation into Development Assistance: Rationale, Institutional Barriers and Opportunities in Mozambique." *Environmental Science and Policy* 14: 493–502. https://doi.org/10.1016/j.envsci.2011.01.001.
- Simpson, Brent M. 2016. Preparing Smallholder Farm Families to Adapt to Climate Change. Pocket Guide 1: Extension Practice for Agricultural Adaptation. Baltimore, MD, USA: Catholic Relief Services:
- SN. 2015. "Rice Lands in Essequibo Flooded after Heavy Rains." Stabroek News, May 25, 2015. https://www.stabroeknews.com/2015/news/guyana/05/25/rice-lands-in-essequibo-flooded-after-heavy-rains/.
- ——. 2016. "Flooding on Essequibo Coast Threatens Rice Crop." *Stabroek News*, December 30, 2016.
- ——. 2018. "Nand Persaud & Company Taking Rice Industry's Challenges in Its Stride." *Stabroek News*, November 30, 2018.
- ——. 2019. "Region Two Farmers Facing Water Woes." *Stabroek News*, January 4, 2019.
- Stage, Jesper. 2010. "Economic Valuation of Climate Change Adaptation." Annals of the New York Acadamy of Sciences 1185: 150–63.
- Stephenson, Tannecia S., Lucie A. Vincent, Theodore Allen, Cedric J. Van Meerbeeck, Natalie Mclean, Thomas C. Peterson, Michael A. Taylor, et al. 2014. "Changes in Extreme Temperature and Precipitation in the Caribbean Region, 1961-2010." *International Journal of Climatology* 34 (9): 2957–71. https://doi.org/10.1002/joc.3889.
- Takahashi, Bruno, Morey Burnham, Carol Terracina-Hartman, Amanda R. Sopchak, and Theresa Selfa. 2016. "Climate Change Perceptions of NY State Farmers: The Role of Risk Perceptions and Adaptive Capacity." *Environmental Management* 58 (6): 946–57. https://doi.org/10.1007/s00267-016-0742-y.
- Thompson, P. R., M. A. Merrifield, E. Leuliette, W. Sweet, D. P. Chambers, B. D. Hamlington, S. Jevrejeva, et al. 2018. "Global Oceans: Sea Level Variability and

Change [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society* 99 (8): S84–85. https://doi.org/10.1175/2018PAMSStateoftheClimate 1

- https://doi.org/10.1175/2018BAMSStateoftheClimate.1.
- Toté, Carolien, Domingos Patricio, Hendrik Boogaard, Raymond Van Der Wijngaart, Elena Tarnavsky, and Chris Funk. 2015. "Evaluation of Satellite Rainfall Estimates for Drought and Flood Monitoring in Mozambique." *Remote Sensing* 7: 1758–76. https://doi.org/10.3390/rs70201758.
- Townsend, Robert M. 1994. "Risk and Insurance in Village India." *Econometrica* 62 (3): 539–91.
- Tripathi, Amarnath, and Ashok K. Mishra. 2017. "Knowledge and Passive Adaptation to Climate Change: An Example from Indian Farmers." *Climate Risk Management* 16: 195–207. https://doi.org/10.1016/j.crm.2016.11.002.
- UNDP, United Nations Development Program. 2018. "Latest Human Development Index (HDI) Ranking." Human Development Indices and Indicators: 2018 Statistical Update. 2018. http://hdr.undp.org/en/2018-update.
- USDA-FAS, United State Department of Agriculture Foreign Agricultural Service. 2018. "Commodity Intelligence Report - 2017/18 Guyana Rice Production Higher than Expected." https://ipad.fas.usda.gov/highlights/2018/02/guyana/index.pdf.
- USGCRP, U.S. Global Change Research Program. 2018. "Impacts, Risks, and Adaptation in the United States: Fourth National Climate Assessment, Volume II [Reidmiller, D. R., C. W. Avery, D. R. Easterling, K. E. Kunkel, K. L. M. Lewis, T. K. Maycock, and B. C. Stewart (Eds.)]." Washington, DC, USA. https://doi.org/10.7930/NCA4.2018.
- Varanasi, Aruna, P. V. Vara Prasad, and Mithila Jugulam. 2016. "Impact of Climate Change Factors on Weeds and Herbicide Efficacy." *Advances in Agronomy* 135: 107–46. https://doi.org/10.1016/bs.agron.2015.09.002.
- Velasco, Marco A. 2014. "Progress and Challenges in Disaster Risk Management in Guyana."
- Vorley, Bill, and Man-kwun Chan. 2012. "Tipping the Balance: Policies to Shape Agricultural Investments and Markets in Favour of Small-Scale Farmers."
- Vose, R. S., R. Adler, A. Becker, and X. Yin. 2018. "Global Climate: Hydrological Cycle -Precipitation [in 'State of the Climate in 2017']." *Bulletin of the American Meteorological Society*. Vol. 99.

https://doi.org/10.1175/2018BAMSStateoftheClimate.1.

- Vulturius, Gregor, and Åsa Gerger Swartling. 2015. "Overcoming Social Barriers to Learning and Engagement with Climate Change Adaptation: Experiences with Swedish Forestry Stakeholders." *Scandinavian Journal of Forest Research* 30 (3): 217–25. https://doi.org/10.1080/02827581.2014.1002218.
- Waggoner, P. E. 1983. "Agriculture and a Climate Changed by More Carbon Dioxide (In Changing Climate: Report of the Carbon Dioxide Assessment Committee)." Washington, DC.
- Wahid, A., S. Gelani, M. Ashraf, and M. R. Foolad. 2007. "Heat Tolerance in Plants: An Overview." *Environmental and Experimental Botany* 61 (3): 199–223. https://doi.org/10.1016/j.envexpbot.2007.05.011.
- Walthall, C. L., J. Hatfield, P. Backlund, L. Lengnick, E. Marshall, M. Walsh, S. Adkins, et al. 2012. "Climate Change and Agriculture in the United States: Effects and

Adaptation." Washington, DC.

http://www.usda.gov/oce/climate_change/effects.htm.

- Wang, Jinxia, Robert Mendelsohn, Ariel Dinar, Jikun Huang, Scott Rozelle, and Lijuan Zhang. 2009. "The Impact of Climate Change on China's Agriculture." Agricultural Economics 40 (3): 323–37. https://doi.org/10.1111/j.1574-0862.2009.00379.x.
- Wassmann, R., S. V. K. Jagadish, S. Heuer, A. Ismail, E. Redona, R. Serraj, R. K. Singh, G. Howell, H. Pathak, and K. Sumfleth. 2009. "Chapter 2 Climate Change Affecting Rice Production. The Physiological and Agronomic Basis for Possible Adaptation Strategies." *Advances in Agronomy* 101: 59–122. https://doi.org/10.1016/S0065-2113(08)00802-X.
- Wassmann, Reiner, and Achim Dobermann. 2007. "Climate Change Adaptation through Rice Production in Regions with High Poverty Levels." *Journal of SAT Agricultural Research* 4 (1): 1–24.
- WB, World Bank. 2010a. "Guyana Agricultural Insurance Component Pre-Feasibility Study Report." Washington, DC.

Welch, Jarrod R., Jeffrey R. Vincent, Maximilian Auffhammer, Piedad F. Moya, and Achim Dobermann. 2010. "Rice Yields in Tropical Subtropical Asia Exhibit Large but Opposing Sensitivities to Minimum and Maximum Temperatures." *Proceedings* of the National Academy of Sciences 107 (33): 14562–14567. https://doi.org/10.1073/pnas.1001222107/-

/DCSupplemental.www.pnas.org/cgi/doi/10.1073/pnas.1001222107.

- Wood, Stephen A., and Robert O. Mendelsohn. 2014. "The Impact of Climate Change on Agricultural Net Revenue: A Case Study in the Fouta Djallon, West Africa." *Environment and Development Economics* 20: 20–36. https://doi.org/10.1017/S1355770X14000084.
- Wooldridge, Jeffery M. 2013. *Introductory Econometrics: A Modern Approach*. Canada: South-Western Cengage Learning.
- World Meteorological Organization (WMO). 2017. "Frequently Asked Questions (FAQ) - What Is the Climate Variability?" 2017.
 - http://www.wmo.int/pages/prog/wcp/ccl/faqs.php.
- Wright, Philip G. 1928. The Tariff on Animal and Vegetable Oils. New York: Macmillan.
- Yohe, Gary, and Richard S. J. Tol. 2002. "Indicators for Social and Economic Coping Capacity - Moving Toward a Working Definition of Adaptive Capacity." *Global Environmental Change* 12 (1): 25–40. https://doi.org/10.1016/S0959-3780(01)00026-7.
- Yoshida, S. 1981. *Fundamentals of Rice Crop Science*. 1st ed. Manila, Philppines: The International Rice Research Institute.
- Yusuf, Haseef. 2018. "Dams Vital for Region Six Rice Farmers in Terrible Shape." Stabroek News, July 12, 2018. https://www.stabroeknews.com/2018/opinion/letters/07/12/dams-vital-for-region-

six-rice-farmers-in-terrible-shape/.

Zamasiya, Byron, Kefasi Nyikahadzoi, and Billy Billiard Mukamuri. 2017. "Factors Influencing Smallholder Farmers' Behavioural Intention Towards Adaptation to Climate Change in Transitional Climatic Zones: A Case Study of Hwedza District in

^{——. 2010}b. "World Development Report: Development and Climate Change." Washington, DC.

Zimbabwe." *Journal of Environmental Management* 198: 233–39. https://doi.org/10.1016/j.jenvman.2017.04.073.

- Zhai, Fan, Tun Lin, and Enerel Byambadory. 2009. "A General Equilibrium Analysis of the Impact of Climate Change on Agriculture in the People's Republic of China." *Asian Development Review* 26 (1): 206–25.
- Zhao, Chuang, Bing Liu, Shilong Piao, Xuhui Wang, David B. Lobell, Yao Huang, Mengtian Huang, et al. 2017. "Temperature Increase Reduces Global Yields of Major Crops in Four Independent Estimates." *Proceedings of the National Academy* of Sciences 114 (35): 9326–9331. https://doi.org/10.1073/pnas.1701762114.
- Zhao, Chuang, Piao Shilong, Wang Xuhui, Huang Yao, Ciais Philippe, Elliott Joshua, Huang Mengtian, Janssens Ivan, Lian Li Tao, and Xu. 2017. "Plausible Rice Yield Losses Under Future Climate Warming." *Nature Plants* 3 (1): 23. https://doi.org/10.1038/nplants.2016.202.
- Zhu, Chunwu, Kazuhiko Kobayashi, Irakli Loladze, Jianguo Zhu, Qian Jiang, Xi Xu, Gang Liu, et al. 2018. "Carbon Dioxide (CO2) Levels This Century Will Alter the Protein, Micronutrients, and Vitamin Content of Rice Grains with Potential Health Consequences for the Poorest Rice-Dependent Countries." *Science Advances* 4 (5): 1–8.
- Ziervogel, Gina, Sukaina Bharwani, and Thomas E. Downing. 2006. "Adapting to Climate Variability: Pumpkins, People, and Policy." *Natural Resources Forum* 30: 294–305. https://doi.org/10.1111/j.1477-8947.2006.00121.x.

APPENDIX A – SURVEY INSTRUMENT: SMALL FARMERS

Consent Statement

Dear Participant,

I invite you to participate in this research study that focuses on how small rice farmers in Guyana respond to weather patterns and variation in climate.

I am a graduate student at Virginia Polytechnic Institute and State University (Virginia Tech) in Blacksburg, Virginia, and I am in the process of writing my Ph.D. dissertation.

The purpose of this research is to determine the impacts of climate change on rice production and how small rice farmers respond and adapt to variations in rainfall, temperature, and extreme weather events such as drought and excess rainfall.

Your participation in this research project is completely voluntary. You may decline altogether or skip any questions you do not wish to answer. Your identity and responses will remain confidential and anonymous. No one other than the researchers will know your individual answers to this questionnaire.

Please give your verbal consent if you agree to participate in this project. It should take approximately 1 hour to complete.

Thank you for your assistance in this important endeavor.

Sincerely,

Omchand Mahdu Co-Investigator

Respondent ID: _____

Name of Respondent	
Village	
Telephone No.	
GPS Coordinates of Farmer's Home	
Administrative Region No.	

Part A – Socio-economic Data

1. Are you the farmer, the person who makes farming decisions on this farm? (0=No, 1=Yes)

If no, who is the head of the farm household and how can I get into contact?

2. Age of Respondent (in years): _____

3. Gender of Respondent: _____ (0=Female, 1=Male)

4. What is the highest level of education you completed? ______ (1=Primary, 2=Community High, 3=High School, 4=University, 5=Other

5. Is rice cultivation your primary occupation? _____ (0=No, 1=Yes)

Now I will ask some questions just about last year, 2016.

6. Primary or secondary occupations/income sources of farmer (if any) in 2016: [check all that applies]

Agricultural	Check (✔)		Non-Agricultural	Check (✔)	
	Primary	Secondary		Primary	Secondary
Rice farming			Carpentry		
Cash			Small business (shop		
crops/vegetables			keeping)		
Fruits			Wholesale/retail (buy and sell)		
Livestock			Service (painting, weeding, taxi, etc.)		
Fishery			Pension		
Ground provision (plantains, eddo, cassava, yams etc.)			Remittances		
Other agricultural (if any)			Other non-agricultural (if any)		

7. In 2016, did other members of the household participate in rice farming activities? (0=No, 1=Yes)

8. In 2016, did you employ any part-time labor to work on the farm? (0=No, 1=Yes

If yes, what activities were part-time laborers engaged?

Part B. Farm Structure and Characteristics

9. In 2016, how many acres of rice did you plant? _____

10. In 2016, who owned the land(s) you planted? What were your rice yields in bags per acre?

	Size of	Yields (bag		
Type of land ownership	plot			Total Yields
	(acres)	Spring Crop	Autumn Crop	
Own land – Self/Family				
Other Owner – Private				
(Individual)				
Other Owner - Other				

11. On the land(s) you own in 2016, do you have legal title? _____ (0=No, 1=Yes)

12. How long have you been a rice farmer in this region (in years)? _____

13. Did you own any farming equipment in 2016? (0=No, 1=Yes)

If yes, what equipment did you own?

14. Did you shared or co-owned any farming equipment with other farmers in 2016? (0=No, 1=Yes)

If yes, what equipment did you shared/co-owned?

15. Did you rent any equipment in 2016? (0=No, 1=Yes)

If yes, what equipment did you rent and from whom?

16. Did you own any livestock in 2016 (cow, sheep, chicken, duck etc.)? ____(0=No, 1=Yes)

If yes, how many of each do you own? _____

17. What is the soil type/quality of land(s) you planted: (1=Clay, 2=Clay-Loam, 3=Loam, 4=Sandy, 5=Other, 6=Do Not Know)

18. Have you ever tested your soil? (0=No, 1=Yes)

If yes, what was the result? _____

Until now, I have been asking questions related to 2016. I will now ask questions related to the last 5 years.

C. Farmers Perceptions, Impacts, and Adaptation

19. Have you observed any major changes in rainfall patterns in your area in recent
years (for example, the last 5 years)?(0=No, 1=Yes). If the answer is yes,
answer question 29.
If yes, how have rainfall patterns changed?

How have the observed changes in rainfall affected the quality and quantity of rice yields?

How have you responded to the observed changes in rainfall?

20. Have you observed any major changes in temperature in your area in recent years (for example, the last 5 years)? ______ (0=No, 1=Yes). If the answer is yes, answer question 29. If yes, how has temperature changed?

How have the observed changes in temperature affected the quality and quantity of rice yields? _____

How	have you	responded	to the	observed	changes in	n temperature?	
	2	1			0	1	

21. Have you observed any changes in insects in your area in recent years (for example, last 5 years)? ______ (0=No, 1=Yes). If the answer is yes, answer question 29. If yes, how have insects' population changed?

How have the observed changes in insects affected the quality and quantity of rice yields?

How have you responded to the observed changes in insects?

22. Have you observed any changes in diseases in your area in recent years (for example, last 5 years)? _____ (0=No, 1=Yes). If the answer is yes, answer question 29. If yes, how have diseases changed? _____

How have the observed changes in diseases affected the quality and quantity of rice yields? _____

How have you responded to the observed changes in diseases?

23. Have you observed any changes in weeds in your area in recent years (for example, last 5 years)? ______ (0=No, 1=Yes). If the answer is yes, answer question 29. If yes, how have weeds changed?

How have the observed changes in weeds affected the quality and quantity of rice yields?

How have you responded to the observed changes in weeds?

24. Have you observed any extreme weather event(s) (severe drought, excess rainfall, wind storms, flooding etc.) in your area in recent years (for example, the last 5 years)? _____(0=No, 1=Yes). If the answer is yes, answer question 29. If yes, describe the extreme weather event(s). _____

How were the quality and quantity of rice yields affected?

How did you respond to the extreme weather event(s) noted above?

25. If you have seen changes in any of the above areas in recent years (for example, the last 5 years), which crop has seen the greater impact? _____ (0=Spring crop, 1=Autumn crop, 2=Both crops equally)

26. How has your overall rice yield changed in recent years (for example, the last 5 years)? ______ (1=Significantly Lower, 2=Lower, 3=Higher, 4=Significantly Higher, 5=No Change)

27. Have you experienced any crop failure in recent years (for example, the last 5 years)? (0=No, 1=Yes).

If yes, which season did the crop failed? (0=Spring, 1=Autumn)

28. If you have seen declines and/or experienced crop failure, how have you coped with reduced yields, grain quality, etc. in recent years (for example, the last 5 years)? _____

29. In recent years (for example, the last 5 years), have you made any changes to your farming practice in relation to the observed changes in rainfall and temperature patterns, insects, diseases, and/or extreme weather events?

_(0=No, 1=Yes)

If yes, which if any of the following practices have you adopted to reduce losses?

Adaptive Measures	Check all that apply
Changing sowing/planting dates	
Use different rice varieties	
Buy new seeds each season	
Pump water	
Treat seeds for insects	
Reduce plant densities	
Agroforestry (example rice and coconut)	
Increase fertilizer application (quantity and frequency)	
Increase herbicide and pesticide application (quantity and frequency)	
Crop rotation	
Cultivation of other crops	
Rent out land to large farmers	
Exited rice cultivation and seek off farm employment	
Leave land idle	
Other (please specify)	
No adaptation	

Adapt to adaptation	Check all that applies
Borrow from family members	
Sell livestock	
Use savings	
Input credit (fertilizer, herbicide, pesticide, seeds)	
Sell productive assets	
Borrow from other farmers	
Obtain institutional microcredit	
Borrow from commercial bank	
Off farm employment	
Government support/subsidy	
Remittances	
Pension	
Other (please specify)	

30. How did you pay for these changes in farming practice?

31. How difficult was it to pay for these activities? Explain

32. How important was each of the following barriers in preventing you from engaging in the activities listed in question 29?

Barriers	1=Very	2=Important	3=Slightly	4=Not
	Important	_	Important	Important
Availability of information about extreme				
weather events, changes in rainfall and				
temperature				
Access to adequate irrigation facilities				
Access to knowledge regarding adaptation				
practices				
Access to money/savings/credit				
Ownership of land				
Availability of labor				
Price received for rice paddy				
Subsidized inputs				
Tax exemption on inputs and equipment				
Availability of markets for rice				
Government assistance/subsidies				
Other (please specify)				

Who did you expect to help? Explain

33. Have you ever heard of the term climate change/global warming? _____(0=No, 1=Yes)

If yes, what does it mean to you?

34. Are you familiar with the Guyana Rice Development Board (GRDB) 6-point practices? (0=No, 1=Yes)

35. If yes, what have you heard about these practices? _____

36. Which of these practices have you employed in recent years (for example, the last 5 years)? [Check all that applies]

Practice	Check (✔)
1=Time of sowing (adjusting planting dates)	
2=Seed rate (optimal plant density)	
3=Treatment of seeds (with insecticides)	
4=Weed control (pre and post emergence)	
5=Balance nutrition (fertilizer application)	
6=Water management (timing and adequacy of water application).	

Part D. Institutional Accessibility

37. In recent years (for example, the last 5 years), did you participate in any extension training/activities (farmer field schools, field visits, field demonstrations), or received any advice and/or guidance about rice production? _____ (0=No, 1=Yes). If yes, answer questions 38 and 39.

38. What type of extension training, advice and/or guidance did you receive and how did you use the new knowledge? _____

39. From what source(s) did you receive extension training, advice and/or guidance? (1=Government (GRDB), 2=NGO, 3=Other farmers, 4=Farmers' association; 5=Other (please specify)____)

40. In 2016, did you receive advance weather (temperature, rainfall, humidity, wind etc.) information from any source? (0=No, 1=Yes)

If yes, from what source? ____(1=Agricultural Extension Office, 2=Television, 3=Radio, 4=Newspapers, 5=Other (please specify____)

41. In 2016, did you receive any financial assistance or agricultural/input credit? (0=No, 1=Yes)

If yes, from what source(s) did you receive financial assistance or access any agricultural credit? ____(1=Government, 2=NGOs, 3=Commercial banks, 4=Microfinance institution, 5-Family, 6-Rice Miller, 6=Other)

42. In 2016, did you have access to any form of agricultural insurance? ____(0=No, 1=Yes)

If yes, did you buy insurance? (0=No, 1=Yes). If no, why not?

43. In 2016, were you able to access adequate irrigation facilities (trench, canal, conservancy)? _____ (0=No, 1=Yes)

44. In 2016, did you use a pump to flood your field? _____ (0=No, 1=Yes)

45. In 2016, what rice seed variety(ies) did you use? _____ (1=GRDB 9, 2=GRDB 10, 3=GRDB 11, 4= GRDB 12, 5=GRDB 14, 6=Other (please specify)

46. What was the source of your seed? ____(1=Retained, 2=Targeted farmers, 3=Other farmers, 4=GRDB, 5=RPA, 6=Rice Miller, 7=Other (please specify____)

47. How would you rate the supply of inputs and equipment rental used in rice cultivation?

Inputs	1=Very Good	2=Good	3=Acceptable	4= Poor	5=Very Poor
Seeds					
Fertilizer					
Pesticides and Herbicides					
Labor					
Equipment:					
combines/tractors					

49. In 2016, were you a member of any other agricultural related institution/group/organization/farmers' cooperative, water management group etc.? (0=No, 1=Yes)

If yes, what is the name of the institution/group/organization/farmers' cooperative that you were a member of? _____

APEENDIX B – SURVEY INSTRUMENT: KEY INFORMANTS

Consent Statement

Dear Participant,

I invite you to participate in this research study that focuses on how small rice farmers in Guyana respond to weather patterns and variation in climate.

I am a graduate student at Virginia Polytechnic Institute and State University (Virginia Tech) in Blacksburg, Virginia, and I am in the process of writing my Ph.D. dissertation.

The purpose of this research is to determine the impacts of climate change on rice production and how small rice farmers respond and adapt to variations in rainfall, temperature, and extreme weather events such as drought and excess rainfall.

Your participation in this research project is completely voluntary. You may decline altogether or skip any questions you do not wish to answer. Your identity and responses will remain confidential and anonymous. No one other than the researchers will know your individual answers to this questionnaire.

Please give your verbal consent if you agree to participate in this project. It should take approximately 45 minutes to complete.

Thank you for your assistance in this important endeavor.

Sincerely,

Omchand Mahdu Co-Investigator
Respondent ID: _____

Name of Respondent	
Position	
District	
Telephone No.	
GPS Coordinates	
Administrative Region No.	

1. Age of Respondent (in years): _____

2. Gender of Respondent: _____ (0=Female, 1=Male)

3. What is the highest level of education you completed? ______(1=Primary, 2=Community High, 3=High School, 4=University, 5=Other

4. How long have you been a rice extension officer (in years)? _____

5. Have you observed any major changes in rainfall patterns in your district in recent years (for example, the last 5 years)? _____(0=No, 1=Yes).

If yes, how have rainfall patterns changed?

How have the observed changes in rainfall affected the quality and quantity of rice yields in your district?

How have farmers in your district responded to the observed changes in rainfall?

6. Have you observed any major changes in temperature in your district in recent years (for example, the last 5 years)? _____ (0=No, 1=Yes).

If yes, how has temperature changed?

How have the observed changes in temperature affected the quality and quantity of rice yields in your district?

How have farmers in your district responded to the observed changes in temperature?

7. Have you observed any changes in insects in your district in recent years (for example, last 5 years)? _____ (0=No, 1=Yes).

If yes, how have insects' population changed?

How have the observed changes in insects affected the quality and quantity of rice yields in your district?

How have farmers in your district responded to the observed changes in insects?

8. Have you observed any changes in diseases in your district in recent years (for example, last 5 years)? _____ (0=No, 1=Yes). If yes, how have diseases changed? _____

How have the observed changes in diseases affected the quality and quantity of rice yields in your district? _____

How have farmers in your district responded to the observed changes in diseases?

9. Have you observed any changes in weeds in your district in recent years (for example, last 5 years)? _____ (0=No, 1=Yes). If yes, how have weeds changed? _____

How have the observed changes in weeds affected the quality and quantity of rice yields in your district?

How have farmers in your district responded to the observed changes in weeds?

10. Have you observed any extreme weather event(s) (severe drought, excess rainfall, wind storms, flooding etc.) in your district in recent years (for example, the last 5 years)? _____(0=No, 1=Yes). If yes, describe the extreme weather event(s). _____

How were the quality and quantity of rice yields affected in your district?

How did farmers in your district respond to the extreme weather event(s) noted above?

11. Given the changes you have observed, have you been supporting efforts to adapt to these changes? ____(0=No, 1=Yes).

If yes, what kind of support have you provided?

12. Are there any specific resources that are needed to more effectively support adaptation now and/or in the future?_____

13. What barrier(s)/challenge(s) do you currently observe in the rice sector?_____

APPENDIX C – CODEBOOK

Variable ID	Variables Name	Variable Description
		Numerical number of the administrative region and
RESPID	Respondent ID	the numerical order the farmer was interviewed
NARESP	Name of Respondent	The name of the farmer
VILLAGE	Village	The name of the village that the farmer resides
TELE	Telephone	The telephone number of the farmer
		The degrees latitude where the interview was
LAT	Latitude	conducted based on WGS84 coordinate system
		The degrees longitude where the interview was
LONG	Longitude	conducted based on WGS84 coordinate system
		The administrative region of Guyana where the
REGION	Region	interview was conducted
		The person interviewed is the person that makes the
FARMER	Farmer	farming decisions
AGE	Age	Age of the farmer interviewed
GENDER	Gender	The gender of the farmer interviewed
EDUC	Education	The education level of the farmer interviewed
		Rice farming is the primary occupation of the
PRIMOCCU	Primary Occupation	farmer
		Rice farming was the primary source of income for
RICEFARM	Rice Farming	the farmer
		The farmer had a secondary source of income from
SECAGOCCU	Secondary Ag Occupation(s)	other agriculture related activities
		Cash crops were a secondary agricultural source of
CASHCROP	Cash Crops	income for the farmer
		Fruits were a secondary agricultural source of
FRUITS	Fruits	income for the farmer
		Livestock was a secondary agricultural source of
LIVESTOCK	Livestock	income for the farmer
		Fishery was a secondary agricultural source of
FISHERY	Fishery	income for the farmer
		Ground provisions were a secondary agricultural
GRNDPROV	Ground Provisions	source of income for the farmer
		Other agricultural sources were a secondary
OTHERAGRI	Other Agri	agricultural source of income for the farmer
SECNIA COCC		The farmer had a secondary source of income from
SECNAGOCC	Secondary Non-Ag Occupation(s)	non-agriculture related activities
CADDENT	C	Carpentry was a secondary non-agricultural source
CARPENI	Carpentry	of income for the farmer
CMI DUC	Surall Duringer	Small business was a secondary non-agricultural
SMLBUS	Small Business	Source of income for the farmer
WUODETI	Wholesele/ Poteil	wholesale/retail was a secondary non-agricultural
WHOKETL	wholesale/ Ketall	Source of income for the familier
SEDVICE	Samilaa	services were a secondary non-agricultural source
SERVICE	Service	
DENISION	Dension	A government pension was a secondary non-
r EINSIOIN		Partitionage ware a secondary non-activity 1
DEMIT	Pemittances	source of income for the former
	Kennitances	Other non agricultural sources were a secondary
OTDNONAG	Other Non Agric	non agricultural sources of income for the former
UTINUNAU	Outer Non-Agric	non-agricultural source of income for the farmer

		Anyone in the household other than the farmer help
HOUSEPART	Household Participate	with rice farming
	•	The farmer hired part-time labor to help with
PARTLAB	Part-time Labor	sowing, applying fertilizer, spraying etc.
		The activities the part-time laborers were hired to
PRTLABACT	Part-time Labor Activities	perform
ACREPLT	Acreage Planted	The number of acres planted
OWNLAND	Own Land	The farmer/family owned the rice farm
		A private individual(s) owned the rice farm that the
OWNPVT	Own Private	farmer is planting
		Any other person or entity owned the rice farm that
OTROWN	Other Owner	the farmer is planting
		Number of bags of paddy harvested in Spring Crop
SPRINGVI D	Spring	2016
51 Idi (G I ED	Spring	Number of bags of paddy harvested in Autumn
AUTYLD	Autumn	Crop 2016
TICTTLD		Total number of hags of naddy harvested in Cron
TITVID	Total Vields	2016
SPRGSACRE	Spring (Bags per Acre)	Spring paddy yields in bags per acre in 2016
	Autumn (Bags per Acre)	Autumn paddy yields in bags per acre in 2016
AVRBGSACRE	Average (Bags per Acre)	Total paddy vields in bags per acre in 2016
HECTPLANT	Hectare Planted	Number of bectares planted in 2016
SPPADVI DMT	Spring Paddy (MT)	Spring paddy yields in Metric Tons in 2016
	Autumn Paddy (MT)	Autumn paddy yields in Metric Tons in 2016
SDD VI DMT	Spring Dice (MT)	Spring Dice vields in Metric Tons in 2016
	Autumn Biog (MT)	Autumn Rice yields in Metric Tons in 2016
AUKILDMI	Autumin Kice (MT)	Autumin Rice yields in Metric Tons nor bestere in
SDDVI DMTUA	Spring Viold (MT/Ha)	2016
SIKILDMIIIA		Autumn Diag violds in Matria Tons per bestere in
	Autumn Vield (MT/Ha)	2016
AUKTLDWITTA	Autumin Tield (MT/ Ha)	Total Rice vields in Matrie Tons per bestere in
ΤΡΥΙ ΟΜΤΠΑ	Total Dian Viold (MT/Ha)	2016
TKTLDMIIIA		Average Dieg violds in Matrie Tons per bestere in
Α ΥΡΥΙ ΌΜΤΗ Α	Average Vield (MT/Ha)	2016
	L agal Titla	The former has a legal title for the rise form
LEGITILE		The number of years the former has been a rice
FARMEYP	Farming Experience	farmer
	Owned Equipment	The former owned any forming equipment
FOLIPOWN	Equipment Owned	The actual equipment owned
EQUIIOWIN		The former shared any forming equipment with
SHARFOUIP	Shared Equipment	others
FOLUPSHAR	Equipment Shared	The actual equipment shared
OWSHED	Owned and Shared Equipment	The former owned and shared forming equipment
DENTEQUID	Pent Equipment	The farmer rant any farming equipment
EQUIDDENT	Fauinment Pented	The natual agginment rented
	Oumod Livesteelt	The former owned any livesteels
NOLIVESTV	Ne. of Livestock	The number and two of the start
NULIVESIK	INO. OI LIVESIOCK	The number and type of ilvestock owned
SUILIYPE		The type of soil on the land(s) planted by the farmer
SOILTEST	Soil Tested	The farmer tested the soil
TESTRSLT	Soil Test Results	The result of the soil test, if applicable
DADICIDIC		I he farmer observed any major changes in rainfall
KAINCHNG	Kainfall Changed	in the area
DESRAIN	How have rainfall patterns changed?	The tarmer description of how rainfall changed

	How did the observed changes affect	The farmer description of how the change in
RAINYIELDS	the quality and quantity of rice yields?	rainfall affected the quantity and quality of yields
	How have you responded to the	The farmer description of how he/she responded to
RESPRAIN	observed changes in rainfall?	the change in rainfall
	<u>v</u>	The farmer observed any major changes in
TEMPCHNG	Temperature Changed	temperature in the area
DESTEMP	How has temperature changed?	The farmer description of how temperature changed
	How have the observed changes in	The farmer description of how the change in
	temperature affected the quality and	temperature affected the quantity and quality of
TEMPYIELDS	quantity of rice yields?	yields
	How have you responded to the	The farmer description of how he/she responded to
RESPTEMP	observed changes in temperature?	the change in temperature
		The farmer observed any major changes in insects
INSECCHNG	Insects and Pests Changed	in the area
	How have insects and pests'	
DESINSECTS	population changed?	The farmer description of how insect(s) changed
	How have the observed changes in	
	insects and pests affected the quality	The farmer description of how the change in
INSECYIELDS	and quantity of rice yields?	insect(s) affected the quantity and quality of yields
	How have you responded to the	The farmer description of how he/she responded to
RESPOINSEC	observed changes in insects and pests?	the change in insect(s)
		The farmer observed any major changes in diseases
DISEACHNG	Diseases Changed	in the area
DESDISEASE	How have diseases changed?	The farmer description of how disease(s) changed
	How have the observed changes in	
	diseases affected the quality and	The farmer description of how the change in
DISYIELDS	quantity of rice yields?	disease(s) affected the quantity and quality of yields
	How have you responded to the	The farmer description of how he/she responded to
RESPDISEASE	observed changes in diseases?	the change in disease(s)
		The farmer observed any major changes in weeds in
WEEDSCHNG	Weeds Changed	the area
DESWEEDS	How have weeds changed?	The farmer description of how weed(s) changed
	How have the observed changes in	
	weeds affected the quality and	The farmer description of how the change in
WEEDSYIELDS	quantity of rice yields?	weed(s) affected the quantity and quality of yields
DECOMPEDO	How have you responded to the	The farmer description of how he/she responded to
RESPWEEDS	observed changes in weeds?	the change in weed(s)
		The farmer observed any major changes in extreme
EXIREWEAT	Extreme Weather Events	weather events in the area
DECEVIUEAT	Describe the extreme weather	The farmer description of the extreme weather
DESEAWEAT	U and the soulite and sound the of	The former description of here the entropy much and
EVTDEVIEI DO	How were the quality and quantity of	I he farmer description of now the extreme weather
EATRETIELDS	Here yields affected?	The former description of how he/she responded to
DESTOEVTDE	Now did you respond to the extreme	the extreme weather event(a)
RESTOEATRE	weather event(s) noted above:	The season that has seen the greater impact from
CROPAFFECT	Crop Affected	the changes observed
VIDCHNG	Vields Changed	How did the overall rice yield change
CROPEAU	Crop Failure	The farmer experienced a crop failure
FAILSEAS	Cron Failure Season	The season the cron failure occurred
TAILOLAO		How did the farmer cone with reduced yields arain
COPSTRG	Coping Strategies	mality etc?
		The farmer made changes to his/her farming
CFARMPRAC	Changed Farming Practices	practices
PLANTDATE	Changing sowing/planting dates	The farmer changed the sowing/planting dates

DIFFRICE	Use different rice varieties	The farmer used different rice varieties		
NEWSEED	Buy new seeds each season	The farmer bought new seeds each season		
PUMPWATER	Pump water	The farmer pumped water		
TREATSEEDS	Treat seeds for insects	The farmer treated seeds for insects		
REPLTDEN	Reduce plant densities	The farmer reduced plant densities		
	Agroforestry (example rice and			
AGROFOR	coconut)	The farmer engaged in agroforestry		
	Increase fertilizer application	The farmer increased fertilizer application (quantity		
INCREFERT	(quantity and frequency)	and frequency)		
	Increase herbicide and pesticide	The farmer increased herbicide and pesticide		
INCREHERB	application (quantity and frequency)	application (quantity and frequency)		
CROPROT	Crop rotation	The farmer engaged in crop rotation		
CULTOTHR	Cultivation of other crops	The farmer cultivated other crops		
RENTOUT	Rent out land to large farmers	The farmer rented out the land to large farmers		
	Exited rice cultivation and seek off	The farmer exited rice cultivation and seek off-farm		
EXITRICE	farm employment	employment		
LANDIDLE	Leave land idle	The farmer left the land idle		
SPRAYWAT	Sprayed water in field	The farmer sprayed the water in the field		
		The farmer engaged in any other adaptation		
OTHER	Other (please specify)	practices		
NOADPT	No adaptation	The farmer did not engage in adaptation		
		The farmer paid for the changes in farming practice		
BORROWFAM	Borrow from family members	by borrowing from family members		
		The farmer paid for the changes in farming practice		
SELLLIVE	Sell livestock	by selling livestock		
		The farmer paid for the changes in farming practice		
USESAV	Use savings	by using savings		
		The farmer paid for the changes in farming practice		
	Input credit (fertilizer, herbicide,	by obtaining input credit (fertilizer, herbicide.		
INPUTCRED	pesticide, seeds)	pesticide, seeds)		
		The farmer paid for the changes in farming practice		
SELLPROD	Sell productive assets	by selling productive assets		
		The farmer paid for the changes in farming practice		
BORROWFARM Borrow from other farmers		by borrowing from other farmers		
		The farmer paid for the changes in farming practice		
MICROCRED	Obtain institutional microcredit	by obtaining institutional microcredit		
		The farmer paid for the changes in farming practice		
COMBANK	Borrow from commercial bank	by borrowing from a commercial bank		
		The farmer paid for the changes in farming practice		
OFFFARM	Off farm employment	by seeking off-farm employment		
	~ /	The farmer paid for the changes in farming practice		
GOVTSUPP	Government support/ subsidy	by using government support and subsidy received		
		The farmer paid for the changes in farming practice		
REMITTANCE	Remittances	by using remittances received		
	D	The farmer paid for the changes in farming practice		
PENS	Pension	by using pension received		
		The farmer paid for the changes in farming practice		
OTHERADPT	Other (please specify)	by other means		
		The farmer found paying for the activities was		
DIFCULTPAY	Difficult to Pav	difficult		
	Availability of information about	Availability of information about extreme weather		
	extreme weather events changes in	events, changes in rainfall and temperature was a		
AVAILINFO	rainfall and temperature	barrier to adaptation faced by the farmer		

ADQIRRIG	Access to adequate irrigation facilities	Access to adequate irrigation facilities was a barrier to adaptation faced by the farmer			
KNOWADPT	Access to knowledge regarding adaptation practices	Access to knowledge regarding adaptation practices was a barrier to adaptation faced by the farmer			
SAVCREDIT	Access to money/savings/credit	Access to money/savings/credit was a barrier to adaptation faced by the farmer			
OWNERLAND	Ownership of land	Ownership of land was a barrier to adaptation faced by the farmer			
AVAILLAB	Availability of labor	Availability of labor was a barrier to adaptation faced by the farmer			
PDYPRICE	Price received for rice paddy	Price received for rice paddy was a barrier to adaptation faced by the farmer			
SUBINPUT	Subsidized inputs	The lack of subsidized inputs was a barrier to adaptation faced by the farmer			
TAXEXEMP	Tax exemption on inputs and equipment	The lack of tax exemption on inputs and equipment was a barrier to adaptation faced by the farmer			
AVAILMKT	Availability of markets for rice	Availability of markets for rice was a barrier to adaptation faced by the farmer			
GOVTASSIST	Government assistance/subsidies	The lack of government assistance/subsidies was a barrier to adaptation faced by the farmer			
OTHERBAR	Other (please specify)	Other barrier(s) to adaptation faced by the farmer			
EXPECT	Expectation	The farmer expected any help and from where			
CLIMCHG	Climate Change	The farmer had heard of climate change			
CCMEAN	What does CC mean to you?	Farmer describe what they understood about climate change			
SIXPNTKNOW	GRDB 6-Point Practice	The farmer heard about the 6-point practice			
		Farmer describe what they have heard about the 6-			
HEARDSIX	What have you heard about 6-point?	point practice			
	Time of sowing (adjusting planting				
TIMESOW	dates)	The farmer adjusted planting dates			
SEEDRATE	Seed rate (optimal plant density)	The farmer optimized plant density			
TREATSEED	Treatment of seeds (with insecticides)	The farmer treated seeds before sowing			
	Weed control (pre and post				
WEEDCRTL	emergence)	The farmer control weeds pre and post emergence			
	Balance nutrition (fertilizer				
BALNUTRN	application)	The farmer used a balanced fertilizer			
WATERMNGT	Water management (timing and adequacy of water application)	The farmer ensure timing and adequacy of water application			
		The farmer participated in any training concerning			
EXIIKAIN	Extension Training				
	Type of Training	The second of the training			
ADVWEATU	A dyor og Woother	The source of the training the farmer participated			
ADVWEATH	Advance weather	The farmer received advanced weather information			
SOURCEWEATH	Source of Advance weather	The source of the advance weather information			
FINASSIST	Financial Assistance	assistance in 2016			
SOURCEEN	Source of Financial Assistance	Source of financial assistance			
AGINSUR	A gricultural Insurance	The farmer has access to agricultural insurance			
ADEOIRBIG	A deguate Irrigation	The farmer has access to adequate irrigation			
PUWATER	Pump Water	The farmer numbed water in 2016			
SFEDVAR	Seed Variety	The seed variety that the farmer used in 2016			
SOURCESEED	Source of Seed	The source of the seed variety used			
SEEDS	Seeds	The farmer rating of the supply of seeds			
FERTILIZER	Fertilizer	The farmer rating of the supply of seeds			

		The farmer rating of the supply of pesticides and		
PESTHERB	Pesticides and Herbicides	herbicides		
LABOR	Labor	The farmer rating of the supply of labor		
COMTRACT	Equipment: Combines and Tractors	The farmer rating of the supply of equipment		
		The farmer was a member of the Rice Producer		
RPAMEM	RPA Membership	Association		
		The farmer was a member of any other agricultural		
OTHRAGMEM	Other Agricultural Membership	organization		
COMMENTS	Comments	General comments made by the farmer		
PRICELOW	Paddy Price Low	The farmer noted that the price of paddy is low		
		The farmer noted that they received late payment		
LATEPAY	Late Payment	from millers		
		The farmer noted that there is poor drainage in the		
POORDRAIN	Poor Drainage	area		
		The farmer noted that the cost of production has		
COSTPROD	Cost of Production Increased	increased		
SIXPOINT	Fully Practiced 6-point	The farmer fully employed the 6-point practice		

APPENDIX D – MEAN, STANDARD DEVIATION, AND COEFFICIENT OF

	Ave	erage Rain	fall	Ave	age Minir	num	Average Maximum		
Year	0			Temperature			Temperature		
	Mean	SD	CV (%)	Mean	SD	CV (%)	Mean	SD	CV (%)
1905	2417.04	215.27	8.91	21.52	0.16	0.76	30.26	0.15	0.50
1906	2463.90	191.17	7.76	21.44	0.21	0.97	30.16	0.19	0.65
1907	2500.82	233.22	9.33	21.36	0.21	0.97	30.10	0.19	0.62
1908	2471.44	209.86	8.49	21.26	0.09	0.42	30.02	0.08	0.28
1909	2550.38	225.03	8.82	21.26	0.09	0.42	30.02	0.08	0.28
1910	2681.98	163.58	6.10	21.24	0.09	0.42	30.00	0.07	0.24
1911	2695.36	143.87	5.34	21.26	0.09	0.42	30.02	0.04	0.15
1912	2550.38	315.38	12.37	21.32	0.13	0.61	30.08	0.13	0.43
1913	2475.42	356.31	14.39	21.36	0.11	0.53	30.10	0.12	0.41
1914	2376.88	331.53	13.95	21.38	0.13	0.61	30.14	0.15	0.50
1915	2273.08	187.46	8.25	21.50	0.19	0.87	30.26	0.23	0.76
1916	2262.86	169.77	7.50	21.58	0.16	0.76	30.34	0.18	0.60
1917	2321.24	117.43	5.06	21.58	0.16	0.76	30.34	0.18	0.60
1918	2372.24	92.91	3.92	21.62	0.13	0.60	30.40	0.12	0.40
1919	2382.14	77.93	3.27	21.66	0.11	0.53	30.42	0.11	0.36
1920	2316.84	179.03	7.73	21.62	0.08	0.39	30.38	0.04	0.15
1921	2307.84	169.01	7.32	21.58	0.08	0.39	30.34	0.09	0.29
1922	2339.04	188.55	8.06	21.54	0.15	0.70	30.28	0.18	0.59
1923	2278.48	198.85	8.73	21.46	0.21	0.97	30.20	0.20	0.66
1924	2260.44	200.67	8.88	21.38	0.16	0.77	30.14	0.17	0.56
1925	2230.54	249.36	11.18	21.34	0.11	0.53	30.08	0.08	0.28
1926	2152.80	225.29	10.46	21.34	0.11	0.53	30.10	0.12	0.41
1927	2262.74	451.82	19.97	21.28	0.19	0.90	30.06	0.18	0.60
1928	2320.90	449.33	19.36	21.28	0.19	0.90	30.04	0.19	0.65
1929	2325.46	448.02	19.27	21.24	0.21	0.98	30.00	0.20	0.67
1930	2392.36	384.96	16.09	21.26	0.23	1.08	30.02	0.22	0.72
1931	2453.06	339.99	13.86	21.28	0.26	1.22	30.02	0.22	0.72
1932	2323.62	95.93	4.13	21.38	0.22	1.01	30.12	0.20	0.68
1933	2410.92	202.18	10.87	21.30	0.24	1.13	30.12	0.20	0.08
1934	2339.44	249 74	13.20	21.30	0.24	1.15	20.12	0.20	0.08
1935	2303.94	340.74	14.75	21.32	0.23	0.84	30.10	0.20	0.00
1930	2392.24	375.67	15.35	21.28	0.18	0.84	30.08	0.18	0.59
1938	2448.10	453.03	18.13	21.20	0.16	0.04	30.00	0.13	0.30
1939	2576.86	336.85	13.07	21.30	0.10	0.74	30.12	0.15	0.45
1940	2463 50	496.36	20.15	21.50	0.11	1 21	30.24	0.00	0.20
1940	2405.30	500.07	20.13	21.40	0.20	1.21	30.34	0.27	1 11
1942	2384 22	482.76	20.02	21.00	0.51	1.87	30.46	0.38	1.11
1943	2381 50	477.67	20.06	21.72	0.28	1.02	30.56	0.28	0.91
1944	2447.62	480.57	19.63	21.92	0.15	0.68	30.66	0.11	0.37
1945	2679.54	325.15	12.13	21.90	0.16	0.72	30.64	0.11	0.37
1946	2704.36	293.07	10.84	21.82	0.19	0.88	30.58	0.15	0.49
1947	2648.24	367.22	13.87	21.78	0.13	0.60	30.56	0.11	0.37
1948	2572.28	280.55	10.91	21.76	0.15	0.70	30.52	0.16	0.54
1949	2695.88	403.29	14.96	21.66	0.19	0.90	30.44	0.21	0.68
1950	2622.12	383.60	14.63	21.60	0.19	0.87	30.36	0.21	0.68

VARIATION (1905 – 2015)^a

1951	2738 14	399.97	14.61	21.58	0.19	0.89	30.34	0.21	0.68
1952	2756.14	337.76	12.16	21.50	0.17	0.07	30.30	0.12	0.00
1053	2821.68	3/2/18	12.10	21.54	0.11	0.55	30.30	0.12	0.40
1954	28/0 10	384 70	13.50	21.52	0.11	0.51	30.26	0.12	0.40
1055	2047.10	3/8 80	11.06	21.50	0.14	0.00	30.20	0.10	0.00
1056	2917.14	362 53	12.32	21.77	0.19	1.13	30.22	0.22	0.72
1057	2943.14	574.33	20.22	21.30	0.24	0.07	30.14	0.23	0.63
1957	2639.90	697.56	26.00	21.34	0.21	1.52	20.16	0.19	0.02
1930	2044.02	573.15	20.00	21.42	0.33	1.53	30.10	0.29	0.90
1939	2417.40	506.10	23.71	21.42	0.33	1.33	30.18	0.28	0.92
1900	2021.30	242.97	12.06	21.40	0.30	1.42	20.20	0.20	0.00
1901	2021.44	243.07	12.00	21.54	0.23	1.07	20.26	0.20	0.08
1902	2034.00	227.97	11.10	21.32	0.25	1.00	20.20	0.21	0.09
1903	2102.00	290.57	13.44	21.48	0.15	0.09	30.24	0.17	0.55
1904	2115.24	297.87	14.08	21.62	0.24	1.10	30.38	0.28	0.91
1905	2017.38	292.71	14.51	21.66	0.21	0.96	30.44	0.23	0.76
1966	2143.48	278.66	13.00	21.74	0.21	0.95	30.52	0.24	0.78
1967	2290.00	3/8.90	16.55	21.74	0.21	0.95	30.52	0.24	0.78
1968	2346.64	434.78	18.53	21.70	0.23	1.08	30.48	0.26	0.85
1969	23/4.20	409.76	17.26	21.74	0.30	1.40	30.50	0.29	0.96
1970	2562.12	301.23	11.76	21.76	0.30	1.36	30.52	0.29	0.94
1971	2633.20	286.17	10.87	21.56	0.47	2.17	30.32	0.44	1.45
1972	2587.38	278.43	10.76	21.52	0.48	2.24	30.30	0.45	1.48
1973	2514.08	262.50	10.44	21.48	0.49	2.29	30.26	0.46	1.51
1974	2585.84	156.34	6.05	21.24	0.31	1.47	30.04	0.31	1.04
1975	2566.48	131.24	5.11	21.12	0.18	0.85	29.92	0.18	0.60
1976	2610.88	211.52	8.10	21.16	0.13	0.63	29.94	0.15	0.51
1977	2521.72	318.82	12.64	21.20	0.20	0.94	29.96	0.18	0.61
1978	2473.86	354.87	14.34	21.24	0.24	1.13	29.98	0.20	0.68
1979	2493.54	358.11	14.36	21.40	0.30	1.40	30.12	0.28	0.92
1980	2412.56	352.01	14.59	21.54	0.29	1.34	30.24	0.29	0.95
1981	2350.36	245.06	10.43	21.66	0.15	0.70	30.36	0.15	0.50
1982	2371.60	219.97	9.28	21.70	0.12	0.56	30.40	0.12	0.40
1983	2293.74	341.94	14.91	21.76	0.05	0.25	30.48	0.08	0.27
1984	2317.78	370.15	15.97	21.64	0.25	1.16	30.38	0.23	0.75
1985	2303.04	373.88	16.23	21.58	0.24	1.11	30.34	0.22	0.72
1986	2196.28	328.25	14.95	21.68	0.37	1.71	30.46	0.37	1.22
1987	2187.04	328.97	15.04	21.86	0.55	2.54	30.66	0.55	1.81
1988	2311.56	246.72	10.67	21.90	0.56	2.54	30.70	0.56	1.81
1989	2300.60	225.45	9.80	22.00	0.43	1.96	30.80	0.43	1.40
1990	2526.48	498.16	19.72	22.14	0.33	1.48	30.94	0.33	1.06
1991	2560.78	466.60	18.22	22.08	0.34	1.55	30.88	0.34	1.11
1992	2477.42	581.78	23.48	21.80	0.38	1.75	30.60	0.38	1.24
1993	2530.40	585.98	23.16	21.68	0.40	1.83	30.48	0.40	1.30
1994	2475.24	585.44	23.65	21.62	0.41	1.92	30.42	0.41	1.36
1995	2275.50	341.39	15.00	21.64	0.45	2.08	30.44	0.45	1.48
1996	2344.20	374.08	15.96	21.64	0.45	2.08	30.44	0.45	1.48
1997	2434.86	205.23	8.43	21.86	0.45	2.06	30.64	0.43	1.40
1998	2389.60	159.28	6.67	22.12	0.49	2.22	30.90	0.48	1.57
1999	2433.00	180.58	7.42	22.26	0.30	1.33	31.02	0.31	1.00
2000	2590.92	340.94	13.16	22.20	0.32	1.42	30.96	0.32	1.04
2001	2408.70	519.63	21.57	22.26	0.27	1.21	31.00	0.29	0.94
2002	2413.70	517.05	21.42	22.12	0.40	1.79	30.88	0.40	1.28
2003	2324.80	547.21	23.54	22.08	0.33	1.48	30.82	0.29	0.93

2004	2285.78	530.58	23.21	22.04	0.34	1.53	30.80	0.29	0.95
2005	2237.70	439.29	19.63	22.04	0.34	1.53	30.80	0.29	0.95
2006	2455.16	375.15	15.28	22.02	0.33	1.49	30.78	0.29	0.93
2007	2605.06	401.33	15.41	22.08	0.25	1.13	30.84	0.21	0.67
2008	2837.02	284.12	10.01	21.96	0.09	0.41	30.72	0.08	0.27
2009	2792.38	374.73	13.42	22.00	0.10	0.45	30.76	0.11	0.37
2010	2753.08	373.78	13.58	22.08	0.20	0.93	30.84	0.23	0.75
2011	2760.10	375.07	13.59	21.98	0.29	1.34	30.76	0.30	0.99
2012	2647.94	384.41	14.52	21.90	0.37	1.68	30.68	0.37	1.21
2013	2527.88	255.80	10.12	21.88	0.37	1.69	30.66	0.38	1.23
2014	2442.00	423.71	17.35	21.86	0.36	1.64	30.62	0.36	1.16
2015	2340.94	412.36	17.61	21.88	0.40	1.81	30.62	0.36	1.16

^a Computations are based on data obtained from the Climate Research Unit (CRU) country dataset (CRU CY v.3.24.01) at East Anglia University for the period 1901-2015. A five-year moving average was used to compute mean and standard deviation (SD). As a result, the time period is reduced to 1905-2015. Coefficient of Variation (CV) = (SD/Mean) *100.